# Southampton

# University of Southampton Research Repository

Copyright © and Moral Rights for this thesis and, where applicable, any accompanying data are retained by the author and/or other copyright owners. A copy can be downloaded for personal non-commercial research or study, without prior permission or charge. This thesis and the accompanying data cannot be reproduced or quoted extensively from without first obtaining permission in writing from the copyright holder/s. The content of the thesis and accompanying research data (where applicable) must not be changed in any way or sold commercially in any format or medium without the formal permission of the copyright holder/s.

When referring to this thesis and any accompanying data, full bibliographic details must be given, e.g.

Thesis: Author (Year of Submission) "Full thesis title", University of Southampton, name of the University Faculty or School or Department, PhD Thesis, pagination.

Data: Author (Year) Title. URI [dataset]

### **University of Southampton**

Faculty of Social Sciences

School of Economic, Social and Political Sciences

# Caregivers and Loneliness in the English Longitudinal Study of Ageing

By

### Lisa Davison

ORCID ID 0000-0002-9874-7713

Thesis for the degree of **Doctor of Philosophy** 

August 2021

# UNIVERSITY OF SOUTHAMPTON <u>ABSTRACT</u>

#### FACULTY OF SOCIAL SCIENCES

School of Economic, Social and Political Sciences Thesis for the degree of <u>Doctor of Philosophy</u> CAREGIVERS AND LONELINESS IN THE ENGLISH LONGITUDINAL STUDY OF AGEING By Lisa Davison

Informal caregivers represent around 26% (13.6 million) of the UK population. Alongside the known health and wellbeing consequences of caregiving, caregivers have been identified as being at higher risk of loneliness. However, current research among caregivers is inadequate. Mostly, specific subgroups of caregivers are analysed in isolation (e.g. dementia caregivers or spousal caregivers), and therefore the diverse caregiving population is under-represented in the literature. Specifically, there is a lack of evidence on the determinants of loneliness among caregivers. What limited research that has been conducted, suggests that the typical determinants applicable to the general population are not consistent amongst caregivers. Therefore, because caregiving-specific risk factors are not clearly understood, support services and policy-makers cannot work to effectively prevent loneliness. Furthermore, individually, both caregiving and loneliness are associated with adverse health and wellbeing outcomes, but there is an absence of research investigating the impact on health and wellbeing for caregivers who are lonely. It is hypothesised that the combined impact of loneliness and caregiving could have worse health and wellbeing outcomes than either characteristic alone.

Using quantitative secondary data analysis of the English Longitudinal Study of Ageing, this thesis explored, both cross-sectionally and longitudinally, the effects of caregiving on loneliness. This thesis addressed the aforementioned research gaps by comparing a wide range of caregivers, including those providing care at different intensities, and to a variety of care recipients. Additionally, caregiving-specific determinants of loneliness were investigated, as well as examining how the health and wellbeing impacts of loneliness differed between caregivers and non-caregivers. The results indicated that caregiving was associated with loneliness, but depended on the loneliness measure used. As such caregivers were more likely to be lonely using the UCLA loneliness scale, but not a direct question. Additionally, the determinants differed for caregivers and non-caregivers: health appeared to play a larger role in non-caregivers' loneliness, whereas within caregivers, care provision to adult-children was a determinant of loneliness. Finally, as predicted, the combined impact of caregiving and loneliness was associated with worse health and wellbeing outcomes across a number of measures.

Overall, there was limited evidence of a long-term effect of caregiving on loneliness, which implies that the impact of caregiving is acute rather than chronic, highlighting a need for early intervention. Finally, the evidence suggests that caregiving may have more substantial effects on mental health and wellbeing than for physical health, which provides direction for policy development.

Table of Co	ntents	i
Table of Tal	bles	vii
Table of Fig	gures	ix
Research Th	nesis: Declaration of Authorship	xi
Acknowledg	gements	xi
Abbreviatio	ns	XV
Chapter 1	Introduction	1
1.1	Aims, objectives and research questions	2
1.1.1	Research questions	3
1.2	Definitions and key concepts	4
1.2.1	Caregiving	4
1.2.2	Loneliness	7
1.2.3	Health and wellbeing	11
1.3	Thesis structure	14
Chapter 2	Theoretical understanding	17
2.1.1	Caregiving theories	17
<mark>2.1.2</mark>	Stereotype embodiment theory	19
2.2	Theoretical and conceptual framework	21
2.3	Chapter summary	22
Chapter 3	Literature review	23
3.1	Chapter outline	23
3.2	Determinants of loneliness	_23
3.2.1	Are females lonelier?	24
3.2.2	Relationships as a protective factor	26
3.2.3	Social networks	29
3.2.4	Socioeconomic status	31
<mark>3.2.5</mark>	Summarising the literature on determinants of loneliness	33
3.3	Caregiving literature	33
3.3.1	Caregiver characteristics	34
3.3.2	Caregiving relationship	39
3.3.3	Factors relating to the caregiving role	41
3.3.4	Burden and satisfaction	45
3.3.5	Summary of caregiving literature	48

3.4	Loneliness, health, and wellbeing	49					
3.4.1	Associations between loneliness, health, and wellbeing						
3.4.2	Loneliness, health, and wellbeing in a caregiving context5						
3.5	Chapter summary	60					
3.5.1	What are the research gaps?						
Chapter 4	Methodology	63					
4.1	Chapter outline	63					
4.2	Research strategy	63					
4.2.1	Research design	63					
4.2.2	English Longitudinal Study of Ageing (ELSA)	67					
4.3	Key variables in ELSA	70					
4.3.1	Caregiving variables	70					
<mark>4.3.2</mark>	Loneliness variables	73					
4.3.3	Health variables	77					
4.3.4	Wellbeing variables	81					
<mark>4.3.5</mark>	Sociodemographic variables	82					
4.3.6	Socioeconomic variables	83					
<mark>4.3.7</mark>	Social network variables	84					
4.3.8	Variables computed for the follow-up analysis	85					
4.3.9	Weighting	89					
4.4	Quantitative methods	89					
4.4.1	Bivariate analysis	89					
4.4.2	Multivariate analysis	90					
<mark>4.4.3</mark>	Missing data	91					
4.4.4	Multicollinearity	92					
4.5	Chapter summary	93					
Chapter 5	Cross-sectional results	95					
5.1	Chapter outline	95					
5.2	RQ1: how does loneliness differ for caregivers and non-caregivers?	95					
<mark>5.2.1</mark>	Descriptive statistics for caregivers and non-caregivers	95					
<mark>5.2.2</mark>	RQ1a: is caregiving associated with loneliness?	99					
5.2.3	RQ1b: are the determinants of loneliness different for caregivers and non-caregivers?	110					
5.2.4	RQ1c: how does the interaction between caregiving and loneliness affect the relation	onships					
between lon	eliness and health and wellbeing?	117					
<mark>5.2.5</mark>	Summary: how does loneliness differ for caregivers and non-caregivers?	133					
5.3	RQ2: how does caregiving affect loneliness, health, and wellbeing?	134					
<mark>5.3.1</mark>	Descriptive statistics for the caregiving sample	134					
5.3.2	RQ2a: how does caregiving affect loneliness?	138					
5.3.3	RQ2b: how does caregiving affect health?141						

5.3.4	RQ2c: how does caregiving affect wellbeing?	150
<mark>5.3.5</mark>	Summary: how does caregiving affect loneliness, health, and wellbeing?	155
5.4	Chapter summary	156
Chapter 6	Twelve year follow-up	159
6.1	Chapter outline	159
6.2	RQ3: What are the longer-term effects of caregiving on loneliness?	159
<mark>6.2.1</mark>	Descriptive statistics for the longitudinally-weighted sample	159
6.2.2	RQ3a: is caregiving in wave two associated with loneliness at wave eight?	163
<mark>6.2.3</mark>	RQ3b: is caregiving at any wave associated with later loneliness?	169
<mark>6.2.4</mark>	RQ3c: how are the caregiving relationship, and role, associated with loneliness for	or short-term
versus long-to	erm, or intermittent caregivers?	173
<mark>6.2.5</mark>	Summary: what are the longer-term effects of caregiving on loneliness?	176
6.3	Chapter summary	178
Chapter 7	Discussion and conclusion	179
7.1	Chapter outline	179
7.2	Association between caregiving and loneliness	179
7.3	Determinants of loneliness	181
7.4	Health and wellbeing consequences of loneliness	184
7.5	Limitations	185
7.5.1	Variable limitations	185
7.5.2	Sample limitations	188
<mark>7.5.3</mark>	Methodological limitations	189
7.5.4	Limitations summary	190
7.6	Conclusion	190
7.6.1	Original contribution	191
<mark>7.6.2</mark>	Contribution to theory	192
7.6.3	Recommendations for policy and further research	194
Appendix A	Confirmation of ethical approval	197
Appendix B	Bivariate associations between sociodemographic characteristics, socioecond	omic factors,
social networ	cks, loneliness, health, wellbeing and the UCLA-LS and direct loneliness question	n in the full
sample		199
Appendix C	RQ1a is caregiving associated with directly reporting loneliness?	201
Appendix D	RQ1a is caregiving associated loneliness on the UCLA loneliness scale?	203
Appendix E	RQ1a does caregiving impact loneliness expectations?	205
Appendix F	RQ1b what are the determinants of caregivers' loneliness?	207
Appendix G	RQ1b what are the determinants of non-caregivers' loneliness?	209

Appendix H	RQ1c how does the interaction between caregiving and loneliness affect the relation	aships
between caregiv	ing, loneliness and physiological health?	_211
Appendix I	RQ1c how does the interaction between caregiving and loneliness affect the relation	nships
between caregiv	ing, loneliness and alcohol consumption?	_213
Appendix J	RQ1c how does the interaction between caregiving and loneliness affect the relation	<mark>1ships</mark>
between caregiv	ing, loneliness and health expectations?	_215
Appendix K	RQ1c how does the interaction between caregiving and loneliness affect the relation	<mark>1ships</mark>
between caregiv	ing, loneliness and mental health (CES-D)?	_217
Appendix L	RQ1c how does the interaction between caregiving and loneliness affect the relation	<mark>1ships</mark>
between caregiv	ing, loneliness and limiting long-term illness?	_219
Appendix M	RQ1c how does the interaction between caregiving and loneliness affect the relation	<mark>1ships</mark>
between caregiv	ing, loneliness and QoL?	_221
Appendix N	RQ1c how does the interaction between caregiving and loneliness affect the relation	nships
between caregiv	ing, loneliness, and life satisfaction?	_223
Appendix O	Bivariate associations between caregivers' characteristics, the caregiving relation	nship,
caregiving role,	social networks, loneliness, health, wellbeing, and loneliness (wave eight) in the cares	<mark>jiving</mark>
sample		_225
Appendix P	RQ2a how does caregiving affect loneliness on the UCLA Loneliness Scale?	_227
Appendix Q	RQ2b how does caregiving affect physiological health?	_229
Appendix R	RQ2b how does caregiving affect alcohol consumption?	_231
Appendix S	RQ2b how does caregiving affect health expectations?	_233
Appendix T	RQ2c how does caregiving affect QoL?	_235
Appendix U	RQ2c how does caregiving affect life satisfaction?	_237
Appendix V	Crosstabulation of UCLA loneliness responses at waves two and eight	_239
Appendix W	Bivariate associations between baseline (wave two) variables and wave eight lonelin	<mark>ess in</mark>
the longitudinall	y-weighted sample	_241
Appendix X	Bivariate associations between the variables computed for the follow-up analyse	<mark>s and</mark>
loneliness occur	ring after the start of care provision, in the longitudinally-weighted caregiving sample $\_$	_243
Appendix Y	RQ3a(i) is caregiving associated with loneliness 12-years later, excluding those lon	ely at
<mark>baseline</mark>		_245
Appendix Z	RQ3a(ii) is caregiving associated with loneliness 12-years later, controlling for loneling	iess at
baseline		_247
Appendix AA	Crosstabulation of responses to loneliness expectations and UCLA-LS loneliness at	wave
<mark>two</mark>		_249
Appendix BB	RQ3b is caregiving at any wave associated with later loneliness?	_251

Appendix CC	RQ3c how are the caregiving relationship, and role, as	ssociated with loneliness for short-term
versus long-tern	n, or intermittent caregivers?	253
Glossary of Ten	ms	255
References		257

## **Table of Tables**

Table 1: The number of core and total respondents at each wave in ELSA waves 1-8
Table 2: Percentage of respondents providing care within the past week/month in ELSA waves 2-8 70
Table 3: Variables relating to the caregivers' relationship to the care recipient
Table 4: Crosstabulation of responses to the UCLA-LS and direct loneliness questions
Table 5: Crosstabulation of responses to two loneliness expectation questions
Table 6: Recoding of the behavioural health variable, frequency of alcohol consumption
Table 7: Crosstabulation of responses to two health expectation questions       81         7: Crosstabulation of responses to two health expectation questions       81
Table 8: Computing the loneliness occurring after the start of care provision variable
Table 9: Descriptive statistics for caregivers and non-caregivers in ELSA wave eight
Table 11: Regression models for RO1a: is caregiving associated with loneliness?       100
Table 12: Summary of significant results for RO1a: is caregiving associated with loneliness?         109
Table 13: Regression models for RQ1b: are the determinants of loneliness different for caregivers and non-
caregivers?
Table 14: Summary of significant results for RQ1b: are the determinants of loneliness different for caregivers
and non-caregivers?117
Table 15: Regression models for RQ1c: how does the interaction between caregiving and loneliness affect the
relationships between loneliness and health and wellbeing?
Table 16: Fully-adjusted linear regression model examining how the interaction between caregiving and
loneliness affects the relationships between caregiving, loneliness and quality of life 129
Table 17: Fully-adjusted linear regression model examining how the interaction between caregiving and
loneliness affects the relationships between caregiving, loneliness and life satisfaction 130
Table 18: Summary of significant results for RQ1c: how does the interaction between caregiving and loneliness
affect the relationships between loneliness and health?
Table 19: Summary of significant results for RQ1c: how does the interaction between caregiving and loneliness
affect the relationships between loneliness and wellbeing?
Table 20: Descriptive statistics for caregivers in ELSA wave eight       135
Table 21: Regression models for RQ2a: how does caregiving affect loneliness?

#### **Table of Tables**

Table 23: Regression models for RQ2b: how does caregiving affect health?       142         Table 24: Summary of significant results for RQ2b: how does caregiving affect health?       150         Table 25: Regression models for RQ2c: how does caregiving affect wellbeing?       150         Table 26: Fully-adjusted linear regression model examining how caregiving-specific variables affect caregivers' quality of life.       151         Table 27: Fully-adjusted linear regression model examining how caregiving-specific variables affect caregivers' life satisfaction       153         Table 28: Summary of significant results for RQ2c: how does caregiving affect wellbeing?       155         Table 29: Descriptive statistics of the longitudinal sample at baseline (ELSA wave two)       160         Table 30: Frequencies of caregiving-specific variables, and changes to mental health and loneliness expectations for caregivers in the longitudinal sample       162         Table 31: Regression models for RQ3a: is caregiving in wave two associated with loneliness at wave eight?       163         Table 32: Summary of significant results for RQ3a: is caregiving in wave two associated with loneliness at wave eight?       168         Table 33: Regression models for RQ3b: is caregiving at any wave associated with ever being lonely?      70	Table 22: Summary of significant results for RQ2a: how does caregiving affect loneliness?141
Table 24: Summary of significant results for RQ2b: how does caregiving affect health?       150         Table 25: Regression models for RQ2c: how does caregiving affect wellbeing?       150         Table 26: Fully-adjusted linear regression model examining how caregiving-specific variables affect caregivers' quality of life       151         Table 27: Fully-adjusted linear regression model examining how caregiving-specific variables affect caregivers' life satisfaction       153         Table 28: Summary of significant results for RQ2c: how does caregiving affect wellbeing?       155         Table 29: Descriptive statistics of the longitudinal sample at baseline (ELSA wave two)       160         Table 30: Frequencies of caregiving-specific variables, and changes to mental health and loneliness expectations for caregivers in the longitudinal sample.       162         Table 31: Regression models for RQ3a: is caregiving in wave two associated with loneliness at wave eight?       163         Table 32: Summary of significant results for RQ3a: is caregiving in wave two associated with loneliness at wave eight?       168         Table 33: Regression models for RQ3b: is caregiving at any wave associated with ever being lonely?      70	Table 23: Regression models for RQ2b: how does caregiving affect health?         142
Table 25: Regression models for RQ2c: how does caregiving affect wellbeing?       150         Table 26: Fully-adjusted linear regression model examining how caregiving-specific variables affect caregivers' quality of life       151         Table 27: Fully-adjusted linear regression model examining how caregiving-specific variables affect caregivers' life satisfaction       153         Table 28: Summary of significant results for RQ2c: how does caregiving affect wellbeing?       155         Table 29: Descriptive statistics of the longitudinal sample at baseline (ELSA wave two)       160         Table 30: Frequencies of caregiving-specific variables, and changes to mental health and loneliness expectations for caregivers in the longitudinal sample       162         Table 31: Regression models for RQ3a: is caregiving in wave two associated with loneliness at wave eight?       163         Table 32: Summary of significant results for RQ3a: is caregiving in wave two associated with loneliness at wave eight?       168         Table 33: Regression models for RQ3b: is caregiving at any wave associated with ever being lonely?       .170	Table 24: Summary of significant results for RQ2b: how does caregiving affect health?
Table 26: Fully-adjusted linear regression model examining how caregiving-specific variables affect         caregivers' quality of life	Table 25: Regression models for RQ2c: how does caregiving affect wellbeing?
<ul> <li>Table 20. Fully-adjusted linear regression model examining how caregiving-specific variables affect caregivers' quality of life</li></ul>	Table 26: Fully-adjusted linear regression model examining how caregiving-specific variables affect
Table 27: Fully-adjusted linear regression model examining how caregiving-specific variables affect         caregivers' life satisfaction         153         Table 28: Summary of significant results for RQ2c: how does caregiving affect wellbeing?         155         Table 29: Descriptive statistics of the longitudinal sample at baseline (ELSA wave two)         160         Table 30: Frequencies of caregiving-specific variables, and changes to mental health and loneliness         expectations for caregivers in the longitudinal sample         162         Table 31: Regression models for RQ3a: is caregiving in wave two associated with loneliness at wave eight?         163         Table 32: Summary of significant results for RQ3a: is caregiving in wave two associated with loneliness at wave eight?         163         Table 33: Regression models for RQ3b: is caregiving at any wave associated with ever being lonely?	Table 20. Fully-adjusted linear regression model examining now categoring-specific variables affect
Table 27: Fully-adjusted linear regression model examining how caregiving-specific variables affect         caregivers' life satisfaction         Table 28: Summary of significant results for RQ2c: how does caregiving affect wellbeing?         Table 29: Descriptive statistics of the longitudinal sample at baseline (ELSA wave two)         Table 30: Frequencies of caregiving-specific variables, and changes to mental health and loneliness         expectations for caregivers in the longitudinal sample         162         Table 31: Regression models for RQ3a: is caregiving in wave two associated with loneliness at wave eight?	caregivers' quality of life151
Table 27. Fully-adjusted linear regression model examining now caregiving-specific variables affect         caregivers' life satisfaction         Table 28: Summary of significant results for RQ2c: how does caregiving affect wellbeing?         Table 29: Descriptive statistics of the longitudinal sample at baseline (ELSA wave two)         Table 30: Frequencies of caregiving-specific variables, and changes to mental health and loneliness         expectations for caregivers in the longitudinal sample         162         Table 31: Regression models for RQ3a: is caregiving in wave two associated with loneliness at wave eight?	Table 27. Evily adjusted linear regression model and in the second state of the state of the
caregivers' life satisfaction       153         Table 28: Summary of significant results for RQ2c: how does caregiving affect wellbeing?       155         Table 29: Descriptive statistics of the longitudinal sample at baseline (ELSA wave two)       160         Table 30: Frequencies of caregiving-specific variables, and changes to mental health and loneliness expectations for caregivers in the longitudinal sample       162         Table 31: Regression models for RQ3a: is caregiving in wave two associated with loneliness at wave eight?       163         Table 32: Summary of significant results for RQ3a: is caregiving in wave two associated with loneliness at wave eight?       168         Table 33: Regression models for RQ3b: is caregiving at any wave associated with ever being lonely?       170	Table 27: Fully-adjusted linear regression model examining now caregiving-specific variables affect
Table 28: Summary of significant results for RQ2c: how does caregiving affect wellbeing?	caregivers' life satisfaction153
Table 29: Descriptive statistics of the longitudinal sample at baseline (ELSA wave two)	Table 28: Summary of significant results for RQ2c: how does caregiving affect wellbeing?
Table 30: Frequencies of caregiving-specific variables, and changes to mental health and loneliness expectations for caregivers in the longitudinal sample	Table 29: Descriptive statistics of the longitudinal sample at baseline (ELSA wave two)
expectations for caregivers in the longitudinal sample	Table 30: Frequencies of caregiving-specific variables, and changes to mental health and loneliness
Table 31: Regression models for RQ3a: is caregiving in wave two associated with loneliness at wave eight?	expectations for caregivers in the longitudinal sample162
Table 32: Summary of significant results for RQ3a: is caregiving in wave two associated with loneliness at wave eight?	Table 31: Regression models for RO3a: is caregiving in wave two associated with loneliness at wave eight?
Table 32: Summary of significant results for RQ3a: is caregiving in wave two associated with loneliness at wave eight?         Wave eight?         Table 33: Regression models for RQ3b: is caregiving at any wave associated with ever being lonely?	163
Table 32: Summary of significant results for RQ3a: is caregiving in wave two associated with loneliness at wave eight?         wave eight?       168         Table 33: Regression models for RQ3b: is caregiving at any wave associated with ever being lonely?       170	
wave eight?	Table 32: Summary of significant results for RO3a; is caregiving in wave two associated with loneliness at
Table 33: Regression models for RQ3b: is caregiving at any wave associated with ever being lonely?170	
Table 33: Regression models for RQ3b: is caregiving at any wave associated with ever being lonely?170	wave eight?
	Table 33: Regression models for RQ3b: is caregiving at any wave associated with ever being lonely?170
Table 34: Regression models for RQ3c: how do changes to the caregiving relationship and role impact	Table 34: Regression models for RQ3c: how do changes to the caregiving relationship and role impact
loneliness among caregivers?	loneliness among caregivers?

# **Table of Figures**

Figure 1: Caregiver Identity Theory: phases of caregiver identity change	
Figure 2: Conceptual framework	
Figure 3: Odds ratio plot, factors associated with directly reporting loneliness	102
Figure 4: Odds ratio plot, factors associated with UCLA-LS loneliness	103
Figure 5: Odds ratio plot, factors associated with disagreeing with loneliness expectations	106
Figure 6: Odds ratio plot, factors associated with neither agreeing or disagreeing with loneliness	expectations
	108
Figure 7: Odds ratio plot, determinants of caregivers' loneliness	113
Figure 8: Odds ratio plot, determinants of non-caregivers' loneliness	115
Figure 9: Odds ratio plot, factors associated physiological health	119
Figure 10: Interaction plot, alcohol consumption by loneliness and caregiving status	
Figure 11: Odds ratio plot, factors associated with alcohol consumption	121
Figure 12: Odds ratio plot, factors associated with disagreeing with health expectations	
Figure 13: Odds ratio plot, factors associated with neither agreeing nor disagreeing with health	expectations
Figure 13: Odds ratio plot, factors associated with neither agreeing nor disagreeing with health	expectations
Figure 13: Odds ratio plot, factors associated with neither agreeing nor disagreeing with health	expectations 124 125
Figure 13: Odds ratio plot, factors associated with neither agreeing nor disagreeing with health Figure 14: Interaction plot, depressive symptomatology by loneliness and caregiving status Figure 15: Odds ratio plot, factors associated with depressive symptomatology	expectations 124 125 126
<ul> <li>Figure 13: Odds ratio plot, factors associated with neither agreeing nor disagreeing with health</li> <li>Figure 14: Interaction plot, depressive symptomatology by loneliness and caregiving status.</li> <li>Figure 15: Odds ratio plot, factors associated with depressive symptomatology</li> <li>Figure 16: Odds ratio plot, factors associated with LLTI.</li> </ul>	expectations 124 125 126 127
<ul> <li>Figure 13: Odds ratio plot, factors associated with neither agreeing nor disagreeing with health</li> <li>Figure 14: Interaction plot, depressive symptomatology by loneliness and caregiving status.</li> <li>Figure 15: Odds ratio plot, factors associated with depressive symptomatology</li> <li>Figure 16: Odds ratio plot, factors associated with LLTI</li> <li>Figure 17: Interaction plot, LLTI by loneliness and caregiving status.</li> </ul>	expectations 124 125 126 127 127
<ul> <li>Figure 13: Odds ratio plot, factors associated with neither agreeing nor disagreeing with health</li> <li>Figure 14: Interaction plot, depressive symptomatology by loneliness and caregiving status</li> <li>Figure 15: Odds ratio plot, factors associated with depressive symptomatology</li> <li>Figure 16: Odds ratio plot, factors associated with LLTI</li> <li>Figure 17: Interaction plot, LLTI by loneliness and caregiving status</li> <li>Figure 18: Interaction plot, quality of life by loneliness and caregiving status</li> </ul>	expectations 124 125 126 127 127 129
Figure 13: Odds ratio plot, factors associated with neither agreeing nor disagreeing with health Figure 14: Interaction plot, depressive symptomatology by loneliness and caregiving status Figure 15: Odds ratio plot, factors associated with depressive symptomatology Figure 16: Odds ratio plot, factors associated with LLTI Figure 17: Interaction plot, LLTI by loneliness and caregiving status Figure 18: Interaction plot, quality of life by loneliness and caregiving status Figure 19: Interaction plot, life satisfaction by loneliness and caregiving status	expectations 124 125 126 127 127 127 129 131
<ul> <li>Figure 13: Odds ratio plot, factors associated with neither agreeing nor disagreeing with health</li> <li>Figure 14: Interaction plot, depressive symptomatology by loneliness and caregiving status</li> <li>Figure 15: Odds ratio plot, factors associated with depressive symptomatology</li> <li>Figure 16: Odds ratio plot, factors associated with LLTI</li> <li>Figure 17: Interaction plot, LLTI by loneliness and caregiving status</li> <li>Figure 18: Interaction plot, quality of life by loneliness and caregiving status</li> <li>Figure 19: Interaction plot, life satisfaction by loneliness and caregiving status</li> <li>Figure 20: Odds ratio plot, caregiving-specific factors associated with caregivers' loneliness</li> </ul>	expectations 124 125 126 127 127 127 129 131 139
Figure 13: Odds ratio plot, factors associated with neither agreeing nor disagreeing with health Figure 14: Interaction plot, depressive symptomatology by loneliness and caregiving status Figure 15: Odds ratio plot, factors associated with depressive symptomatology Figure 16: Odds ratio plot, factors associated with LLTI Figure 17: Interaction plot, LLTI by loneliness and caregiving status Figure 18: Interaction plot, quality of life by loneliness and caregiving status Figure 19: Interaction plot, life satisfaction by loneliness and caregiving status Figure 20: Odds ratio plot, caregiving-specific factors associated with caregivers' physiological Figure 21: Odds ratio plot, caregiving-specific factors associated with caregivers' physiological Figure 21: Odds ratio plot, caregiving-specific factors associated with caregivers' physiological Figure 21: Odds ratio plot, caregiving-specific factors associated with caregivers' physiological Figure 21: Odds ratio plot, caregiving-specific factors associated with caregivers' physiological Figure 21: Odds ratio plot, caregiving-specific factors associated with caregivers' physiological Figure 21: Odds ratio plot, caregiving-specific factors associated with caregivers' physiological Figure 21: Odds ratio plot, caregiving-specific factors associated with caregivers' physiological Figure 21: Odds ratio plot, caregiving-specific factors associated with caregivers' physiological Figure 21: Odds ratio plot, caregiving-specific factors associated with caregivers' physiological Figure 21: Odds ratio plot, caregiving-specific factors associated with caregivers' physiological Figure 21: Odds ratio plot, caregiving-specific factors associated with caregivers' physiological Figure 21: Odds ratio plot, ca	expectations 
<ul> <li>Figure 13: Odds ratio plot, factors associated with neither agreeing nor disagreeing with health</li> <li>Figure 14: Interaction plot, depressive symptomatology by loneliness and caregiving status</li> <li>Figure 15: Odds ratio plot, factors associated with depressive symptomatology</li> <li>Figure 16: Odds ratio plot, factors associated with LLTI</li> <li>Figure 17: Interaction plot, LLTI by loneliness and caregiving status</li> <li>Figure 18: Interaction plot, quality of life by loneliness and caregiving status</li> <li>Figure 19: Interaction plot, life satisfaction by loneliness and caregiving status</li> <li>Figure 20: Odds ratio plot, caregiving-specific factors associated with caregivers' loneliness</li> <li>Figure 21: Odds ratio plot, caregiving-specific factors associated with caregivers' alcohol consum</li> </ul>	expectations 
<ul> <li>Figure 13: Odds ratio plot, factors associated with neither agreeing nor disagreeing with health</li> <li>Figure 14: Interaction plot, depressive symptomatology by loneliness and caregiving status.</li> <li>Figure 15: Odds ratio plot, factors associated with depressive symptomatology</li> <li>Figure 16: Odds ratio plot, factors associated with LLTI</li> <li>Figure 17: Interaction plot, LLTI by loneliness and caregiving status.</li> <li>Figure 18: Interaction plot, quality of life by loneliness and caregiving status.</li> <li>Figure 19: Interaction plot, life satisfaction by loneliness and caregiving status.</li> <li>Figure 20: Odds ratio plot, caregiving-specific factors associated with caregivers' loneliness.</li> <li>Figure 21: Odds ratio plot, caregiving-specific factors associated with caregivers' alcohol consump</li> <li>Figure 23: Odds ratio plot, caregiving-specific factors associated with caregivers disagreeing</li> </ul>	expectations 

### **Table of Figures**

Figure	24:	Odds	ratio	plot,	caregiv	ing-specif	ic factor	s associated	d with	caregivers	neither	agreeing nor
			disag	greeing	<mark>g with h</mark>	ealth expe	ctations					148
<b>Figure</b>	25:	Odds	ratio	plot,	wave tv	vo factors	associat	ed with wa	ive eig	ht loneline	ss, exclu	iding baseline
			lone <sup>l</sup>	liness	•••••							164
<b>Figure</b>	26:	<mark>Odds 1</mark>	atio p	lot, w	<mark>ave two</mark>	factors a	ssociated	with wave	eight l	oneliness, c	ontrollin	g for baseline
			lone <sup>l</sup>	liness								166
Figure	<mark>27:</mark> (	<mark>Odds r</mark> a	<mark>atio pl</mark>	<mark>ot, ass</mark>	ociatior	<mark>is betweer</mark>	loneline	ss and durat	ion of o	care		171
<b>Figure</b>	28:	<mark>Odds 1</mark>	atio p	<mark>lot, as</mark>	sociatio	ns betwee	n lonelin	ess and dur	ation o	of care, care	<mark>giving-s</mark>	pecific factors
			and o	change	es to lor	eliness ex	pectation	s and menta	l healt	<mark>h</mark>		175
Figure	29: .	Amend	led cor	nceptu	al fram	ework						191

### **Research Thesis: Declaration of Authorship**

Print name: Lisa Davison

Title of thesis: Caregivers and Loneliness in the English Longitudinal Study of Ageing

I declare that this thesis and the work presented in it are my own and has been generated by me as the result of my own original research.

I confirm that:

- This work was done wholly or mainly while in candidature for a research degree at this University;
- Where any part of this thesis has previously been submitted for a degree or any other qualification at this University or any other institution, this has been clearly stated;
- Where I have consulted the published work of others, this is always clearly attributed;
- Where I have quoted from the work of others, the source is always given. With the exception of such quotations, this thesis is entirely my own work;
- I have acknowledged all main sources of help;
- Where the thesis is based on work done by myself jointly with others, I have made clear exactly what was done by others and what I have contributed myself;
- None of this work has been published before submission.

Signature:

Date: 06/08/2021

#### Acknowledgements

### Acknowledgements

Foremost, I would like to thank the Economic and Social Research Council for their financial support in funding this research project, and my PhD studentship (grant number ES/P000673/1). Additionally, I extend this recognition to the South Coast Doctoral Training Partnership for their excellent provision of support, training, and opportunities throughout my candidature.

I would also like to express my gratitude to my supervisory team: Dr. Rosalind Willis and Dr. Gloria Chepngeno Langat, for their ongoing contributions. Their detailed and timely provision of advice, and feedback on my work, over the past few years has been greatly appreciated. I have learned a lot from their efforts as supervisors, and hope they have enjoyed working with me, as much as I have them. Alongside this, honourable mention goes to Prof. Asghar Zaidi, for acting as my secondary supervisor for the first seven months of my candidature.

My next expression of gratitude is to Dr. Kathryn Wicks, without whom, this PhD would not have been possible. Kathryn acted as both my personal tutor and dissertation supervisor throughout my MSc degree, and I cannot thank her enough for encouraging me to apply for this PhD studentship. Without a doubt, my number one supporter throughout this process, Kathryn has helped in more ways than imaginable, including, but not limited to, preparation for my confirmation exam and proof-reading my final thesis.

During my candidature I was awarded a UK Research and Innovation Research Councils Policy Internship, which I completed by way of a thirteen-week internship with Age UK, in London. I consider this internship a pivotal part of my personal development, as it helped me to improve many transferrable skills, but predominantly my confidence in communicating and presenting. Therefore, I extend my sincerest appreciation to the Wellbeing and Services teams at Tavis House, the local Age UK teams involved in the project, and in particular my supervisor, Sarah Stott.

Furthermore, I would like to formerly thank my examiners: Prof. Emily Grundy (University of Essex) and Dr. Elisabeth Schröder-Butterfill (University of Southampton) for their commitment to examining the final thesis, conducting the viva voce examination, and reviewing the submitted corrections. By extension, I would also like to thank Elisabeth for her previous contributions to both my first, and second (confirmation) milestones, and Prof. Athina Vlachantoni for her feedback and guidance during my confirmation exam.

I am grateful to the wider University of Southampton Gerontology department, both staff and students, for the excellent support network we have built over the years. Also to the Centre for Population Change, for the welcome distraction of different data to analyse during the final year of my candidature, and for the initiation of my academic career. Last, but by no means least, I thank my dogs, Chester and Mylo, for, unknowingly, providing the best emotional support one could ask for during a PhD.

Additional post-viva acknowledgements go to the wonderful research fellow team I now work in at the University of Stirling, for the provision of excellent advice and support during the completion of my thesis corrections.

# Abbreviations

ADLs	Activities of Daily Living
BBC	British Broadcasting Corporation
BHPS	British Household Panel Survey
CASP19	Control Autonomy Self-realisation and Pleasure (19 item)
CCA	Complete Case Analysis
CES-D	Centre for Epidemiological Studies Depression Scale
CIT	Caregiver Identity Theory
CLS	Community Life Survey
CV	Cardiovascular
CVD	Cardiovascular disease
DCMS	Department of Culture, Media, and Sport
DD	Developmental disabilities
DEAS	German Ageing Survey
DF	Degrees of freedom
DHSC	Department of Health and Social Care
DJGLS	de Jong Gierveld Loneliness Scale
DV	Dependent variable
ELSA	English Longitudinal Study of Ageing
FINDS	Family and Individual Needs for Disability Supports
HPD	Hours per day
HPM	Hours per month
H/PP/Y	Hours per person per year
HPW	Hours per week
HRS	Health and Retirement Study
HSE	Health Survey for England
IADLs	Instrumental Activities of Daily Living
ICIM	Informal Caregiving Integrative Model
IV	Independent variable
LLTI	Limiting long-term illness
NHS	National Health Service
NICE	National Institute for Health and Care Excellence

#### Abbreviations

NSHAP	National Social Life, Health and Ageing Project
ONS	Office for National Statistics
OR	Odds ratio
QoL	Quality of life
RQ	Research question
SARS-CoV-2	Severe acute respiratory syndrome coronavirus 2
SD	Standard deviation
SDA	Secondary data analysis
SEM	Social Ecological Model
SES	Socioeconomic status
SET	Stereotype Embodiment Theory
SF-36	QoL Short Form
SHARE	Survey of Health, Ageing and Retirement in Europe
SWLS	Satisfaction with life scale
TILDA	The Irish Longitudinal Study on Ageing
UCLA-LS	University of California, Los Angeles Loneliness Scale
UK	United Kingdom
UKHLS	United Kingdom Household Longitudinal Study
US	United States (of America)
VIF	Variance inflation factor
WHO	World Health Organisation
95CI	95% Confidence Interval

### **Chapter 1** Introduction

Informal caregivers, hereafter referred to as caregivers, are an expanding, but largely undetected subgroup of the population, who play an essential role in supporting health and social care systems. Prior to the coronavirus (SARS-CoV-2) outbreak, there were an estimated 9.1 million caregivers across the United Kingdom (UK), representing as many as 17% of adults within the general population, although, due to the initial lockdown period beginning in March 2020, this increased by almost 50% to 13.6 million (Carers Week, 2020). Data from Understanding Society collected 2016-2018, indicated around 20% of the population were caregivers; 23% of females and 17% of males (Zhang and Bennett, 2019). Furthermore, caregiving is known to impact health and wellbeing. Schulz et al. (2020) highlight that evidence on negative outcomes are more prevalent than positive, listing psychological distress and poor physical health as common consequences. Moreover, caregiving has been linked to loneliness, particularly among dementia caregivers (Bramboeck et al., 2020; Leszko et al., 2020; Victor et al., 2020). Caregivers are 37% more likely to report loneliness than noncaregivers (Pyle and Evans, 2018). It is considered that due to the increased time spent caregiving, social activity, interaction, and consequently, social networks may be reduced amongst caregivers (Vasileiou et al., 2017). Alternatively, the loss of relationship between caregiver and care recipient may also contribute towards increased loneliness (Leszko et al., 2020). However, it is not yet fully understood why the prevalence of loneliness is higher amongst caregivers, or whether the negative consequences of loneliness are greater for caregivers compared to non-caregivers.

The prevalence of loneliness is dependent on sociodemographic characteristics (Groarke *et al.*, 2020). Data from the Office for National Statistics (ONS) indicates that in 2016/17 amongst individuals in the general population (aged  $\geq 16$ ) in England, 5% were lonely often or always, and 16% sometimes (Pyle and Evans, 2018). Moreover, evidence from Victor and Yang (2012) suggests a U-shaped distribution of loneliness, highest among younger (aged <25) and older (aged  $\geq 55$ ) adults. Using data from the English Longitudinal Study of Ageing (ELSA), the percentage of lonely people in England, aged  $\geq 50$ , was shown to be relatively stable over a decade (2006/07 to 2016/17); 6-9% were often lonely, and 24-26% were sometimes lonely (Age UK, 2018). These figures are slightly elevated in comparison to the aforementioned ONS data, but this is likely due to the reported U-shaped age distribution, it could reasonably be expected to see a higher prevalence of loneliness among a sample aged  $\geq 50$ .

The impact of loneliness on mortality risk is comparable to factors such as daily smoking, and excessive alcohol consumption; Holt-Lunstad *et al.* (2010) concluded that individuals with adequate relationships (i.e. not lonely) experience a 50% greater likelihood of survival than those with inadequate relationships. Furthermore, loneliness has previously been identified as a public health issue (Groarke *et al.*, 2020), but, since the outbreak of SARS-CoV-2, loneliness has become an increasing concern among policy-makers. Evidence suggests that risk factors for loneliness since the

#### Chapter 1

outbreak are virtually identical to beforehand; however those previously at risk of loneliness may be at higher risk (Bu *et al.*, 2020). Therefore, interventions should be aimed at groups already known to be vulnerable to loneliness (*ibid*). As such, the UK Government has identified caregivers as being a group vulnerable to loneliness (Department of Culture, Media, and Sport [DCMS], 2018; Victor *et al.*, 2020).

Although both caregiving and loneliness have been linked to adverse health and wellbeing outcomes, there is little research investigating the impact on health and wellbeing for caregivers who are lonely. Individually, loneliness is associated with physical and psychological morbidities (Ong et al., 2016), and reduced wellbeing (Emerson et al., 2020). Similarly, caregiving has been linked to negative health consequences (Caputo et al., 2016), and detrimental effects on wellbeing (Dahlrup et al., 2015). It is likely that the combined impact of loneliness and caregiving could have worse health and wellbeing outcomes than either characteristic alone. This is because, firstly, there is potential for overlap between the separate health and wellbeing consequences. As such, both caregiving and loneliness have been associated with a wide range of similar health and wellbeing outcomes, including, but not limited to, cardiovascular (CV) conditions (Friedler et al., 2015; Bouchard et al., 2019), depression (Caputo et al., 2016; Ong et al., 2016) and effects on wellbeing (Dahlrup et al., 2015; Emerson et al., 2020). Therefore, the presence of loneliness could further exacerbate depression, poor CV health, or reduced wellbeing amongst caregivers. Secondly, because loneliness is often reported as a consequence of caregiving (Beeson, 2003; Bramboeck et al., 2020), there may be sequential or cumulative effects on health and wellbeing. Additionally, evidence has demonstrated that the impact of caregiving on physical health may affect older caregivers (aged  $\geq$ 50) more than younger caregivers (Danilovich et al., 2017). Therefore, to make appropriate policy recommendations to support and protect caregivers, it is essential to investigate how the caregiving population differs in relation to loneliness, health and wellbeing; particularly among older caregivers, where health outcomes may be more prominent.

The following sections outline the proposed aims, objectives, and research questions (RQs) (section 1.1). Subsequently, key concepts and relevant definitions for this thesis (section 1.2), including caregiving, loneliness, health, and wellbeing are introduced, before outlining the structure of the thesis (section 1.3).

#### 1.1 Aims, objectives and research questions

This thesis examines the relationships between caregiving, loneliness, health, and wellbeing, for which the collective association is largely unknown. The subsequent aims, objectives, and research questions detail how this thesis investigates these areas, in order to disentangle the relationships.

The main aims are to investigate the association between caregiving and loneliness, identify the determinants of loneliness, and to evaluate the health and wellbeing consequences. From the analysis,

policy recommendations can be designed to support caregivers within England. In order to meet these aims, secondary analysis was conducted with ELSA to determine the relationship between caregiving and loneliness. Moreover, the available data were explored to ascertain the determinants of loneliness; this involved the analysis of sociodemographic characteristics, socioeconomic factors and health and wellbeing variables, contrasted between caregivers and non-caregivers. To investigate the impact of loneliness on health and wellbeing for caregivers and non-caregivers, different pathways of health were explored, namely physiological, psychological, and behavioural, alongside general physical health, mental health, and aspects of wellbeing. Primarily, this thesis contrasted caregivers and non-caregivers to determine the association between caregiving and loneliness, and in turn, the health and wellbeing impact. However, to progress the understanding of these associations, the effects of caregiving characteristics, the caregiving relationship, and the caregiving role were explored.

#### **1.1.1** Research questions

- 1. How does loneliness differ for caregivers and non-caregivers?
  - a. Is caregiving associated with loneliness?
  - b. Are the determinants of loneliness different for caregivers and non-caregivers?
  - c. How does the interaction between caregiving and loneliness affect the relationships between loneliness and health and wellbeing?
- 2. How does caregiving affect loneliness, health, and wellbeing?
  - a. How does caregiving affect loneliness?
  - b. How does caregiving affect health?
  - c. How does caregiving affect wellbeing?
- 3. What are the longer-term effects of caregiving on loneliness?
  - a. Is caregiving in wave two associated with loneliness at wave eight?
  - b. Is caregiving at any wave associated with later loneliness?
  - c. How are the caregiving relationship, and role, associated with loneliness for short-term versus long-term, or intermittent caregivers?

#### **1.2** Definitions and key concepts

#### 1.2.1 Caregiving

An informal caregiver provides unpaid assistance to someone with disabilities, or chronic physical or mental health conditions (Vasileiou et al., 2017; Gérain and Zech, 2019). Recipients of care are typically family members, friends or neighbours (Vasileiou et al., 2017). However, the circumstances of caregiving can differ considerably; as such, caregivers may cohabit with, or live separately from the recipient, and responsibilities may be regular, occasional or continuous (Brodaty and Green, 2002). The roles of a caregiver vary significantly, often categorised into activities of daily living (ADL) and instrumental activities of daily living (IADL), examples of which include: assistance using the toilet (ADL) and meal preparation (IADL) (Lyons et al., 2015). Personal care, such as assistance with bathing (Bergman and Pulling, 2020), is provided by around one-third of caregivers across Europe (Broese van Groenou and de Boer, 2016). Provision of personal care can increase physical and mental demand on the caregiver (Hiel et al., 2015). Additionally, more intimate personal care tasks have been linked to higher emotional burden (Schrank et al., 2016). This is particularly relevant for caregivers of individuals with dementia, where care needs are typically more complex. In contrast to caregivers of older adults with physical limitations, dementia caregivers reported a lack of rest, higher irritability, and relationship difficulties (Sequeira, 2013). The physical and emotional effects of providing care are typically referred to as caregiver burden (section 1.2.1.4).

Caregiving, while central to this thesis, is also a recurring theme within Gerontological research, though research often overlooks care provided by older adults, fixating on care provided by younger or middle-aged adults (Phillips *et al.*, 2010). This thesis expands upon that criticism, focusing on caregivers aged  $\geq$ 50 (middle-aged and older adults). Thomas (1993) and Phillips *et al.* (2010) advocate that care cannot be researched as a stand-alone term; as it is a multi-dimensional concept, several factors of care must be considered for a reliable analysis. Within this thesis, the factors of care are separated into three logical categories: caregiver characteristics, the caregiving relationship, and intensity of the caregiving role. Additionally, caregiving appraisal, including burden and satisfaction, is thought to mediate the relationship between caregiving and related outcomes (Gérain and Zech, 2019). Moreover, burden and satisfaction are often shown to impact upon caregiver health and wellbeing (Chang *et al.*, 2010; Mochari-Greenberger and Mosca, 2012; Dahlrup *et al.*, 2015), therefore these aspects are also discussed.

#### 1.2.1.1 Characteristics of caregivers

Thomas (1993) uses the term 'social identity' regarding the caregiver and recipient; under these circumstances, the identity of a caregiver can refer to the familial role (e.g. spouse) or the sex. Alternatively, Phillips *et al.* (2010) considers social identity to mean a 'type' of caregiver, including factors such as age or ethnicity. For the purpose of this thesis, sociodemographic and socioeconomic

characteristics are examined, aligning with Phillips *et al.* (2010) definition of identity. The familial role, as suggested by Thomas' (1993) definition, is examined as the caregiving relationship (section 1.2.1.2).

Some of the key caregiver characteristics identified within research (section 3.3.1) include age, sex, marital status, and socioeconomic status (SES). To illustrate, caregiving at older ages is linked to providing higher-intensity care (Dahlberg *et al.*, 2007) and poorer health outcomes (Danilovich *et al.*, 2017). Moreover, although females are more commonly associated with caregiving roles (van Campen *et al.*, 2013), 2011 England and Wales Census data shows that more older males than older females are caregivers (Milligan and Morbey, 2016); this suggests that the age and sex of caregivers are likely to impact outcomes such as loneliness, health, and wellbeing.

#### 1.2.1.2 Caregiving relationship and co-residential care

The inter-personal relationship refers to the existing relationship prior to caregiving; typically, these would be familial (e.g. parent) but could also include friends and neighbours (Thomas, 1993; Phillips *et al.*, 2010). Another aspect of caregiving relevant to the relationship is whether the caregiver is paid, and if the care is considered formal or informal. The concept of care is substantially different for formal and informal caregivers; financial involvement can alter the dynamic of the relationship considerably (Thomas, 1993). Nevertheless, these concepts (financial involvement and formal care) are less relevant for this thesis, as the focus is solely on informal (unpaid) caregivers.

Furthermore, location is also important; caregiving usually takes place within the caregivers' or recipients' homes, although caregiving roles often continue if the recipient is hospitalised or institutionalised (Montgomery and Kosloski, 2009; Montgomery *et al.*, 2016). The location of the caregiving tasks may also influence intensity. To illustrate, co-residential care, whereby the care recipient lives with the caregiver, is more likely to be continuous or higher-intensity (Kumagai, 2017); whereas if the recipient lives separately from the caregiver (non-residential), caregiving can be carried out over fewer hours (e.g. visiting to assist with shopping or household chores). For this thesis, both the inter-personal relationship, and provision of co-residential care are investigated as factors related to caregiving.

#### 1.2.1.3 Caregiving role: intensity of care

The caregiving role includes factors such as nature of care, duration, and intensity. Nature of care can denote the specific tasks of caregiving (Thomas, 1993), for example, assisting the recipient to dress. However, more broadly, the nature can refer to the type of support provided (Phillips *et al.*, 2010), such as personal care, or instrumental support. Nature of care is often associated with intensity, as more complex care may require longer caregiving hours.

Caregiving intensity can be classified by the number of ADL and IADL tasks the recipient requires support with, and the perceived difficulty of each task (Lyons *et al.*, 2015; Anthony *et al.*, 2017).

#### Chapter 1

However, there are weaknesses associated with this method. To illustrate, Lyons *et al.* (2015) categorise  $\geq 2$  ADL tasks, or  $\geq 6$  IADL tasks, as high-intensity; however, each task is equally weighted. Therefore, caregivers helping with two less complex ADL tasks would be categorised as high-intensity caregivers, alongside those aiding with more challenging tasks. Additionally, some caregivers might find the same task more difficult than others, for example, recipient cooperation could influence the intensity of the task. In contrast, duration could be considered a more consistent measure; difficult ADL or IADL tasks may take longer to complete, therefore longer durations would reflect higher-intensity caregiving.

Intensity is most commonly measured as the number of hours spent providing care per week, utilised in the Dutch National Population Survey (van Campen *et al.*, 2013), ONS Longitudinal Study (Robards *et al.*, 2015), ELSA (Clemens *et al.*, 2019) and UK Household Longitudinal Study (UKHLS) (Lacey *et al.*, 2018). The hours per week (hpw) method of measuring intensity is adopted in this thesis, because it is more comparable across individuals. To illustrate, caregiver A may find assisting the recipient to dress difficult due to limited mobility, whereas caregiver B might find this task simple, because the recipient is able to assist. Therefore, it would be illogical to classify this as the same level of intensity.

However, in research, intensity categories are inconsistent. Low-intensity caregiving is typically recognised as 1-19hpw (Dahlberg et al., 2007; Robards et al., 2015; Vlachantoni et al., 2016); though Lacey et al. (2018) divide this further into <5 hpw and 5-19 hpw. It is generally acknowledged that  $\geq$ 20hpw is high-intensity caregiving (Vlachantoni *et al.*, 2016; Lacey *et al.*, 2018); yet some studies introduce additional categories separating 20-49 hpw and  $\geq$  50 hpw (Dahlberg *et al.*, 2007; Robards *et* al., 2015). For context, a full-time job typically equates to 40 hpw, and many jobs do not permit working in excess of 48hpw (Gov.UK, 2018). Therefore, caregiving for  $\geq$ 50hpw is comparable to roles such as emergency services (*ibid*). Overall, current research is largely unanimous that  $\geq 20$  hpw is high-intensity; however, this poses a certain inequity when comparing within high-intensity care. The total number of potential caregiving hours ranges between 0-168hpw. Thus, if the most common division of intensity was assumed, 1-19hpw (low-intensity) versus ≥20hpw (high-intensity), all caregivers providing care for 20-168hpw would be grouped together. Despite providing over eight times as many hours of care, round-the-clock caregivers (168hpw), would be in the same classification as those providing 20 hpw. For this reason, it is logical to separate round-the-clock caregivers into a discrete category for comparison with high-intensity and low-intensity caregivers. In this thesis, intensity is determined by caregiving hpw and classified as low-intensity ( $\leq$ 19hpw) high-intensity ( $\geq$ 20hpw), or round-the-clock (168hpw).

The duration of the caregiving role considers the time spent as a caregiver, i.e. short-term or long-term (Phillips *et al.*, 2010), but it is also important to consider whether caregiving is continuous over this period or intermittent. Evidence suggests the average duration of a caregiving role is over four-years (Boling, 2010). However, duration of care can differ depending on recipient care needs

(Barnhart *et al.*, 2020; Fuino and Wagner, 2020). For example, care provision to individuals with developmental disabilities (DD), such as Down syndrome, is typically over a longer duration than to older adults (Anderson *et al.*, 2018).

#### 1.2.1.4 Caregiver burden and satisfaction

Early definitions considered burden to be negative consequences resulting from caregiving activities (Chou, 2000), but more recent definitions are more specific. As such, Chiao *et al.* (2015, p.341) define burden as "physical, psychological, emotional, behavioural and financial burden that may be experienced by informal caregivers"; although a flaw in this definition is that the term burden is used within. Alternatively, Lee *et al.* (2017a) suggest that burden is a multidimensional outcome of negative appraisal. Evidently, burden is a negative experience, which can be encountered in a variety of ways; it can affect the physical health, mental health, wellbeing, and social engagement of caregivers (*ibid*). However, it is important to note the criticisms of research into burden. Caregiving can be both emotionally demanding and rewarding (Dahlrup *et al.*, 2015), therefore it is plausible that most caregivers encounter a combination of negative and positive experiences. However, negative consequences of caregiving continue to be more extensively researched than positive ones (de Labra *et al.*, 2015; García-Mochón *et al.*, 2019).

Caregiver satisfaction may be perceived to contrast with burden, but this is not explicit (García-Mochón *et al.*, 2019), as caregivers can experience both feelings of burden and satisfaction simultaneously. Research is largely in agreement that satisfaction is derived intrinsically, for instance, strengthening relationships (Kietzman *et al.*, 2008), or personal growth (Ekwall and Hallberg, 2007); thus satisfaction could have beneficial effects on health, wellbeing, or loneliness. Additionally, evidence suggests that satisfaction has implications for both caregiver and recipient wellbeing (Kietzman *et al.*, 2008).

It is often speculated that increased time spent caregiving is linked to burden and satisfaction, as such caregiving  $\geq$ 5-hours per day (hpd) is linked to lower satisfaction among caregivers (Marcén and Molina, 2012). In contrast, interviews with adult-daughters providing care to their mothers (N=133) suggested that, in around half of the sample, caregiving was beneficial to their relationship, 57% of those attributed this to additional time spent together (Walker *et al.*, 1990). While these relationship improvements were correlated with satisfaction, the findings contradict the idea of higher-intensity caregiving equating to greater burden, as increased time resulted in increased satisfaction. This would suggest that burden is not only linked to caregiving intensity, but that there are other influential or mediating factors, such as the inter-personal relationship, or the nature of care.

#### 1.2.2 Loneliness

There are several definitions and interpretations of loneliness, ultimately emphasising a lack of satisfaction with one's relationships. Seminal definitions include that of Perlman and Peplau (1981,

#### Chapter 1

p.31), who delineate loneliness as an "unpleasant experience that occurs when a person's network of social relations is deficient in some important way, either quantitatively or qualitatively", and Weiss (1973, p.17), who articulates: "loneliness is not caused by being alone but by being without some definite needed relationship or set of relationships". Overall loneliness is a discrepancy between actual and desired relationships, in that networks are smaller (quantitatively) or less satisfying (qualitatively) than desired. Moreover, these early definitions are still widely cited in contemporary research (Barreto *et al.*, 2020; Groarke *et al.*, 2020; Hajek and König, 2020).

Loneliness is a multidimensional concept, often categorised into social and emotional loneliness (Weiss, 1973; Dahlberg and McKee, 2014; Tiilikainen and Seppanen, 2017). Simply, social loneliness reflects a lack of socially inclusive relationships, whereas emotional loneliness manifests as the absence of significant emotional attachments (Weiss, 1973). The Weiss distinction suggests that social and emotional loneliness are separate; however, it could be argued that there is an overlap. As such, unmarried, childless, individuals may develop significant attachments within friendship groups, which may also constitute inclusive relationships. Among older, unmarried, childless women, relationships with siblings, friends, and neighbours were stronger than those who were married or with children (Wenger *et al.*, 2000). Therefore, the Weiss distinction may not be a distinction, but more of an identification of two possible sources of loneliness. Attachment theory suggests that new attachments may be established after loss (Lang and Fowers, 2019). Therefore, as people gain and lose relationships throughout their lifecourse, emotional loneliness is likely to occur, until significant attachments are re-established or transferred.

The definition of loneliness adopted within this thesis is a discrepancy between desired and actual relationships (Perlman and Peplau, 1981, Barreto *et al.*, 2020). However, different dimensions, including absence of significant attachment (emotional) and lack of inclusive relationships (social) are also acknowledged, alongside the notion that one can be lonely without being alone (Weiss, 1973).

#### 1.2.2.1 Related terminlogy

Alongside loneliness, there are terms such as solitude, social isolation, and social exclusion. Often in literature, these are used synonymously with loneliness, but the definitions suggest that they represent individual constructs. Solitude refers to the experiences of being or feeling alone (Burger, 1995; Long *et al.*, 2003). Similarly, Galanaki (2004) defines solitude as a temporary state of voluntary, beneficial aloneness; in contrast to the aforementioned sources, this suggests solitude is exclusively positive. Nguyen *et al.* (2018) highlight that as solitude (the experience of being alone) is related to other psychological experiences, such as loneliness, there is little focus on solitude within research. Solitude is conceptualised as being alone without communication or other activities (*ibid*), therefore, is considered a deliberate action, rather than an emotional or psychological response. In contrast, social isolation is an objective measure of aloneness, typically a quantification of relationships, inferring that fewer relationships result in isolation (Jopling, 2015). Tilvis *et al.* (2012) expand on this by separating the size of networks from the frequency of interactions within. Moreover, de Koning *et al.* (2017) specify that to be socially isolated, contact with friends or family should be less frequent than once per week. As an objective measure, a standardised value for social isolation should be employed to ensure consistency when monitoring individuals, and to promote comparability for research purposes. However, Victor *et al.* (2003) criticised this approach of quantifying weekly interactions, identifying that by determining social isolation using a specific value within a certain timeframe (e.g. once per week), this infers that all interactions are equal, and a greater number is always preferable, which may be inaccurate.

Victor et al. (2003) consider that social isolation could predict loneliness. However, as indicated when discussing the definitions (section 1.2.2), loneliness is not always the result of being alone (Weiss, 1973). The evidence on the association between isolation and loneliness is inconsistent (Smith and Victor, 2019), thus it is important to consider the possibility of having contact with individuals and still experiencing loneliness. This highlights the incongruity between social isolation and loneliness, whereby to be isolated, an individual must have infrequent social interactions, or a small social network, whereas loneliness reflects individuals' psychological response to a discrepancy between the relationships they want and currently have. Social isolation, which could be equated to reduced social networks (de Jong Gierveld and van Tilburg, 2016) has been linked to loneliness (Hawkley et al., 2008; Domènech-Abella et al., 2017a). However, small social networks do not necessarily cause loneliness; fewer relationships are often indicative of isolation, but not always loneliness (de Jong Gierveld, 1998). This highlights that social isolation and loneliness are separate phenomena. Within this thesis, social isolation is not measured as an outcome; but, measures of social networks (section 4.3.7), which arguably reflect isolation by determining the quantity of individuals' networks (Tilvis et al., 2012; Jopling, 2015), are included within the conceptual framework (section 2.2) and analysed as a potential moderator between caregiving and loneliness.

Another concept is social exclusion. Similar to loneliness, social exclusion is subjective; Hajek and König (2017) differentiate between loneliness and social exclusion, indicating that the latter is a perception of not belonging within society. Social exclusion is also used to depict disadvantaged states experienced by marginalised groups, who perceive themselves to be removed from society (O'Donnell *et al.*, 2018). Exclusion is more commonly perceived than observed, i.e. individuals feel excluded from society, rather than being actively excluded by members of societal groups. Therefore, it could be considered that the perception of being left out could lead to loneliness, but the two are not automatically linked. For instance, if individuals feel excluded from one community, but accepted within another, this may alleviate feelings of loneliness, but would not reduce the exclusion from the initial community. Social exclusion involves detachment not only from relationships, but also institutions and groups (Hajek and König, 2017), additionally exclusion is influenced by health and resources (Kristensen *et al.*, 2019). This suggests that, for social exclusion, larger emphasis

would be placed on examining personal and community resources, rather than a responsibility such as caregiving. Overall, a larger body of evidence links caregiving to loneliness than to social exclusion, and one of the main aims of this thesis (section 1.1) is to determine the health and wellbeing consequences of loneliness among caregivers. For this reason, social exclusion is not investigated further within this thesis. Nevertheless, defining these key terms is essential prior to discussing the current literature, to enhance the clarity and comparability of the results from this thesis with other studies.

Identified within this section, the terms social isolation, solitude, and social exclusion, despite frequently being discussed in conjunction with loneliness, are all distinct. For the purpose of this thesis, loneliness is the focus. However, social isolation is included in the analyses as a covariate, determined using measures of familial and friendship networks. Given the circumstances of caregiving, caregivers' networks may reduce due to caregiving responsibilities (Vasileiou *et al.*, 2017), but caregivers are unlikely to be completely isolated due to connection with the recipient. For this reason, loneliness was considered more appropriate to research, while considering the effects of a reduced network. Additionally, there is a higher quantity of evidence supporting the link between social isolation and health, than between loneliness and health (Blazer, 2020), which emphasises a need for further research on loneliness, in relation to health and wellbeing.

#### 1.2.2.2 Measuring loneliness: loneliness scales

The most common methods to measure loneliness are: single-item direct questions, the University of California Loneliness Scale (UCLA-LS), and de Jong Gierveld Loneliness Scale (DJGLS) (Campaign to End Loneliness, no date; ONS, 2018b). Single-item questions ask directly about loneliness, typically including the word 'lonely'; for example: 'how often do you feel lonely?' (ONS, 2018b). One of the main disadvantages of direct questions is the social stigma attached to loneliness, making individuals less likely to report feeling lonely (Shiovitz-Ezra and Ayalon, 2012). Indirect scale measures are an alternative approach, the questions do not include 'lonely' or 'loneliness', therefore are less likely to result in under-reporting (*ibid*).

The 20-item UCLA-LS contains positively (N=10) and negatively (N=10) worded items (Russell *et al.*, 1980). Despite this, the UCLA-LS is characterised as a global scale, not as two separate scales (Dodeen, 2015). There are several short-form versions, such as 3-item (Campaign to End Loneliness, no date), 4-item (Hays and DiMatteo, 1987), 6-item (Neto, 2014) or 8-item (Wu and Yao, 2008). However, the items used in each scale are not always reported, nor are they consistent. As such, the numbering of items in the 4-item scale reported by Hays and DiMatteo *et al.*, (1987) did not correspond with the numbering on the original 20-item UCLA-LS (Russell *et al.*, 1978). To illustrate, item 15 according to Hays and DiMatteo (1987) related to companionship, whereas originally this was reported as "No one really knows me well" (Russell *et al.*, 1978, p.292). The items in the 3-item UCLA-LS ask about companionship, isolation and feeling left out, and responses include 'hardly

ever or never', 'some of the time', or 'often', scored as 1-3 respectively; the overall loneliness score is determined from the sum (ONS, 2018b).

In contrast, the 11-item version (DJGLS-11) of the DJGLS has been described as the most utilised loneliness instrument across Europe (Tomás *et al.*, 2017). Moreover, unlike the UCLA-LS, which emphasises social loneliness, DJGLS-11 incorporates the Weiss distinction (Penning *et al.*, 2014). Therefore, the DJGLS-11 can be used as an overall measure of loneliness, but also as individual, social (N=5) and emotional (N=6) loneliness subscales (de Jong Gierveld and van Tilburg, 2010). The 6-item version (DJGLS-6) was developed specifically for large surveys (*ibid*). Although most commonly used for older adults (Penning *et al.*, 2014), the DJGLS-6 is appropriate for all adults aged 18-99 (Leung *et al.*, 2008). As with the DJGLS-11, the DJGLS-6 can also be used as an overall loneliness measure, or two shorter social (N=3) and emotional (N=3) subscales (de Jong Gierveld and van Tilburg, 2006). Emotional loneliness indicators ask about emptiness, having people around and feelings of rejection, whereas the social indicators include relying on, trusting, and feeling close to others (*ibid*); responses include 'yes', 'more or less', and 'no'.

In this thesis, initially, a direct loneliness question is compared with the UCLA-LS to identify any differences in reporting loneliness among caregivers and non-caregivers. However, for the majority of the analyses, the UCLA-LS is the primary loneliness measure. This is because firstly, shorter versions of scales are considered more appropriate in large surveys (ONS, 2018a), therefore, in large surveys such as ELSA, the UCLA-LS 3-item would be favoured over the DJGLS-6. Secondly, the UCLA-LS is well-established (*ibid*), widely used among older adults, and applicable in both questionnaire and interview scenarios (Campaign to End Loneliness, no date).

#### **1.2.3** Health and wellbeing

Health and wellbeing are distinct but connected; health influences wellbeing, but health is also impacted by wellbeing (Department of Health, 2014). Therefore, it is difficult to distinguish between the two and often, in research, they are combined without clear distinction between the concepts. Nevertheless, it is important to recognise the similarities and differences between health and wellbeing, and to identify clear definitions to guide the interpretation of the findings. Previous research on wellbeing is founded on descriptions of wellbeing, as opposed to a coherent definition (Dodge *et al.*, 2012); this diverse approach has fostered ambiguity and broad definitions (Forgeard *et al.*, 2011). The following sections discuss common definitions for both health and wellbeing, identifying which are the most relevant for this thesis. To aid clarity, key indicators, quality of life (QoL) and life satisfaction (which are often used interchangeably with wellbeing), are also defined.

#### 1.2.3.1 Health

The World Health Organisation (WHO) published a definition of health in 1948, which is still widely used: "a state of complete physical, social and mental wellbeing, and not merely the absence of

#### Chapter 1

disease or infirmity" (WHO, 1998, p.1; McCartney *et al.*, 2019, p.23). Although WHO (1998) contended that the most comprehensive understanding of health involves collective health and wellbeing, the lack of clarity surrounding the constitution of 'social wellbeing' (as opposed to physical or mental wellbeing) contradicts this. Alternatively, Card (2017) proposed a definition whereby health was not defined by presence of disease or disability, but considered on a spectrum from more to less healthy, determined by physical and psychological wellbeing. However, this definition is still largely weighted on wellbeing. In contrast, health status, a measurement of individual or population health compared to specific criteria (WHO, 1998), may be a more appropriate perspective. Health status can be determined by using either professional diagnoses or self-reported measures (Ngamaba *et al.*, 2017). A significant positive association was observed between health status and subjective wellbeing, highlighting that health and wellbeing are distinct components, but inevitably influence one another (*ibid*).

The medical model of health – health as absence of disease – is an objective measure where an individual must meet specific parameters (Victor, 2005). Similarly, health as functional normality, as proposed by Tengland (2007), focuses on areas of limited function, such as disabilities. These perspectives are particularly reductionist, considering solely medical diagnoses, irrespective of how individuals feel. For example, people can be healthy, whilst living with disability or chronic conditions (McCartney et al., 2019). In comparison to the medical model and WHO definition, the sociological perspective adopts a more person-centred approach (Victor, 2010). The sociological perspective – health as absence of illness – allows an individual to determine their physical and mental health status (Victor, 2005; 2010). Alongside perceived health status, the presence of any pain or self-reported disabilities are acknowledged (Victor, 2005). Although this method is entirely subjective, it may be a more accurate way of determining an individual's health. To clarify, the individual can offer a conclusive overall health status, incorporating specific conditions that they have been diagnosed with, but also any acute or undiagnosed problems that may be directly affecting their health. One limitation of this definition is the subjective nature; self-rated health status may fluctuate significantly depending on personal circumstances and may not necessarily follow an expected pattern.

The definition of health, for the purpose of this thesis, is a combined approach, considering both the medical model (absence of disease, e.g. diagnosed conditions) and sociological perspective (absence of illness, e.g. self-rated health and the individuals' perceptions of health). Furthermore, a key component to defining health is the recognition that health is multidimensional (McCartney *et al.*, 2019), this will be further applied within the thesis by including both physical (e.g. reported illnesses) and mental health measures (e.g. depressive symptomatology). The thesis also measures health expectations, behavioural health, and physiological health conditions by incorporating specific health pathways derived from theory (chapter 2).

#### 1.2.3.2 Wellbeing

Wellbeing is a broad concept: "wellbeing encapsulates how we are faring, in all domains of life, including financial, health, social, personal and the local environment" (Green *et al.*, 2017, p.3). Alternatively, national wellbeing includes individuals' perceptions of their wellbeing, external factors which impact wellbeing (e.g. health or education), and contextual influences, such as the economy (Beaumont, 2011). These two perceptions incorporate an extensive range of factors, and the lack of clarity has resulted in a disorganised approach to measuring wellbeing. As such, researchers use contrasting approaches to assess wellbeing (Forgeard *et al.*, 2011). This ultimately impacts comparability of research and limits the understanding of the concept.

The two main domains are subjective and psychological wellbeing; but, similarly to the definitions of overall wellbeing, there are several interpretations for both. Subjective wellbeing is expressed as a balance between positive and negative emotions (Dolan *et al.*, 2011). However, that is a simplistic definition; Diener *et al.* (1999) define subjective wellbeing differently, advocating three components: pleasant affect (e.g. happiness), unpleasant affect (e.g. stress) and life satisfaction. Moreover, subjective wellbeing reportedly contains hedonic (feeling good) and eudaimonic (functioning well) aspects (Chanfreau *et al.*, 2013). Correspondingly, psychological wellbeing is also considered to compromise hedonic and eudaimonic perspectives (Winefield *et al.*, 2012). Therefore some ambiguity surrounds the contrast between subjective and psychological wellbeing, although Linley *et al.* (2009) argue that subjective wellbeing is hedonic, whereas psychological wellbeing is eudaimonic.

An alternative, and arguably clearer, distinction of psychological wellbeing is provided in Ryff's sixdimension wellbeing model (Ryff, 1989). The dimensions are: self-acceptance, personal growth, purpose in life, positive relations with others, environmental mastery, and autonomy (*ibid*). In a more recent evaluation of the model, correlations were identified between specific dimensions (e.g. purpose in life) and conditions such as dementia, highlighting the link between health and wellbeing (Ryff, 2014). Moreover, although 'positive relations with others' could be compared to social loneliness, arguably, the perspectives differ significantly. To clarify, loneliness focuses on a lack of relationships (negative outlook), whereas the eudaimonic perspective of psychological wellbeing adopts a positive approach. Therefore, social loneliness and 'positive relations with others' are contrasting perceptions on a similar experience, and reasonably, this may make them compatible with one another.

Finally, QoL is frequently used synonymously with wellbeing. However QoL reflects the degree to which an individual feels their needs are met and, if they have adequate opportunities to achieve fulfilment (WHO, 1998). This definition highlights that QoL is closely linked to areas of psychological wellbeing (e.g. self-acceptance), identified in the psychological wellbeing model (Ryff, 1989). In contrast, life satisfaction is an aspect of subjective wellbeing linked closely to

#### Chapter 1

happiness (Diener *et al.*, 1985). Life satisfaction is a cognitive process involving an overall judgement about one's life. However, the criteria are vague and rely on individuals to compare their lives to an ideal standard (*ibid*). Within this thesis, the division of subjective and psychological wellbeing is adopted, utilising measures of QoL and life satisfaction to determine wellbeing within the sample. See section 4.3.4 for further discussion on QoL and life satisfaction scales.

#### **1.3** Thesis structure

The first chapter has introduced the main concepts relevant to the thesis, and clarified definitions. The discussion on loneliness also introduced key measurement tools and details of the UCLA-LS used within this thesis. Moreover, the aims, objectives and research questions were explicated.

#### Chapter two: theoretical understanding

In the second chapter, key theories are discussed in relation to caregiving motivation, and the impact of caregiving. Furthermore, the theoretical, and conceptual knowledge from chapters one and two are combined to formulate a conceptual framework for this thesis.

#### Chapter three: literature review

This chapter is divided into sections discussing the determinants of loneliness, caregiving literature, and loneliness in relation to health and wellbeing, for both the general population and within a caregiving context. Determinants of loneliness are addressed with consideration to the demographics of this thesis' sample (aged  $\geq$ 50), looking particularly at studies that examine middle-aged or older adults. Caregiving-specific literature examines characteristics of caregivers, the caregiving relationship, the caregiving role, and burden and satisfaction. The final literature review section recognises that the relationship between loneliness, health, and wellbeing is multidimensional, therefore literature is examined on the associations in multiple directions, and within the caregiving context.

#### Chapter four: methodology

The methodology outlines the research strategy, including cross-sectional and longitudinal research design and an overview of the research materials, which are the ELSA dataset. The chapter outlines the key variables from ELSA that are analysed within the thesis, and variables computed for the follow-up analysis. Finally, the statistical techniques applied within the thesis are outlined.

#### Chapter five: cross-sectional results

Chapter five reports on the cross-sectional results, carried out on ELSA wave eight. This chapter addresses research questions one and two, as outlined previously (section 1.1.1). The methods applied are primarily binary logistic regressions, with additional multinomial logistic regressions and
linear regressions where appropriate. Initially, caregivers are compared with non-caregivers, using variables relevant to both groups; subsequently, research question two builds on this by adding caregiving-specific variables.

# Chapter six: twelve year follow-up

Follow-up analyses are conducted on a sample of respondents who provide data from waves two to eight of ELSA, the results of which are reported in chapter six. Binary logistic regressions are used to determine if caregiving at wave two, caregiving in any wave, or changes to the caregiving relationship and role between waves two and eight, are linked to loneliness.

# Chapter seven: discussion

The final chapter elaborates on the interpretations of each research question, in relation to the theoretical and conceptual framework, and published evidence. The limitations of the thesis are also outlined, and to conclude, a statement of original contribution is made alongside recommendations for policy and further research.

# **Chapter 2** Theoretical understanding

The application of Gerontological theory connects current findings to previous observations, and encourages new perspectives (Biggs *et al.*, 2003). Theoretical and conceptual frameworks in Gerontology provide structure for interpreting new ideas, which can prevent ageism within research, aid policy development, and enhance practical applications of the findings (Hasworth and Cannon, 2015). In social Gerontology, and more specifically family Gerontology, a division of social Gerontology focused on ageing families (Humble *et al.*, 2020), it was identified that stress and coping theories were commonly applied in research over the past few decades (1990-2018) (Roberto *et al.*, 2006; Alley *et al.*, 2010; Humble *et al.*, 2020). These theories are relevant to this thesis on loneliness among caregivers because, predominantly, caregivers within these analyses are providing care to a family member, thus representing family Gerontology. The following sections discuss caregiving-relevant theories, alongside the key concepts (section 1.2), will formulate a suitable conceptual framework for this thesis (section 2.2).

## 2.1.1 Caregiving theories

It is important to discuss caregiving-specific theory because, as outlined previously, theories are used to interpret research findings (Biggs *et al.*, 2003). Therefore, to develop and address caregiver-focused research questions, a deeper understanding of current knowledge of the caregiving role is beneficial. As such, historical evidence suggests caregiving may be an evolutionary characteristic, due to its long-standing and multi-cultural nature (Lang and Fowers, 2019).

The lifecourse perspective is fundamental to social theories of ageing, assuming occurrences throughout the lifecourse have a cumulative effect (Hasworth and Cannon, 2015). The lifecourse has multiple pathways that involve experiences, lifelong development, and sociocultural contexts (Elder Jr., 1994; Crosnoe and Elder Jr., 2015). The lifecourse can be analysed using either a two-stage or a three-stage model, the latter separates the lifecourse into childhood, adulthood, and older age, whereas two-stage models contrast older age with the rest of the lifecourse (Vincent *et al.*, 2006). A fundamental aspect of the lifecourse perspective is linked lives (Elder Jr., 1994; Crosnoe and Elder Jr., 2015). Linked lives are direct and indirect effects of resources across multiple generations within a family (Gilligan *et al.*, 2018); resources are not limited to tangible items such as money but can also include knowledge and social networks. As the nature of change throughout one's life can vary considerably, lifecourse perspective encompasses several factors, for example biological, developmental, or geographic. Therefore, it is applicable to a diverse range of research topics (Hendricks, 2012).

The less diverse, more caregiving-specific, theories tend to focus on understanding the motivation behind caregiving, or the impact of caregiving (Lang and Fowers, 2019). Theoretical explanations,

such as the aforementioned linked lives or other exchange theories (e.g. reciprocity), are common in research examining why caregivers provide care; whereas the popular stress, coping and appraisal theories are more relevant for caregiving impact (section 2.1.1.2).

While theoretical frameworks on caregiving motivation are important for understanding caregiving as a concept, they are less relevant to the work within this thesis, which is largely interested in the outcomes of caregiving. That is exploring the 'how', not the 'why'. Motivation to care will not be analysed within this thesis; the intention of this thesis is to consider the impact of caregiving on loneliness, health and wellbeing, thus the following section highlights relevant theories on the impact of caregiving.

# 2.1.1.1 How does caregiving impact the caregiver?

Stress, coping and appraisal theories are prevalent in underpinning research into the impact of caregiving (Gérain and Zech, 2019). Caregiving appraisal, defined as the cognitive and emotional evaluation of stressors and coping (Wang *et al.*, 2020a), was introduced to expand the understanding of caregiver burden, by establishing that not all caregiving appraisal is negative (Lawton *et al.*, 1989). Moreover, appraisal is multifaceted and represents satisfaction, impact, mastery, ideology, and burden (*ibid*). Despite this conceptualisation, burden is still frequently used synonymously with appraisal. Therefore, in response to these inconsistencies, Wang *et al.* (2020a) proposed three levels of a Social Ecological Model (SEM): individual (caregiver and recipient factors), interpersonal (family function and support), and community (social support or pressure), which arguably provides a clearer approach to understanding appraisal. A similar approach was adopted by Gérain and Zech (2019), through the development of the Informal Caregiving Integrative Model (ICIM). Similar to SEM, ICIM includes factors inside and outside of the caregiving context and considers appraisal to be a key component. However, ICIM largely reflects the negative bias of appraisal, by focusing on burnout (*ibid*). That said, the structure of ICIM is more detailed than SEM, including appraisal and relationships as mediators of burnout.

Alternatively, Caregiver Identity Theory (CIT) depicts the dynamic process of adaptations throughout the caregiving role, for example, accommodating the recipient's changing health status (Montgomery *et al.*, 2016). To clarify, adaptations may include increasing caregiving hours, or new, caregiving tasks. Caregiver identity can refer simply to whether individuals consider themselves caregivers, but also the extent to which they associate with the role. For example, although caregiving tasks are recognised, many people may not identify as caregivers, due to stigma surrounding the terms 'carer' or 'caregiver' (Hughes *et al.*, 2013). Montgomery and Kosloski (2009) detail the identity change process using the example of a spousal caregiver (Figure 1); however, the process is not exhaustive, individuals may not experience all phases, or phases may occur in different orders (Montgomery *et al.*, 2016).



Figure 1: Caregiver Identity Theory: phases of caregiver identity change Source: author's adaptation from Montgomery and Kosloski, 2009, p.51; Montgomery et al., 2016, p.454

Transitions between phases may relate to changes in both the caregiving context and caregiver identity (Montgomery, 2007; Montgomery and Kosloski, 2009). Using the example of a spousal dementia caregiver, identity may change as the dementia progresses. In phase one, caregiving tasks are less noticeable, such as assisting with shopping. Often, caregivers may not identify as a caregiver until phase two, when tasks surpass typical family roles (Montgomery *et al.*, 2016). Phase three commonly involves the introduction of personal care, as dementia severity is likely to have increased: caregiving now forms a substantial part of the relationship. Within the fourth phase, caregiving is dominant as care needs have considerably increased, and here, the caregiver often identifies with the caregiver role more than as a spouse. The fifth phase (repeat phase two) may reflect a transition to formal care, or institutionalisation, which often occurs in advanced stages of dementia, but informal care roles frequently continue at reduced intensities.

CIT outlines the process of adaptations that a caregiver may experience, highlighting how identity and caregiving behaviours can change throughout the role. This is particularly relevant when researching loneliness among caregivers; for example, identity change from spouse to caregiver could contribute to reduced relationship quality, which is known to influence loneliness (Hsieh and Hawkley, 2018; Leszko *et al.*, 2020). However, aspects of SEM and ICIM are also relevant to this thesis, highlighting the importance of factors both within, and external to, the caregiving role, and the mediating impact of appraisal.

#### 2.1.2 Stereotype embodiment theory

Stereotype embodiment theory (SET) demonstrates that as people age, age-stereotypes become increasingly relevant (Kornadt, 2016). Although age-stereotypes can be positive, they are predominantly negative, for example, considering physical decline a normal part of ageing. Internalised negative age-stereotypes result in self-fulfilling prophecies, such as frailty or memory-loss (Levy, 2009; Meisner and Levy, 2016).

Perceptions of ageing are determined through personal experience and societal influences. Moreover, often the most vulnerable to negative age-stereotypes are children, because of limited exposure to older adults, or older adults being portrayed as frail in the media (Levy, 2009). A large portion of

societal influence is drawn from culture; the UK was found to have among the strongest negative age-stereotypes, whereas China demonstrated stronger neutral and positive age-stereotypes (Meisner and Levy, 2016). Negative age-stereotypes are linked to poorer health outcomes in older adults (Levy *et al.*, 2000; Levy and Myers, 2004; Levy *et al.*, 2009). However, negative age-stereotypes can also influence whether individuals obtain medical advice or treatment; older adults who consider pain and illness as a typical age-related component may not seek medical attention, assigning their symptoms to age as opposed to illness or injury (Hasworth and Cannon, 2015).

Although SET appears to relate predominantly to individuals' expectations, SET operates through multiple pathways: psychological, behavioural and physiological (Levy, 2009; Meisner and Levy, 2016). The psychological pathway relates to expectations; negative age-stereotypes are linked to lower self-efficacy, whereas positive age-stereotypes are linked to higher self-efficacy (Meisner and Levy, 2016). An earlier study demonstrated the effect of age-stereotype priming on memory in individuals from the United States (US), aged 60-90 (Levy, 1996). Individuals were exposed to twelve negative (e.g. senile) or twelve positive (e.g. wise) words prior to completing memory tasks. Those exposed to positive age-stereotypes performed significantly better than the negative terminology group.

The second SET pathway is behavioural, related to healthy practices (Meisner and Levy, 2016). As most individuals consider health decline as unavoidable (Levy, 2009), engagement in healthy practices is often limited. Individuals with positive age-perceptions were found to engage in more preventative health behaviours (e.g. regular exercise), than those with negative perceptions (Levy and Myers, 2004).

Thirdly, the physiological pathway is connected to the autonomic nervous system (Levy, 2009), specifically stress and CV activity (Meisner and Levy, 2016). Negative age-stereotypes are associated with increased CV response to stress, which increases likelihood of CV events (Levy *et al.*, 2000). Negative age-stereotypes can also inhibit recovery from acute CV events (Levy *et al.*, 2006). Comparison of cardiac recovery patients (mean age 71) showed that physical performance of the positive age-stereotype group was significantly greater than the negative group at follow-up.

Within this thesis expectations for both future loneliness and health are analysed, thus SET is highly relevant. The psychological pathway aids the interpretation of the results and connects the findings from this thesis with current understanding on expectations. Additionally, SET is applicable to other areas within this thesis, as by mapping out behavioural, physiological, and psychological pathways, this gives structure to how the health and wellbeing consequences of loneliness are examined among caregivers and non-caregivers.

# 2.2 Theoretical and conceptual framework

The main theories underpinning this thesis are CIT and SET. CIT demonstrates that caregiving is adaptive and shows transitions between the pre-caregiving relationship and the caregiver-recipient relationship. The phases represent how the caregiver identifies with the role but could also be attributed to evolving care needs. While changes in identity are not explicitly measured in this thesis, the understanding that perhaps more complex care needs (higher-intensity roles) could reflect less of a connection to the pre-caregiving relationship, would highlight potential connections to loneliness. As suggested by Weiss' (1973) social and emotional loneliness, identifying more as a caregiver than, for example a spouse, caregivers may feel the significant emotional attachment is lost, or weaker, or the relationship is less inclusive because they are responsible for more of the marital tasks (e.g. finances) which may have previously been shared. Furthermore, to provide a more structured approach to the health aspect of the thesis, a health-related theory was deemed important. SET identifies three pathways of health, which implements a specific structure for interpreting health impacts. Providing evidence on the association between caregiving, loneliness and, for instance, a self-rated general health measure would identify if caregiving and loneliness were broadly connected to health. However, there would be limited detail into which particular aspects of health were most affected. A more specific approach, through the application of theoretical knowledge, can provide more comprehensive results, and detailed policy recommendations.

Additional theories (caregiving appraisal, SEM and ICIM) have guided the formulation of the conceptual framework. SEM and ICIM identify the importance of factors inside, and outside of the caregiving role, which have been integrated into the proposed framework. Moreover, certain aspects of appraisal are relevant to this research. This thesis focuses on the impact of caregiving on loneliness, health, and wellbeing, so satisfaction and burden (shown to be linked to health and wellbeing, section 3.3.4), are likely to play integral roles in the observed relationships.



Figure 2: Conceptual framework Source: author's own

The conceptual framework (Figure 2) depicts a consolidation of the aforementioned theories and concepts, whereby factors related to caregiver characteristics, the caregiving relationship and role shape the caregiver, and in turn, contribute to caregiving outcomes. While there is likely to be an overlap between the determinants of loneliness and the characteristics of caregivers (e.g. sociodemographic characteristics), this is not exclusive. For example, there is contrasting evidence on the association between sex and loneliness (section 3.2.1), and sex and caregiving (section 3.3.1.2) therefore, sex may be a determinant of both caregiving and loneliness, either caregiving or loneliness, or neither caregiving nor loneliness. Moreover, factors such as reduced social networks of caregivers, identification with the role (CIT), and appraisal are likely to moderate the impact of caregiving on loneliness. The pathways to health, as identified by SET, also form an integral part of the framework, encouraging investigation into specific health outcomes to form more precise policy recommendations.

# 2.3 Chapter summary

Chapter two has discussed relevant caregiving theories and introduced the conceptual framework. One of the main aims of this thesis is to identify the impact of caregiving upon loneliness, health and wellbeing, thus theories on motivation to care are less applicable. Moreover, the long-term effects of caregiving are considered, including changes to the caregiving role, therefore CIT aids the understanding of how caregivers adapt to changing caregiving circumstances. Finally, the application of SET allows a structured interpretation of health consequences, to promote more specific health policy recommendations. The next chapter critically discusses current literature in relation to the key concepts caregiving, loneliness, health and wellbeing. Specifically, the literature review addresses the determinants of loneliness, caregiving-specific literature, and loneliness, health and wellbeing in the context of both the general population and amongst caregivers.

# **Chapter 3** Literature review

# **3.1** Chapter outline

This chapter critically discusses relevant literature on the determinants of loneliness (section 3.2), considering which predictors may be the most relevant to a caregiving population. Moreover, caregiving-specific literature (section 3.3), in addition to literature on burden and satisfaction, will be discussed. Finally, the relationship between loneliness, health and wellbeing is known to be complex, whereby bi-directional associations have been reported. Section 3.4 disentangles this literature and discusses evidence in a caregiving context. The summary of the literature is presented in section 3.5, where fundamental research gaps are outlined, clearly addressing how this thesis augments the current knowledge on caregiving and loneliness.

The literature search was conducted using several databases including Elsevier's ScienceDirect® and Scopus®, Clarivate analytics' Web of Science, EBSCOhost's APA PsycINFO, and DelphiS which is specific to the University of Southampton (also powered by EBSCOhost). Searches were conducted using key terms, and combinations of key terms using Boolean operators such as 'caregiving AND loneliness', and 'caregiving AND (health OR wellbeing)'. Initially articles from the search results were excluded based on the relevance of their titles and abstracts, and any duplicates were removed before articles were fully-reviewed. Supplementary literature was sourced from the reference lists of reviewed articles and additional searches to provide evidence for specific definitions, policy documents or publications from third sector organisations. Finally, additional literature searches were conducted towards the end of writing up the thesis to obtain any new evidence published since the initial literature review. These searches involved the aforementioned databases alongside the use of Google Scholar to locate specific articles.

# **3.2 Determinants of loneliness**

A determinant of loneliness is a factor deemed to have an influential connection to loneliness. De Jong Gierveld (1998) suggests that the exact identification of loneliness determinants is unattainable due to their extensive and interdependent nature. As such, it would be inaccurate to assume that determinants act separately; for example, health may have direct effects on loneliness, but also indirect, by affecting social activity, capacity to work, and wellbeing, which in turn could impact loneliness. Victor *et al.* (2020) identifies and addresses a key research gap, highlighting that evidence on prevalence and predictors of caregiver loneliness is largely inadequate. Few studies have addressed loneliness among caregivers, but focus instead on outcomes, rather than understanding the determinants. Typically, studies investigating the determinants of loneliness in the general population identify similar factors, including sociodemographic and socioeconomic characteristics,

social networks and resources, and aspects of physical and mental health (Honigh-de Vlaming *et al.*, 2014; de Jong Gierveld *et al.*, 2015; Hajek and König, 2020). Less commonly, neighbourhood factors such as public transport use have been linked to loneliness (Kemperman *et al.*, 2019). Moreover, Victor *et al.* (2020) investigated the determinants of loneliness among dementia caregivers, the most influential factors were: social isolation and networks, living arrangements, depression, life satisfaction and QoL.

The forthcoming sections discuss literature on sex (section 3.2.1), relationships (section 3.2.2), social networks (section 3.2.3), and SES (section 3.2.4) as determinants of loneliness. These factors were selected due to their relevance for older caregiving populations. As such, a larger proportion of caregiving roles are fulfilled by females (Rutherford and Bowes, 2014), except among older adults (aged  $\geq$ 65) whereby England and Wales 2011 Census data revealed a higher percentage of older male caregivers, compared to older females (Milligan and Morbey, 2016). However, due to the gendered nature of the caregiving role, whether actual or stereotypical, it could be expected that sex should play a large role in the loneliness of caregivers, as sex is well-reported as a predictor of loneliness in the general population. Moreover, evidence suggests that caregiving is linked to a reduction in social activities; caregiving responsibilities often restrict caregivers' availability for social interaction, impacting relationships and social networks, which, subsequently, lead to loneliness (Vasileiou *et al.*, 2017). Care provision is also known to impact employment, due to demands of the caregiving role, often caregivers cease paid work, or substantially reduce working hours (Ciccarelli and van Soest, 2018). Other known associations such as caregiving burden, satisfaction, health, and wellbeing are addressed subsequently (sections 3.3.4 and 3.4.2).

## **3.2.1** Are females lonelier?

It is commonly conceived that females are lonelier, or more likely to report loneliness, than males (Pinquart and Sörensen, 2001; Nicolaisen and Thorsen, 2014a). However, some evidence argues being female does not predispose an individual to loneliness (Borys and Perlman, 1985; Barreto *et al.*, 2020), therefore inconsistencies within the literature must be addressed. Females are more likely to experience widowhood and undertake a caregiving role, compared to men (Pinquart and Sörensen, 2001), both of which are linked to loneliness. Therefore, it is also important to consider other contextual or sociodemographic factors that may mediate the significant relationship between sex and loneliness. When Norwegian middle-aged individuals, aged 40-59 (N=1,882), were compared with older individuals, aged 60-80 (N=1,181), sex was not significantly associated with loneliness in the middle-aged group. However, females were significantly more likely to report loneliness in the older group (Nicolaisen and Thorsen, 2014a). As sex was not significant across both age groups, this highlights that the relationship between sex and loneliness is likely influenced by other characteristics. Using a lifecourse perspective, childhood events were used to predict loneliness; significant differences were observed across age groups and between sexes. This further

emphasises the importance of contextual factors in relation to sex differences, but also suggests that age may mediate the relationship between sex and loneliness.

The literature provides limited evidence to definitively conclude that females are lonelier than males. Often, sources cited for higher prevalence of loneliness among females are quite dated, typically using data from the 1980/90s. More contemporary evidence advocates inconsistent sex differences. A meta-analytical review found that the magnitude of sex differences was smaller in more recent literature (Pinquart and Sörensen, 2011); publications ranged from 1948-1999, although sex differences were still present in the 'recent' studies (1995-1999), a declining trend was evident. This suggests that contemporary literature (>2000s) comparing male and female loneliness should observe fewer sex differences, however evidence remains inconsistent. From early 2000s data, in a sample of 999 individuals aged  $\geq 65$  (53% female), females were more likely to be lonely than males; however, after adjusting for age, marital status and household composition, sex was not significantly associated with loneliness (Victor et al., 2005). Loneliness was more stable among males, 75% reported no change in loneliness over the past decade, compared to 60% of females (*ibid*). However, this was not a longitudinal comparison, only based on individuals' recall of the 10-year period, therefore subject to recall bias. Furthermore, 2006 data on UK individuals (N=2,393), indicated females were more likely to report loneliness across all ages (15-97 years), although the magnitude varied dependent on age and loneliness frequency (always versus sometimes) (Victor and Yang, 2012). This evidence also supports the hypothesis that factors such as age may mediate the relationship between sex and loneliness.

In contrast, recent evidence from the British Broadcasting Corporation (BBC) Loneliness Experiment found that males were lonelier than females across all ages (N=46,054, aged 16-99), but the effect of sex was weaker among older ages (Barreto *et al.*, 2020). This finding contradicts most other evidence, for example, 2016/17 ONS data aligned with aforementioned research suggesting a higher prevalence of female loneliness (Pyle and Evans, 2018). It is argued by Barreto *et al.* (2020) that the context of an online survey, as with the BBC experiment, is likely to reduce perceived stigma, therefore males may feel more able to report loneliness. However, if this were the case, whereby reduced stigma contributed towards a male majority, this should also have been evident in ONS data (Pyle and Evans, 2018), as the Community Life Survey (CLS) 2016/17 was administered in online and paper formats only (DCMS, 2017).

Alongside contextual factors, sex differences may also be attributable to the method of measuring loneliness. Often, studies that employ indirect scale measures report no significant sex differences, alternatively, studies using direct loneliness measures generally report higher female prevalence (Heinrich and Gullone, 2006; Pinquart and Sörensen, 2011). Similarly, Nicolaisen and Thorsen (2014a) found that females (aged 60-80) were more likely to report loneliness, than age-matched males using a single-item measure of loneliness; thus this may have contributed towards the observed

sex differences. Overall, evidence suggests that the prevalence of loneliness may not necessarily be higher among females, but that females may face fewer barriers to directly report loneliness.

In conclusion, the literature is inconsistent in reporting sex differences in loneliness. Loneliness appears to be mediated by several factors, including the measurement tool (direct *versus* indirect), or contextual factors (e.g. age). Overall, females appear more likely to report loneliness, therefore it could be expected that, when using single-item measures, sex differences may be observed. Arguably, methods which do not require interviews or directly-reported loneliness, challenge the stigma of loneliness, thus using self-completion questionnaires and scale measures may give a more accurate depiction of loneliness prevalence. Finally, despite the indication that females were more likely to be lonely and to undertake caregiving roles, there was little discussion in the reviewed literature comparing loneliness between male and female caregivers. Victor *et al.*, (2020) attempted to address this by comparing loneliness predictors among dementia caregivers, but no significant association was found between sex and loneliness. Those findings may be limited to dementia caregivers, however, the lack of evidence among alternative caregiving samples (e.g. non-dementia), and the inconsistency in the general population literature, makes it difficult to hypothesise whether sex differences would be observed among other caregiving groups.

### **3.2.2** Relationships as a protective factor

Conceptually, relationships cover a broad spectrum and can be measured through marital status, living arrangements, or factors such as relationship quality. Marital status is frequently reported as an important predictor of loneliness (de Jong Gierveld *et al.*, 2015; Hajek and König, 2020) with emphasis on cohabitation or living arrangements among caregivers (Victor *et al.*, 2020). As definitions of loneliness frequently refer to both qualitative and quantitative aspects of relationships (section 1.2.2), it is likely that different aspects of relationships are influential on loneliness.

#### 3.2.2.1 Marital status

Marriage is considered a protective factor against loneliness (Hsieh and Hawkley, 2016; 2018), although Dafoe and Colella (2016) argue that simply being married does not prevent loneliness, marital quality and satisfaction are highly influential. Moreover, using data from the US-based National Social Life, Health and Ageing Project (NSHAP), higher-quality marriages were indicative of lower loneliness (Warner and Kelley-Moore, 2012). Additionally, previous marriage was positively associated with loneliness, suggesting that those married multiple times were more likely to be lonely. This might be interpreted that longer (first) marriages equate to higher-quality relationships, and therefore less loneliness. However, the majority of the respondents were in long-term first marriages (average 44-years), which therefore suggests that the quality of the marriage is more influential than the duration on loneliness, but that longer marriages do not necessarily reflect higher-quality.

In studies that compare marital status, never married, divorced, and widowed respondents are frequently grouped as unmarried (Victor and Bowling, 2012; Dahlberg *et al.*, 2018b), which makes it particularly difficult to differentiate between the effects of widowhood and divorce on loneliness. The experience of being single is different to being widowed, both of which are contrasting experiences to divorce. It is unlikely that, collectively, these respondents would experience loneliness in the same way. Some studies focus particularly on widowhood, for example, Cavallero *et al.* (2007) compare widowed and married Italian older adults (N=956), reporting that married individuals were significantly less lonely than widowed. No divorced or never married respondents were included, so further comparison cannot be made. Alternatively, Ben-Zur (2012) contrasted married, divorced, and widowed Israeli individuals (N=196). Loneliness was significantly higher among widowed than married respondents, but no significant difference was observed between widowed and divorced, or divorced and married. Arguably, the mean loneliness score used within this study is unreliable; the scoring system for the UCLA-LS was not designed to quantify loneliness, thus the scale should only be used to categorise as lonely or non-lonely (Campaign to End Loneliness, no date).

Interviews with 'often or always' lonely, older Finnish adults (N=10, aged 70-83) revealed that the interviewees considered loneliness unavoidable, attributing this to age-related health declines which limit social interaction (Tiilikainen and Seppanen, 2017). This would suggest that loneliness was an internalised stereotype; SET conveys that this perception is likely to be a contributing factor towards their experiences of loneliness. One of the key themes identified from the interviews was the loss, or lack of, a partner (*ibid*); however, all but one of the respondents were without a partner (divorced N=4, widowed N=4, never married N=1, married N=1). Although this was identified as a substantial determinant of loneliness among the sample, and is supported by additional evidence indicating that widowhood and divorce are predictive of loneliness (Pinquart and Sörensen, 2011; Ben-Zur, 2012), arguably, a sample skewed towards those who have lost, or are without, a partner would be expected to produce such a strong result. The key themes arising in the interviews may have been largely different among a predominantly married sample, or a mixed sample of married, widowed, and divorced individuals.

Longitudinal analysis of loneliness among older adults is much less common; Victor and Bowling (2012) compare 1999 ONS omnibus data (N=999) with follow-up postal questionnaires in 2007/08 (N=287). Bivariate analysis demonstrated that marital status and living arrangements at follow-up were associated with change in loneliness; overall, 12% experienced a reduction in loneliness, but over twice as many (25%) showed an increase in loneliness. Changes in marital status, but not living arrangements, were significant in predicting loneliness change (*ibid*). It is possible that individuals experiencing a change in marital status (e.g. married to widowed), simultaneously experience a change in living arrangements, unless living with other family members, therefore it is plausible that the significant association between marital status and loneliness accounts for change in living arrangements.

The literature search revealed an absence of evidence examining the relationship between marital status and loneliness among caregivers. However, Smith *et al.* (2020) analysed changes to the psychosocial status of caregivers (N=777) using ELSA waves 1-5. Over the eight-year period, prevalence of loneliness increased to a greater extent for caregivers than non-caregivers (N=1,463). Simultaneously, the proportion of married and cohabiting relationships significantly reduced among the caregiving sample. However, although both trajectories were statistically significant (i.e. different for caregivers and non-caregivers), there was little consideration into whether they were linked. This raises the question of whether the reduction in marital relationships contributes towards increased loneliness among caregivers.

# 3.2.2.2 Relationship quality

Hsieh and Hawkley (2018) also analysed NSHAP data, categorising marriages as aversive, indifferent, ambivalent, and supportive. Compared to those in supportive marriages, females in indifferent or aversive marriages were more likely to be lonely; whereas among males, only aversive marriages were significantly lonelier than supportive. Descriptive data identified that more males reported being in ambivalent marriages, whereas females most commonly reported supportive marriages. Given that the sample was dyadic, this suggests that females were less likely to report marital strain, or that males over-reported marital strain, which could have contributed to the gender differences. The sample comprised 953 heterosexual couples, limiting the generalisability of the results only to heterosexual relationships. This is a common theme within the literature, where there is little evidence on marital status, or quality of marriages, among same-sex couples. It is therefore unknown whether the gender differences identified within marital literature would be relevant. Relationships outside of marriage were also considered, but, regardless of quality, relationships with family and friends did not influence the direct effect of marital quality on loneliness (*ibid*). This suggests individuals with poor quality marriages are likely to be lonely, even if other relationships are high-quality. One explanation for this might be: marriages are potentially more intense relationships, and therefore could be expected to have a larger weighting on loneliness. As such, couples are likely to be cohabiting and therefore spend more time together than with friends or other relatives; moreover, exposure to a higher-quality relationship over a short time-period (e.g. visiting friends) is unlikely to counteract a poor quality relationship over a longer period (e.g. living with spouse).

Additional research on marital quality was conducted by de Jong Gierveld *et al.* (2009), among a sample of 755 respondents (60% male, aged 64-92) from the 2001/02 wave of Longitudinal Ageing Study Amsterdam. Many marital quality factors were investigated for associations with loneliness; larger social networks, infrequent contact with children, and lack of spousal emotional support were associated with higher incidence rates of loneliness. Furthermore, instrumental support provided to someone other than a spouse was associated with lower incidence of loneliness; the tasks reported as instrumental support included helping with daily chores or around the house, which could

be likened to some caregiving tasks. Therefore, this implies that caregiving may be linked to lower loneliness.

Overall, the literature appears to support aforementioned claims that being married is protective of loneliness (Hsieh and Hawkley, 2016; 2018). It is evident that factors such as relationship quality play an important role, thus the presence or absence of a spouse alone may not determine loneliness. Additionally, it is difficult to draw comparison between most other categories of marital status, as they are commonly aggregated. Due to the lack of evidence, it is unknown whether caregiving would impact the relationship between marital status and loneliness. Evidence from Smith *et al.* (2020) implies an association between marital status and caregiver loneliness, but this was not tested.

The main research gaps, other than the dearth of research on the impact of relationships on caregiver loneliness, is research focusing on the impact of, for example, divorce in comparison to widowhood. Particularly, when evidence suggests that being unmarried is related to loneliness, additional evidence comparing the different classifications of 'unmarried' would augment this conclusion. Moreover, alternative aspects of relationships, such as quality, or cohabitation, appear to be largely influential on loneliness. Additional research emphasising the importance of relationship features outside of marital status would largely expand the current knowledge.

#### **3.2.3** Social networks

Reduced networks have been linked to social isolation (de Jong Gierveld and van Tilburg, 2016) and loneliness (Domènech-Abella *et al.*, 2017a). Additionally, evidence from a National Institute for Health and Care Excellence (NICE) quality standard suggests that increasing or maintaining social participation can prevent loneliness in older adults (NICE, 2016). Diminished social networks, often through reduced social participation, are common with age (Böger and Huxhold, 2018); moreover, due to caregiving responsibilities, social networks of caregivers are at particular risk of disruption (Davies *et al.*, 2019).

Social networks can be measured in many ways, most commonly: network size, relationship quality, or satisfaction with social connections. Among a sample of Spanish adults aged  $\geq$ 50 (N=3,535), several aspects of social networks, including size, frequency of contact, and quality, were associated with loneliness at bivariate level (Domènech-Abella *et al.*, 2017a). However, in the multivariate analyses, only network size remained significant, those with larger networks displayed lower odds of loneliness (*ibid*). These results appear contradictory to other published evidence, which tend to favour network satisfaction or quality, over size as determinants of loneliness.

By illustration, in a sample of Dutch older adults (N=182, aged  $\geq$ 65), network satisfaction was directly associated with loneliness (Kemperman *et al.*, 2019). In contrast, network size was only significantly associated with loneliness when mediated by satisfaction. Network satisfaction was derived from three items (social contacts, network quality and network size), therefore provided a

multifaceted approach. Conversely, Domènech-Abella *et al.* (2019) used a social network index to assess loneliness within the Irish Longitudinal Study on Ageing (TILDA), they reported significant loneliness differences between the most isolated and most integrated respondents. However, this association attenuated after controlling for other covariates. The index encompassed a combination of closer and wider network aspects, including marriage, cohabitation, close friends and relatives, volunteering or club memberships, and frequent church attendance. Respondents were categorised as integrated or isolated, based on these variables, although arguably, these may be restricted to a specific context and not applicable to everyone. As such, non-religious individuals would score lower due to not attending church, but may be integrated in other ways.

There appears to be no standardised measure for quality of social networks; Böger and Huxhold (2018) compared distressing and pleasant relationships, whereas Štípková (2019) considered that individuals who list their partner in their close network demonstrate high-quality relationships. Analysis of the German Ageing Survey (DEAS) (N=10,900) found that quantity of distressing and pleasant relationships, alongside relationship satisfaction, were indicative of loneliness among adults aged 40-84 (Böger and Huxhold, 2018). More pleasant relationships, and less distressing relationships, were associated with lower levels of loneliness, these relationships were not significantly different for middle-aged and older adults. This suggests that regardless of age, lowerquality relationships are linked to loneliness. Alternatively, Štípková, (2019) did not directly measure the quality of social networks; but Czech data from wave six of the Survey of Health, Ageing and Retirement in Europe (SHARE) included the size and composition of respondents' (N=3,949) networks. The majority (>80%), mentioned their partner within their network. The author suggests that this was indicative of a high-quality relationship, and regression analysis revealed that loneliness was lower for those with high-quality relationships (partner mentioned in network). This evidence on the quality of social networks corresponds with evidence on marital quality (section 3.2.2), thus implying that regardless of whom the relationship is with, quality is still an influential factor for loneliness.

There was a lack of substantive evidence demonstrating the relationship between social networks and caregiver loneliness. In a review examining the social consequences of caregiving, only two of the reviewed studies (N=66) referred to loneliness (Keating and Eales, 2017). The first indicated that the changing relationship between spouse and caregiver contributed towards loneliness among Swedish male caregivers. This relates to the adaptations outlined by CIT (section 2.1.1.1), whereby caregivers begin to identify more as a caregiver than a spouse. The second suggested that reduced social engagement was particularly difficult for caregivers reporting high levels of burden, depression, and loneliness. Overall, neither gave insight into the impact of social networks on loneliness among caregivers, but suggested that reduced networks, or changes to relationship quality, were linked to loneliness.

Focus groups, conducted with older adults on the causes of loneliness, proposed that caregivers experience isolation due to lack of conversation with others outside of the caregiving situation (Cohen-Mansfield *et al.*, 2016). This would suggest that an external support network (e.g. non-family members), might be advantageous to caregivers. However, qualitative evidence from Davies *et al.* (2019) posed contradictory results on support groups as part of a social network, many of the caregivers perceived this as being separate from their typical network. Caregivers were able to talk freely about caregiving within the group, whereas conversations within their friendship network rarely included caregiving. Moreover, some did not engage with the group, as they found it depressing and did not wish to be burdened with other people's problems (*ibid*). This evidence would suggest that how individuals engage with, or perceive, their network is more important than the presence of a network. Additionally, aligning with aforementioned evidence from Keating and Eales (2017), relationship losses (e.g. family members, care recipients and their own identity), were represented through social isolation and loneliness.

Overall, there is evidence to suggest that multiple aspects of social networks: size, satisfaction, and quality, are linked to loneliness. However, the methods used to determine social networks vary substantially, assessing either overall social network (size and satisfaction), or just the closest individuals within a social network (living arrangements or relationship quality).

#### **3.2.4** Socioeconomic status

SES can be represented by one, or a combination of factors, including education, income, and employment (Braveman *et al.*, 2005). Arguably, due to the multifaceted disposition of SES, a single measurement (e.g. only education) would not capture an accurate picture of an individual's socioeconomic activity. Different aspects of SES are likely to have varying influence at different life stages, as such, income may be less valid among retirees, and thus wealth may be a more reliable criterion among retired older adults. Measurement of SES also differs across countries; occupation is common across Europe, whereas in the US, income and education are more common (*ibid*).

Within the literature, there is evidence that lower SES is linked to loneliness. A meta-analytical study highlighted income and education as the most common SES indicators of loneliness (Pinquart and Sörensen, 2001). The association between loneliness and SES was stronger for single-item measures (N=73) compared to the UCLA-LS (N=40), and among income as opposed to education, although both were significant (*ibid*). Arguably, education is a weaker measure of SES among older adults, compared to wealth (Darin-Mattsson *et al.*, 2017). Education is less likely to be affected by health (Smith and Goldman, 2007), and in previous decades, the majority of UK and European older adults reportedly left school with few or no qualifications, and at younger ages than later birth cohorts (Grundy and Holt, 2001). When comparing cohort studies (birth cohorts: 1958, 1970, 1980 and 1990), early entry into the labour market declined across the cohorts (Anders and Dorsett, 2017). Simultaneously, accumulation of human capital (i.e. continuing education) substantially increased

across the cohorts. This implies that older adults are likely to **report** lower education, therefore education should be considered in addition to other SES measures, rather than in isolation.

Loneliness was more prevalent in deprived Danish neighbourhoods, than in the general population (Algren *et al.*, 2020). Within the deprived population, loneliness significantly differed by employment status, highest among those unable to work due to disability, and lowest among the employed. However, loneliness did not significantly differ by education. Alternatively, data from a representative sample of Spanish individuals aged  $\geq$ 50 (N=971), demonstrated that among lonely individuals (N=100) loneliness differed by SES (Domènech-Abella *et al.*, 2017b). Individuals with higher SES were significantly less likely to be lonely than those with lower SES. The measure of SES in this study was computed using combined education and household income, and is therefore less comparable with other studies using education and income separately. The authors argue that occupation-based methods were not used because they have different meanings for different birth cohorts, but as discussed previously, this argument is also applicable to education (Anders and Dorsett, 2017).

Wealth was the primary measure of SES in a study analysing wave five of SHARE (Niedzwiedz *et al.*, 2016); among the respondents (N=29,795), prevalence of loneliness was highest in the poorest quintile, more so for females than males. The logistic regressions indicated that, for both males and females, all wealth quintiles demonstrated lower odds of loneliness in comparison with the poorest. Whereas, medium or high education were only negatively associated with loneliness among females. Similar to Domènech-Abella *et al.* (2017b), Niedzwiedz *et al.* (2016) measured loneliness using the UCLA-LS, which suggests that SES is also linked to loneliness when using scale measures, not just direct questions.

The review of evidence on SES and loneliness emphasises the importance of selecting an appropriate context-specific variable, education is reportedly most common in particular populations (Pinquart and Sörensen, 2001; Braveman *et al.*, 2005) but, as identified above, across a range of contexts, the relationship between education and loneliness was frequently non-significant (Niedzwiedz *et al.*, 2016; Domènech-Abella *et al.*, 2017b; Algren *et al.*, 2020). Overall, the use of education as a measure of SES appears complex, the relationship between education and loneliness is variable, with little evidence supporting a link.

Research with caregivers (N=70) found education to be negatively associated with loneliness (McRae *et al.*, 2009); respondents were predominantly female (74%) and spousal caregivers (N=67), average education level was 14-years, ranging from 8-18 and 23% were employed alongside caregiving. The authors suggest that higher education equated to greater resources to deal with the challenges of a caregiving role, thus protective of loneliness. No other socioeconomic factors were included in the analysis, such as wealth or income, therefore it is difficult to generalise to SES as a whole. The overwhelming evidence would suggest that the relationship between education and loneliness is largely inconsistent. Alternatively, wealth is considered an appropriate measure of SES,

particularly among older, retired populations (Demakakos *et al.*, 2016; Niedzwiedz *et al.*, 2016) as it is reflective of accumulated assets. Wealth may be considered more appropriate among caregivers aged  $\geq$ 50, as caregivers are shown to have lower levels of employment, often due to their caregiving commitments (Hendricks-Lalla and Pretorious, 2020), which, correspondingly, reduces income. Wealth has also been shown to have strong links to health and mortality (Torres *et al.*, 2016). The connection between loneliness, health and wellbeing is discussed in section 3.4.

### 3.2.5 Summarising the literature on determinants of loneliness

Largely, the highlighted evidence is quantitative and identifies recurring predictors of loneliness, such as sex, marital status, poor social networks, and SES. However, there was limited research discussing the determinants of loneliness among caregivers. From the research that has been conducted, typical loneliness determinants do not appear to be consistent for caregiving populations. For example, Victor et al. (2020) found no significant association between sex and loneliness among caregivers. Additionally, over a period of eight-years, it was demonstrated that for a sample of English caregivers, the proportion of married individuals reduced more substantially than among non-caregivers. Meanwhile, the prevalence of loneliness increased more for caregivers than noncaregivers, yet there was no evidence in the study to confirm the two occurrences were linked (Smith et al., 2020). This may suggest that the established determinants of loneliness are not necessarily applicable to caregivers. Alternatively, qualitative evidence from focus groups with older adults suggested that caregivers may experience loneliness and isolation, as they have no one else to talk to besides the care recipient (Cohen-Mansfield et al., 2016). This aligns with evidence from Victor et al. (2020) which highlights social networks and isolation being among the most influential factors for caregiver loneliness. Due to the lack of research into the predictors of loneliness among caregivers, and the indication that traditional predictors may not apply, further research is needed to identify which determinants are applicable to caregiving and non-caregiving populations. Furthermore, if there are different predictors of loneliness for caregivers, the identification of these are highly relevant for policy-makers to implement prevention and support services.

# **3.3** Caregiving literature

The literature discussed in the previous section examined the determinants of loneliness, and although this was largely within the general population, reference was made to caregivers where available. The following sections examine literature exclusively in caregiving samples, to address sociodemographic characteristics associated with caregiving, and factors related to the caregiving relationship and role. Furthermore, in the conceptual framework, several factors were identified that could impact the relationship between caregiving and loneliness; these included caregiver burden, satisfaction, social networks, and identity. Caregiver burden and satisfaction are discussed in more depth in section 3.3.4. However, literature on social networks was addressed previously as a

determinant of loneliness (section 3.2.3), as this was deemed applicable to loneliness in general, not just among caregivers. Finally, the identity of a caregiver shifts continuously, as described by CIT (section 2.1.1.1). This topic is not well-researched, and typically, identity has been determined by a respondent categorising themselves as providing care. For this reason, it was challenging to address the impact of caregiver identity on loneliness.

# **3.3.1** Caregiver characteristics

Conceptual literature identified caregiver characteristics as a key area for caregiving research. This section discusses age, sex, marital status, and SES as characteristics of caregivers. The literature addresses the peak age of caregiving and care provision among older adults, in addition to discussing the stereotypical assumption that caregivers are mostly female. Marital status is likely to form a large part of the caregivers' networks and has previously been linked to loneliness. Therefore, it is important to understand the marital demographic of caregivers, identifying if being married, or unmarried, is associated with likelihood to provide care. Finally, other characteristics influential to caregiving are socioeconomic factors such as education or employment, which are likely to restrict availability to provide care and the intensity in which it is delivered.

### 3.3.1.1 Age

The peak age for caregiving across the UK is 50-years old. In 2001 UK Census data, caregiving was most common between ages 45-59 (Dahlberg *et al.*, 2007). In comparison, the largest percentage of caregivers were aged 50-59 (Nomis, 2018a) or 50-64 (Nomis, 2018b) in 2011 England Census data. However, discrepancies between 2001 and 2011 are likely due to inconsistencies when dividing the age groups. Similarly, Vasileiou *et al.* (2017) indicates that across Britain, the peak age for caregiving is 50-64 years. In a geographically-stratified sample of caregivers from the 2001 UK Census (N=182,664) caregiver age was significantly associated with higher-intensity caregiving; approximately 50% of male and 45% of female caregivers aged 80-89 provided  $\geq$ 50hpw of care. Therefore, those aged  $\geq$ 70 spend the most time caregiving due to increasing intensity. Furthermore, data from the Carers Trust indicated that 13% of individuals aged  $\geq$ 75 were caregivers, rising to 19% among ages  $\geq$ 85 (Greenwood and Smith, 2016).

The age of caregivers appears to differ by sex, whereby female caregivers are, on average, younger than males. Nevertheless, by age 50, half of people in England will have provided care (Zhang and Bennett, 2019). Stratified by sex, 50% of females are assumed to have provided care by age 47 and by age 57 for males (*ibid*). The trend of older male caregivers is present in other countries too, in a sample of Spanish caregivers (N=6,923), the majority of female caregivers were aged 45-64, whereas the largest proportion of male caregivers were aged  $\geq 65$  (Abajo *et al.*, 2017). Further sex differences in caregiving are discussed in the next section.

#### 3.3.1.2 Sex

Caregiving is most commonly associated with females (Dahlberg *et al.*, 2007; van Campen *et al.*, 2013); analysis of 2016-2018 Understanding Society data indicated that 58% of current caregivers were female (Zhang and Bennett, 2019). However, evidence from 2011 England and Wales Census data highlights that of those aged  $\geq$ 65, 15% of males and 14% females identified as caregivers (Milligan and Morbey, 2016). These figures suggest a greater prevalence of male caregivers, compared to female among those aged  $\geq$ 65, although it could be considered that these males are more likely to identify as caregivers, compared to older females. To clarify, due to the stereotypical nurturing disposition of females, they may consider helping a spouse as an extension of roles they have had throughout the lifecourse, such as childcare. Familiarity with the tasks may act as a buffer for recognising their caregiving role. Alternatively, this could be due to greater longevity of females. In other words, due to discrepancies in life expectancy, females are more likely than males to be widowed and therefore less likely to have a spouse to care for (Pinquart and Sörensen, 2011).

It is possible that the prevalence of male caregivers may be higher than reported. As such, due to the female caregiving stereotype, males may be deterred from identifying as caregivers. Sanders and Power (2009) interviewed only male caregivers for wives with dementia (N=17). Two themes emerged from the interviews: adapting roles from a marital to caregiving relationship, and changes to the inter-personal relationship. The adaptation of roles aligns with CIT, suggesting that, as dementia progresses, care needs change and simultaneously, the caregiving role changes, including tasks such as personal care. No comparison was drawn with female caregivers, but the authors propounded that the caregivers identified with their male "protective role" (Sanders and Power, 2009, p.44). This could be linked to SET, whereby stereotypes about protective males influenced their marital and caregiving relationships. Caregiving was considered to pose an unnecessary challenge to male caregivers' masculinity (Milligan and Morbey, 2016), with many justifying the role by highlighting manual tasks. A qualitative investigation with caregiving husbands (N=9) reported a similar methodical approach in the way males provided spousal care (Calasanti and King, 2007), for example, by blocking emotions or exerting force, which could be considered stereotypical male traits. These articles suggest males focus only on the physical tasks of caregiving, and not the emotional aspects (Swinkels et al., 2017); but also emphasise the relevance of CIT and SET.

#### 3.3.1.3 Marital status

Unlike section 3.2.2.1, where there was a range of evidence to suggest significant associations between marital status and loneliness in the general population, there is limited research evaluating the effects of marital status among caregivers. However, there is evidence of marital status differences among the caregiving population. Among a sample aged 40-59 in England and Wales (N=110,464), non-married individuals were the most likely to report caregiving, compared to married, remarried, separated or divorced, and widowed (Young and Grundy, 2008). The evidence

was drawn from the ONS Longitudinal Study in 2001, so these figures may not be reflective of the current population. Moreover, the authors include only caregivers providing high-intensity care (≥20hpw) which represents 5% of the total sample, an additional 14% reported providing care for <20 hpw. Marital status may affect an individual's availability to provide care. As such, if they have responsibilities to uphold within their own family, they may be more likely to provide lower-intensity care, therefore larger proportions of the caregivers providing <20 hpw may have been married. Evidence from Finch and Mason (1991) suggests that obligation to care tends to be stronger among close family than other relatives, but females tended to have stronger obligation to care than males. For example, regardless of the direct relationship with the recipient, more people expected females to cease employment for care responsibilities than males (i.e. daughter-in-law instead of a son). This could also be extended to marital status, where obligation to care may be higher among unmarried family members, but that married daughters may be more likely to provide care than unmarried sons due to the gendered stereotypes of care. This is supported by evidence from Pillemer and Suitor (2014), who compared 537 adult-children within 139 families. The analyses showed that competing responsibilities (marital status, employment or parenthood) were not significantly associated with care provision, suggesting that among filial caregivers, marital status has little influence on the decision to provide care. However, odds of caregiving were twice as high for females, compared to males, indicating that sex was a stronger determinant of care than marital status or other responsibilities. Marital status data were collected in 2001-2003, whereas care provision occurred within the two-years preceding the follow-up interview (2008-2011). Therefore, it is possible that baseline marital status may have differed to marital status during care provision (e.g. unmarried individuals may have married). Arguably, current marital status would be expected to have greater influence on the uptake of caregiving responsibilities than marital status recorded several years previously.

Among filial caregivers, marital status may impact co-residential caregiving. When comparing married (N=38) and unmarried (N=10) male filial caregivers, a substantially larger percentage of non-married caregivers provided co-residential care than married caregivers (Campbell, 2010). This suggests that marriage introduces additional responsibilities, which may supersede the responsibilities that individuals feel towards older parents. Some respondents expressed that their single status meant they were more available than married siblings, and therefore assumed the caregiving role. In contrast, evidence from Bauer and Sousa-Poza (2015) found no association between the caregivers' marital status and the likelihood of co-residential care provision to a parent. However, marital status did influence the location of the care; for married respondents, co-residential care typically occurred within the caregivers' home. As highlighted by Campbell (2010), unmarried children may have greater availability, this is reflected by unmarried caregivers moving in with their parents to provide care (Bauer and Sousa-Poza, 2015). Alternatively, being married does not make an adult-child less likely to provide care, but there may be greater time limitations. This may reduce

availability for caregiving hours, or, as Wong and Shobo (2017) depict, reduce time spent on leisure activities outside of the caregiving role.

### 3.3.1.4 Socioeconomic status

It could be considered that those with higher education levels would be more likely to be in higher levels of employment (e.g. managerial) and therefore receive higher incomes. This would imply greater financial stability, with disposable income to pay for formal care, or alternatively, they may rely on other family members to provide care. Low-wage workers, or those providing a smaller proportion of overall household income are considered more likely to cease paid employment for care responsibilities (Carr *et al.*, 2018). Furthermore, Finch and Mason (1990) suggest that family responsibilities have become weaker over time and may take less precedence than other commitments such as employment. A stronger presence in the labour market, particularly among women, suggests less assistance is available to family members.

Largely, it appears the connection between SES and caregiving differs across countries. Full-time paid work was indicative of lower likelihood of informal activity (e.g. care provision) in Denmark, France, Italy and England, whereas part-time work was only negatively associated with care provision in England (Di Gessa and Grundy, 2017). This analysis was conducted cross-sectionally, therefore it is unknown whether the employment causes lack of informal activity, or informal activity is taken up because of lack of employment. When investigated longitudinally, ceasing employment was linked to increased likelihood of undertaking a caregiving role at follow-up (2-years later) in Italy and England but not Denmark or France (*ibid*). This further suggests country differences, but also that uptake of care provision could be due to availability.

Among Japanese females aged 40-60, caregivers (N=1,417) had lower levels of education and were less likely to be in full-time employment, but more likely to work part-time or be unemployed than non-caregivers (Tokunaga and Hashimoto, 2017). In the full sample, those with a high school education, in contrast to university graduates, were more likely to be caregivers, and both full-time and part-time workers were less likely to provide care than unemployed. This evidence suggests that less educated individuals and those not in employment are more likely to provide care. Similarly, pooled 1991-2005 data from the British Household Panel Survey (BHPS), demonstrated among 8,652 males and 13,983 females, that care provision was more likely among unemployed or low-wage employees (Carmichael *et al.*, 2010).

In contrast, in a Chinese sample of filial caregivers, divided into low-intensity (N=586), highintensity (N=450), and non-caregivers (N=1,705), caregivers were more highly educated and reported higher incomes than non-caregivers (Wang *et al.*, 2020b). The evidence also suggests that SES is associated with caregiving intensity, among those with high school or university degrees, there was a higher likelihood of high-intensity care provision, but not low-intensity care, compared to non-caregiving. The authors suggest this provision of filial care was related to the repayment of

their upbringing and education. Arguably, the division of intensity is not comparable with other studies, as  $\geq 10$  hpw was classified as high-intensity. Chen *et al.* (2017) examined the association between caregiving intensity, labour force participation, and working hours; when intensity was defined using 10 hpw (<10 hpw or  $\geq 10$  hpw), no significant differences in labour force participation or working hours were observed. Alternatively, significant differences were observed in labour force participation when intensity was divided at 15 hpw or 20 hpw. To illustrate, caregiving overall was associated with lower labour force participation, but caregivers of  $\geq 15$  hpw or  $\geq 20$  hpw were significantly less likely to be in the labour force (*ibid*). Typically high-intensity caregiving is recognised as  $\geq 20$  hpw (Vlachantoni *et al.*, 2016; Lacey *et al.*, 2018). Therefore, the evidence from Chen *et al.* (2017) may be more reflective of the high-intensity caregivers in other contexts, than those from Wang *et al.* (2020b).

Furthermore, within a representative US sample aged  $\geq$ 55 (N=3,005), data suggests that caregivers (N=413) were more highly educated than non-caregivers: 55% of female and 63% of male caregivers reported receiving college education, compared to 50% (female) and 59% (male) of non-caregivers (Wharton and Zivin, 2017). Contradictorily, household income trends did not match. Female caregivers reported considerably lower income than female non-caregivers, whereas male caregivers reported higher income than male non-caregivers. This income discrepancy may be due to differences in care provision, i.e. those providing lower-intensity care may work part-time, which would be reflected in their income. Additionally, higher percentages of males were married, which would increase average household income, compared to single households.

Discrepancies between age ranges in the sample may explain why the literature appears to show contrasting evidence. For example, Tokunaga and Hashimoto (2017) suggest caregivers have lower education levels (aged 40-60), whereas, Wharton and Zivin (2017) (aged  $\geq$ 55) and Wang *et al.* (2020b) (aged <52) found that caregivers were more highly educated. This could also be due to cultural differences, as the three studies were conducted in Japan, America, and China, respectively. Moreover, both Tokunaga and Hashimoto (2017) and Wang *et al.* (2020b) analysed female-only samples, therefore there may be educational differences due to sex. Additionally, the direction of relationships between care provision and employment is largely unclear, i.e. were working hours or unemployment make the individual more readily available than other family members to provide care? Evidence from Tokunaga and Hashimoto (2017) suggests that employment level is linked to care provision.

Within the UK context, the main predictor of ceasing full-time employment, for both male and female caregivers (aged 50-75), was the provision of co-residential care (Carr *et al.*, 2018). Amongst female caregivers, the provision of spousal care, and the uptake of a new caregiving role of >10hpw were positively associated with ceasing employment, although these factors were not significant for male caregivers. Conversely, intensity of a current caregiving role was not significantly associated with

an exit from the labour force (*ibid*). This implies that if a routine is established, for example balancing care and work responsibilities, the caregiver is likely to continue, perhaps using work as a respite from care responsibilities. Ceasing employment due to a new higher-intensity care role, but not a current one, could also suggest that if caregiving needs have increased over time (i.e. low- to high-intensity), the caregiver may feel able to cope and thus less likely to stop working, but initiating a high-intensity role without previous caregiving may feel overwhelming while working, and thus prompt an exit from the labour market.

#### 3.3.2 Caregiving relationship

The caregiving relationship can refer to the inter-personal relationship between caregiver and recipient, but also differentiates between co-residential caregivers, and caregivers to recipients in other households. The following sections highlight comparisons between caregivers based on the relationship with the recipients, and contrast evidence on co-residential and non-residential caregivers.

### 3.3.2.1 Inter-personal relationship between caregiver and care recipient

Inter-personal relationship depicts the relationship outside of the caregiving context (e.g. spouse, rather than care recipient). Predominantly, research has been conducted with female caregivers, typically the recipients' wife or daughter. A review of 56 research papers on dementia caregivers identified that, although limited, research involving male caregivers was predominantly on spousal caregivers, with little focus on filial care (McDonnell and Ryan, 2013). Comparison between spousal, filial, and other (sibling, friend, or neighbour) caregivers from the Netherlands (N=660) suggested that, typically spousal caregivers were older, male, less educated, and more commonly co-residential caregivers, compared to the other groups (Oldenkamp et al., 2018b). Parental caregivers were excluded, indicating this type of care was different; although arguably, all caregiving differs, as spousal care is unlikely to be a similar experience to caregiving for a neighbour. Qualitative evidence among spousal caregivers suggests that experiences of caregiving may be linked to the quality of previous relationships (Shim et al., 2012). For example, those with negative relationships with the recipient prior to care provision reported negative current relationships and negative overall experience. Although this was only examined among spousal caregivers, it is likely that a prior negative relationship with another relative would also result in negative relationships during care provision.

Alternatively, Yu *et al.* (2020) compared parental (N=151) and spousal (N=114) caregivers in China. On average, spousal caregivers were younger, male, highly educated, employed, and provided care for longer durations, than parental caregivers. This contradicts previous evidence, which suggests that providing care to a child, often with DD, is a lifelong role (Weeks *et al.*, 2008; Byram, 2018; Barnhart *et al.*, 2020). However, parental caregivers in this sample were providing care to individuals

with schizophrenia (average age 38), for an average of 14-years (Yu *et al.*, 2020), thus much shorter than caregivers to a child with DD. Multivariate analyses demonstrated that, compared to spousal caregivers, parental caregivers experienced higher levels of burden, depression, and anxiety. Despite the lack of consistency with other literature, overall this evidence would suggest that spousal caregivers differ significantly from all other caregiving subgroups. Furthermore, it has previously been reported that spousal caregivers demonstrate distinct characteristics, greater levels of burden, and often higher-intensity roles (Friedemann and Buckwalter, 2014). Comparison between US spousal (N=208) and filial (N=325) caregivers also indicated that spousal caregivers, regardless of sex, experienced a significantly higher obligation to care, compared to filial caregivers (*ibid*). This is consistent with Finch and Mason (1991), which indicates obligation is highest among closest family (e.g. spouse). Therefore, spousal caregivers may often experience the most negative health effects (Bédard *et al.*, 2005; Pinquart and Sörensen, 2011).

Additionally, care can be provided to multiple generations simultaneously. Falkingham et al. (2020) demonstrate from the China Health and Retirement Longitudinal Study that of the sample aged 45-64 (N=12,552), 33% were at risk of 'sandwich' caregiving (caring concurrently for a grandchild and older parent), but only 15% of those reported providing care to both. Most commonly, the respondents reported providing grandchild care only, or no care provision. Care to grandchildren contrasts to typical informal care, it is usually non-custodial and more representative of childcare. Schmidt et al. (2016) indicate that care to grandchildren is lower-intensity and comparable to social engagement. Evidence from Quirke et al. (2019) supports this, demonstrating that among German grandparent caregivers (N=1,125), care to grandchildren was associated with increased network size and reduced social isolation. For the purpose of this thesis, individuals reporting providing care to a grandchild were excluded. Moreover Falkingham et al. (2020) indicate that individuals aged ≥65 are increasingly less likely to have surviving parents; therefore, as the ELSA sample includes both middle-aged and older-adults, and with the exclusion of grandchild care, the relevance of 'sandwich' caregiving to this thesis is negligible. In contrast, 'compound' caregiving, which also reflects multiple caregiving roles, involves undertaking an additional caregiving role, while already providing care to a dependent child with DD (Perkins, 2009). This differs from 'sandwich' caregiving, because the additional role is not always a parent, and could be a spouse or sibling (*ibid*); caregivers within the ELSA sample reporting care to a child and another recipient are likely to be 'compound' caregivers.

### 3.3.2.2 Provision of co-residential care

Typically, co-residential caregivers are compared with caregivers of recipients in other households (non-resident). Caregiving intensity is considerably higher among co-residential caregivers, which has been linked to detrimental health consequences (Kumagai, 2017). To illustrate, Mentzakis *et al.* (2009) report data from BHPS: co-residential caregivers (N=7,212) most commonly provided care for >100hpw, whereas the majority of non-residential (N=17,366) caregivers reported providing

<5hpw. Co-residential care may also be linked to the caregiving relationship; Bédard *et al.* (2005) highlighted 98% and 97% of spousal caregivers respectively lived with the recipient, compared to 11% of son and 19% of daughter caregivers.

However, co-residential care was negatively correlated with role burden, and positively correlated with personal burden (Bédard *et al.*, 2005). These results indicate that co-residential care may reduce the demands of caregiving to some extent. For example, within the same household, caregivers do not have to travel to the recipients' homes for a specific time, this is therefore less demanding, as tasks can be carried out when appropriate with less scheduling required. Alternatively, co-residential care contributed to increased levels of personal burden (*ibid*), this may be due to increased intensity. Living with the recipient could indicate that care might be required 24hpd, with the caregiver available at all times. Moreover, US caregivers of recipients who lived within 20 minutes of the caregivers' home, were less likely to report negative health consequences, compared to co-residential caregiver did not have substantial travelling to organise, and therefore tasks could be performed more conveniently, also as the caregiver was not with the recipient all of the time, it was less demanding and less likely to negatively impact health.

### **3.3.3** Factors relating to the caregiving role

The nature of care, alongside the duration and intensity are the main factors related to the caregiving role. These factors substantially alter the caregiving dynamic, as such, personal care may be more physically and emotionally demanding for caregivers, compared to providing financial assistance. Moreover, long-term, and higher-intensity care often requires the largest adjustments from caregivers, such as reducing paid employment, or ceasing hobbies and social activities; therefore, it is likely that these factors have considerable impact on loneliness, health, and wellbeing.

## 3.3.3.1 Nature of care

Informal care can involve a range of tasks from more physical aspects, to psychological (Zwar *et al.*, 2018). Provision of personal care, which typically involves assistance with tasks such as dressing, or bathing, is considered to be more physically and mentally demanding than other aspects of care (Hiel *et al.*, 2015). In a study on the provision of personal care in SHARE (N=997 caregivers), caregiving was linked to poorer physical and mental health outcomes over an eight-year follow-up (*ibid*).

Types of care are inconsistent within the literature. When analysed within DEAS (Zwar *et al.*, 2018), caregiving was divided into: 'helping around the house' (e.g. cleaning), 'looking after' (e.g. emotional support), and 'nursing care' (e.g. personal care). Among caregivers aged  $\geq 65$ , 'looking after' someone was positively associated with cognitive function, but once stratified by sex, this association remained significant only for females. This suggests that females experienced greater cognitive benefit from caregiving and could be linked to previous evidence (section 3.3.1.2) which

suggests males focus on physical tasks and not emotional (Calasanti and King, 2007; Swinkels *et al.*, 2017). Arguably, emotional aspects of caregiving may require further cognitive processes than physical tasks, which may be more familiar, thus promoting cognitive benefits. Zwar *et al.*, (2018) reported no data on the frequency or intensity of caregiving, but 'looking after' could be considered lower-intensity than 'nursing care'; lower-intensity care has previously been reported as beneficial to caregiver health (Vlachantoni *et al.*, 2016; Lacey *et al.*, 2018).

Alternatively, in a Dutch study, caregiving was divided into personal and emotional care; Hwang et al. (2010) contrasted spousal caregivers of individuals with heart failure and spouses of individuals without. Significant sociodemographic and health differences were identified between the groups at baseline; therefore, it could be questioned how comparable the two groups were throughout the study. Personal and emotional care were more common in the heart failure group. A key difference between the groups was that a substantially higher percentage of heart failure caregivers were required to provide 24 hour care (41%), compared to 9% of healthy spouses (*ibid*). However, it could be considered among the healthy spouses, many may be non-caregivers. Nevertheless, this implies that the presence of a specific medical condition alters the nature and intensity of care required. Caregiving types identified within a UK study were 'visible' and 'invisible' care (Clark et al., 2008). This separates behaviours (visible), such as personal care, from observations (invisible), such as symptom monitoring (*ibid*). The caregivers identified both types of care as essential, and despite being a constant task, even throughout the night, 'invisible' care was rarely recognised by recipients or external parties. This research highlights how personalised care provision is, with regard to recipients' needs. However, similarities arose with the results of Hwang et al. (2010), whereby more advanced medical conditions required increased levels of personal care. Although no comparable studies were found on caregivers for recipients with conditions other than heart failure, thus, little comparison can be drawn. Additionally, Clark et al. (2008) identified that as care needs increased among the heart failure recipients, adaptation was required to increase the level of care. Therefore, as care needs increase, care provision may change; this aligns with CIT whereby the caregiving role, and therefore the cargiving identity, change over time.

#### 3.3.3.2 Duration of care

Informal caregiving is the most common source of long-term care (Ciccarelli and van Soest, 2018). However, within literature, the total duration of caregiving is not as widely researched as caregiving intensity (section 3.3.3.3). This may be because where research is conducted with current caregivers, care provision is ongoing, and thus total duration of care is unknown. Additionally, in situations where caregivers have provided care intermittently, it would be difficult to determine total duration of the caregiving role. Moreover, because duration of care is collected retrospectively, typically as self-reported data, there may be recall errors, and caregivers are reported to often overestimate (Bremer *et al.*, 2017).

Duration of care has been shown to differ depending on the care recipient. Caregiving for individuals with DD was associated with longer caregiving durations than caregivers to individuals with mental illnesses; 69% of DD caregivers provided care for >5-years, compared to 48% of caregivers to those with mental illness (Barnhart *et al.*, 2020). Moreover, evidence from US-based Family and Individual Needs for Disability Supports (FINDS) and Caregiving in the US surveys indicated large percentages of caregivers providing longer-term care. Twenty-four percent of caregivers in Caregiving in the US were providing care >5-years, whereas in FINDS 35% had provided care for >25-years (Anderson *et al.*, 2018). However, the comparability of the two surveys is debateable; FINDS exclusively interviewed caregivers of those with DD, whereas Caregiving in the US included caregivers of older adults, physical disabilities, and dementias. Additionally, the age ranges of the care recipients differed substantially: a large percentage of the care recipients in FINDS (44%) were aged  $\leq$ 21, compared to Caregiving in the US which included no recipients aged <18, and only 14% of the sample were aged below 50.

Similarly, when considering the comparison between caregivers of recipients with DD or mental illnesses, DD included congenital conditions such as autism and Down syndrome, which are most commonly diagnosed at birth or in early childhood, whereas examples of mental illness were schizophrenia or depression, which typically do not occur at such early ages and are acquired during a person's lifetime (Barnhart *et al.*, 2020). Therefore, the potential care duration would likely always be longer for DD, because the caregiving occurs throughout the recipients' lives as opposed to later onset conditions. Overall, the evidence suggests that caregiving for individuals with DD is associated with longer duration of care, compared to that of older adults (Anderson *et al.*, 2018) and those with mental illness (Barnhart *et al.*, 2020). Care is typically provided to individuals with DD for the duration of the recipients' life. Conversely, care to older adults occurs later in life, typically after the onset of physical or mental health conditions, thus expected to be of a shorter duration than for a lifelong caregiving scenario.

### 3.3.3.3 Intensity of care

As outlined in section 1.2.1.3, caregiving intensity is typically determined by care provision in hpw. Evidence suggests females provide higher-intensity care, through both an increased number of caregiving hours, and more caregiving tasks than males (Cook and Cohen, 2018; Asi and Williams, 2020; Schulz *et al.*, 2020). Estimated from US data, on average, caregivers aged <65 provide around 53-hours per person per year (h/pp/y), compared to those aged  $\geq$ 65, providing on average 90h/pp/y (Chari *et al.*, 2015). Limitations of these estimates are that they are based only on US caregivers (N=1,383) who had provided care within 24-hours prior to the survey. Equivalent data on caregiving hours was not available with UK samples, so little comparison can be made. However, using 2011 England and Wales Census data, Milligan and Morbey (2016) demonstrate that male caregivers aged  $\geq$ 65 provide more caregiving hours than females of the same age. This contrasts with the US evidence, whereby aged  $\geq$ 65, females provided a greater number of caregiving hours than males

(Chari *et al.*, 2015). However, it is reported that caregivers are often undetected within the population (Carers UK, 2018), and are largely under-represented in population-based surveys, because many do not recognise their responsibilities as caregiving and therefore may not identify as caregivers.

In some studies, care is recorded in hours per month (hpm), rather than hpw, as per the majority. As such, 92 Swedish dementia caregivers (78% spousal) reported spending on average, 299hpm providing care (Wimo *et al.*, 2002). When divided into activities, the majority of hours (51%) were spent surveilling (e.g. preventing falls). It could be considered that for non-dementia caregivers, the proportion of time spent on surveillance may be considerably less, and additional time may be spent supporting with ADL/IADLs. Additionally, data from the National Study of Caregiving reports average caregiving intensity, for filial caregivers (N=993), at 85hpm (Cohen *et al.*, 2019). No division of high- or low-intensity is made within the study, but Schulz *et al.* (2020) suggest that >100hpm is a risk factor for adverse outcomes. Therefore, the hpm measure does not appear directly comparable to hpw, as  $\geq$ 20hpw has been linked to poor health (Asi and Williams, 2020), which would equate to approximately 80hpm.

Costa *et al.* (2013) reviewed 21 studies for caregiving time across different age-related dementias. The majority (N=15) involved Alzheimer's disease, so little comparison could be drawn between dementia types, although severity of dementia was compared. Time spent caregiving ranged from 1-56hpw for mild, 12-77hpw for moderate, and 21-120hpw for severe dementia. Although there appears to be a clear increase in intensity relative to dementia severity, the studies used different instruments to measure caregiving time (e.g. diary methods or activity surveys). The authors suggest that diary methods are preferred and typically yield fewer errors due to shorter recall periods (*ibid*). Nevertheless, there is additional evidence to support that caregiving intensity is related to the care needs of the recipient. Caregivers to individuals with DD were significantly more likely to provide >40hpw, compared to caregivers to individuals with mental illness (Barnhart *et al.*, 2020). Moreover, Wang *et al.*, (2020b) found that providing care to individuals aged  $\geq$ 50 (compared to <50) was associated with a higher likelihood of providing both low- and high-intensity care. However, high-intensity in this study was classified as  $\geq$ 10hpw, therefore not reflective of the same high-intensity (e.g.  $\geq$ 20hpw) in other studies.

Caregiving intensity may also differ due to caregivers' circumstances. As such, among employed filial caregivers (N=333), those with organised caregiving arrangements (e.g. care leave or reduced hours) were able to provide higher-intensity care, than those without workplace arrangements (Oldenkamp *et al.*, 2018a). Caregiving hours ranged from 1-25hpw, on average, those with workplace arrangements provided 4hpw, and those without, 3hpw; although a minor difference, it was statistically significant. Moreover, the low-intensity care and the small difference in caregiving hours may be explained by both caregiving groups being in employment; as demonstrated by Chen *et al.* (2017) when comparing <10hpw caregiving *versus*  $\geq$ 10hpw, no significant difference was

found in labour force participation, suggesting that lower-intensity caregivers can manage both work and caregiving responsibilities.

#### **3.3.4** Burden and satisfaction

Caregiving is most commonly portrayed as a negative experience, resulting in emotional distress, poor health, and reduced QoL (Casado and Sacco, 2012). The result of stress from the caregiving role is termed as caregiver burden (Lee *et al.*, 2017a). Studies have investigated the link between caregiver burden and recipient loneliness (Iecovich, 2016), or researched caregiver burden and caregiver loneliness within the same study (Bramboeck *et al.*, 2020) but there is an absence of research examining the association between caregiver burden and caregiver loneliness. Moreover, negative aspects of caregiving are more widely researched than positive aspects (de Labra *et al.*, 2015; García-Mochón *et al.*, 2019). Satisfaction represents a positive aspect of caregiving: a subjective measure of perceived gains from caregiving (de Labra *et al.*, 2015). Amongst caregivers, Kietzman *et al.* (2008) relates satisfaction to intrinsic feelings (e.g. pride), and growth within the relationship (e.g. closeness to the recipient).

#### 3.3.4.1 Caregiver burden

A key caregiver characteristic shown to impact burden is sex. Telephone interviews conducted with US caregivers (N=1,002, 54% female) suggested that females were more likely than males to experience difficulties providing care (Navaie-Waliser *et al.*, 2002). However, due to the nature of data collection, it is essential to consider under-reporting of these difficulties, especially from male caregivers. The results highlighted that males were more likely than females to be employed alongside caregiving, and less likely to be primary caregivers (*ibid*). These factors would suggest lower-intensity care, and, combining caregiving with other activities has been linked to higher satisfaction (van Campen *et al.*, 2013).

Moreover, Akpinar *et al.* (2011) found overall burden to be significantly higher in females, in addition to all sub-scales of burden, except emotional burden. The absence of differences in emotional burden contrasts with research that suggests males are less emotionally attached within the caregiving role (Calasanti and King, 2007), as this would imply that males and females experience similar levels of emotional burden. Notably, intensity within this study was extremely high: 21hpd for females, and 19hpd for males. Therefore, the high-intensity may have influenced burden, alongside the caregivers' sex.

The caregiving relationship is also an important factor contributing towards caregiver burden. A study comparing Korean-American spousal caregivers (N=47) and other relatives (N=8) with filial caregivers (N=91) reported significantly higher burden among spousal caregivers, than filial (Casado and Sacco, 2012). No significant difference was observed between other relatives and filial, although this may have been due to the substantially smaller sample of caregivers to other relatives. Similar

comparison was made amongst Finnish caregivers: whereby perceived burden was highest among spousal caregivers (N=660), than filial (N=143) and parental (N=259) caregivers (Juntunen *et al.*, 2018). However, differences in burden were only reported for the female caregivers within the sample (72%), which was foreseeable because there was little male representation for filial and parental in comparison with spousal caregivers. Nevertheless, there is evidence to suggest there are sex differences in caregiver burden.

In contrast, comparison was drawn between spousal (N=12) and filial (N=12) dementia caregivers in Australia (Tatangelo *et al.*, 2018). Although the interviews did not contain burden-specific questions, 'feeling overwhelmed' was commonly cited by filial caregivers arising from physical burden, emotional strain, and isolation. A greater percentage of filial caregivers identified unmet mental health needs compared to spousal caregivers, which would suggest higher levels of burden among filial caregivers. Moreover, concerns raised by spousal caregivers were related to worrying about recipients, rather than self-directed, which could be indicative of emotional burden. The authors conveyed that greater adjustments are often made by filial caregivers, as they are most likely to be employed, with their own family responsibilities, in addition to the caregiving role (*ibid*). Additionally, spousal caregivers tended to be older (aged 62-89), compared to filial (aged 36-70), which would suggest they may be retired, and are unlikely to have additional responsibilities (e.g. childcare), alongside their caregiving role. Overall, the quantitative evidence suggests spousal caregivers experience higher levels of burden than other caregiving groups. The qualitative study from Tatangelo *et al.* (2018) did not directly explore burden in the interviews, and therefore the findings are not considered robust enough to directly contradict what previous studies have found.

Finally, factors associated with the caregiving role, (e.g. caregiving intensity), have been shown to influence burden. Caregivers to cancer patients experienced similar levels of burden to dementia caregivers (Bevans and Sternberg, 2012). This suggests that additional care needs associated with chronic illness or cognitive impairments result in increased burden, compared to caregivers of older adults without chronic conditions. This could also be related to the relationship with the recipient. As such, when caregivers of children (N=100) were compared with equal groups of caregivers of adults, and older adults, using the Zarit Caregiver Burden Interview, caregivers of older adults provided the lowest intensity care at 15hpd, but exhibited significantly higher burden than caregivers of children and adults (de Oliveira et al., 2015). Alternatively, caregivers of children provided the highest intensity care (20hpd), but burden was significantly lower than that of older adults. Participants were recruited from a Brazilian medical centre, meaning care recipients were likely to have complex health needs. Thus, contradictory to the evidence from Bevans and Sternberg (2012), factors other than medical conditions may be differentiating caregiver burden. Additionally, although caregiving to older adults was identified as the lowest intensity, typically high-intensity is classified as  $\geq$ 20 hpw, so if they were providing care over multiple days, all caregivers in the sample were highintensity caregivers. It is likely that caregiving intensity contributed towards the burden scores, but that daily hours are less influential than the overall impact of high-intensity ( $\geq 20$  hpw) care.

#### 3.3.4.2 Satisfaction with caregiving

Satisfaction (defined in section 1.2.1.4) is often perceived as the opposite to burden. However, it is more accurate to articulate that caregivers experience burden and satisfaction simultaneously, as the role can be both difficult, but fulfilling (Dahlrup *et al.*, 2015). Iecovich (2016) reported that the correlation between caregiver satisfaction and burden was non-significant; the lack of association suggests the two may not be directly related. In contrast, García-Mochón *et al.* (2019) suggest that burden and satisfaction are related, but with distinct interpretations, and typically lower burden scores are indicative of higher satisfaction. However, it is possible to experience burden and satisfaction together, individually, or not at all.

As with burden, there is evidence of sex differences in the experience of satisfaction. Amongst **Spanish** caregivers (N=610), significant sex differences were recorded for certain domains of satisfaction (e.g. self-esteem), but not for general satisfaction (García-Mochón et al., 2019). However, in separate male (N=265) and female (N=345) analyses, differences in the predictor variables were apparent. Age and location were significantly associated with male satisfaction but not female, and among females but not males, education and use of social care services were significant predictors of satisfaction. However, males were older than females (mean age 62 versus 58), thus could have skewed the results. Evidence on Swiss caregivers (N=118) reported that females (N=86) experienced lower levels of satisfaction than males (Tough *et al.*, 2017). This aligns with the literature on burden, which suggests females experience higher levels of burden (Akpinar et al., 2011) and supports the conclusion by García-Mochón *et al.*, (2019) that higher burden is linked to lower satisfaction. Similarly to the discussion on burden, caregiving relationships were significantly associated with satisfaction. Caregivers with shared ancestry (e.g. filial or sibling) experienced greater satisfaction than those without (e.g. spousal or in-laws), after controlling for factors such as the caregiving relationship, burden, and dementia severity (de Labra et al., 2015). This aligns with previously discussed research that suggests spousal caregivers experience higher levels of burden, which could be indicative of lower role satisfaction.

Among Spanish caregivers (N=1,107) providing care <2hpd, 3-5hpd, and >5hpd, higher-intensity care was significantly associated with reduced odds of satisfaction, but only once controlled for other factors such as sociodemographic characteristics (Marcén and Molina, 2012). In contrast, when intensity was the only predictor, there was no significant association. Typically, caregiving intensity is measured in hpw, as opposed to hpd, which limits the comparability with other studies. Additionally, the authors speculate that caregiving tasks are based on the division of hours, but give little explanation about the significance of two, or five hours of care.

Overall, research investigating burden and satisfaction is largely focused on the caregiving relationship and role. Moreover, there is often a large female bias in the studies. Thus, it is not always clear what impact is due to caregiver characteristics (e.g. sex). There is evidence to suggest that male

and female caregivers experience different levels of burden, but much less focus on age differences, for example. Additionally, there is considerably more literature discussing burden than satisfaction, which arguably, may strengthen negative perceptions and stereotypes surrounding caregiving. Iecovich (2016) found associations between burden, satisfaction, and loneliness, but this referred to care recipient loneliness. As highlighted, there is little evidence to clarify the relationship between burden, satisfaction, and caregiver loneliness.

# 3.3.5 Summary of caregiving literature

Overall, the literature identifies that caregivers are most commonly aged around 50-years old (Vasileiou *et al.*, 2017; Zhang and Bennett, 2019), and are typically female, but with a higher percentage of male caregivers at older ages (Milligan and Morbey, 2016). There is limited evidence suggesting a direct relationship between marital status and care provision, but evidence suggests that marital status might influence the type of care provided (i.e. co-residential) or the recipient. Finally, the association between SES and caregiving appears to differ across countries; in a UK context, provision of co-residential care was an important factor linked to caregivers ceasing employment (Carr *et al.*, 2018).

Evidence suggests that spousal caregivers provide higher-intensity care (Oldenkamp et al., 2018b) and are more likely to provide co-residential care, compared to filial caregivers (Bédard et al., 2005). However, the literature review revealed that, largely, research is conducted with spousal caregivers, and there are few studies comparing spousal caregivers with other caregiving subgroups. Further research is therefore needed to compare caregiving subgroups; thus, this thesis contributes towards this gap by comparing spousal, parental, filial, and other caregivers. There is a larger body of evidence comparing intensity of care than its nature, or duration. For instance, research is typically conducted with current caregivers, therefore total duration is unknown. Nevertheless, periods of caregiving have been shown to be longer among caregivers to individuals with DD, compared to older adults, or individuals with mental illness (Anderson et al., 2018; Barnhart et al., 2020). Additionally, nature of care is not always clearly defined or is inconsistent within the literature. Care has been categorised as helping, looking after, or nursing (Zwar et al., 2018), or referred to as 'visible' and 'invisible' (Clark et al., 2008). Arguably, 'visible' care would cover all three delineations outlined by Zwar et al. (2018), making comparison difficult. Moreover, evidence suggests caregiving intensity is dependent on other factors, such as sex (Schulz et al., 2020), employment (Oldenkamp et al., 2018a), and recipients' needs (Costa et al., 2013).

Finally, caregiver burden is more widely researched than satisfaction. The literature suggests that both burden and satisfaction are influenced by caregiver characteristics, caregiving relationships and factors relating to the caregiving role. For example, burden was shown to differ by intensity and care recipient (de Oliveira *et al.*, 2015).

Literature would suggest that caregiving is not limited to one particular scenario. Caregivers differ by sociodemographic and socioeconomic characteristics, provide care to a range of recipients with variable needs, and can be high- or low-intensity. The aforementioned variations were also shown to influence perceived caregiver burden and satisfaction. Overall, when researching caregiving populations, these multiple factors should be considered. Accordingly, within this thesis, controls were included for caregiver characteristics, the caregiving relationship, and factors relating to the caregiving role to account for potential influences on caregiver loneliness, health and wellbeing.

# 3.4 Loneliness, health, and wellbeing

There is a vast range of literature reviewing the impact of loneliness on physical and/or mental health. However, the relationship between loneliness and health is complex. Evidence suggests loneliness influences health and wellbeing, but also that health and wellbeing are determinants of loneliness. Within this section, the relationship between loneliness, health and wellbeing are critically discussed. The aim is to disentangle whether loneliness contributes to poor health and wellbeing, if poor health and wellbeing leads to loneliness, or the extent to which simultaneity1 occurs.

### 3.4.1 Associations between loneliness, health, and wellbeing

Loneliness has been linked to increased morbidity of several chronic diseases, for example, hypertension, heart disease and stroke (Friedler *et al.*, 2015; Ong *et al.*, 2016), which reflect physiological pathways (see SET, section 2.1.2). Research across 52 countries suggests that psychosocial risk factors (e.g. loneliness) contribute to over 32% of cardiovascular disease (CVD) risk (Dafoe and Colella, 2016). In comparison, CVD risk from smoking is around 36%. However, more commonly, research investigates loneliness in relation to more general health measures.

There are several review articles examining the relationship between loneliness and health, identifying a large range of health outcomes. Alongside being a risk factor for mortality, loneliness is linked to psychological and physical morbidity, depression, functional limitations, poor physical health, and cognitive decline in older adults (Ong *et al.*, 2016). Additionally, a review on loneliness among the general population, not just older adults, reported associations between loneliness and depression, dementia, alcoholism, chronic stress, suicide, sleep disturbance and physical health

<sup>1</sup> Simultaneity: "the fact of something happening or being done at the same time as something elsewhere" (Oxford University Press, 2020) for example, both loneliness and health influence each other at the same time, loneliness contributes towards poor health, but poor health also impacts loneliness.

(Mushtaq et al., 2014). However, in a review on loneliness, social isolation and health in which 68 articles (53%) focused on loneliness in relation to health, results were inconclusive (Courtin and Knapp, 2017). Several articles found loneliness, but not social isolation, to be significantly associated with increased mortality risk in older adults, while others reported both loneliness and social isolation as mortality risk factors. Examples of these associations have also been investigated in large representative studies across a range of countries. In Denmark, loneliness had different effects for middle-aged and older adults, whereby greatest impacts on physical ability was observed for middleaged individuals (Jessen et al., 2017). Once controlled for sociodemographic characteristics and SES, lonely middle-aged individuals showed the greatest risk of limited physical ability, compared to nonlonely individuals of the same age. Moreover, results from the Swiss National Survey indicate that the highest risk of poor health was for lonely individuals aged 30-59, but the association was also significant for ages 15-29 and  $\geq 60$  (Richard *et al.*, 2017). This would indicate that loneliness has greater impact on health among middle-aged rather than older adults, and the reasoning for this may be relevant to the health measures employed. Many older adults consider illness or health decline to be a normal part of ageing (Hasworth and Cannon, 2015), which therefore may reduce their likelihood to report poorer health in self-report measures. Associations between loneliness and poor health were also present in a sample of older New Zealanders (N=332); separated into three groups (non-lonely, moderately lonely and severely lonely), both physical and mental health were worse among the severely lonely (La Grow et al., 2012). A limitation of this cross-sectional study is that the loneliness and health data were obtained at the same time-point. Therefore, it is unclear whether loneliness may have resulted in poor health, or whether poor health was a factor for increasing loneliness severity. Alternatively, Losada et al. (2012) focused solely on mental health implications of loneliness for community dwelling, Spanish older adults. The results followed a similar pattern to other literature, whereby loneliness was negatively associated with mental health.

#### 3.4.1.1 Simultaneity: the impact of health on loneliness

A review by Ong *et al.* (2016) highlighted key research on loneliness and health. Its focus was predominantly on the impact of loneliness on health, however, it was identified that accumulation of chronic illness, mobility impairments, and poor physical health are risk factors for loneliness. Cacioppo *et al.* (2002) conducted two US-based studies, one with undergraduate students (aged 18-24), the other with middle-aged and older adults (aged 53-78). In both studies, health behaviours (e.g. alcohol consumption) were not significantly different among lonely and non-lonely respondents. This would suggest that loneliness has little influence on the behavioural SET health pathway. However, blood pressure was significantly higher in the lonely-older (aged 65-78), compared to lonely-younger group (aged 53-64). These results suggest, particularly among older adults, CV activity (physiological pathway) was associated with loneliness, but this may be accounted for by age differences. Analysis of TILDA also supported the link between health and loneliness: in a large, nationally representative sample of Irish older adults (N=6,613), the direct
effect of poor health on loneliness was significant (Burholt and Scharf, 2014). However, the association was stronger when variables such as depression and social resources were included. This indicates health directly affects loneliness, but it also has indirect effects, as the combined impact of health and other variables was stronger than health alone. Contrasting evidence from a large representative US sample (N=2,261) suggested that health was not significantly associated with changes in loneliness (Hawkley and Kocherginsky, 2018). Longitudinal evidence demonstrated that self-reported health was not significantly predictive of loneliness, but functional limitations were. However, further analysis indicated that functional limitations were only significant predictors of becoming lonely, not recovering from loneliness. Therefore, difficulty with ADLs can lead to loneliness in older adults but improving functional health does not significantly reduce loneliness. It could be interpreted that poor health leads to irreversible loneliness outcomes, perhaps suggesting that the link between health and loneliness is stronger than that between loneliness and health.

Furthermore, expectations and stereotypes about loneliness (SET psychological health pathway) have been linked to loneliness outcomes. In a sample from ELSA (N=4,465), negative stereotypes and expectations about future loneliness collected in wave two were positively associated with loneliness in later waves (3-6) (Pikhartova *et al.*, 2016). Self-rated health and depression were also indicative of developing loneliness. Lonely respondents at baseline were excluded, this clarified the direction of the relationship, in that poor health led to loneliness, as opposed to loneliness negatively affecting health. This emphasises the importance of the psychological health pathway.

In summary, a greater body of evidence investigates the directional relationship between loneliness and health, compared to health and loneliness. However, there are many inconsistencies within the results. One reason could be the different measurement tools used to quantify health and loneliness: validity of measurement tools differs among populations (e.g. older adults). It is evident in the literature that mediating factors, such as sociodemographic characteristics, significantly affect relationships, in many cases strengthening the association. For this reason, it is important to consider any factors that may contribute towards loneliness and/or health.

#### 3.4.1.2 Wellbeing and loneliness

Largely, in loneliness literature, aspects of physical health and mental health are more broadly researched than wellbeing. Despite this, evidence on the association between loneliness and wellbeing is consistent in reporting that loneliness is negatively associated with wellbeing (Tough *et al.*, 2018; Emerson *et al.*, 2020).

Loneliness appears to have similar effects on wellbeing regardless of sociodemographic characteristics (e.g. age). Data from three waves of CLS demonstrated loneliness was more strongly related to personal wellbeing (including life satisfaction) among ages 16-64 (N=17,723) than social interaction or perceived social support (Emerson *et al.*, 2020). Similarly, for a sample of Irish older adults (N=1,299, aged  $\geq$ 65), loneliness was linked to lower happiness and life satisfaction. Another

common theme was that loneliness and wellbeing were strongly linked to disability (Golden *et al.*, 2009; Emerson *et al.*, 2020). Moreover, Arslantas *et al.* (2015) measured the impact of loneliness on QoL for Turkish older adults (N=4,170, aged  $\geq$ 65), where loneliness was significantly associated with all subscales of the QoL Short Form (SF-36) including general health perceptions, social functioning, and mental health. Respondents with chronic disease or physical disability reported higher loneliness scores, which would imply lower QoL, given the significant negative association.

In contrast Gerino *et al.* (2017) found no significant direct effect of loneliness on wellbeing, as measured by the WHO Quality of Life Instruments. Loneliness was significantly associated with QoL, but only through mediators, such as mental health and resilience. This suggests that the relationship between loneliness and wellbeing is complex and interrelated with other factors, such as health. However, it must be considered that a key issue with wellbeing research is a lack of uniformity in wellbeing measures. Both measures and content vary substantially, some including subscales on health-related aspects, such as physical functioning and mental health (e.g. SF-36, Arslantas *et al.*, 2015), others focusing more on life satisfaction and happiness measures (Tough *et al.*, 2018; Emerson *et al.*, 2020), while others measure QoL (Gerino *et al.*, 2017). As there is little clarity over the definition of wellbeing (section 1.2.3.2), it is therefore difficult to compare wellbeing studies. Nevertheless, loneliness is consistently reported as detrimental to a range of wellbeing measures.

# 3.4.2 Loneliness, health, and wellbeing in a caregiving context

The following sections discuss caregiving literature, in relation to loneliness (section 3.4.2.1), health (section 3.4.2.2), and wellbeing (section 3.4.2.3) outcomes. The review considers different aspects of caregiving (i.e. characteristics, relationship, or role) reported to influence loneliness, health, and wellbeing and, where the health literature permits, relate this to the SET pathways.

# 3.4.2.1 Caregiver loneliness

Previous research suggests that females are more likely to experience loneliness or other difficulties associated with caregiving (Navaie-Waliser *et al.*, 2002; Vitaliano *et al.*, 2003). Moreover, evidence from Beeson (2003) highlights that loneliness and depression are significantly higher among female than male caregivers, aligning with the evidence suggesting that females are more susceptible to negative experiences of caregiving. However, male and female caregivers did not differ in relation to negative relationship changes with the recipient, or in experiencing loss of identity (*ibid*). Therefore, as outlined by CIT, changes that occur to the caregiver-recipient relationship, and adaptations to identity due to the caregiving role do not appear to discriminate by sex. Similarly, female caregivers were lonelier in samples of Turkish caregivers (N=100) of recipients with cancer (Soylu *et al.*, 2016) and British dementia caregivers (N=1,283) (Victor *et al.*, 2020). Soylu *et al.* (2016) reported significantly higher loneliness scores for female compared to high school and

university educated caregivers. Qualitative investigation on loneliness among English caregivers (N=16, 69% female) would also suggest that loneliness was more common among the female caregivers (Vasileiou *et al.*, 2017). When discussing the key themes, issues arose around reduced personal space, relational losses, social interactions, and sole-responsibility. However, it is important to note that there were a greater number of female caregivers, and all male caregivers were spousal, whereas female caregivers included spousal, filial, and parental. Therefore, as discussed below, relationship type may have influenced their caregiving experiences and loneliness level.

Moreover, Bramboeck *et al.* (2020) measured loneliness using UCLA-LS and German Multidimensional Loneliness Questionnaire (MEF). Both measures were associated with caregivers' age, sex, and living circumstances. In contrast to previous findings, among the caregivers (N=40) of people with Alzheimer's disease, male caregivers experienced higher levels of loneliness than females. Little is known about the caregiver-recipient relationship, other than that they were related, and sociodemographic information was not reported by sex; therefore, it is not possible to determine whether other relevant factors, were contributing towards the higher loneliness scores. Overall, the majority of the evidence suggests that female caregivers typically experience greater loneliness than male caregivers. Research conducted with non-caregiving populations report mixed findings (section 3.2.1), whereby significant sex differences often rely heavily on contextual factors, such as the loneliness measure, or other sociodemographic characteristics. This could be relevant among caregiving populations too, whereby in addition to sociodemographic discrepancies, caregiving relationship and role differences may contribute towards observed sex differences in loneliness.

Alternatively, there is evidence to suggest that factors related to the care recipient contribute towards caregiver loneliness. Longitudinal research from France investigating daughters (N=557) providing care to widowed parents suggests that care recipients' health was linked to caregiver loneliness, but that care provision was not significantly associated with caregiver loneliness (van den Broek and Grundy, 2018). It could be argued that health limitations of the care recipients (parent) means a lower likelihood of support provided to their adult-children (*ibid*), ensuing stress and increased need for a support network, thus resulting in feelings of loneliness. Other factors shown to contribute towards caregivers were lonelier than non-residential caregivers (Bramboeck *et al.*, 2020), and spousal caregivers were lonelier than caregivers to other family and friends (Victor *et al.*, 2020).

Bramboeck *et al.* (2020) recruited their sample from an inpatient ward, thus the prevalence of coresidential care was much lower than typically reported in other studies (section 3.3.2.2). Nevertheless, the results showed that co-residential care was associated with increased loneliness compared to non-residential care. Moreover, the experience of loneliness may have been greater than usual for those providing co-residential care: because at the time of the study, the care recipients were hospital inpatients, therefore not residing at home with the recipient, which may have influenced feelings of loneliness. Furthermore, qualitative interviews with Polish spousal caregivers (N=30)

indicated that behavioural changes linked to the recipients' Alzheimer's disease, contributed towards breakdown in communication and reduced relationship quality (Leszko *et al.*, 2020). Consequently, the caregivers reported experiences of loneliness due to the loss of the friendship and partnership. Additionally, the caregivers suggested that as the condition progressed, care needs and time spent caregiving increased accordingly, which exacerbated feelings of loneliness by reducing the caregivers' ability to socialise outside of the caregiving role.

Victor et al. (2020) found significant associations between factors such as depression, stress, life satisfaction and wellbeing and caregiver loneliness. Higher levels of depressive symptoms and stress were associated with greater loneliness severity, whereas lower life satisfaction and wellbeing were linked to more severe loneliness. The evidence would suggest that similar to the non-caregiving population, loneliness among caregivers is linked to poorer health and wellbeing. For example, loneliness among caregivers has been linked to frailty, which has been shown to predict other health outcomes in older adults (dos Santos-Orlandi et al., 2019). In a sample of predominantly spousal caregivers (aged  $\geq 60$ ) who had been caregiving for an average of 6hpd over 10-years, odds of frailty among lonely caregivers were over four times higher than non-lonely caregivers (*ibid*). However, there were very few studies that examined the link between loneliness, health and wellbeing among caregivers, therefore, currently little comparison can be drawn with other findings. Largely, the concepts have been investigated in isolation. However, as discussed in the subsequent sections (3.4.2.2. and 3.4.2.3), caregiving has been linked to many health and wellbeing consequences. Therefore, further research is needed to investigate loneliness, health, and wellbeing measures simultaneously to aid the understanding within a caregiving context. This thesis therefore aims to help fill that knowledge gap. The subsequent section discusses literature on further health implications of caregiving.

# 3.4.2.2 Caregiver health

Health consequences of caregiving have been shown to differ by age. Among US caregivers, caregivers aged 18-64 experienced significantly more mental distress and reported lower life satisfaction compared to caregivers aged  $\geq$ 65 (Anderson *et al.*, 2013); however, caregivers aged  $\geq$ 65 reported poorer self-rated health and more frequent physical distress. The different outcomes suggest that caregiving has a greater impact on physical aspects of health in older caregivers, but appears to have greater psychological impact on younger caregivers. Similarly, Danilovich *et al.* (2017) found that US caregivers aged 50-64 were 65% more likely to perceive that caregiving had a negative impact on self-rated health, compared to younger caregivers (aged 18-49), also suggesting that the physical consequences of caregiving are more pronounced as caregivers age.

Alternatively, Tosi and Grundy (2019) hypothesised that career breaks to provide care to family would benefit the health of women, compared to those not working and those working continuously, however their results provided no support for this. The latent class analysis highlighted two groups

of women likely to have taken breaks to provide family care (>5-years): (i) those categorised as being in a lower socioeconomic position who started an early (aged <20) large family ( $\geq$ 3 children), and (ii) those categorised as being in a low socioeconomic position who took work breaks and provided family care. Of the two groups, the percentage providing family care was 58% and 93% respectively. The physical and mental health of the second group with the larger percentage of caregivers did not differ significantly from the reference group (women in a low socioeconomic position with a long working life and two children), but those in the first group demonstrated significantly lower physical and mental health, compared to the reference group. The results suggest that the negative effect on health was not necessarily due caregiving, as there were a lower percentage of caregivers in the first group compared to the second, or the lower socioeconomic position given that all three aforementioned groups shared this characteristic. This therefore suggests caregiving during an individual's working life had little impact on health in later-life, which would imply that the effect of caregiving on health is short-term rather than long-term.

In addition, caregiver health has been linked to the caregiving relationship and role. As such, significant differences in depressive symptoms and mortality were reported between co-resident, non-resident, and non-caregivers in the US (Caputo *et al.*, 2016). All co-residential caregivers, regardless of recipient (spouse, parent, or other), experienced significantly higher levels of depression than non-caregivers. In comparison, non-resident caregiver depression scores were not significantly different to non-caregivers. However, mortality did not reflect the same pattern: the mortality rate for non-resident caregivers was significantly lower than non-caregivers, and only co-residential caregivers of 'other' recipients experienced significantly higher mortality rates, compared to non-caregivers. The results infer that non-residential caregiving may be protective against mortality; however, it is important to consider that low-intensity caregiving has been linked to health benefits among caregivers, in comparison with higher-intensity care (Vlachantoni *et al.*, 2016; Lacey *et al.*, 2018), and non-residential care is typically indicative of lower-intensity care, than co-residential care.

Increased duration of caregiving roles has previously been linked to psychological morbidities, such as depression (O'Reilly *et al.*, 2008). Conversely, a comparison drawn from UK Censuses over a 10-year period suggest that those in long-term, high-intensity caregiving roles were less likely to report poor health, compared to persistent non-caregivers (Vlachantoni *et al.*, 2016). Upon reflection, it could be considered that poorer health is reported by non-caregivers, compared to long-term caregivers, due to their capacity to care. To clarify, caregivers consistently provide care because they are physically able to do so, and if a caregiver experiences poor health, they may need to cease their caregiving role due to their own health limitations. As shown by Vlachantoni *et al.* (2016), those transitioning from high-intensity caregivers. Similarly, in a sample of US caregivers, duration of caregiving role was associated with poorer self-reported health (Danilovich *et al.*, 2017), compared to caregivers who had provided care for <6 months. Caregiving for 1-4 years, 5-9 years and  $\geq$ 10 years significantly

increased the risk of caregivers reporting worse health. It is noteworthy that this relationship only appears to take effect after a year of caregiving, as caregiving for 6-12 months, compared to <6, was not significantly associated with an increased risk of poorer health. This suggests that short-term caregiving roles <12 months have little impact on the health of caregivers, neither positively or negatively.

Finally, Lyons *et al.* (2015) evaluated stress and depression among a sample of 992 US females. Perceived stress was significantly higher among high-intensity caregivers, compared to low-intensity and non-caregivers, who exhibited similar scores. Additionally, depression scores were lowest among low-intensity caregivers in comparison with non-caregivers and high-intensity caregivers (*ibid*). These studies correspond with previously discussed research that low-intensity caregiving is beneficial to the health of caregivers. Vlachantoni *et al.* (2016), and Lacey *et al.* (2018) argue that caregiving for  $\geq$ 20hpw resulted in poorer health outcomes compared to fewer caregiving hours. Similarly, Northern Irish Census data highlighted that male caregivers providing  $\geq$ 50hpw were more likely to report a limiting long-term illness (LLTI), compared to non-caregivers (O'Reilly *et al.*, 2008). Further analysis considering self-rated general health indicated that high-intensity caregivers were more likely to report poorer health, compared to non-caregivers regardless of sex. This evidence would suggest that caregiving intensity has a more substantial effect on caregiver health, after accounting for sociodemographic characteristics.

#### Stereotype embodiment theory: pathways to health

Little research focuses on caregiving and health in relation to SET pathways, although Bevans and Sternberg (2012) highlight the importance of recognising the effects of physiological, psychological, and behavioural health outcomes among caregivers to prevent more serious health conditions and premature mortality.

Caregiver stress has been linked to increased risk of heart disease and stroke among females (Bouchard *et al.*, 2019). The impact of caregiving on physiological health was reportedly similar for both spousal and parental caregivers. Compared to caregivers reporting low or no stress, strained spousal and parental caregivers more likely to experience fatality as a result of CVD. Moreover, using pooled data from Swedish and UK-based surveys, high-intensity caregiving (>20hpw) was linked to increased CVD risk among caregivers (Mortensen *et al.*, 2017). When combined with working hours, a combination of higher-intensity care (>8hpw) and part-time work ( $\leq$ 40hpw) was found to significantly increase CVD risk, compared to lower-intensity care and part-time work. In contrast, full-time working (>40hpw) regardless of caregiving intensity was not significantly associated with CVD. This suggests that stressors associated with caregiving may have a more substantial impact on physiological health than work-related stressors (*ibid*). Within the UK sample, longer-term care (caregiving at both time-points three-years apart) was associated with increased CVD risk for caregivers providing high-intensity care (>20hpw), compared to  $\leq$ 8hpw. Risk was not significantly different for short-term caregivers at any intensity, or long-term caregivers providing

9-20hpw care. These results suggest that high-intensity care over a prolonged period may contribute to higher levels of caregiver stress, and therefore increase the risk of poor physiological health among caregivers. As discussed previously, the evidence of the relationship between the caregiving role and health is largely inconsistent, which suggests that other contextual variables, such as differences in the caregiver relationship, or caregiver characteristics, may moderate the relationship between caregiving role and physiological health.

With regard to the psychological pathway, stereotypes about competence in ageing can act as a self-fulfilling prophecy (Fernández-Ballesteros *et al.*, 2016). As such, caregiving experiences have been shown to influence planning future care needs. Among Canadian respondents (N=180 non-caregivers, N=74 caregivers, N=92 caregivers with experience transferring their recipient to formal long-term care), both groups of caregivers were significantly less likely to avoid planning their own care needs, in comparison with non-caregivers. However, caregivers with formal care experience were also significantly less likely to avoid planning their own future care, than caregivers without formal care experiences. This indicates that personal experience with formal long-term care influences the decision to plan for future care needs more so than informal caregiving experience, perhaps due to the increased complexity of the recipient care needs for those transferred to formal care transfers, influence the caregivers' own health perceptions and increase the likelihood of expecting to require care.

Finally, caregiving has also been associated with behavioural health outcomes. World Health Survey data across 38 low- and middle-income countries (N=204,315) in 2002-04 indicated that caregivers were more likely to be physically active than non-caregivers (Jacob et al., 2020). This was consistent with evidence from Gottschalk et al. (2020) who reported from a large US sample of caregivers (N=12,044) and non-caregivers (N=45,925) that caregivers were significantly less likely to be physically inactive compared to non-caregivers. Arguably, the level of physical activity among caregivers may be directly related to the caregiving role, in that they may consider manual caregiving tasks to form the majority of this activity, rather than exercising outside of the caregiving context. Nevertheless, caregiving appears to have a positive impact on behavioural health when measured by physical activity. Similarly, caregivers consume alcohol more responsibly than non-caregivers (*ibid*). Within the literature, the body of research mainly concentrates on the consequences of over-drinking, with less research on the positive aspects, such as the release of endorphins and increased functionality in social situations (Dunbar et al., 2017). Moreover, there is evidence to suggest that moderate consumption of alcohol, particularly those who drink in a community atmosphere at a 'local' may be more socially engaged, happier and have higher life satisfaction. A simultaneous link may also exist between life satisfaction and drinking, in that those with higher satisfaction are more likely to visit a pub, and thus experience the social benefits (*ibid*). However, it is important to consider that the aforementioned study was not conducted with caregiving populations, and caregivers are less likely to have leisure time alongside their caregiving role to drink in pubs, so are

less likely to experience the social benefits outlined in the study. Studies suggest that alcohol consumption among lonely individuals may be a coping mechanism (Wilson and Moulton, 2010), and among caregivers, burden was linked to problematic drinking (Rospenda *et al.*, 2010).

In contrast, other aspects of behavioural health, such as obesity and smoking were more common among caregivers (Gottschalk *et al.*, 2020). However, factors pertaining to the caregiving relationship and role had significant effects on the behavioural health of caregivers. Compared to spousal caregivers, care to parents, children, grandparents, grandchildren, and other relatives were associated which higher likelihood of obesity and excessive alcohol consumption. Additionally, high-intensity care ( $\geq$ 40hpw) was linked to greater likelihood of obesity and smoking compared to caregivers of <9hpw. Principally, the literature suggests that caregiving positively influences the physical activity levels of caregivers, but this does not necessarily equate to good behavioural health, because negative health behaviours such as smoking, alcohol consumption, and obesity are prevalent within the caregiving population. This would emphasise a need for a multifactorial approach to measuring behavioural health to determine an overall effect. As such, evidence considering physical activity alone might conclude that caregiving has positive effects on behavioural health without investigating a more complete picture.

# 3.4.2.3 Caregiver wellbeing

This final section discusses the literature on the wellbeing implications of caregiving. There is less evidence connecting caregiving to wellbeing than that of health, and the relationship appears more consistent: caregiving is frequently shown to be linked to poorer wellbeing (Ekwall *et al.*, 2005; Dahlrup *et al.*, 2015; Sin *et al.*, 2016). However, there is also evidence to suggest that the positive aspects of caregiving, when researched in isolation, can have positive effects on wellbeing (Quinn *et al.*, 2019; Quinn and Toms, 2019).

Compared to the general population, caregivers exhibited significantly lower wellbeing scores (Sin *et al.*, 2016). Wellbeing was substantially lower among a sample of caregiving siblings (N=90, aged 16-58), than a comparable sample (N=2746, aged 16-54) of the general population from Health Survey for England (HSE) (*ibid*). However, once stratified by sex, caregiver wellbeing scores were significantly lower than the general population for females, but not males. Despite this, further analysis suggested that sociodemographic characteristics (including sex) were not influential on wellbeing scores among caregivers. This means male and female wellbeing was not significantly different within caregivers, but sex differences were apparent, compared to the general population.

Qualitative evidence, collected from caregivers of individuals with advanced cancer indicated that caregiver QoL was indistinguishable from care recipients' QoL (McDonald *et al.*, 2018). A key theme that emerged from the study was that caregivers were "living in the patient's world" (*ibid*, p.71), caregiving burden and that of the cancer diagnosis influenced the caregivers emotionally, physically, socially and financially, thus having a large impact on their QoL. However, this study

was conducted with caregivers of patients receiving, or eligible for palliative care, therefore it could be considered that caregivers to recipients without a terminal condition (e.g. caregivers to older adults at earlier stages of health conditions) may experience less burden and thus less consequence on their QoL.

Additionally, among Swedish caregivers (N=369) and non-caregivers (N=2,233) caregiver burden (high- and low-strain) significantly impacted wellbeing, but the impact of caregiving itself was not significant (Dahlrup et al., 2015). To illustrate, high-strain caregivers demonstrated significantly lower life satisfaction scores, compared to both non-caregivers and low-strain caregivers, whereas low-strain caregivers demonstrated higher life satisfaction than both non-caregivers and high-strain caregivers. Overall, the low-strain group could also be considered low-intensity, as they spent less time caregiving than high-strain caregivers. Therefore, the implications of the wellbeing research supplement the findings of Vlachantoni et al. (2016) and Lacey et al. (2018), suggesting that lowintensity caregiving may also be beneficial to caregiver wellbeing as well as health. Furthermore, in Switzerland, Tough et al. (2018) examined disabled (spinal injury) respondents (N=123) and their caregiving partners. Disabled respondents reported lower average loneliness and life satisfaction scores than their caregivers. Both loneliness and relationship quality with the care recipient were associated with caregiver life satisfaction scores. This suggests that disability, or perhaps health in general, is strongly associated with wellbeing. However, the results also imply that aspects of social networks (e.g. relationship quality), often measured alongside loneliness, appear more relevant in caregiving populations. As discussed in section 3.2.2.2 relationship quality is a key determinant of loneliness among caregivers, but this additional evidence suggests it may also be linked to caregiver wellbeing.

Alternatively, positive aspects of caregiving (PAC) are under-reported in comparison to negative aspects (de Labra et al., 2015; García-Mochón et al., 2019). Moreover, PAC may have beneficial effects on caregiver wellbeing. Quinn and Toms (2019) produced a systematic review (N=59 publications) exploring the impact of PAC on wellbeing among dementia caregivers. PAC were linked to higher QoL, increased life satisfaction, greater self-efficacy, and reduced burden. However, PAC were determined using a range of twenty different scales, therefore comparability between the studies may be limited. Additional evidence among dementia caregivers supported the link between PAC, QoL, and life satisfaction (Quinn et al., 2019). Furthermore, in some European countries it has been highlighted that state provision of long-term care may influence caregiver wellbeing: evidence using SHARE data (Sweden and Denmark) over multiple time-points demonstrates that caregiving was negatively associated with QoL among all caregivers, compared to non-caregivers, with negligible difference in QoL between those caregiving at higher-intensities (daily/every week) and those providing care less often (van den Broek and Grundy, 2020). This suggests that caregiving intensity was not linked to caregiver wellbeing. However, the study suggested that as state provision changed between the waves, the effect of care on caregivers' QoL was weaker when state provision was more readily available. The authors suggest this could be linked to obligation to care, as in

countries with generous long-term care provision, informal care is likely provided through choice, not necessity. As suggested by Li and Lee (2020) increased choice around caregiving is linked to fewer psychological symptoms among caregivers.

Overall, the evidence suggests a strong link between caregiving and wellbeing, but the concept of wellbeing is poorly measured within caregiving research. In the systematic review by Quinn and Toms (2019), burden was considered a wellbeing outcome, whereas Dahlrup *et al.* (2015) employed burden as a predictor of QoL. Some studies considered health and mental health as wellbeing outcomes, but, as discussed in section 1.2.3, although linked, health and wellbeing are separate constructs. Typically, it is understood that positive aspects of caregiving have beneficial effects on wellbeing, and negative aspects of caregiving, such as stress or burden, can have negative wellbeing consequences.

# 3.5 Chapter summary

This chapter identified and critically discussed relevant literature to this thesis, including: determinants of loneliness, key sociodemographic and socioeconomic characteristics of caregivers, the caregiving relationship, factors related to the caregiving role, and loneliness, health and wellbeing in the general population and among caregivers. Research into determinants of loneliness suggests that sex, marital status, and social networks are significant predictors of loneliness. However, the impact of sex appears, in some cases, to be conditional upon mediating factors such as age, and the association can easily be influenced by the measurement tools used to determine loneliness. Moreover, marriage appears largely protective against loneliness, but evidence on social networks suggest that all aspects contribute significantly to a loneliness outcome, including network size, relationship quality and satisfaction. Therefore, this would imply that the existence of a marriage alone does not necessarily protect against loneliness, and that quality or satisfaction within the marriage could be an important factor. This evidence could be an indication why prevalence of loneliness is higher among caregivers, and for potential discrepancies between the determinants of loneliness. To illustrate, large percentages of caregivers are female, and caregiving is shown to impact relationship quality and social networks. This would therefore suggest that caregiving may amplify these determinants of loneliness. Moreover, the literature suggests that caregiver characteristics, the caregiving relationship and the caregiving role are interlinked. For example, significant associations were reported between the age of a caregiver and the intensity of care provided, and in particular, spousal caregivers broadly differ from other caregiving subgroups, in relation to both characteristics and role.

Loneliness has proven detrimental to multiple aspects of health and wellbeing, while health has also been linked to loneliness outcomes, suggesting a high level of simultaneity between the concepts. The measures of health and wellbeing vary considerably within the literature, but typically focus on similar aspects, such as mental health and depression, general self-rated health, or areas such as life satisfaction. Arguably, there is less detailed evidence on health outcomes, for example, the impact on specific health conditions. Finally, research within the caregiving population examines the link between caregiving, loneliness, health, and wellbeing predominantly as distinct relationships. Negative aspects of caregiving (e.g. burden) have been linked to poorer health and wellbeing consequences. However, although less researched, positive aspects have demonstrated promise in areas of health and wellbeing. For example, highly satisfied caregivers are typically those who provide lower-intensity care, which has been reported to have positive health effects.

#### **3.5.1** What are the research gaps?

Fundamentally, there is a substantial lack of cross-sectional, and longitudinal research carried out in caregiving populations. Where caregiving populations are investigated, this is often limited to one recipient health condition (e.g. dementia), or one type of caregiver (e.g. spousal). Therefore, diversity within the caregiving population is largely under-researched and subsequently, under-represented in the literature.

More specifically, there is a lack of evidence on the determinants of loneliness among caregivers. Thus, despite caregivers being frequently reported as vulnerable to loneliness, there is little robust evidence examining whether typical determinants of loneliness within the general population are also applicable to caregivers. As concluded in section 3.2.5, the body of literature is limited, but the available evidence suggests that many of the determinants of loneliness in the general population may not be applicable to caregivers. This means that caregiving support services cannot work to effectively prevent loneliness, because risk factors and indicators are not clearly understood within this subgroup of the population. Loneliness prevention would, arguably, be more beneficial than supporting caregivers to overcome loneliness, because the likelihood of negative consequences on health and wellbeing would be significantly less.

Moreover, greater clarity is needed around the division of short- and long-term caregivers or lowand high-intensity caregivers. The latter, considerably more researched than the former, has proven to be largely influential on caregiver loneliness, health, and wellbeing. However, within the literature, what constitutes high-intensity ranges from  $\geq 10$  hpw to  $\geq 20$  hpw, which is misleading when comparing the implications. Finally, there is considerable research on the health implications of caregiving, and lesser, but expanding bodies of evidence on loneliness and wellbeing among caregivers. However, little research has been conducted on the concepts simultaneously. Despite evidence from the general population suggesting strong links between loneliness and health, loneliness and wellbeing, and health and wellbeing, few studies have explored whether the health and wellbeing implications of loneliness differ for caregivers. As caregiving is known to impact loneliness, health, and wellbeing, it could be expected that the combined effect of caregiving and loneliness would have greater impacts on health and wellbeing than caregiving or loneliness individually, which highlights a substantial research gap.

Moreover, there are a vast range of health and wellbeing measures applied within the literature. Often these are general health measures (e.g. self-reported general health) or omnipresent factors such as depression and other mental health outcomes. Further evidence is needed on more specific and scarcer health measures, such as provided by the SET pathways. Additional research on the impact of caregiving and loneliness on specific areas of health, such as physiological health conditions, health behaviours and perceptions about future health can expand the understanding of the link between loneliness and health and provide greater scope for support services and policy recommendations for areas of health to monitor.

# 3.5.1.1 Original contribution

This thesis seeks to address the aforementioned research gaps by including a diverse range of caregivers, including spousal, parental, filial, and other, without restricting the analyses to caregivers of people with a particular health condition or disease. Moreover, comparing determinants of loneliness in caregiving and non-caregiving subgroups seeks to clarify which factors are associated with loneliness among caregivers, and whether these differ from the general population. Reported prevalence of loneliness can differ, depending on the loneliness measure used, which has largely been evidenced by comparing males and females. It could be hypothesised that a similar division would be observed among caregivers, whereby the stereotype around caregiving and loneliness (e.g. that caregivers should not be lonely because they spend time with the care recipient) could act as a barrier for caregivers directly reporting loneliness; alternatively, because caregivers are predominantly female until older ages, direct loneliness measures may show little contrast to indirect scale measures within this subgroup. This thesis aims to investigate whether discrepancies between loneliness measures are also relevant among caregivers. Finally, by way of uncovering and clarifying relationships between loneliness, health, and wellbeing among caregivers, the concepts will be examined simultaneously, and in various combinations, to contribute towards the knowledge gaps. Moreover, health theories (SET) are applied to gain a more structured insight into specific health outcomes. This aids the disentanglement of the relationships and provides a clearer framework for the development of policy and practice recommendations. As such, identifying significant impacts on behavioural health may lead to more targeted recommendations than those from self-rated general health measures. The next chapter outlines the methodology applied to address these research gaps.

# Chapter 4 Methodology

# 4.1 Chapter outline

This research adopts a quantitative approach, using secondary data analysis (SDA) to analyse the ELSA dataset cross-sectionally and through follow-up analysis. Chapter four explains the methodological decisions, justifying the methods and materials using published literature. The selected research strategies are outlined and research materials discussed in relation to comparable datasets (section 4.2). Moreover, key variables are identified (section 4.3), including those computed for the analysis, and weights. Subsequently (section 4.4), the quantitative methods for both the cross-sectional and follow-up analyses are presented.

# 4.2 Research strategy

The quantitative methods of this thesis are deductive, primarily following a positivist epistemology (Bryman, 2016). As the main aims of this thesis are to examine associations and identify determinants, this is most effectively achieved in larger samples using numerical data; therefore, a quantitative approach is the most appropriate. Quantitative research can be used to test predetermined hypotheses, whereas qualitative methods often seek to develop deeper understanding, such as understanding why something occurs (McCusker and Gunaydin, 2015). Section 4.2.1 discusses the advantages and disadvantages of secondary analysis, cross-sectional and follow-up methods. Moreover, section 4.2.2 introduces the data, and the study population examined within this thesis.

# 4.2.1 Research design

Secondary analyses are conducted on multiple waves of ELSA (section 4.2.2) through two research designs: cross-sectional and follow-up. These designs are complimentary, while the first can be used to identify associations between variables and prevalence of a phenomenon (e.g. loneliness among caregivers), the latter can further investigate the time-order of associated variables (e.g. is caregiving in an earlier wave, indicative of reporting loneliness in a later wave?).

# 4.2.1.1 Secondary analysis

The main advantages of SDA include high-quality data, reduced burden on potential participants, cost and time effectiveness, and the opportunity for new interpretations (Bryman, 2016). Many of the existing SDA datasets are high-quality (ESRC, 2019) with rigorous sampling procedures; this means that the study sample is often representative of the general population, thus conclusions are frequently more generalisable than results from smaller studies using less diverse primary data. To

clarify, time and cost associated with collecting a large representative sample often means that primary data collection is limited to desired characteristics. Smaller studies may focus less on matching demographic characteristics to the general population (e.g. proportions of age categories), and collect participants with a specific attribute (e.g. caregivers for individuals with certain health conditions). This would restrict the generalisability of the results to the specific characteristic, or context, within the study.

For data collected with the intention of SDA, informed consent is obtained from the respondents for the re-use of their data. The respondents can withdraw their data at any time, but obtaining initial consent for re-use is essential; otherwise, respondents would need to be contacted each time a researcher analysed the dataset. This is likely to be infeasible and result in excessive burden on the respondents, potentially reducing the likelihood of future participation. However, the re-use of collected data, such as that in ELSA, can reduce this burden on potential participants; individuals are aware that data are collected on a biennial basis, and therefore do not have to go through the recruitment process at each wave. Furthermore, existing SDA datasets often have a larger number of participants than would be available to individual researchers collecting primary data (Windle, 2010); this reduces the costs associated with data collection and is substantially less time-consuming, thus more efficient. Finally, Bryman (2016) suggests that social research is often under-analysed, meaning that the full potential of data resources are yet to be explored. New interpretations could occur through using a more exhaustive range of the available variables, analysing different combinations of variables, or exploring different subgroups. Particular advantages that are pertinent to the research in this thesis, that is SDA of the ELSA dataset, are the opportunities for follow-up and subgroup analysis. The availability of large quantities of longitudinal data means that time-order can be investigated, which may lead to conclusions that can suggest causality. Large samples also permit subgroups to be explored (e.g. by age, or other characteristics); in the case of this research, caregivers are contrasted with non-caregivers.

Nevertheless, SDA is not without limitations. Bibb (2007) highlights three key limitations: locating data, compatibility, and data quality. Locating data is not too critical a limitation for this thesis, as the UK Data Archive holds a large range of high-quality, impactful data available for secondary analysis, including ELSA, and is updated regularly. Compatibility refers to the extent to which available data fits the proposed research questions, which can be addressed by altering the research approach (Bibb, 2007). Designing research questions to fit an existing dataset increases the likelihood of securing compatible data; alternatively, existing data can be found to fit the research questions. However, the latter approach is often more difficult and, depending on the available data, may require multiple adaptations to the original research questions. Arguably, research questions should be modelled from a research gap rather than from available data; for this thesis, the research questions were designed around previous literature and the most suitable dataset was identified to answer those research questions. Finally, data quality is highlighted as a limitation of SDA by several authors (Bibb, 2007; Windle, 2010; Bryman, 2016). Data quality may be compromised, due to incomplete

or inaccurate records (Windle, 2010), or a lack of key variables between waves of data (Bryman, 2016). For example, in ELSA, loneliness expectations questions were asked in waves two, seven and eight, but were missing in waves three through six, which restricted availability to analyse changes over all of the waves. Nonetheless, adaptations were made to analyse changes over the available waves (section 4.3.8.2). To assess data quality, data familiarisation is essential, and, as with primary analysis, data organisation and cleaning (Bibb, 2007); this allows the researcher to identify and resolve issues with missing data. During the familiarisation period for this thesis, several variables were recoded (section 4.3).

#### 4.2.1.2 Cross-sectional research design

This thesis uses cross-sectional analysis to identify associations between caregiving, loneliness, health, and wellbeing in wave eight of ELSA. Cross-sectional research occurs at a single time-point (Sedgwick, 2014); characteristically this involves a large number of cases and variables (Bryman, 2016). Cross-sectional studies are particularly suited for exploring prevalence within a specific population (Sedgwick, 2014); in this thesis, the prevalence of loneliness is examined among caregivers and non-caregivers. Additionally, cross-sectional studies often provide the foundations for further analysis (*ibid*). In this thesis, some of the associations uncovered through cross-sectional analysis are investigated further within the follow-up analysis. As such, cross-sectionally, caregiving was significantly associated with loneliness, this was further investigated in the follow-up analyses to identify longer-term associations between caregiving and loneliness.

Typically, internal validity in cross-sectional research is weak, as causality cannot be established (Levin, 2006; Bryman, 2016); however, external validity can be strong, when the sample is collected randomly, as the results can often be generalised to wider populations. Cross-sectional research is widely used, as associations can be uncovered with minimal time commitment, in comparison to longitudinal or repeated methods, and there is relatively little expense (Levin, 2006). However, particularly within health research, cross-sectional methods are subject to prevalence-incidence bias, where risk factors associated with mortality may be under-represented within the respondents with the condition (*ibid*). To illustrate, when investigating health conditions that have high risk of mortality (e.g. strokes (Donkor, 2018)), the sample will predominantly include those who experienced mild to moderate strokes, as it is likely that the most severe strokes resulted in fatality. Therefore, severe, or advanced stages of conditions are often under-represented within research. As loneliness has been linked to increased physical and psychological morbidity, and is reported as a risk factor for mortality (Friedler et al., 2015; Ong et al., 2016), there is potential for underrepresentation within the sample. The response rate at wave eight across all cohorts is reported at 82%, with significant differences between responders and non-responders including, age by sex, and education (Banks et al., 2018).

In this thesis, cross-sectional design was applied to investigate associations between caregiving and different loneliness measures (RQ1a), to identify the determinants of loneliness (RQ1b), and to examine associations between caregiving, loneliness, health, and wellbeing (RQ1c). Furthermore, in RQ2, cross-sectional analysis was conducted to determine the associations between different caregiving factors and loneliness (RQ2a), health (RQ2b), and wellbeing (RQ2c). All of the variables analysed using cross-sectional research design were collected at a single time-point (wave eight).

# 4.2.1.3 Follow-up research design

This thesis uses follow-up analysis to identify associations between caregiving (wave two) and loneliness twelve-years later (wave eight). Follow-up studies are a branch of longitudinal research, which is characterised as an extension of cross-sectional research; data are collected over two or more time-periods (waves), using the same variables and respondents (Bryman, 2016). In contrast to cross-sectional research, longitudinal designs offer understanding into the time-order and potential causality between variables, as the variables arising first can be identified. Cohort and panel studies are the two main types of data for longitudinal research (ibid). Cohort studies are carried out on samples of people with shared characteristics or experiences that differentiate them from other cohorts (e.g. born at similar times) (Mason and Wolfinger, 2001). Whereas panel studies are initiated with a single cross-sectional study, followed by repeated data collections with the same individuals, although unlike cohort studies these individuals do not need a shared characteristic (ibid). Panel studies can identify ageing and cohort effects, whereas cohort studies are limited to ageing effects only, as the entire sample is from the same cohort (Bryman, 2016). ELSA is a panel study including representative cohorts of individuals aged  $\geq 50$  in England (Steptoe *et al.*, 2013a). Panel studies can result in skewed age distributions if new (younger) respondents are not introduced (Mason and Wolfinger, 2001); however, this is not the case with ELSA, as refreshment samples are included in multiple waves (section 4.2.2.1).

The main limitation of longitudinal data is attrition (Bryman, 2016; Kelfve *et al.*, 2017), which can be due to mortality, withdrawal, or selective attrition. In ELSA, 55% of the respondents from cohort one also responded in wave eight (Banks *et al.*, 2018), indicating an attrition rate of 45%. Approximately one-third of eligible cohort one responders in wave eight were listed as non-response due to 'other reasons'; it is reported that 'other reasons' including poor health or hospitalisation were most common among this cohort due to age-related circumstances (*ibid*). Additionally, among older samples, alongside higher mortality rates, respondents may have been unable to participate due to cognitive impairment (Kelfve *et al.*, 2017). However, if the respondents lost to follow-up differ significantly from the remaining respondents, this would introduce bias into the sample (*ibid*). To clarify, if a certain characteristic is highly prevalent among the dropouts (e.g. non-employment), this reduces the proportion of non-employment within the remaining sample, introducing potential bias towards employed respondents. This can reduce how representative the remaining sample is of the target population, and is particularly detrimental if the characteristics are key measures (Lynn, 2011).

Moreover, attrition due to mortality also introduces bias (Kelfve *et al.*, 2017). As the waves of the study progress, the sample ages, and mortality naturally reduces sample size. Within ELSA, to maintain a representative sample of those aged  $\geq$ 50, continuous recruitment of younger participants (aged 50/51) is essential to counteract the ageing sample (NatCen, 2018b).

However, this can lead to higher proportions of characteristics associated with higher life expectancy within the sample (e.g. females, and higher SES). Kelfve *et al.* (2017) term this as cohort inversion, where the sample gradually becomes healthier, as mortality reduces the groups with disadvantaged characteristics (healthy survivor effect). Within the data, it would appear that disadvantages such as lower SES reduce over time or with age, but this is not the case; reduced life expectancy among these characteristics skews the sample. The use of proxy responses, for example, for those unable to respond due to poor health, or incentives for participation, can combat attrition within longitudinal research (Lynn, 2011), and frequent refreshment samples ensure that the youngest ages are not underrepresented (NatCen, 2018b). In accordance with this approach, the sample size and characteristics of the ELSA sample have remained relatively stable across the waves.

The follow-up research design is applied in RQ3 of this thesis. Initially, caregiving and relevant control variables from wave two were analysed in relation to loneliness in wave eight (RQ3a), which identified whether previous caregiving was linked to longer-term loneliness outcomes. Additionally, variables across multiple waves (2-8) were combined, to determine if caregiving, at any point within the 12-year period, was linked to loneliness within the same 12-year period (RQ3b). Finally, changes to caregiving variables across the waves were analysed in relation to loneliness (RQ3c), which examined how changes over the duration of the caregiving role impacted caregivers' loneliness.

# 4.2.1.4 Design summary

A combination of both cross-sectional and longitudinal research designs were employed in this thesis. By utilising both approaches, this combats the main disadvantages of cross-sectional research, whereby causal inference cannot be made and the order of events are unknown. Cross-sectional analyses were conducted on wave eight of ELSA to identify associations between variables, and follow-up analysis between waves two and eight, to suggest the time-order of variables and indicate potential causality. For further discussion on the statistical methods, see section 4.4, and corresponding results can be found in chapters five and six, respectively.

# 4.2.2 English Longitudinal Study of Ageing (ELSA)

Recruitment for ELSA began in 1998, based on characteristics to build a representative sample of individuals aged  $\geq$ 50 in England; respondents from the 1998, 1999 and 2001 waves of HSE were selected (NatCen, 2018b). The study was designed as a companion study to the US-based Health and Retirement Study (HRS), and included data relevant to policies about ageing (Steptoe *et al.*, 2013a). Waves of data are collected biennially using interviews and questionnaires, with nurse visits for

selected respondents every alternate wave, to collect biological samples and conduct physical examinations (NatCen, 2018b). A primary focus of the study is to evaluate policy-relevant data on ageing (Steptoe *et al.*, 2013a); therefore, a vast range of data are collected, including but not limited to, health, social participation, work and pensions, and housing (NatCen, 2018b).

ELSA is therefore the most appropriate dataset, in a UK context, to conduct analysis on loneliness, health, and wellbeing among caregivers. UKHLS is a comparable population-based survey; however, the availability of longitudinal analysis of loneliness was not possible. Throughout UKHLS, objective measures for social isolation are available (e.g. proximity of social networks), but the inclusion of a subjective measure for loneliness (3-item UCLA-LS) was not included until wave nine (data collected 2017-2019) (Understanding Society, 2019). When data analysis for this thesis began, wave nine was the most recent wave of UKHLS, and therefore no longitudinal analysis could be conducted on loneliness among caregivers. ELSA is also the most appropriate dataset due to the age range of respondents. In comparison to ELSA respondents, who are aged  $\geq$ 50, UKHLS interview respondents aged  $\geq 16$ , with an additional youth questionnaire for children aged as young as 10 (*ibid*). As discussed in section 3.3.1.1, the peak age for caregiving occurs around 50-years old (Vasileiou et al., 2017). Moreover, evidence suggests that caregiving more commonly occurs in mid- to later-life (Dahlberg et al., 2007; Nomis, 2018a, 2018b), therefore it could be argued that younger adult caregivers should be analysed separately, and not in combination with older caregivers. Additionally, research and policy for caregivers typically considers young caregivers (aged <18) differently to adult caregivers (Powell et al., 2020). Thus caregivers <18 may not be comparable with adult caregivers, and the inclusion of child caregivers would be beyond the scope of this thesis.

# 4.2.2.1 Study population of ELSA

For the purpose of this thesis, waves two and eight of ELSA were analysed, however some data were obtained from the between waves to compute covariates; further discussion on key variables can be found in section 4.3. At the time of conducting the analysis, wave eight was the most recent wave of ELSA data and contained all the relevant variables to analyse caregiving and loneliness. Wave eight was therefore selected for the cross-sectional analysis, and also as the second time-point in the follow-up analysis. Using the most recent wave in the follow-up, this allowed for the longest-term analysis to be conducted using the available data. Wave one was excluded from the analyses because several key variables were not present, most importantly, UCLA-LS. Therefore, wave two was selected as the earliest time-point for the follow-up analyses. Wave two was chosen over the other waves for a number of reasons; firstly, variables examining expectations for future health and loneliness were only included in waves two, seven and eight. Previous evidence using ELSA indicated that negative expectations for future loneliness were positively associated with experiencing loneliness 2-8 years later (Pikhartova *et al.*, 2016). Therefore, comparing waves two and eight meant longer-term effects (12-years) were examined. Secondly, there is little research on the long-term effect of caregiving; while caregiving has been linked to loneliness, the evidence is

largely cross-sectional (section 3.4.2.1). Thus, investigating the long-term effects of caregiving in a 12-year follow-up identified whether loneliness is enduring among caregivers.

Table 1 shows sample sizes for each ELSA wave. In addition to core members, total respondents include partners of core members, some of which aged <50. However, younger partners are not ageeligible sample members and are not intended for analysis as respondents (Taylor *et al.*, 2007). In waves where the number of core member respondents increased (waves three, four, six and seven), a refreshment sample was added (NatCen, 2018b).

Table 1: The number	• of core and	d total resp	pondents a	t each wav	e in ELSA	waves 1-8	8	
	Wave 1 2002	Wave 2 2004	Wave 3 2006	Wave 4 2008	Wave 5 2010	Wave 6 2012	Wave 7 2014	Wave 8 2016
	/03	/05	/07	/09	/11	/13	/15	/17
Core member respondents	11,391	8,780	8,810	9,886	9,090	9,169	8,249	7,223
Total respondents (including partners)	12,099	9,432	9,771	11,050	10,274	10,601	9,666	8,445

Source: author's adaptation from NatCen (2018b: p16)

Interviews with younger partners enabled more complete data collection for core members and provided a representative sample of mixed-age couples (aged >50 and <50) (Institute of Fiscal Studies, no date). Partner responses are not weighted, and it is suggested that where weighted analyses are conducted, non-sample (e.g. partners) are removed from the baseline, as partner data is supplementary to core member data (*ibid*). Furthermore, proxy responses are also excluded from this thesis, due to the subjective nature of the questions on loneliness and other variables (e.g. health expectations), it cannot be guaranteed that proxy responses accurately reflect perceptions of core respondents.

#### 4.2.2.2 Ethical considerations and consent

Ethical approval for all waves was obtained by the ELSA research team through the Research and Ethics Service, which is regulated by the National Health Service (NHS) (NatCen, 2018b). Respondents are advised that their data will be utilised by multiple researchers, but solely for research purposes; they are also assured that all data are anonymised with no possibility of identification (ELSA, 2019a). To uphold confidentiality, certain variables are collapsed, which eliminates the possibility that any respondents are identifiable within the data. As such, the age variable is collapsed at 90-years, with a collective age category for respondents aged  $\geq$ 90. Respondents of ELSA are educated on the longitudinal intent of the research, and therefore no time limit is placed upon the retention of their data, however, participants are informed of their right to withdraw (*ibid*). Specific to this thesis, ethical approval for SDA was obtained on 2<sup>nd</sup> November 2017 (Appendix A), through the University of Southampton Ethics and Research Governance Online (Submission ID: 30736). All data were stored and handled in line with General Data Protection Regulation and the Data Protection Act.

# 4.3 Key variables in ELSA

This section discusses the key variables analysed within this thesis, drawing on the instruments used by the ELSA researchers. Primarily, the variables can be divided into three main categories: explanatory, dependent, and covariates. Caregiving (section 4.3.1) was the main explanatory variable, manipulated to determine the effects on loneliness, health, and wellbeing, for example, caregivers *versus* non-caregivers, although caregiving-specific factors (characteristics, relationship, and role) were also used as explanatory variables (e.g. comparing caregivers by intensity). Variables which were used as dependent variables (DVs) included: loneliness (section 4.3.2), health (section 4.3.3) and wellbeing (section 4.3.4). The DV was contingent on the research question. To illustrate, in the follow-up portion of this thesis, the loneliness variable differed. In some regressions, loneliness at wave eight was the DV, whereas in others, a computed variable measuring loneliness across all waves was used (section 4.3.8). The covariates included sociodemographic (section 4.3.5), socioeconomic (section 4.3.6), and social network variables (section 4.3.7).

# 4.3.1 **Caregiving variables**

Within the ELSA survey (Clemens *et al.*, 2019), respondents were asked about activities they had engaged in within the past month, an option of which was 'cared for someone'. Across waves 2-5, only respondents who mentioned providing care within the past month were eligible to answer a question about caregiving within the past week. However, in waves 6-8, the monthly activities question was no longer a prerequisite, and all respondents (excluding those residing in an institutionalised setting) were asked if they had looked after anyone in the past week (NatCen, 2014). In waves 6-8, a higher number of respondents identified as providing care within the past week, compared to the past month (Table 2), which is likely due to the aforementioned changes in the survey.

<u>Table 2: Per</u>	centage of respondents providu	ing care within the past week	/month in ELSA waves 2-8
	Care provision: past week	Care provision: past month	
	N (%)	N (%)	
Wave 2	1036 (11.0)	1344 (14.3)	
Wave 3	1000 (10.2)	1270 (13.0)	
Wave 4	999 (9.0)	1369 (12.4)	
Wave 5	1006 (9.8)	1365 (13.4)	
Wave 6	1962 (18.5)	1325 (12.5)	
Wave 7	1647 (17.0)	1239 (12.9)	
Wave 8	1480 (17.5)	1120 (13.3)	

Notes: % percentage of the total sample at each wave, source: author's analysis of ELSA waves 2-8

Within the monthly activities question, alongside caregiving, another option was 'looked after home or family'. It is plausible that some respondents, particularly those whose care recipients were family members, may have selected this option over 'cared for someone'. This links to the phase of CIT that the individual identifies with. As such, if they identified more as a family member than as a caregiver,

they may have considered 'looked after' as more appropriate than 'cared for' to describe their role. In waves 2-5, those for whom 'looked after' resonated more than 'cared for' would have been ineligible for the weekly care provision question. However, in waves 6-8, this is irrelevant, as all respondents were eligible to answer the weekly care provision question. Additionally, the weekly care provision question is worded as "Did you look after anyone in the past week (including your partner or other people in your household)?" supplemented by the interviewer explaining "By 'look after' we mean the active provision of care." (NatCen, 2018a, p.413). Therefore, it is possible that, due to the use of 'looked after' in the weekly caregiving question, those who responded with 'looked after home or family' in the monthly activity question may have identified as caregivers. Moreover, responses to the monthly activities remains relatively stable throughout the waves. This suggests that it is not a change in the interpretation of the questions that contributed towards the increased weekly caregiving in later waves. Rather, placing less restriction on who was eligible for the question allowed those who did provide care, but did not necessarily identify as a caregiver in the monthly activities question, the opportunity to be included within the caregiving sample.

Although data suggest that there may be missing information in waves 2-5 from those who failed to mention caregiving activities within the past month, the responses to the weekly caregiving variable were used to determine the caregiving sample. This decision was made based on additional caregiving variables, such as caregiving intensity, which were weekly measurements (hpw), therefore, the weekly caregiving variable was deemed most appropriate to correspond with these.

Evidence suggests a continuous increase of the population in the UK who provide care. Data from the most recent (2011) Census identified around 6.5 million caregivers across the UK, (DCMS, 2018; Parliamentary Office of Science and Technology, 2018). In contrast, more recent surveys using Understanding Society data indicated that there were around 7.6 million caregivers (Petrie and Kirkup, 2018). Whereas, data collected in May 2020 suggest that as many as 13.6 million people in the UK could be providing unpaid care (Carers Week, 2020). However, it is important to consider that the age ranges for these sources differ from the ELSA sample. Census and Government data includes individuals aged  $\geq$ 5 (ONS, 2013; Parliamentary Office for Science and Technology, 2018), whereas the 2020 polling involved respondents aged  $\geq 18$  (Carers Week, 2020). A further limitation was identified by Rutherford and Bu (2018), suggesting that measures used in surveys may be underestimating the scale of care provision. It is highlighted that due to discrepancies in question wording (as discussed above), and inconsistencies with recognition of caregiving activities (as identified by CIT), the rate of reporting caregiving roles is likely to be reduced. Table 2 previously identified that the percentage of caregivers (provision of care in the past week) ranged from 9-18.5% across the waves, although being a larger range, this is not inconsistent with alternative sources, which suggests that the proportion of caregivers in ELSA is representative of the general population of England.

Changes to caregiver identity cannot be directly examined within this thesis. There are no ELSA variables on identity; this therefore limits the exploration of caregiver identity in line with how CIT suggests. However, as proposed in section 2.2, higher-intensity care roles are likely to represent the later stages of CIT and lower-intensity roles the earlier stages. Whereby caregivers providing higher-intensity care are more likely to relate to their caregiver-recipient relationship rather than their pre-caregiving relationship. Therefore, caregiving intensity (see section 4.3.1.1) will be examined both in respect of the caregiving role, but also discussed as a proxy of caregiver identity. To examine identity in more depth, questions about how the caregiver feels about their role, specifically with regard to relationship changes, would be required in the data.

# 4.3.1.1 Caregiving-specific variables

In RQ2, caregiver characteristics include age, sex, marital status, employment status, health, and wealth. However, these variables represent sociodemographic, socioeconomic, and health characteristics in RQ1. These are outlined in subsequent sections (sections 4.3.5 to 4.3.7).

In the ELSA questionnaire, the subsequent question from looking after anyone in the past week, asked the respondent what relation this person was to them. The options were coded as spouse or partner, child, grandchild, parent, parent-in-law, other relative, friend or neighbour, and other; a maximum of eight responses were coded per respondent. Due to relatively small samples (relative to other categories) of respondents providing care to a parent-in-law (N=87) and other relatives (N=51), for this thesis, they were collapsed and merged with parent, and friend or neighbour, respectively. The variables for care recipients are displayed in Table 3. Provision of care to grandchildren is not typically recognised as informal care, therefore individuals who reported providing care to a grandchild were excluded from these analyses.

Table 3: Variables relating to the caregivers' relationship to the care recipient				
Variable name	Variable label			
Care_spouse	Provides care to partner/spouse			
Care_child	Provides care to adult-children			
Care_parents	Provides care to parent/parent-in-law			
Care_others	Provides care to friends, neighbours or other relatives			

Notes: value labels 0 = No, 1 = Yes, source: author's own

To determine the location of care, co-resident caregivers were compared with non-residential caregivers. The original question was phrased as "Does the person, or do any of the people you care for, live with you?" (NatCen, 2018a, p.414) with responses of 'yes' and 'no'. This variable remained unchanged, other than reordering the responses to use non-residential caregivers as the control group.

The caregiving intensity variable was computed from data on the number of hours the respondent had reported providing care during the past week. A specific hourly value (e.g. 12hpw) was reported, or if caregiving had occurred 'all of the time', this was coded as the maximum value (168hpw). In accordance with published literature and the earlier discussion (section 3.3.3.3), intensity was

grouped into 1-19 hours,  $\geq$ 20-hours, and a separate category for 168-hours (round-the-clock care). Similarly to caregiving intensity, the number of care recipients was collected as a continuous variable in ELSA; respondents provided a numerical value (range 1-20) for the number of recipients they provide care to. Of the caregivers in wave eight, the most common response was providing care to one person (86%). Respondents providing care to two or more people were grouped into 'Multiple care recipients' due to the increasingly small sample sizes for the higher care recipient values. To illustrate, 93 respondents indicated providing care to two people, and only a further 42 respondents provided care to three or more recipients. Thus, in the analyses for this thesis, caregivers to single recipients are compared with caregivers who provide care to two or more recipients.

As discussed in section 4.3.8.1 ahead, for the follow-up analyses, duration of care is calculated using the 'care provision within the past week' variable from each wave. This identifies caregivers who have provided care for a single wave (short-term) or for two or more waves (long-term and intermittent). This therefore cannot be accurately computed cross-sectionally, and the duration of care cannot be measured in the cross-sectional analyses. As a caregiving-specific factor, which conceptually should be strongly associated with loneliness among caregivers, this has potential implications and limitations for the caregiver-only cross-sectional analyses (RQ2) whereby it must be acknowledged that certain important factors related to the caregiving role cannot be accounted for.

As part of the caregiving questions, ELSA respondents were asked how strongly they agreed with the following statements: "Considering all the efforts that I have put into caring for someone, I am fully satisfied with what I have achieved so far" and "I have always received adequate appreciation from others" (NatCen, 2018a, pp.419-420). There are no direct questions on burden or satisfaction, but these could be used as proxy measures. As such, a lack of appreciation would contribute to increased stress and potential burden for the caregiver. However, as several established measures of burden already exist (e.g. Zarit Burden Interview), the use of these proxy measures would restrict any comparability with previous studies, and there would be limitations around how representative of burden and satisfaction the responses were. As such, while some caregivers may not receive appreciation from others, this does not automatically equate to experiences of burden. To accurately determine burden among caregivers, specific burden measures could be added to future waves of ELSA.

#### 4.3.2 Loneliness variables

For the purpose of this thesis, three distinct measures were identified to determine loneliness: a multiple-item loneliness scale, a single-item direct question and loneliness expectations. As discussed in section 1.2.2.2, the UCLA-LS is commonly used to measure loneliness. For the purpose of this thesis, the 3-item UCLA-LS is the main variable used to measure loneliness. This version is suitable for use with older adults and relevant for a UK context (Campaign to End Loneliness, no

date), which, as the 20-item version was developed using US students, arguably makes the shorter version more appropriate for the ELSA sample. The wording and scoring of the 3-item UCLA-LS was as follows: how often do you feel that you lack companionship, how often do you feel left out, and how often do you feel isolated from others? The responses to which are 'hardly ever or never', 'some of the time', or 'often' scored 1-3 respectively.

To determine overall score for UCLA-LS, the three items are summed (Hughes *et al.*, 2004; Steptoe *et al.*, 2013b): scores of 6-9 would suggest the presence of loneliness. While the scoring system allocates numerical categories, it does quantify loneliness (Campaign to End Loneliness, no date), as such classifying the UCLA-LS as ordinal, not interval. Therefore, when utilising this scale, calculating averages (e.g. mean scores) would be unreliable. To clarify, the difference between scoring six or seven does not necessarily indicate a standardised increase in loneliness. Both scores would suggest the presence of loneliness, but, the degree to which loneliness changes between scores is ultimately unknown. Numerical scores are assigned to subjective categories, the interpretation of which likely differs for each respondent. Moreover, the difference between 'hardly ever or never' and 'some of the time' may not equate to the difference between 'some of the time' and 'often', despite the assigned numerical differences being the same (+/-1). Therefore, mean scores are not a reliable interpretation of UCLA-LS.

# 4.3.2.1 Alternative loneliness measures

In addition to UCLA-LS, there are single-item direct questions for loneliness in ELSA. The first question asks whether the respondent has felt lonely much of the time during the past week. This is a dichotomous variable, asked during the face-to-face interview, with responses of yes or no. The second asks how often the individual feels lonely, without a defined timescale, and is scored on a 3-point scale: 'hardly ever or never', 'some of the time', or 'often'. Restricting the experience of loneliness to the past week could exclude individuals who may typically feel lonely but have been feeling less so this week. However, answering questions about a specific timeframe (e.g. past week), and a more typical period (e.g. typical week), are both exposed to certain levels of bias (Chang and Krosnick, 2003). Typical week responses are often overestimated, whereas past week responses are often under-reported (*ibid*). Alternatively, Angrisani *et al.*, (2015) and Araujo *et al.*, (2017) indicate that past week questions are less predisposed to recall error and reduce cognitive load on respondents. This could imply that a respondent's report of loneliness in the past week may be more accurate than questions with vague or no timescales. For example, the direct question without a timeframe may require the respondent to consider loneliness over a longer period of time and calculate an average, whereas loneliness in the past week is a simpler recall task.

Table 4: Crossta	bulation of responses to	the UCLA-LS and	d direct loneliness	
Direct questions		UCLA-LS		
		Not Lonely	Lonely	
		N (%)	N (%)	
Loneliness (no timescale)	Hardly ever or never	3878 (83.4)	151 (12.9)	
	Some of the time	741 (15.9)	650 (55.7)	
	Often	31 (0.7)	367 (31.4)	
Loneliness (past	Not Lonely	4472 (96.3)	681 (58.3)	
week)	Lonely	172 (3.7)	487 (41.7)	
Loneliness (no timescale) Loneliness (past week)	Hardly ever or never Some of the time Often Not Lonely Lonely	3878 (83.4) 741 (15.9) 31 (0.7) 4472 (96.3) 172 (3.7)	151 (12.9) 650 (55.7) 367 (31.4) 681 (58.3) 487 (41.7)	

*questions* 

Notes: % of UCLA-LS, source: author's analysis of ELSA wave 8

Crosstabulation between the two direct measures and UCLA-LS (Table 4) indicate that there is substantial affinity between responses to direct loneliness (no timescale) and UCLA-LS, whereby around 87% of respondents who score as lonely (UCLA-LS) indicate they feel lonely some of the time or often. In contrast, only 42% of respondents who score as lonely (UCLA-LS) indicate feeling lonely much of the time during the past week. The direct question about loneliness in the past week is asked during the face-to-face interview, whereas the direct question without a timeframe is included as part of the self-completion questionnaire, preceded immediately by 3-item UCLA-LS. Therefore, it is reasonable to expect higher levels of similarity between this direct loneliness question and UCLA-LS. This is supported by findings from ONS (2018c), whereby an analysis was conducted of the loneliness questions in ELSA and found strong significant correlations between the direct loneliness (no timeframe) question and the 3-item UCLA-LS. Consequently, this collinearity may influence the analysis if this direct measure, and UCLA-LS, were included. Due to the identified differences in loneliness responses and the different contexts (face-to-face interview versus selfcompletion questionnaire), direct loneliness in the past week is the most appropriate direct question to contrast with UCLA-LS. Although both measure loneliness, it is indicative that different aspects of loneliness are measured, or that there are evident differences between directly reporting loneliness and using a measurement scale.

The previous crosstabulation of direct loneliness and UCLA-LS (Table 4) indicated the majority of respondents were 'not lonely' across both measures. However, there were a number of instances where the measures did not correspond: 4% of respondents who directly-reported loneliness were not lonely on the UCLA-LS, and 59% of UCLA-LS lonely respondents did not directly report loneliness. By comparing both measures of loneliness, the aim was to include as many individuals as possible, who may be experiencing loneliness. Additionally, due to the limitations of direct questioning, whereby the negative stigma surrounding loneliness is linked to under-reporting (Campaign to End Loneliness, no date), it is important to include alternative measures of loneliness, such as UCLA-LS. The larger percentage of respondents not directly reporting loneliness, but being categorised as lonely on the UCLA-LS suggests that there is an element of under-reporting present when directly asked, which may be related to the different contexts (self-report compared to face-toface interview).

Waves two, seven and eight included questions on expectations for loneliness. Each of the waves contained similarly-worded questions; as such, in wave eight, respondents were asked both whether

they expected to get lonelier as they got older and, if they considered old age to be a time of loneliness. Responses to which were: agree strongly, agree slightly, neither agree nor disagree, disagree slightly, and disagree strongly. The main reason for including this variable was to determine if negative expectations for loneliness were linked to increased likelihood of loneliness. For this reason, it was deemed unnecessary to differentiate between slight and strong positive or negative perceptions, as such, slightly agreeing and strongly agreeing both implied negative expectations for loneliness. Therefore, the strongly and slightly categories were collapsed to form a single 'agree' or 'disagree' response. This meant that positive, negative, and neutral expectations could be compared.

It was hypothesised that respondents 'agreeing' that they expect to get lonelier with age would also think old age was a time of loneliness, but this was not necessarily the case (Table 5). As shown, only 58% of respondents who agreed they expect to get lonelier as they age also agreed that old age was a time of loneliness.

Table 5: Crosstabulation of responses to two loneliness expectation questions				
	As I get older, I expect to become more lonely			
	Neither agree nor			
	Agree	disagree	Disagree	
Old age is a time of loneliness	N (%)	N (%)	N (%)	
Agree	1397 (57.9)	280 (16.3)	146 (8.7)	
Neither agree nor disagree	686 (28.4)	896 (52.0)	228 (13.6)	
Disagree	330 (13.7)	546 (31.7)	1304 (77.7)	

Notes: % of 'as I get older, I expect to become more lonely', source: author's analysis of ELSA wave 8

As there were evident discrepancies in the respondents' interpretation of the questions, issues arose for the author when selecting variables. Further considerations were made around question wording to ensure the most appropriate variable was selected to represent loneliness expectations. It was considered that expecting to become lonelier as they age as opposed to believing old age was a time of loneliness was more personal and thus demonstrates that the stereotype was internalised, which aligns with the SET framework.

To measure loneliness, the predominant variable throughout this thesis is the UCLA-LS, however in RQ1a (section 5.2.2) three loneliness measures are compared to identify differences between the measurement tools. Additionally, in the follow-up analyses, loneliness expectations in earlier waves are examined in relation to UCLA-LS loneliness in wave eight and across all waves.

While ELSA includes a range of loneliness measures, such as the aforementioned UCLA-LS, two direct loneliness questions, and loneliness expectations, another common loneliness scale is the DJGLS (section 1.2.2.2). The main advantage of the DJGLS is the ability to examine the social and emotional subscales of loneliness, which is not possible with UCLA-LS. Therefore analysis of loneliness using ELSA data cannot separate the different loneliness types. Nevertheless, as demonstrated in this thesis (section 5.2.2) there is scope for comparing different methods of measuring loneliness, i.e. direct single-item, and indirect scale measures (UCLA-LS).

# 4.3.3 Health variables

Health is multi-dimensional. Within published literature, the measurement of health ranges, including: self-rated general health, self-reported health conditions, scales for mental health and the use of medical professionals' diagnoses. Within this thesis, a number of health measures are used, including variables that align with SET, mental health scales, and self-reported limiting illnesses.

Self-rated general health is a widely used health measure (Garbarski, 2016), which is reportedly related to health outcomes and mortality (Bailis *et al.*, 2003; Feenstra *et al.*, 2020). However, self-rated health was excluded from these analyses because of its generic nature; as such, the included health measures focus on specific areas of health, whereas self-rated health asks broadly about overall health. Moreover, an individual could also perceive their health as 'good', in relation to someone they know with poorer health than themselves, but they still may have chronic conditions, or poor behavioural health. Therefore this 'good' response would not necessarily reflect good health.

Garbarski (2016) indicates that often the purpose of self-rated health measures are to act as proxies for more objective measures, such as self-reported conditions. Arguably, therefore, self-rated general health is redundant since ELSA data includes more objective measures. Additionally, when identifying policy implications from the results of this thesis, the recommendations can be tailored more specifically to certain areas of health, if more objective measures are used. The following sections discuss the health measures used within this thesis: ones that align with SET and additional health measures.

#### 4.3.3.1 Stereotype embodiment theory: health-related measures

As identified in section 2.1.2, SET outlines three distinct health pathways: physiological, behavioural, and psychological. For each of these pathways, the most appropriate variables were sourced within ELSA. This section will discuss the selection of each of these variables.

Physiological health considers the impact of internalised age-stereotypes on CV activity, and the stress response (Meisner and Levy, 2016). To measure physiological health, diagnoses of specific CV conditions are used. SET typically refers to particular outcomes, including CV responses such as heart rate and blood pressure measurements (Levy *et al.*, 2000), or specific CV events including angina, heart failure, heart attacks, and strokes (Levy *et al.*, 2009). However, Levy (2009) indicated that the heightened CV responses can also increase susceptibility to further CV problems and have negative consequences on recovery from CV events (Levy *et al.*, 2006). Within ELSA, there are data available on a large range of chronic CV conditions, including the aforementioned SET CV responses and events, but also additional data on arrhythmias and other diseases. In this thesis, a binary variable is created to identify individuals with a relevant health condition; which included the presence of one of the following: hypertension, angina, congestive heart failure, heart attacks, strokes, arrhythmias, other heart diseases and dementia. Most of these conditions were selected because they

were specifically named in previous SET literature, as above. However, due to the highlighted susceptibility to other CV conditions it was deemed relevant to include all available CV conditions from the ELSA data.

Alternatively, dementia could be considered less appropriate for inclusion in the physiological pathway. However, dementia, specifically vascular dementia, which is the second most common form, is caused by reduced blood flow to the brain (NHS, 2017; British Heart Foundation, 2019) and is often medicated using drugs for hypertension (British Heart Foundation, 2019), so could theoretically be grouped with other CV conditions. Additionally, de Roos *et al.* (2017) advocate that dementia is CV-driven, primarily, problems with the proximal aorta are an indication of not only CV health, but also cerebrovascular damage, including memory loss, cognitive decline and dementia. Evidence also suggests that with progressing CV symptoms, ranging from risk factors to CV disease, the risk of developing cognitive impairment and consequently dementia and loneliness in both directions; for instance loneliness is associated with increased risk of cognitive decline and dementia (Holwerda *et al.*, 2014; Mushtaq *et al.*, 2014) and individuals with dementia are likely to report loneliness (Holmen *et al.*, 2000; Moyle *et al.*, 2011). Therefore, it was considered an important and highly relevant condition to include.

Conditions that the respondent 'still has' from previous waves, and any new diagnoses in the current wave were combined. A wide range of CV conditions were included in the physiological health variable, because research suggests that validity is higher among self-reported measures when a broader definition is adopted (Valtorta *et al.*, 2018). Arguably, several of the CV conditions listed in ELSA represent CVD risk factors (e.g. diabetes or high cholesterol). Individuals with diabetes are at higher risk of CVD (Diabetes UK, no date). Moreover, recent statistics from the American Heart Association categorise diabetes and high cholesterol as CVD risk factors (Benjamin *et al.*, 2019). The International Classification for Disease also states that metabolic conditions (i.e. diabetes), are excluded from the circulatory disease category (a component of CVD) (WHO, 2006). Although hypertension is also considered to be a risk factor (Benjamin *et al.*, 2019), simultaneously there are multiple hypertensive diseases that are categorised as CVD (WHO, 2006, Benjamin *et al.*, 2019). For these reasons, diagnoses of diabetes and/or high cholesterol are excluded. Furthermore, due to a high prevalence of hypertensive diseases (*ibid*), and the accepted classification of hypertensive diseases as CVD, hypertension was included in the physiological health variable.

Behavioural health refers to healthy practices, or preventative health behaviours. Examples of behavioural health variables in ELSA include alcohol consumption, smoking, and physical activity. The questions on alcohol consumption in ELSA relate to the frequency, as opposed to the volume. This contrasts with published guidelines, which recommend weekly unit limits (NHS, 2018; Drink Aware, 2020). The aforementioned guidelines suggest that those regularly consuming 14 units per week should spread this over three days rather than frequently consuming excessive volumes of

alcohol, and that individuals should have several alcohol-free days each week (*ibid*). ELSA variables measure frequency of consumption over the past twelve months, and past seven days. Data over the past year gives a better impression of typical drinking patterns in comparison to the past week; for example, an individual may report that they have not had an alcoholic drink this week, but this may not be indicative of their usual drinking behaviours. Moreover, data for a larger proportion of the sample were collected for the past twelve months compared to the past week (N=7,134 and N=4,470 respectively in wave eight), as individuals who consumed alcohol on a less-than-weekly basis were ineligible for the question about the past week.

Furthermore, in ELSA, data were collected to identify respondents who had 'ever smoked' (i.e. previous smokers) and those who 'smoked at all nowadays' (i.e. current smokers). However, the ambiguity surrounding the 'ever smoked' variable made it difficult to quantify behavioural health. Theoretically, this would include individuals who had smoked only once, individuals who smoked briefly at a younger age but had not smoked for a substantial period of their life, and those who continuously smoked throughout their life but had recently quit, with no indication of which category the respondent might fall under. Moreover, there were a substantial amount of missing data, whereby a large proportion of the respondents were not asked about having ever smoked, with a sizeable majority of respondents marked as not applicable across the waves. The data were more populated for current smokers, the majority of respondents had provided a response. However, again, there were limitations, in that quantity or frequency of smoking behaviours could not be identified. Individuals who occasionally smoked, for example, socially with friends, would be categorised with individuals who smoked full packets of cigarettes daily.

With regard to physical activity, ELSA variables measure the frequency of mild, moderate and vigorous sports or activities; the responses to which are 'more than once per week', 'once per week' 'one to three times per month' and 'hardly ever or never' (Clemens *et al.*, 2019). Aligning with exercise guidelines, which recommend that 150 minutes of moderate aerobic activity, 75 minutes of vigorous aerobic activity, or an equivalent combination of both, are completed each week alongside strength training exercises on at least two days per week (NHS, 2019) the variables for moderate and vigorous activity were used to compute a new variable, which identifies all respondents participating in either moderate or vigorous activity more than once per week.

Initially, a behavioural health variable was constructed similar to the physiological health variable above, in that a binary variable identified respondents who presented with one or more measure of poor behavioural health (smoker, frequent alcohol consumption, or limited physical activity). Preliminary results returned non-significant results, and upon review it was deemed that aggregating the behavioural health variables was ineffective due to loss of information. Additionally, over 90% of the overall sample exhibited at least one negative health behaviour, skewing the sample towards poor behavioural health. In relation to loneliness, smoking, alcohol consumption and physical activity were likely to have substantially different relationships so further analyses were conducted

including all three behavioural health variables as controls. However, changes to confidence intervals and fit statistics suggested that overfitting may become a concern, particularly due to large numbers of health-related controls in some models. Therefore, for these analyses, a single example of behavioural health was selected.

The example of behavioural health utilised in these analyses was frequency of alcohol consumption. This was selected due to firstly, a stronger conceptual link with both caregiving and loneliness, known as a common coping mechanism. But also previous analysis suggested stronger links between alcohol and loneliness in comparison to the other two behavioural health variables when included in the models individually. A new variable was computed to reflect the guidelines of spreading consumption over a few days and having several alcohol free-days (Table 6). Although these groups do not reflect the volume of alcohol consumed, individuals consuming alcohol more than three to four days per week are not having several alcohol-free days, thus suggestive of high-risk drinking.

Table 6: Recoding of the behavioural h	ealth variable, frequency of alcohol consumptio
How often the respondent had an alcoholic drink in the last 12 months	Over the past year, did the respondent regularly consume alcohol ≥3 days per week?
Almost every day	
5-6 days per week	Yes
3-4 days per week	
1-2 days per week	
Once or twice per month	
Once every couple of months	No
Once or twice per year	
Not at all in the last 12 months	

Source: author's analysis of ELSA wave 8

While psychological health may often be used interchangeably with mental health, in this thesis psychological health refers to health expectations and stereotypes within the SET pathway. Mental health is determined using depressive symptomatology (section 4.3.3.2). Similarly to the expectations for loneliness, corresponding questions were asked about health. In wave eight, respondents were asked both whether they worry about their health getting worse as they get older, and if they consider old age to be a time of ill health, responses to which were: agree strongly, agree slightly, neither agree nor disagree, disagree slightly, and disagree strongly (ELSA, no date [b], question 36). Corresponding with the loneliness expectations variable, the strongly and slightly categories were collapsed to form a single 'agree' or 'disagree' response, regrouping to form single positive, negative, and neutral categories. Table 7 shows the crosstabulation of responses to both questions: 61% of respondents who agreed that they expected health to get worse as they age, also agreed that old age was a time of ill health. The more personal question 'I worry that my health...' was selected as the measure for psychological health, as this reflected an internalised stereotype, in comparison with 'old age is...' which is arguably more generic.

Table 7: Crosstabulation of responses to two health expectation questions					
	I worry that my health will get worse as I grow older				
	Agree	Disagree			
Old age is a time of ill health	N (%)	N (%)	N (%)		
Agree	2482 (60.7)	262 (25.7)	163 (22.3)		
Neither agree nor disagree	955 (23.4)	412 (40.4)	175 (23.9)		
Disagree	651 (15.9)	345 (33.9)	393 (53.8)		

Notes: % of 'I worry that my health will get worse as I grow older', source: author's analysis of ELSA wave 8

# 4.3.3.2 Other health measures

In addition to SET health pathways, self-rated LLTI is used as an overall health measure. This variable is applied as part of the caregiver characteristics, a determinant of loneliness and a health control variable in different research questions within the thesis. Long-term illnesses, particularly limiting ones, could be seen as restrictive. LLTI is also suggestive of disability, and may be indicative that the respondent requires assistance or receives care. This may influence the caregiving role, if an individual is receiving care, it is unlikely they will have capacity to provide care for others. Additionally, LLTIs may reduce physical and social activities outside of the caregiving role, which, in turn, may lead to loneliness.

The Centre for Epidemiological Studies Depression Scale (CES-D) measures depressive symptomatology. This scale is used to assess mental health amongst the respondents. CES-D is a valid and reliable measure amongst the general population and older adults, and is used commonly in large population-based studies such as ELSA and HRS (Karim et al., 2015). The full version of CES-D is administered as a 20-item scale; however, an 8-item version is often favoured by researchers as the validity and reliability are comparable to the original version (*ibid*). Typically, CES-D is scored on a 4-point Likert scale. However, in ELSA, CES-D questions are binary, scored as yes/no. Therefore, instead of scores ranging from 4-32, scores ranged from 0-8, but higher scores remained suggestive of depression. This method of scoring is common in other population surveys, as CES-D results in ELSA are comparable with HRS (Kapteyn, 2008; Crimmins et al., 2010), which also utilises dichotomous scoring (Steffick, 2000). Moreover, recent publications from Kobayashi and Steptoe (2018) and White *et al.* (2018) analyse CES-D in ELSA, and advocate a score of  $\geq 3$  is suggestive of depressive symptomatology. One of the questions within CES-D asks whether the respondent has felt lonely much of the time during the past week, which is utilised as the direct loneliness question. To avoid overlap, and potential multicollinearity, this question was excluded when calculating depressive symptomatology score; this exclusion was also applied in other literature analysing both loneliness and CES-D (Kobayashi and Steptoe, 2018).

### 4.3.4 Wellbeing variables

The CASP19 scale is a common measure of QoL among older adults (Wiggins *et al.*, 2008; Sim *et al.*, 2011; Howel, 2012). CASP refers to four life domains (Control, Autonomy, Self-realisation, and Pleasure), and 19 indicates the number of items. CASP19 includes a combination of negatively and

positively worded items, and is typically scored from 0-57, with higher scores indicating greater QoL (Wiggins *et al.*, 2008; Sim *et al.*, 2011; Howel, 2012). In ELSA, the responses, 'often', 'sometimes' 'not often' and 'never', are all coded 1-4 respectively. Therefore, to reflect the 0-57 scoring system and to ensure that higher scores reflect greater QoL, recoding was required. The negatively worded questions were reverse coded (0-3 often to never), and the remaining positively worded questions were recoded from 3-0, often to never. For these analyses, the sum of all CASP19 variables was calculated to reflect the total score.

Finally, within literature, the Satisfaction with Life Scale (SWLS) is typically scored from 5-35 so that higher scores are indicative of greater life satisfaction. To reflect this, the seven-point Likert scale ranges from strongly disagree (1) to strongly agree (7) (Diener *et al.*, 1985). In the original ELSA data, these responses are reverse coded (strongly disagree=7, strongly agree=1), therefore recoding was necessary. Similarly to the CASP19, a variable was computed to determine total score. The score should remain continuous (*ibid*), thus SWLS reflects the sum of the seven variables.

### 4.3.5 Sociodemographic variables

Largely, sociodemographic characteristics act as control variables within the analyses. The inclusion of control variables allows insight into the effect of each variable after accounting for all of the other variables. Moreover, when characteristics such as sex are included in regression analyses, the results indicate effects for males *versus* females in isolation of all the other variables, in addition to the overall model. The sociodemographic characteristics included within this analysis are: age, sex, and cohabitation.

To reduce the likelihood of respondents being identified, age is collapsed at 90-years; therefore, the range within the analytical sample is 50-90 years. To analyse differences between age groups, this was split into 50-59, 60-69 and  $\geq$ 70. The decision was made not to separate 70-79 from  $\geq$ 80 because only a small minority of caregivers were aged  $\geq$ 80-years (N=91, 10%), thus, in an attempt to minimise small cell counts throughout the regression analyses, this variable was collapsed further. The sex of respondents was determined using a dichotomous variable, permitting responses of either male or female. Within this thesis, this is termed as sex, not gender, due the lack of diversity in the response options. To clarify, the distinction between sex and gender is that sex refers to biological difference, whereas gender reflects a social identity of the individual (Mitchell, 2004). Therefore, to constitute gender, it additional responses should be available (e.g. non-binary or transgender); it is proposed that ELSA should include additional gender options in future waves.

Marital status is readdressed for all respondents at each wave. Therefore, changes such as widowhood or divorce can be observed longitudinally. The responses for marital status include: (i) single never married, (ii) first marriage, (iii) second or later marriages, (iv) legally separated, (v) divorced, and (vi) widowed. For the purpose of this thesis, legally separated and divorced were merged into a single

category, as representation of the legally separated variable was limited and conceptually, this was deemed most similar to divorced. Gov.UK (no date) suggest that legal separation is considered an alternative to divorce for individuals with religious beliefs that prohibit divorce. Furthermore, it is common within loneliness literature to merge non-married categories (e.g. separated and divorced), as they are reported to have similar psychological, social, or economic outcomes (Ben-Zur, 2012). In this thesis, widowhood remained separated from legally separated and divorced, because evidence suggests significant differences in loneliness between divorced, widowed, and married individuals (Dahlberg et al., 2018a). In addition to marital status, cohabitation was measured by determining if respondents lived with a partner or spouse. Evidence suggests that research on cohabitation and loneliness is limited, compared to that on marital status (Buecker et al., 2020), thus identifying a need for research that considers cohabitation. Arguably, the presence of a close relationship (cohabiting partner, married or unmarried) would have a more substantial effect on loneliness, than the legal status of the relationship. Research from Perelli-Harris and Styrc (2017) compared married and cohabiting individuals in relation to mental wellbeing, and suggested that relationship type had little effect. Furthermore, particularly throughout Europe, cohabitation is often considered an alternative to, or precursor to, marriage (Sassler and Lichter, 2020). As such, single, cohabiting individuals should experience similar benefits of a significant emotional attachment as for married cohabiting individuals. Von Soest et al. (2018) highlight that relationship quality is considered integral to loneliness, and therefore further emphasis should be placed on close relationships or spousal/cohabiting partners, in relation to loneliness. Within their study, cohabitation and marriage was combined suggesting the two are comparable.

A limitation of the ELSA dataset is the lack of ethnic diversity, with >95% of the total sample at each wave identifying as White. Moreover, the remaining ethnic categories are consolidated into a single category to create a dichotomous variable: White *versus* Non-White. Alongside the small sample of Non-White respondents, a further limitation is that the diversity of the Non-White ethnicities is lost by aggregating these respondents. To clarify, this category contains all Asian, Black, and other ethnic groups. Due to lack of a representative sample of ethnic minorities, ethnicity was excluded from the analyses; the omission of the ethnicity variable is common among studies using the ELSA dataset (Torres *et al.*, 2016; Wade *et al.*, 2017; Valtorta *et al.*, 2018).

#### 4.3.6 Socioeconomic variables

There are several variables in ELSA which can be used to determine SES, such as education, employment status, and wealth. ELSA is known for having high-quality wealth variables (Banks *et al.*, 2011). Torres *et al.* (2016) indicate that wealth is the strongest indicator of SES in ELSA, and amongst older adults, as the association between wealth and mortality is greater than other SES indicators. To determine wealth, 'total non-pension net wealth', is commonly used (Demakakos *et al.*, 2016; Torres *et al.*, 2016); this reflects financial and physical wealth and wealth from business and property assets, minus any debt. The wealth variable refers to the benefit unit, or household and

provides an aggregate value of wealth (Oldfield, 2018). The ELSA researchers indicate in the financial variables user guide that the wealth variables are not equivalised because, there is no universally accepted method to adjust wealth in accordance with household size (ELSA, no date [a], p.8). This therefore could place single-person households at a disadvantage when measuring wealth, but arguably wealth is still more appropriate as a measure of SES than income, which is less relevant for older adults due to high prevalence of retirement. Moreover, Demakakos *et al.* (2016) highlight that this wealth variable is the most appropriate for use within an older adult population, derived from 22 wealth components, as well as reflecting assets accumulated over the lifecourse. Other SES variables, such as education, employment, or income do not consider assets accumulated over the lifecourse. Moreover, as a minority of older adults tend to be in paid employment, as opposed to retired, wealth is arguably more relevant to determine current SES. Within this thesis, quartiles of total non-pension net wealth were calculated.

Furthermore, in ELSA, employment status is determined by asking if the respondent was in paid employment during the last week, including temporary leaves of absence from their regular job. Arguably employment is not as relevant among older adults, due to the high prevalence of retirement; therefore, the number of non-employed respondents in the sample is likely to be considerably higher than those in employment. However, although not as strong an indicator of SES as wealth, employment status was deemed an important socioeconomic variable to include in these analyses. Firstly, as of 2017 (ELSA wave eight data were collected in 2016/17), the state pension age in the UK was 65-years (Age UK, 2017), thus there is a considerable portion of the ELSA sample still below statutory retirement age in England, i.e. those aged 50-64 years. This is also set to increase over the coming years, so employment will become increasingly more relevant for older samples. Additionally, evidence suggests non-employed caregivers experience higher levels of depression, mortality and functional limitations (Caputo *et al.*, 2016). Moreover, high-intensity caregiving has been shown to be negatively correlated with employment (Ciccarelli and van Soest, 2018). Thus employment was considered highly relevant when examining the impact of caregiving, even among older adults.

#### 4.3.7 Social network variables

Stephens *et al.* (2011) suggest that both the size of social networks, and the members that constitute the network, impact perceived social support and loneliness. Moreover, many of the loneliness definitions emphasise the importance of both quality of relationships, alongside quantity (section 1.2.2). Therefore, to gain a more detailed insight into the social networks of respondents, it was necessary to acknowledge the number of people in an individual's network, but also consider the quality of these relationships.

In the self-completion questionnaires (ELSA, 2019b), respondents were asked about relationships with their children, immediate family, and friends. To determine average frequency of contact the

questions included several forms of communications (e.g. face-to-face, phone calls, and written communications). Although increased frequency of contact might suggest greater relationship quality, the multiple modes of contact could be construed differently. As such, an individual who lives within close geographical proximity of their family may have more face-to-face contact. Whereas an individual with family members in different locations may maintain regular contact via. phone, but have limited availability to meet in-person. Thus, the mode of contact is not necessarily reflective of relationship quality without additional context. Alternatively, there were questions which asked how many of their children, family members, and friends, the respondent considered themselves to hold a close relationship with. These responses provide insight into the number of higher-quality relationships that an individual perceives themselves to have. For these analyses, the children and family members variables were merged, as both were considered reflective of immediate family, which also reduced the likelihood of respondents with no children being excluded from the analyses. However, it should be acknowledged, that while this variable allows the quantification of reported close relationships, it does not provide much detail about the quality of each. For example, those that listed being close with multiple friends and family members, it is unlikely that all these relationships were of the same quality. Therefore, the level of detail within this variable is focussed more on the size of someone's network than the quality. Additionally, while close family relationships can be separated from friendships, there is no supplementary information on who that person is, i.e. sibling or parent. This reflects both a limitation of the ELSA dataset, whereby specific questions about relationships were not included in the questionnaires or interviews in order to examine social networks in more detail, but this also limits the interpretation of the variable. To illustrate, using this variable, very little inference can be drawn on the link between loneliness and relationship quality or the members that constitute the network other than comparing family with friends.

Household demographics have also been linked to loneliness, such as whether a respondent has a spouse/partner (Kemperman *et al.*, 2019). As this information can be determined from marital status or cohabitation, additional household demographics were obtained using household size, that is, the number of people living in the household. Hill and Dunbar (2003) indicate that household members are fundamental to an individual's social network. In ELSA, household size was determined using a continuous variable, whereby in wave eight, the valid responses ranged from 1-11. The modal value was two, indicating that dual-person households were the most common. Due to small cell counts for the larger households, particularly for >5 people, the variable was collapsed into three categories: single person household, dual-person household, and large household ( $\geq$ 3 people).

# 4.3.8 Variables computed for the follow-up analysis

In RQ3 analyses were conducted using wave two as the first time-point, and wave eight as the second time-point. Using waves 3-7, variables were computed to control for changes that may have occurred

over the 12-year period between waves two and eight. The key areas of change were caregiving, loneliness, and mental health.

# **4.3.8.1** Defining the caregiving relationship and role across the waves

To analyse the long-term associations between caregiving and loneliness, it was important first to quantify caregiving over the 12-year period. Using the caregiving variable from each wave (2-8), 1,445 respondents had reported providing care in at least one wave. It was of interest to test whether caregiving for longer periods of time, rather than for shorter periods, was more likely to be associated with loneliness. Of the caregivers, most (N=633, 44%) provided care in only one wave, which were categorised 'short-term'. Only 14 caregivers (<1%) reported providing care in all waves. Thus, due to this small sample, it was not possible to statistically examine consistent caregivers (waves 2-8) as a distinct group. Moreover, there is limited definitive criteria surrounding what constitutes 'longterm' caregiving. US data suggests that the average length of a caregiving role is four-years, with 24% of caregivers providing care for  $\geq$ 5-years (National Alliance for Caregiving and AARP, 2015; Barnhart et al., 2020). A Canadian study investigating the impacts of the caregiving role defined short- and long-term as less than, or more than two-years respectively (Williams et al., 2014). Whereas Lacey *et al.* (2019) divided their UKHLS sample at  $\geq$ 3-years for long-term care. As ELSA data are collected every two-years, it is not possible to know what occurs between the data collection points, that is, whether caregivers continued to provide care consistently, or not. However, for the purpose of this analysis, long-term caregivers are defined as providing care for two or more consecutive waves (N=594, 41%). Some caregivers did not align with either the short-term or longterm classification; 15% (N=218) of caregivers reported providing care in more than one wave, but not consecutively (e.g. care provision in waves two and six). In this thesis, these individuals were categorised as 'intermittent caregivers'. However, caregivers providing multiple episodes of care have been labelled differently within the literature, Larkin (2009) initially developed the term 'serial caregiver', also adopted by Corey and McCurry (2018); whereas Lacey et al. (2019) refer to them as intermittent caregivers.

Dummy variables were used to identify whether caregivers had (i) ever provided care to any of the recipient groups (see below), (ii) ever provided co-residential care, (iii) ever provided high-intensity **care**, (iv) ever provided round-the-clock care, and (v) ever provided care to multiple recipients. These variables were created because, short-term caregivers were included in the analyses for RQ3b and RQ3c, therefore, as care provision occurred in only one wave for many of the sample, changes between waves could not be observed. To identify the care recipient, variables were created from reported relationships across all waves, and grouped into those who had 'ever cared for adult-child', 'ever cared for parent/parent-in-law' and 'ever cared for others', with spousal caregivers as the reference category. Due to individuals providing care to multiple recipients, which may be from the same, or different recipient groups (e.g. two parents, or a spouse and a neighbour), and missing data, it was not possible to accurately compute a variable to track care recipient changes between
caregiving episodes. Additionally, changes between care recipients within the same category would not be identified. For example, if a respondent was providing care to their mother-in-law in wave two, then to their own mother in wave six, both recipients would be classified as 'parent/parent-inlaw'.

#### 4.3.8.2 Loneliness and changes to loneliness expectations

In RQ3b and RQ3c of the follow-up regressions, the DV represents loneliness occurring after the start of care provision. This was computed by identifying which waves caregiving and loneliness were first reported. Those who indicated providing care prior to reporting loneliness were compared with those never reporting loneliness. For example, for caregivers who started providing care in wave two, their loneliness was reported at wave three or later. An implication of this variable was therefore that all caregivers reporting loneliness at wave two were excluded, because it was not possible to identify if they provided care prior to reporting loneliness, as wave two was the first wave whereby all relevant variables for these analyses were included. Furthermore, caregivers from wave eight were excluded, because at the time of conducting the analyses, ELSA wave nine data were not available, so future loneliness could not be determined. Table 8 demonstrates the waves in which caregiving and loneliness could be reported for each respondent.

T 11 0	<u> </u>	. 1	1 1.	•	С.	.1	C	• •	
lahle X.	( omnutu	10 the	loneliness	occurring	atter	the start	ot care	provision	variable
aute o.	Company	is inc	<i>ionenness</i>	occurring	ajici	ine sieni	of cure	provision	<i>runuon</i>

Wave in which loneliness was first reported
Waves 3-8
Waves 4-8
Waves 5-8
Waves 6-8
Waves 7-8
Wave 8

Source: author's own

Loneliness expectations were only asked at waves two, seven, and eight. To control for loneliness expectations in the follow-up analyses, comparison was drawn between these time-points and individuals were grouped into six categories: (i) stable– always agree, (ii) stable– always neither, (iii) stable– always disagree, (iv) negative (disagree to agree/neither, or neither to agree), (v) positive (agree to neither/disagree, or neither to disagree) and (vi) fluctuating. It was deemed important to distinguish between the stable categories because individuals who persistently agreed (expected to get lonelier with age), in line with the theory of self-fulfilling prophecies in SET, should have different loneliness outcomes to those who persistently disagreed (did not expect to get lonelier with age). Moreover, it is unclear how individuals who responded with neither perceive their future loneliness, therefore there was a lack of evidence available to collapse this into another category.

#### 4.3.8.3 Changes to mental health: depressive symptomatology

Research on obesity transitions, over an 8-year period in ELSA, computed a variable to account for changes to weight between the time-periods (Hamer *et al.*, 2015). To compute this variable, data were compared at baseline (wave two), a mid-point (wave four), and at follow-up (wave six). This method was applied when computing the mental health change variable for RQ3c (how are the caregiving relationship, and role, associated with loneliness for short-term *versus* long-term, or intermittent caregivers?). CES-D variables were compared at wave two, wave five, and wave eight, to identify how caregivers' mental health changed over the 12-year period. The respondents were classified according to whether their CES-D score was indicative of depressive symptomatology at each wave: (i) stable– no depressive symptoms, (ii) decrease in depressive symptoms, (ii) increase, or stable depressive symptoms, (iii) fluctuating. Due to a small number of respondents who had depressive symptoms at all data collection points (N=49, 1.7%), these were collapsed with the 'increase' group. It was deemed most appropriate to include those with persistent depressive symptoms with those who had developed depressive symptoms, as all other groups involved either absence or reduction in depressive symptoms at some point between waves two, five, and eight. Full descriptive information is included in section 6.2.1.2.

#### **4.3.8.4** Control variables in the follow-up analyses

After reviewing published research using ELSA data, and similar follow-up methods over a range of different time-periods, the most common means of including control variables was to adjust for baseline measurements (Shankar *et al.*, 2013; Fancourt and Steptoe, 2018). As such, for this thesis, aside from the computed change variables for loneliness expectations (section 4.3.8.2) and mental health (section 4.3.8.3), the control variables included sociodemographic, SES, social network, health, and wellbeing variables from wave two (sections 6.2.2 and 6.2.3). This method of including only baseline controls, and treating them as time-invariant, was selected to avoid complex regression models whereby the interpretations would be difficult. However, changes to loneliness expectations and mental health were included to better understand the effect of these conceptually important variables on loneliness. For discussion on the limitations of this approach, see section 7.5.3.

There was conflicting evidence (using ELSA data) about whether controlling for the outcome variable at baseline was appropriate. For example, Jackson *et al.* (2019) examined the prospective association between age discrimination at wave five, and health outcomes at wave eight, and in their analyses, the outcome variable was excluded at baseline. Alternatively, Shankar *et al.* (2013) investigated the relationship between loneliness and isolation at wave two with cognitive function at wave four. These regressions were conducted with a control for baseline cognitive function. For the purpose of this thesis, both methods were adopted and contrasted. In RQ3a separate regressions were conducted (i) when lonely individuals at baseline were excluded, and (ii) controlling for baseline loneliness.

#### 4.3.9 Weighting

Cross-sectional weighting was calculated by the ELSA research team for core sample members across all waves of the study. Weighted respondents include proxy and partial interviews, but do not include individuals living in institutionalised settings, those who live outside of England, and partners of core members (NatCen, 2018b). However, within this thesis, proxy responses were also excluded, alongside the non-weighted individuals. Cross-sectional weights were calculated separately for each cohort using ONS household data, and account for refreshment samples, and non-responders who may have responded in some waves, but not all. This ensures that all cohorts are represented in the same proportions. Weighting was also introduced for the self-completion questionnaire between waves 4-7, and individuals who returned the survey without answering the majority of the questions were allocated a weight of 'system missing' to be excluded from analyses (*ibid*).

To be eligible for longitudinal weighting, a core member must have responded to all of the waves of ELSA, to date. As such, longitudinally-weighted respondents in wave four were core members who had responded in waves 1-4 of the survey. Moreover, in wave eight, those with longitudinal weights had provided responses in waves 1-8. Unlike the cross-sectional weights, non-responders were not eligible for longitudinal weighting (NatCen, 2018b). The longitudinal weight was calculated for 3,470 core members who responded to all waves and continued living in private households, excluding individuals who reside in institutionalised settings.

The ELSA research team advocates the use of cross-sectional and longitudinal weights when analysing the data to minimise the bias of non-response (NatCen, 2018b). The cross-sectional weight applied to wave eight was 'w8xwgt', and the longitudinal weight, which was used to compare respondents between waves two and eight, was 'w8w1lwgt'.

#### 4.4 Quantitative methods

Within this thesis, several statistical techniques were used: initial linear regressions to obtain collinearity statistics and to rule out multicollinearity, descriptive analyses such as crosstabulations, and bivariate tests including Chi-squared statistics and Spearman's Rho (r). For the multivariate analyses, the primary methods were binary and multinomial logistic regressions and linear regressions. The following sections will address the assumptions and provide an overview of each of the tests.

#### 4.4.1 Bivariate analysis

Bivariate analyses were conducted to determine if any associations existed individually between any of the independent variables (IVs) and the two DVs. Chi-squared tests were used for categorical variables and Spearman's r for continuous variables.

A Pearson's Chi-squared test is used to determine whether two categorical variables are associated; the statistic is calculated by comparing observed with expected frequencies (Stockemer, 2019). The main assumption for a Chi-squared test is that both variables are categorical, and each variable has a minimum of two categories (e.g. caregiver and non-caregiver) (Field, 2013). A greater number of categories increases the degrees of freedom (df) for the test statistic, which equates to a larger critical value, Chi-squared statistics above the critical value are considered significant. Another assumption is that the observations are independent (McHugh, 2013); a Chi-squared test can be used to determine an association between caregiving and age, but should not be used to compare pre- and postobservations, such as self-rated health before and after caregiving, as these variables would be paired. One of the main limitations of the Chi-squared test is the sensitivity to sample size and cell counts; a larger sample size is advantageous and will yield results with greater statistical power. However expected cell counts must be greater than five for the majority of the cells (typically accepted that  $\geq$ 80% of cells should have an expected cell count >5). In all reported Chi-squared analyses, these checks were carried out and, in all tables, less than 20% of cell counts were <5. A further limitation is that the Chi-squared test only indicates whether variables are independent. If an association is present, the nature and direction must be tested using alternate methods.

However, as the Chi-squared test is only appropriate for categorical variables, Spearman's r is used to determine bivariate correlations among continuous variables. A Pearson's correlation would not be an appropriate method, as an assumption of this test is that both variables are continuous. Unlike the Chi-square test, Spearman's r measures both the magnitude and the direction of a relationship between two variables. Spearman's r ranks the X and Y values independently of one another, then performs the correlation based on the ranks (Wilcox, 2016) and can be used with non-parametric data (Field, 2013).

#### 4.4.2 Multivariate analysis

Logistic regressions are used to predict a categorical outcome, using a combination of categorical or continuous predictor variables. In this thesis, the main DV, UCLA-LS, alongside other loneliness and health variables (section 4.3) was categorical, therefore logistic regressions were the most appropriate method. Furthermore, binary logistic regressions were used when the outcome has two categories, or multinomial logistic regressions when the outcome has more than two categories. Although regressions are considered to be linear models, this assumption cannot be met with categorical variables; therefore, a link function of the outcome is given. For logistic regression, this

is typically the logit, or log-odds calculated as:  $P(Y) = \frac{1}{1 + e^{-(b_0 + b_1 X_{1i})}}$ where P(Y) represents the probability of the outcome (e.g. being lonely, Y=1), *e* represents the natural logarithm base, *b*<sub>0</sub> the Y intercept, *b*<sub>1</sub> the predictor-outcome relationship, and X<sub>1i</sub> the predictor variables' value (Field, 2013). In logistic regression output tables, *B* is given in log-odd units, thus it is assumed that the logit is linearly related to the explanatory variables. However, odds ratios (Exp(B)) are more intelligible and are frequently used within research to interpret logistic regressions.

When using SPSS software, as standard, the logistic regression output reports the Nagelkerke R square. This statistic explains the proportion of variance in the outcome variable, which can be explained by the predictors in the model. Aurora and Elena (2019) suggest that Nagelkerke is favoured over Cox and Snell due to its ease of interpretation; the Nagelkerke scale ranges from 0-1 providing a percentage of variance, whereas this is not the case for Cox and Snell. In addition to Nagelkerke, the Hosmer-Lemeshow statistic is an additional measure of model fit. The Hosmer-Lemeshow test sorts observations into ten groups based on the regression model predictions. The observed and estimated frequencies are then compared in a contingency table, and a Chi-squared statistic is used to determine goodness of fit (Fagerland and Hosmer, 2017). If the significance level of the Chi-squared statistic is above 0.05, then the model is deemed a good fit, whereas a significant Chi-squared (p<.05) indicates a poor model fit.

Alternatively, linear regressions are applied when the outcome variable is continuous. A typical

linear regression equation is denoted as:  $Y_i=\beta_0+\beta_1X_i+\varepsilon_i$  whereby:  $Y_i$  represents the outcome variable,  $\beta_0$  the intercept,  $X_i$  the predictor and  $\beta_1$  corresponding coefficient, and  $\varepsilon_i$  is the error associated with the model (Field, 2013). One of the main assumptions of a linear model is that the DV is continuous, and the IVs are continuous or binary (Tranmer *et al.*, 2020). To accommodate this assumption, categorical variables in the analysis were recoded into dummy variables. The SPSS output from a linear regression provides an R<sup>2</sup> value. The R<sup>2</sup> value indicates the variance explained by the model (Field, 2013), similar to the Nagelkerke R<sup>2</sup>, discussed previously. Another main assumption of both regressions is limited multicollinearity, which is discussed in the subsequent section.

#### 4.4.3 Missing data

A complete case analysis (CCA) approach was applied throughout this thesis, therefore all analyses were restricted to respondents with complete data for all included variables. The main limitation to this approach is that the sample size is often substantially affected, which can lead to a loss in statistical power (Lodder, 2013). However, CCA was deemed more appropriate than other methods of dealing with missing data, such as imputation. Largely, imputation methods require the missing data to be missing at random (dependent only on observed data but not unobserved data), or missing completely at random (unrelated to observed or unobserved data) (Hughes *et al.*, 2019; Griswold *et al.*, 2021). Whereas CCA can return unbiased results under most circumstances, for example, even where data are missing not at random (Bartlett *et al.*, 2014; Hughes *et al.*, 2019).

Imputation models should contain all the variables in the regression model, the outcome variable, variables that explain why the data are missing, and variables that predict the values of the incomplete

variables (Hughes *et al.*, 2019). As it was not possible to determine whether the missing data in ELSA was missing at random, or not at random, it would be difficult to be certain about what variables explained the missing data. While methods such as regression imputation can predict the missing values from other non-missing variables and maintain the distribution of the sample, they are still linked to biased estimates if the data are not missing completely at random (Lodder, 2013).

Within the full cross-sectional sample in this thesis, 16% are caregivers, 84% non-caregivers, these proportions were almost mirrored in the sample with missing data, whereby 17% were caregivers and 83% non-caregivers. Additionally, the sociodemographic characteristics of the missing data were similar to that of the full sample, suggesting limited bias was introduced into the remaining sample. Larger percentages of missing data were seen for regressions conducted for RQ1a and RQ1b, ranging from 45-48% of the full sample (N=5937) excluded due to incomplete data. The variable which was most likely to be missing from these cases was the number of close relationships with family, in comparison to key variables, the response rate for which was around 65% for those providing UCLA-LS scores, and 66% for those providing a response to the caregiving question. Furthermore, in all of the regressions conducted for RO2, between 43-48% of the caregiving-only sample (N=962) were excluded due to incomplete data, this was likely due to the same low response rates to the number of close family relationships. Among the caregivers, 65% provided family relationship data. Finally, for the longitudinal caregivers with missing data, most were short-term caregivers (38%) or longterm caregivers (38%) which was similar to the proportions of caregiving durations within the full longitudinal caregiving sample (44% and 41% respectively). Similarly, the baseline sociodemographic characteristics of the missing respondents were similar to that of the full sample, for example, most were aged 50-59 (43%) as in the overall sample (46%). In RQ3c around 44% of the longitudinal caregiving sample (non-lonely caregivers, or caregivers experiencing loneliness after the start of care provision, N=938) were excluded due to incomplete data. The lowest response rate within this sample was for the computed changes to loneliness expectations variable, around 58% of caregivers provided this data over the three waves it was measured (waves two, seven and eight).

#### 4.4.4 Multicollinearity

An assumption of regression models is that no strong linear relationships should exist between the predictor variables (Field, 2013), which is known as multicollinearity. Multicollinearity is problematic in regression analyses, causing large sampling variability among the coefficients reducing the accuracy of the results (Alin, 2010). Multicollinearity is measured using the variance inflation factor (VIF) and tolerance statistic, which can be obtained by running a linear regression. A larger VIF demonstrates higher sampling variability and suggests the presence of strong linear relationships, and tolerance represents the reciprocal of this value. The highest VIF value should not exceed 10 (Alin, 2010; Field, 2013), and the average should not be substantially greater than 1. Alternatively, Daoud (2017) indicates that VIF values of >5 suggest highly correlated variables, values from 1-5 suggest moderate correlation, and VIF equal to one shows no correlation. For

tolerance, it is typically agreed that values <0.1 indicate collinearity (Field, 2013; Daoud, 2017). Prior to conducting the regression analyses, multicollinearity was tested, and all values for VIF and tolerance were within accepted parameters.

### 4.5 Chapter summary

Chapter four has introduced the methods and materials used within this thesis. Justification was made for variable selection, and the bivariate and multivariate methods were outlined. This thesis used bivariate analyses when discussing the descriptive statistics, and multivariate methods, namely binary logistic and linear regressions, for the main analyses. A combination of both cross-sectional (chapter five) and follow-up methods (chapter six) were employed to disentangle the relationships between caregiving, loneliness, health, and wellbeing using ELSA data. The following chapters present the results from the analyses and build a narrative of how the key concepts are interrelated.

## **Chapter 5** Cross-sectional results

### 5.1 Chapter outline

Chapter five presents the cross-sectional results. In section 5.2 the first research question was addressed: how does loneliness differ for caregivers and non-caregivers? Descriptive and bivariate analyses were conducted for the wave eight sample in relation to both UCLA-LS and the direct loneliness question (section 5.2.1). Sections 5.2.2 to 5.2.4 display the results from the regression analyses, comparing loneliness, the determinants of loneliness, and the impact of loneliness on health and wellbeing, respectively, among caregivers and non-caregivers.

Additionally, section 5.3 focuses on the caregiving subsample only, answering research question two: how does caregiving affect loneliness, health, and wellbeing? The questions compare the caregiving sample through the caregiving characteristics, the caregiving relationship, and the caregiving role. Descriptive and bivariate analyses are presented in relation to UCLA-LS loneliness (section 5.3.1) and the subsequent sections (5.3.2 to 5.3.4) present the regression results for the loneliness, health, and wellbeing outcomes.

## 5.2 RQ1: how does loneliness differ for caregivers and noncaregivers?

#### 5.2.1 Descriptive statistics for caregivers and non-caregivers

Once proxy responses were removed and the cross-sectional weight (w8xwgt) applied, the total sample was 5,937. The descriptive characteristics for wave eight include crosstabulations between caregiving and the key variables within the analysis, including sociodemographic characteristics, SES, social networks, health, and wellbeing.

No multicollinearity was observed within the included variables, and the VIF and tolerance values were well within the guidelines (VIF <10 and Tolerance >0.1, (Field, 2013)). VIF ranged from 1-3, suggestive of some moderate correlations. A correlation matrix revealed that the strongest correlations existed between cohabitation and household size (r=-0.62), CASP19 and SWLS (r=-0.61), and age and employment (r=0.50), and all other correlations were considerably weaker. The correlation between cohabitation and household size was expected as individuals cohabiting would have larger households than non-cohabiting (e.g. single households) but it was deemed important for household size to remain in the analyses to determine if differences existed for the larger households ( $\geq$ 3 people) as this would suggest that relationships outside of a spousal relationship were also important for loneliness. The wellbeing variables were correlated, however, previous literature identified an overlap but confirms QoL and life satisfaction are separate measures of wellbeing. For

instance, Dodge *et al.* (2012) indicate that QoL cannot completely define wellbeing because it is a dimension, and similarly, life satisfaction plays a pivotal role in determining subjective wellbeing. This emphasises that the two are distinct, measuring separate, but linked, aspects of wellbeing. Finally, the association observed between age and employment is likely indicative of retirement, whereby under retirement age, employment rates are would be higher, and above retirement age, are expected to be lower. The caregiving role (e.g. caregiving intensity) is often influenced by the caregivers' employment status (Wang *et al.*, 2020b); moreover, caregivers often cease paid employment due to caregiving commitments (Carr *et al.*, 2018). Thus, employment was considered a key variable, not only for SES, but also in relation to the experience of caregivers.

Table 9 compares key variables for sociodemographic characteristics, SES, social networks, loneliness, health, and wellbeing between caregivers and non-caregivers. Respondents who identified as caregivers, having provided care within the past week, represented 16% of the sample (N=962). In addition, bivariate analyses were conducted; each of the variables were tested for associations with directly-reported and UCLA-LS loneliness (Appendix B).

Table 9: Descriptive	e statistics for caregivers and non-c	aregivers in ELSA wave e	ight
		Non-caregivers	Caregivers
		N (%)	N (%)
Caregiving status		4975 (83.7)	962 (16.2)
	50-59	1469 (29.5)	352 (36.6)
Age (years)	60-69	1656 (33.3)	311 (32.3)
	≥70	1849 (37.2)	299 (31.1)
Sev	Male	2489 (50.0)	381 (39.6)
Sta	Female	2487 (50.0)	582 (60.4)
	Married/civil partnership	2672 (53.7)	590 (61.4)
	Single, never married	415 (8.3)	54 (5.7)
Marital status	Remarried	495 (10.0)	121 (12.6)
	Divorced/legally separated	725 (14.6)	146 (15.1)
	Widowed	666 (13.4)	51 (5.3)
Cohabitation	No	1509 (30.5)	189 (19.6)
(spouse/partner)	Yes	3436 (69.5)	774 (80.4)
In paid	No	3082 (62.4)	630 (65.6)
employment	Yes	1859 (37.6)	330 (34.4)
	Quartile 1 (poorest)	1210 (24.6)	257 (26.9)
<b>W</b> 7141-	Quartile 2	1231 (25.1)	237 (24.8)
wealth	Quartile 3	1247 (25.4)	219 (22.9)
	Quartile 4 (wealthiest)	1221 (24.9)	242 (25.3)
	Single person household	1245 (25.0)	92 (9.6)
Household size	Dual-person household	2706 (54.4)	614 (63.8)
	Large household (≥3)	1025 (20.6)	256 (26.6)
Close	with family, mean (SD)	4.6 (2.9)	4.9 (3.0)
relationships	with friends, mean (SD)	3.5 (3.1)	3.7 (2.7)
Direct loneliness	Not lonely	4405 (88.6)	856 (89.2)
question	Lonely	565 (11.4)	104 (10.8)
	Not lonely	3905 (80.0)	746 (79.6)
UCLA-LS	Lonely	977 (20.0)	191 (20.4)
Lonalinasa	Agree	2041 (41.7)	392 (41.1)
concentrations	Neither agree nor disagree	1450 (29.6)	277 (29.2)
expectations	Disagree	1406 (28.7)	279 (29.4)
נו דו	No	3384 (68.0)	635 (66.0)
	Yes	1592 (32.0)	327 (34.0)
Dhysiological	No chronic CV conditions	2702 (54.3)	582 (60.5)
Physiological	$\geq 1$ chronic CV condition	2273 (45.7)	380 (39.5)
Haalth	Agree	3434 (70.0)	669 (70.2)
Avportations	Neither agree nor disagree	861 (17.6)	161 (16.9)
expectations	Disagree	608 (12.4)	123 (12.9)
Alcohol	<3 days per week	3279 (66.7)	680 (71.4)
consumption	≥3 days per week	1634 (33.3)	272 (28.6)
CES D	No depressive symptomatology	4128 (83.5)	725 (76.0)
(E2-D	Depressive symptomatology	814 (16.5)	229 (24.0)
Wallbair	CASP19, mean (SD)	41.7 (9.0)	39.9 (10.1)
wendenig	SWLS, mean (SD)	25.4 (6.4)	24.9 (7.1)

Notes: standard deviation (SD), source: author's analysis of ELSA wave 8

#### 5.2.1.1 Sociodemographic characteristics and socioeconomic status

Typically, caregivers were younger than non-caregivers, as higher percentages of caregivers (37%) were aged 50-59 compared to non-caregivers (30%). Moreover, caregivers were predominantly female (60%), whereas the male and female split was exactly 50% among non-caregivers. This female majority could have been expected, as literature and previous evidence suggests a higher prevalence of caregiving among females compared to males, except in the oldest age groups (Milligan and Morbey, 2016).

Amongst caregivers and non-caregivers, respondents were most commonly married (61% and 54% respectively). However, small cell counts were observed for single, never married (N=54, 6%) and widowed caregivers (N=51, 5%) which resulted in large confidence intervals and empty cells in an initial regression analysis. Measures were taken to collapse these categories, as it was hypothesised that the presence or absence of a relationship would be a substantial factor related to loneliness and should be accounted for. It was assumed that single, never married would refer to individuals without a significant attachment (e.g. partner), however crosstabulation with the cohabitation variable (Table 10) indicated that a large percentage of the single caregivers (40%) reported cohabiting with a partner. Furthermore, 25% of divorced or legally separated and 14% of widowed caregivers also reported cohabitation. This raised complications with collapsing the categories; for instance, collapsing into 'currently married' (married and remarried) and 'not currently married' (single, divorced or separated and widowed) would mean a substantial proportion of the 'not currently married' were still cohabiting with a partner. Therefore, single-cohabiting individuals may experience similar emotional attachment benefits to married-cohabiting individuals, although in the ELSA data, this would not be accounted for by controlling for marital status alone. However, given that the majority of 'currently married' caregivers were cohabiting, the cohabitation variable would account for the aforementioned attachment. Moreover, the small cell counts of 'currently married' caregivers who were not cohabiting prevented the inclusion of both variables in the regressions. Due to the arguments presented above, it was deemed more conceptually appropriate to include cohabitation in the regression analyses, rather than marital status.

Table 10:	Crossiadulation of caregivers respons	es lo martial status ana	<u>conabilation var</u>	abi
		Cohabitat	ion	
	Marital status	No	Yes	
		N (%)	N (%)	
Currently	Married, first and only marriage	2 (0.3)	588 (99.7)	
married	Remarried, second or later marriage	0 (0.0)	121 (100.0)	
Not	Single, never married	33 (60.0)	22 (40.0)	
currently	Divorced or legally separated	109 (75.2)	36 (24.8)	
married	Widowed	44 (86.3)	7 (13.7)	

Notes: % percentage of marital status, source: author's analysis of ELSA wave 8

SES was determined through employment status and wealth. Levels of employment were comparable between the two groups: 38% of non-caregivers and 34% of caregivers were employed. Finally, the distribution of wealth was consistent across the quartiles for both caregivers and non-caregivers, although a slightly larger percentage of caregivers were in the lowest quartile (27%), compared to non-caregivers (25%).

#### 5.2.1.2 Social networks

Caregivers and non-caregivers differed considerably in household size, whereby 25% of noncaregivers lived in single person households, compared to around 10% of caregivers. As the data indicated the majority of caregivers were cohabiting (80%), this suggests that a large percentage of caregivers may live with the care recipient. The descriptive analyses for RQ2 (section 5.3.1) confirmed this, indicating that over half (51%) of caregivers reside with their care recipient. The number of close relationships with family and friends reported by caregivers and non-caregivers were similar. To illustrate, the mean number of close family relationships was 4.9 for caregivers and 4.6 for non-caregivers. Similarly, the mean number of close friendships was 3.7 and 3.5 for caregivers and non-caregivers respectively. This suggests that among both groups, respondents had larger family than friendship networks.

#### 5.2.1.3 Loneliness

Three measures of loneliness were included in the analyses for RQ1a, the first of which was a direct question: whether the respondent felt lonely much of the time during the past week. Within the sample, 11% of both caregivers and non-caregivers directly-reported feeling lonely. Whereas, for the UCLA-LS, an indirect 3-item loneliness scale, the prevalence of loneliness appeared much higher. For both groups, 20% of respondents were lonely. As discussed in section 4.3.2.1, stigmas around loneliness may have influenced the likelihood of directly reporting, therefore this may explain why considerably fewer respondents were lonely when using this measure.

The final method determined the respondents' future loneliness expectations. These expectations were similar amongst caregivers and non-caregivers, as such 41% of caregivers and 42% of non-caregivers 'agreed' that they expected to become lonelier with age. Similar percentages (29%) were also observed for those who 'disagreed', in that they did not expect to become lonelier.

#### 5.2.1.4 Health and wellbeing

Health status was not substantially different for caregivers and non-caregivers, the majority of both groups were healthy across all measures. Furthermore, all health measures were significantly associated with both loneliness measures. The prevalence of LLTIs was slightly higher among caregivers (34%) than non-caregivers (32%), and caregivers showed elevated levels of depressive symptoms (24%) compared to non-caregivers (17%). Among the SET health pathways, similar percentages (70%) of caregivers and non-caregivers 'agreed' that they expected their health to get worse with age. Moreover, for behavioural health, non-caregivers (33%) were more likely to consume alcohol frequently ( $\geq$ 3 days per week), than caregivers (29%). Additionally, non-caregivers reported a higher prevalence of chronic CV conditions (46%), compared to caregivers (40%). When considering only the SET health pathways, caregivers appeared healthier than non-caregivers.

Overall caregivers displayed poorer wellbeing than non-caregivers based on both QoL and life satisfaction. QoL among caregivers was on average 2.2 points lower than non-caregivers. Similarly, caregivers reported lower life satisfaction than non-caregivers.

#### 5.2.2 RQ1a: is caregiving associated with loneliness?

Alongside investigating whether loneliness differed between caregivers and non-caregivers, analyses were conducted using different loneliness measures. Comparison between the results identified

whether reporting of loneliness was consistent across different measurement tools. Binary and multinomial logistic regression analyses were conducted with the three loneliness measures (direct question, UCLA-LS, and loneliness expectations) to determine whether caregiving was associated with loneliness. Table 11 depicts the models applied for each of the regressions in RQ1a.

Table 11: Regres	ssion models for RQ1a: is caregiving associated with loneliness?
Model number	Variables included in the model
1	Caregiving status
2	Model 1 + sociodemographic characteristics and socioeconomic status
3	Model 2 + social networks
4	Model 3 + health and wellbeing

Source: author's own

The initial model (model one) tested whether caregiving individually, was significantly associated with the different loneliness measures, whereas in model two sociodemographic characteristics and SES were included as covariates alongside caregiving. Model two tested whether caregiving was associated with loneliness after controlling for characteristics such as age, sex, employment status and wealth. Some sociodemographic characteristics are considered to be both determinants of loneliness and typical caregiver characteristics. For example, evidence from literature suggests that females are more likely to be caregivers (section 3.3.1.2) and more likely to report loneliness (section 3.2.1). Therefore, by controlling for sex in the analyses, any improvements or attenuation to the relationship between caregiving and loneliness demonstrates whether any of the relationship can be accounted for by sex.

Model three included the addition of social network variables: household size, close family relationships and close friendships. The literature review highlighted that factors, such as social isolation and social networks, are closely linked to loneliness (section 3.2.3), therefore it was important to evaluate the impact of these variables. By adding these in model three, separately from sociodemographic variables and SES, the results would indicate whether social networks better account for the relationship between caregiving and loneliness, after having already controlled for the variables in model two. Health and wellbeing variables were added into model four (fully-adjusted model). Research suggests that poor health and wellbeing contribute to loneliness, thus any attenuation to previously significant relationships (e.g. caregiving and loneliness) would indicate that health and wellbeing were more strongly related to loneliness, after controlling for caregiving. Health and wellbeing were added separately to sociodemographic characteristics, SES, and social networks because, typically, research has shown that health and wellbeing have different effects on loneliness in comparison with the other characteristics.

#### 5.2.2.1 Is caregiving associated with directly reporting loneliness?

The first binary logistic regression was conducted with direct loneliness as the DV; 45% (N=2701) of the sample were excluded from the analyses due to missing data. In this analysis, caregiving was

not significantly associated with directly reporting loneliness, even after controlling for differences in sociodemographic characteristics, SES, social networks, health, and wellbeing (Appendix C).

In the fully-adjusted model (Figure 3), for sociodemographic and socioeconomic variables, cohabitation was negatively associated with loneliness and wealth positively associated. Other factors negatively associated with loneliness were: household size, LLTIs, QoL, and life satisfaction, with depressive symptomatology demonstrating a positive association to loneliness. This suggests that health and wellbeing largely influenced directly-reported loneliness.

LLTIs were negatively associated with loneliness; odds of respondents with LLTIs reporting loneliness were 29% lower than those without LLTIs (odds ratio (OR)=0.71, 95% confidence interval (95CI)=0.51-0.99). Typically, within the literature, health has been shown to be negatively correlated with loneliness. A review conducted by Cohen-Mansfield et al. (2016) identified that poor health and functional limitations were linked to increased loneliness, although the loneliness measurement tool varied across these studies. Moreover, a link was established between long-term health conditions, determined through the receipt of disability pension, and directly reporting loneliness (von Soest et al., 2020). Therefore, it was unexpected that the results of this thesis suggests LLTIs are indicative of lower odds of loneliness, as this contradicts published findings. One explanation for this could be that individuals with LLTIs may be more likely to receive help due to their illness, whether from professionals or family members; they may have people coming into their home or accompanying them to assist with tasks such as shopping. These visits may provide companionship and therefore the respondent may be less likely to feel lonely. Individuals without LLTIs are likely not to receive assistance and thus could feel lonelier, despite better health. Conversely, evidence suggests health is a determinant of loneliness, but, when using a direct measure, health was inferior to partner status (defined as having a spouse/cohabitant) in predicting loneliness (Nicolaisen and Thorsen, 2014b). The findings from this regression may support this conclusion, as cohabitation remained significant after the inclusion of health and wellbeing in the fully-adjusted model, which could indicate a strong relationship between cohabitation and directly-reported loneliness.

The Nagelkerke  $r^2$  indicated that the fully-adjusted model accounted for 39% of variance in loneliness, which was a substantial increase from model three (20%). This indicates that health and wellbeing variables play a large role in the variance of loneliness. However, the Hosmer-Lemeshow statistic advocates that model three (p=.844) was a better fit than the fully-adjusted model (p<.001). Despite the significant results, the reduced model fit would suggest that the remaining relationships in the fully-adjusted model were contributing poorly to the model fit. As such, health and wellbeing may have such a substantial effect that it overshadows the other variables.



Figure 3: Odds ratio plot, factors associated with directly reporting loneliness Source: author's analysis of ELSA wave 8

#### 5.2.2.2 Is caregiving associated with loneliness on the UCLA loneliness scale?

Caregiving was significantly positively associated with UCLA-LS loneliness in the first three models, but the significance attenuated with the inclusion of health and wellbeing variables in the fully-adjusted model (Appendix D). The regression analysis included 54% of the full sample, 46% (N=2724) were excluded from the analyses due to missing data. In the initial model, with no additional controls, odds of caregiver loneliness (score of  $\geq$ 3 on the UCLA-LS) were higher than non-caregivers (OR=1.35, 95CI=1.06-1.71). However, the Nagelkerke r<sup>2</sup> indicated that this initial model accounted for less than 1% of variance in loneliness and the Hosmer-Lemeshow statistic demonstrated that the model was a poor fit. With the inclusion of sociodemographic characteristics, SES, and social networks in model three, the odds of caregivers being lonely were almost 1.5 times higher than non-caregivers (OR=1.46, 95CI=1.13-1.89). The Nagelkerke r<sup>2</sup> indicated that model three accounted for more variance (15%) but overall the model fit did not substantially improve (p=.004). In the fully-adjusted model (Figure 4), caregiving was no longer significant, but the Hosmer-Lemeshow statistic suggests that this model better fits the data (p=.675).



Figure 4: Odds ratio plot, factors associated with UCLA-LS loneliness Source: author's analysis of ELSA wave 8

Similar to the direct loneliness regression (section 5.2.2.1), cohabitation, LLTIs, higher QoL and higher life satisfaction were negatively associated with UCLA-LS loneliness. Wealth and depressive symptomatology were significantly positively associated with loneliness. In contrast, household size was not statistically significant, and frequent alcohol consumption was negatively associated with UCLA-LS loneliness, but not significantly associated with directly-reported loneliness.

As with direct loneliness, after controlling for health and wellbeing in the fully-adjusted model, wealth was positively associated with loneliness. However, unlike direct loneliness, wealth was significant in previous models. Prior to the inclusion of health and wellbeing controls, wealth was negatively associated with loneliness. Wealthier people tend to have better access to healthcare, additionally, the negative association between health and loneliness is well-reported in literature (section 3.4.1). Therefore, prior to controlling for health, it could be predicted that those in higher wealth quartiles may have better health, and are therefore less likely to be lonely. Conversely, once controlled for health and wellbeing differences in the fully-adjusted model, wealth had contrasting effects. Higher wealth was indicative of higher odds of loneliness. This contrasts with previous

literature, as it is commonly reported that lower SES is linked to poorer social relations (Vonneilich *et al.*, 2012).

Respondents who consumed alcohol three or more days per week were found to have significantly lower odds of loneliness, compared to those who consumed alcohol less frequently (OR=0.76, 95CI=0.58-0.99). Canham et al. (2016) investigated the association between loneliness and alcohol, also finding that loneliness was negatively associated with frequent alcohol consumption. Descriptive analyses of their sample showed that among the respondents drinking >4 days per week. only 27% reported feeling lonely often, whereas respondents reporting drinking one day per week were lonelier (39% reported feeling lonely often). There is limited evidence reporting the effect of alcohol consumption on loneliness, however, Dunbar et al. (2017) suggest that moderate alcohol consumption can have social benefits. In particular, those who drink in their 'local' or community pubs had higher social engagement than that of non-drinkers. Nevertheless, due to the cross-sectional nature of this part of the analysis, it is not possible to determine the direction of the relationship, i.e. whether increased alcohol consumption reduced likelihood of loneliness, or whether lonelier people were less likely to frequently consume alcohol. Greenstone et al. (2019) examined the bi-directional effects between loneliness and alcohol and found no significant evidence of causality in either direction. However, it was suggested that stress played an indirect role in risky behaviours (*ibid*), therefore the relationship between alcohol and loneliness may be stronger among the caregiving sample, due to stress from the caregiving role. This is investigated further in section 5.2.4.2.

#### 5.2.2.3 Does caregiving impact loneliness expectations?

The final regression for RQ1a was conducted with loneliness expectations as the DV, 45% (N=2678) of the sample were excluded due to missing data. Respondents either 'agreed', 'disagreed', or 'neither agreed nor disagreed' that loneliness would increase as they grew older. In these multinomial logistic regressions, caregiving was not significantly associated with loneliness expectations. Nagelkerke r<sup>2</sup> indicated that the fully-adjusted model accounted for 19% of variance in loneliness expectations, which suggests that there are additional factors, not included in the model, which may have a substantial impact on loneliness expectations.

Significant factors associated with loneliness expectations varied for 'neither agree nor disagree' ('neither') and 'disagree', in relation to 'agree' (Appendix E). An individual who 'disagrees' does not expect to become lonelier with age, whereas those who responded with 'neither' may be unsure about whether they expected increased loneliness as they age. In the fully-adjusted model for 'disagree' compared to 'agree', sex, wealth, and depressive symptomatology were negatively associated with 'disagreeing', whereas cohabitation, close relationships with family and friends, LLTIs, health expectations, and QoL were positively associated with 'disagreeing'. In contrast, for 'neither' compared to 'agree', sex, employment, wealth, and depressive symptomatology were

negatively associated, and physiological health, health expectations, and QoL positively associated with the 'neither' response.

#### Odds of disagreeing with loneliness expectations, compared to agree

In the fully-adjusted model (Figure 5), odds of 'disagreeing' were significantly lower for females than males (OR=0.80, 95CI=0.66-0.97), which suggests that females expected to become lonelier as they grow older. It is frequently reported that females are more inclined to report being lonely, compared to males (Pinquart and Sörensen, 2011). It is therefore possible that this stereotype (that females are lonelier) has been internalised, influencing their perception of loneliness. Moreover, cohabitation was positively associated with 'disagreeing' (OR=1.53, 95CI=1.02-2.29); individuals living with a spouse or partner were less likely to expect to get lonelier with age, than those not cohabiting. This is could be attributed to the presence of a strong emotional attachment with their spouse. As shown in the previous two analyses, cohabitation was negatively associated with directly-reported and UCLA-LS loneliness, Therefore, those living with their spouse were less likely to be lonely, which in turn is likely to have a positive effect on their loneliness expectations. As such, individuals who are currently lonely may be more likely to expect to stay lonely, or become lonelier, whereas those who are not currently lonely may not expect to become lonely, as they have their spouse and/or other network members and may expect this to continue.



Figure 5: Odds ratio plot, factors associated with disagreeing with loneliness expectations Source: author's analysis of ELSA wave 8

Close relationships play a role in an individual's future loneliness expectations. The number of close family members (OR=1.06, 95CI=1.03-1.10) and close friendships (OR=1.06, 95CI=1.02-1.09) were positively associated with 'disagreeing' with loneliness expectations. This advocates that those with larger close social networks were more positive about future loneliness expectations, possibly because they expect to maintain their networks as they age. Those with smaller, or without close networks may expect to be lonelier with age, as they may already be experiencing loneliness. Alternatively, household size was not significantly associated with loneliness expectations. This indicates it is not the number of people an individual has around them, more so the quality (or closeness) of those relationships which impact perceptions of loneliness. Additionally, health expectations were positively associated with loneliness expectations. For example, respondents who did not expect their health to deteriorate with age, were also less likely to expect increased loneliness (OR=3.25, 95CI=2.45-4.33). This is suggestive of dispositional optimism, whereby individuals' personalities predispose them to more positive expectations (Rius-Ottenheim *et al.*, 2012). Furthermore, optimism is not limited to one area of life (e.g. health) and more likely to be generalised (e.g. health, loneliness, and other aspects).

The results suggest that depressive symptomatology was suggestive of a negative outlook on loneliness and age. The odds of 'disagreeing' with loneliness expectations were considerably lower

among respondents experiencing depressive symptoms than those without depressive symptomatology (OR=0.60, 95CI=0.44-0.82). Evidence suggests that depression is linked to negative self-evaluation, in that individuals are more likely to consider themselves worse off than others (McCarthy and Morina, 2020). Therefore, it is likely that individuals with depressive symptomatology may consider themselves lonelier than others, and thus demonstrate a negative outlook on future loneliness. It was expected that poor physical health (LLTI) would increase the likelihood of loneliness, but, as discussed previously (section 5.2.2.1), potential assistance received due to illness may act as a buffer between poor health and loneliness. In contrast, individuals with poor mental health may be less likely to receive help, as late-life depression is considerably underdiagnosed and undertreated (Mann *et al.*, 2020). These individuals may already experience loneliness and expect this to develop with age.

# Odds of neither agreeing nor disagreeing with loneliness expectations, compared to agree

Females were less likely to be uncertain about future loneliness than males (OR=0.83, 95CI=0.70-0.99). As females were also less likely to 'disagree' (see previous section: disagree *versus* agree), this would indicate that males have a more positive outlook on future loneliness, compared to females. However, as discussed from previous results and published literature, females are more likely to report loneliness, therefore this stereotype could influence their perceptions.

In the fully-adjusted model (Figure 6), the odds of individuals with poor physiological health, determined by chronic CV conditions, reporting 'neither', were 1.2 times higher than those with good physiological health (OR=1.21, 95CI=1.01-1.46). Individuals with chronic CV conditions, (e.g. hypertension), may be unsure about the progression of their condition, and the forthcoming impact it may have. To illustrate, if their condition considerably worsened, this would limit their ability to go out and socialise, but if their condition remained stable, it may involve limited change to their lives, hence the increased likelihood of uncertainty. As discussed in the previous section (disagree *versus* agree), expectations for health and expectations for loneliness were related, although it is unknown whether this was due to a generalised outlook on life, or due to relationships between loneliness and health. For example, if an individual is unsure about future health it may be difficult to make an accurate judgement on how lonely they expect to be, as this will depend on other factors (such as health). Therefore, it was expected that a 'neither' response for health expectations was likely to be linked to higher odds of a 'neither' response for loneliness expectations, which was reflected in the results (OR=1.55, 95CI=1.22-1.96).

Similarly to 'disagree', odds of uncertainty about future loneliness were lower among respondents with depressive symptomatology than those without depressive symptoms (OR=0.52, 95CI=0.39-0.68). As discussed, depression is linked to negative self-evaluation (McCarthy and Morina, 2020), which implies, and as the results show, individuals with depressive symptoms are more likely to 'agree' with loneliness expectations than be uncertain. Finally, QoL was positively associated with

the 'neither' response (OR=1.04, 95CI=1.02-1.05), but life satisfaction was non-significant. The SWLS asks whether individuals have important things in their life and, if they could relive their life, would they change much, whereas CASP19 includes several questions about the future. Therefore, in contrast, CASP19 is more likely to be related to future expectations than SWLS, which is more focused on current perceptions.



Figure 6: Odds ratio plot, factors associated with neither agreeing or disagreeing with loneliness expectations Source: author's analysis of ELSA wave 8

#### 5.2.2.4 Summary: is caregiving associated with loneliness

Caregiving was significantly positively associated with UCLA-LS loneliness across the first three models, prior to the inclusion of health and wellbeing controls. In contrast, caregiving was not significant in any model for direct loneliness or loneliness expectations. This highlights a substantial discrepancy between the loneliness measures, whereby caregivers were more likely to be categorised as lonely using an indirect, but not direct, measure. This may be due to assumptions that caregivers cannot be lonely, because they spend time caregiving for others. Aligning with the psychological SET pathway, if this assumption is internalised, caregivers may overlook feelings of loneliness, or

feel unable to report them because they believe that caregivers cannot be lonely. However, as discussed in the definitions (section 1.2.2) an individual can feel lonely without being alone. Using HRS data, Shiovitz-Ezra and Ayalon (2012) compared the same loneliness measures: the direct CES-D question and 3-item UCLA-LS. The authors reported an overlap between the direct question and UCLA-LS, but also several discrepancies between the measures. A direct measure was considered more appropriate in self-completion questionnaires compared to face-to-face (as it is asked in ELSA) or telephone interviews, due to stigmatisation of loneliness. The two scales captured different samples with distinct characteristics (*ibid*), and therefore comparison is important to determine which scale is the most appropriate in the given sample.

RQ1a: is ca	regiving asso	ciated with lonel	<mark>iness?</mark>	
Direct		Loneliness expectations		
Question	UCLA-LS	Neither agree nor disagree	Disagree	
NS	NS	NS	NS	
NS	NS	NS	NS	
NS	NS	NS	NS	
NS	NS	*	*	
***	***	NS	*	
NS	NS	*	NS	
**	NS	NS	**	
**	*	NS	***	
*	**	**	***	
**	NS	NS	NS	
*	NS	NS	NS	
NS	NS	NS	**	
NS	NS	NS	**	
*	**	NS	***	
NS	NS	**	NS	
NS	NS	***	***	
NS	NS	**	***	
NS	*	NS	NS	
***	**	***	**	
***	***	***	***	
**	***	NS	NS	
	RQ1a: is ca           Direct           Question           NS           NS           NS           ***           **           **           **           **           **           NS           ***           ***	RQ1a: is caregiving assoDirect QuestionUCLA-LSNSNSNSNSNSNSNSNS********NS**NS**NS**NS**NS**NS**NS**NS*NS*************************	RQ1a: is caregiving associated with lonel.Direct QuestionUCLA-LSLoneliness ex nor disagreeNS******NSNS**NSNS****NSNSNS****NSNS***NSNSNSNSNSNSNSNSNSNSNSNSNS**NSNSNS**NSNSNS**NSNSNS**NSNSNS**NSNS**NS******NS**NS***NS	

Notes: NS non-significant, \*p<.05, \*\* p<.01, \*\*\* p<.001, source: author's analysis of ELSA wave 8

In addition to this, the significance of covariates fluctuated between the loneliness measures (Table 12). This further emphasises differences between loneliness when measured directly or indirectly. Arguably, UCLA-LS is more reflective of feeling lonely, compared to directly reporting, because there may be barriers which inhibit direct responses, such as feeling guilty or unable to report loneliness due to circumstances. Moreover, evidence suggests that factorial surveys (such as the UCLA-LS) in comparison with direct single-item questions, can combat social desirability bias (Walzenbach, 2019). Additionally, the UCLA-LS enquires about both companionship and inclusion which incorporates different aspects of loneliness, which may not otherwise be considered when directly asked if feeling lonely.

Furthermore, the multinomial regression for loneliness expectations demonstrated that different factors were associated with the 'neither' and 'disagree' responses in relation to 'agree'. In the fully-adjusted model, close relationships with family and friends were significantly positively associated with higher odds of 'disagreeing' that loneliness would increase with age. However, none of the

social network variables were significantly associated with the 'neither' response in relation to 'agree'. One interpretation of 'neither' is that individuals were unsure whether loneliness will change with age, as respondents may recognise that loneliness is circumstantial and factors, other than age, have influence. Aspects related to social networks, such as close relationships with family, therefore, may be more likely to prompt a positive or negative response; those with close family networks may not expect loneliness, because they recognise that children and grandchildren will be there as they age. Furthermore, close family networks may be unlikely to elicit a neutral response because respondents can be more certain about most of their family network. To clarify, with age, individuals may expect spousal or friend bereavements, however the loss of children and grandchildren is less likely. Thus individuals with larger family networks may not expect to grow lonelier as they can rely on children and grandchildren within the family network.

Overall, it is important to pursue the comparison between caregivers' and non-caregivers' loneliness despite the contrasting results for RQ1a, in which only one of the three loneliness measures identified a significant difference. Firstly, further investigating what is contributing towards the difference in UCLA-LS loneliness for caregivers and non-caregivers, through both shared characteristics (RQ1b) and caregiving-specific factors (RQ2a), may aid understanding as to why no difference was observed for the direct measure. Thus, this contributes towards the body of evidence which discusses differences between direct and indirect approaches to measuring loneliness. Additionally, this research is meaningful because, as identified in the research gaps (section 3.5.1) from the literature review, there is a dearth of studies researching caregiving and loneliness longitudinally. RQ1a suggests that the positive association between caregiving and loneliness attenuates after controlling for health and wellbeing, however, one limitation of cross-sectional evidence is that data are collected at a single time-point, therefore the direction of relationships is unknown. Further analysis (RQ3) identifies whether the connection between caregiving and loneliness is long-term, and how this is impacted by changes to the caregiving role, and caregivers' health.

### 5.2.3 RQ1b: are the determinants of loneliness different for caregivers and noncaregivers?

To assess whether determinants of loneliness differed for caregivers and non-caregivers, binary logistic regression analyses were conducted on caregivers and non-caregivers separately. The analyses were carried out using UCLA-LS loneliness as the DV. This decision was made on the basis that UCLA-LS loneliness significantly differed for caregivers and non-caregivers (prior to controlling for health and wellbeing), whereas direct loneliness did not. Therefore, it would be logical to assume that if the determinants of loneliness were different, this likely would be more evident for UCLA-LS where significant differences were already apparent. Furthermore, as identified in RQ1a the UCLA-LS is arguably a more effective measure of loneliness because the responses are not inhibited by a preconception of whether respondents feel able to say they are lonely. Additionally,

the three items cover different aspects of loneliness and, when collated, give a multifaceted indication of loneliness, in comparison to a singular direct question. Table 13 outlines the regression models applied for RQ1b.

Table 13: Regres	ssion models for RQ1b: are the determinants of loneliness different for caregi	vers and non-
cai	regivers?	
Model number	Variables included in the model	

1	Age, sex, cohabitation, LLTI, loneliness expectations, public transport use.
2	Model 1 + socioeconomic status
3	Model 2 + social networks
4	Model 3 + health and wellbeing

Source: author's own

The initial model (model one) included determinants of loneliness: variables identified through reviewing literature and theory. The variables in these regressions differ from the regressions in RQ1a because, factors such as age, and sex were pivotal within both caregiving and loneliness literature, and were therefore discussed in depth within the literature review (section 3.2), whereas the use of public transport was identified as a determinant of loneliness (van den Berg et al., 2016; Kemperman et al., 2019), but was not a common theme raised within caregiving literature. Therefore the inclusion of this variable in this regression analysis can identify whether public transport use was linked to loneliness among caregivers, as well as the general population. The aim of RQ1a was to establish whether an overall association was present between caregiving and loneliness, when controlling for known covariates; whereas RQ1b seeks to determine whether the determinants of loneliness differ for caregivers and non-caregivers. For this reason, the inclusion of additional variables reported as determinants of loneliness were key in determining the relevance of these factors for both caregivers and non-caregivers. In model two SES variables were added as covariates alongside the determinants of loneliness. This second model differs from the second model in RQ1a because the inclusion of sociodemographic variables was not necessary, because age, sex and cohabitation were all previously used as determinants of loneliness. Model three determined the impact of social networks on the relationships between the aforementioned determinants and loneliness, for both caregivers and non-caregivers. Any attenuation to previously significant relationships would suggest that social networks better account for loneliness than SES or the proposed determinants. Finally, in model four, additional health and wellbeing variables were included as covariates. As health status was identified within the literature as a determinant of loneliness, presence of LLTIs was included from model one, but adding other health and wellbeing variables in the fully-adjusted model (model four), tested whether specific health (e.g. SET pathways) and wellbeing measures were more strongly associated with loneliness for either caregivers or non-caregivers.

#### 5.2.3.1 Determinants of loneliness among caregivers

In the fully-adjusted model (Figure 7), which included the determinants of loneliness and all covariates, being female was positively associated with loneliness among caregivers, whereas

'disagreeing' with loneliness expectations, being in paid employment, close relationships with friends, and higher QoL were indicative of lower odds of caregiver loneliness (Appendix F). Both the Nagelkerke  $r^2$  and Hosmer-Lemeshow (p=.517) demonstrate that the fully-adjusted model was the best fit and accounted for the greatest variance in loneliness in comparison to previous models; accounting for 57% of variance in caregiver loneliness. The regression analysis was conducted on 52% of the caregiving sample, as 48% (N=459) were excluded from the analyses due to missing data.

Odds of female caregivers being lonely were three times higher than for male caregivers (OR=3.06, 95CI=1.49-6.30); however, this was not significant in previous models. The reason for this could be that, among caregivers, owing to the slight female majority in the sample, the relationship between sex and loneliness among caregivers is weaker. However, with the introduction of a stronger association (e.g. between health and/or wellbeing and loneliness), and, in turn, the sex differences that arise between the health and wellbeing variables and loneliness, this may explain the significant result.



Figure 7: Odds ratio plot, determinants of caregivers' loneliness Source: author's analysis of ELSA wave 8

<sup>•</sup>Disagreeing' with loneliness expectations (not expecting to get lonelier with age) was negatively associated with loneliness (OR=0.18, 95CI=0.06-0.54). In accordance with SET, evidence suggested that internalised beliefs can become self-fulfilling prophecies (Levy, 2009; Meisner and Levy, 2016), indicating that expecting to be lonely would result in higher likelihood of actually being lonely, which aligns with the findings in this thesis. Research also suggests that use of transport (e.g. buses) is related to lower loneliness (Kemperman *et al.*, 2019), giving access to social situations outside the local neighbourhood, and providing opportunities for interactions during the journey (van den Berg *et al.*, 2016). However, the results from this thesis suggest otherwise: the odds of loneliness among caregivers who never used public transport were not significantly different from those who used it frequently. Nevertheless, other factors may be influencing this relationship, such as SES. Caregiving roles are reported to negatively impact the financial situation of caregivers (Chiao *et al.*, 2015), thus

public transport may be less accessible. Alternatively, those in higher wealth categories may be more likely to own a car, so measuring public transport use may be largely redundant.

Caregivers in paid employment alongside their caregiving role demonstrated lower odds of loneliness than caregivers without additional employment (OR=0.40, 95CI=0.18-0.89), which is likely due to increased opportunities to socialise outside of the caregiving role. Furthermore, a greater number of close relationships with friends was linked to lower loneliness odds (OR=0.84, 95CI=0.72-0.99). Conversely, close relationships with family members were not significantly associated with loneliness among caregivers. This suggests that caregivers in employment, with a larger friendship network outside of the family and caregiving role may experience greater levels of social support and are less likely to experience loneliness. Finally, QoL (OR=0.85, 95CI=0.81-0.90) was negatively associated with loneliness among caregivers, but none of the health measures were significant, which would suggest that wellbeing has a stronger association with loneliness among caregivers than health.

#### 5.2.3.2 Determinants of loneliness among non-caregivers

The same models were applied within the non-caregiving sample; however, different determinants were associated with loneliness, highlighting inconsistency between determinants of loneliness for caregivers and non-caregivers. Similarly, around 46% (N=2274) of the non-caregiving sample were excluded from the regression analyses due to missing data. The Hosmer-Lemeshow statistic was non-significant (p>.05) across all models, indicating a good fit to the data. Moreover, the Nagelkerke  $r^2$  increases with each model, with the fully-adjusted model accounting for the greatest variance in loneliness (49%).

For non-caregivers (Appendix G), in the fully-adjusted model (Figure 8), cohabitation with a partner was negatively associated with loneliness (OR=0.21, 95CI=0.14-0.33). Furthermore, LLTIs were also negatively associated with loneliness (OR=0.71, 95CI=0.51-0.99). Neither of these were determinants of caregiver loneliness. As discussed in RQ1a, those with LLTIs may receive help from family or friends and therefore may experience social contact due to this. In contrast, caregivers with LLTIs may receive less help for their own illness as, owing to the caregiving role, they may be perceived as more capable or able to look after themselves, therefore the additional social contact may not occur for caregivers. Amongst non-caregivers, both uncertainty about (OR=0.30, 95CI=0.22-0.41) and 'disagreeing' with (OR=0.21, 95CI=0.14-0.32) loneliness expectations were negatively associated with loneliness, compared to 'agreeing' that respondents would be lonelier as they aged. This suggests that, aligning with the SET hypothesis of self-fulfilling prophecies, expecting loneliness is linked to experiencing loneliness, as all other responses, compared to 'agree', were suggestive of lower odds of loneliness. However, as with the caregiving sample, use of public transport was non-significant in all models, suggesting no association with loneliness. As discussed previously (section 5.2.3.1), this contrasts with the literature; however, additional factors, such as

access to a car, may mediate the effect of public transport use. For example, those with a car may not use public transport often, but they would still be able to travel to social events.



Figure 8: Odds ratio plot, determinants of non-caregivers' loneliness Source: author's analysis of ELSA wave 8

Among the health and wellbeing variables, uncertainty about future health, in comparison with 'agreeing' health would get worse with age (OR=1.74, 95CI=1.17-2.58), and depressive symptomatology, compared to no depressive symptoms (OR=1.91, 95CI=1.35-2.70), were indicative of higher odds of loneliness for non-caregivers. In contrast, frequent alcohol consumption ( $\geq$ 3 days per week), compared to less frequent consumption (OR=0.61, 95CI=0.44-0.83), QoL (OR=0.88, 95CI=0.86-0.91) and life satisfaction (OR=0.93, 95CI=0.90-0.95) were negatively associated with loneliness among non-caregivers. Although alcohol consumption may be a coping mechanism for loneliness, and has been shown to be used among adolescents to avoid feelings of loneliness (McKay *et al.*, 2017), the relationship between alcohol consumption and loneliness is largely inconclusive

(Segrin *et al.*, 2018). It is plausible that under these circumstances the context of the alcohol consumption may be accounting for this significant association. To clarify, if the alcohol consumption is occurring in a pub or similar establishment, it is likely that socialising will occur (Dunbar *et al.*, 2017) and therefore the individual may not feel lonely. Alternatively, consuming alcohol frequently alone at home is likely to involve limited social interactions, and therefore the experience of loneliness may be more likely. However, it was not possible to investigate from this data where alcohol was consumed, but it is logical to consider that if an individual is consuming alcohol frequently outside the home, that it may be occurring with friends or family. The health and wellbeing determinants vary substantially from the findings for caregivers, only QoL was significantly negatively associated with caregiver loneliness. This difference between caregivers and non-caregivers indicate that factors related to caregiving (e.g. intensity) may be playing an important role in determining loneliness.

# 5.2.3.3 Summary: are the determinants of loneliness different for caregivers and non-caregivers?

The variables associated with loneliness differed substantially in the caregiver and non-caregiver analyses, indicating that the determinants of loneliness were different for each group. The only corresponding determinants of loneliness for both caregivers and non-caregivers were 'disagreeing' with loneliness expectations and lower QoL (Table 14).

Among caregivers, the odds of female caregivers being lonely were three times higher than male caregivers, however sex was not a significant determinant of loneliness among non-caregivers. Similarly, paid employment was negatively associated with loneliness among caregivers, but no significant loneliness differences were found between employed and non-employed non-caregivers. Finally, a larger close network of friends was indicative of lower odds of loneliness among caregivers, but not non-caregivers. In contrast, cohabitation, LLTI, health expectations, alcohol consumption, depressive symptomatology, and life satisfaction were all determinants of loneliness among non-caregivers.

The considerable differences between caregivers and non-caregivers highlighted that the determinants of loneliness may differ, based on caregiving circumstances. Fewer of the included determinants and covariates were significant among caregivers than the non-caregivers, which could suggest that aspects of caregiving (e.g. relationship to care recipient) may be affecting the loneliness of caregivers. This was further investigated in section 5.3, where caregiving-specific variables were investigated in relation to loneliness, health, and wellbeing.

Variables	Caregivers	Non-caregivers
Age: 60-69	NS	NS
≥70	NS	NS
Sex	**	NS
Cohabitation	NS	***
LLTI	NS	*
Loneliness expectations: neither agree nor disagree	NS	***
Loneliness expectations: disagree	**	***
Public transport use: 1-3 times per month	NS	NS
1-3 times per week	NS	NS
Every day or nearly every day	NS	NS
Paid employment	*	NS
Wealth: quartile 2	NS	NS
Quartile 3	NS	NS
Quartile 4	NS	NS
Household: dual-person	NS	NS
Household: large (≥3 people)	NS	NS
Close family relationships	NS	NS
Close friendships	*	NS
Physiological health	NS	NS
Health expectations: neither agree nor disagree	NS	**
Health expectations: disagree	NS	NS
Alcohol consumption	NS	**
CES-D: depressive symptomatology	NS	***
CASP19	***	***
SWLS	NS	***

 Table 14: Summary of significant results for RQ1b: are the determinants of loneliness different for

 caregivers and non-caregivers?

Notes: NS non-significant, \*p<.05, \*\* p<.01, \*\*\* p<.001, source: author's analysis of ELSA wave 8

# 5.2.4 RQ1c: how does the interaction between caregiving and loneliness affect the relationships between loneliness and health and wellbeing?

To investigate how caregiving interacts with the relationship between loneliness and health, or loneliness and wellbeing, a caregiving-loneliness interaction term was created. The interaction was included because, as identified in RQ1a, caregiving was significantly negatively associated with UCLA-LS loneliness in the first three models prior to the inclusion of health and wellbeing, and in RQ1b fewer of the health and wellbeing variables were associated with loneliness among caregivers compared to non-caregivers. Therefore, it was hypothesised that the relationship between loneliness and health or loneliness and wellbeing should differ between caregivers and non-caregivers.

The interaction was included in model two to determine the combined effect of caregiving and loneliness, in contrast with the conditional main effects of each variable (model one). Using this interaction, binary logistic and multinomial logistic regression analyses were conducted with the health DVs, and linear regressions with the wellbeing DVs. The analyses were carried out using UCLA-LS loneliness and, similar to previous research questions, sociodemographic characteristics, SES, and health or wellbeing variables were included as blocks of covariates (Table 15).

<mark>ti</mark>	he relationships between loneliness and health and wellbeing?
Model number	Variables included in the model
1	Caregiving status and UCLA-LS
2	Model 1 + caregiving-loneliness interaction
3	Model 2 + sociodemographic characteristics and socioeconomic status
4	Model 3 + health or wellbeing*

wellbeing covariates were included and in regressions with wellbeing DVs, health covariates were included. Multicollinearity was tested among all the variables, all tolerance and VIF statistics were within accepted parameters, but strong correlations were observed between the two wellbeing variables. Source: author's own

The results for RQ1c are separated into seven specific health and wellbeing measures: physiological health (chronic CV conditions), behavioural health (alcohol consumption), health expectations, depressive symptomatology, LLTIs, QoL, and life satisfaction. The main objective of this research question was to further investigate which areas of health or wellbeing loneliness and caregiving were more strongly associated, in order to develop targeted policy recommendations. Self-rated general health was not included as a health measure due to its more generic and subjective nature, as such obtaining information about specific health conditions would not be possible, see section 4.3.3 for discussion on this variable.

#### 5.2.4.1 Caregiving, loneliness, and physiological health

This regression included 89% of the sample, with 11% (N=662) excluded due to missing data. Caregiving was significantly negatively associated with poor physiological health, measured by the presence of one or more chronic CV conditions (Appendix H). In contrast, loneliness was found to increase the odds of poor physiological health in models 1-3, but this positive association attenuated with the inclusion of wellbeing covariates. In the fully-adjusted model (Figure 9), odds of caregivers experiencing poor physiological health were lower than non-caregivers (OR=0.71, 95CI=0.59-0.85). This could suggest a positive effect of caregiving on physiological health, or, more likely, it may highlight the capacity to care. For example, individuals with severe CV conditions are unlikely to be healthy enough to provide substantial care to another individual with separate health needs (healthy caregiver effect). The follow-up analyses within this thesis (section 6.2) aids the interpretation of this, determining time-order of caregiving and loneliness, and considering changes to health to identify if a healthy caregiver effect may be occurring. The caregiving-loneliness interaction was not significant in any model, which indicates that the relationship between loneliness and physiological health did not differ, depending on caregiving status.

Chapter 5



Figure 9: Odds ratio plot, factors associated physiological health Source: author's analysis of ELSA wave 8

Similarly to the aforementioned healthy caregiver effect, a comparable association may be observed with employment (healthy worker effect). ORs suggest that those in employment were less likely to have chronic CV conditions, compared to those not in employment (OR=0.80, 95CI=0.69-0.93). The healthy worker effect would infer that those with poor physiological health were less able to work due to their condition, and therefore, those in employment were more physically able. Moreover, wealth was also negatively associated with physiological health. Respondents in the highest quartile were less likely to report CV conditions compared to the lowest quartile (OR=0.60, 95CI=0.50-0.71). This could be due to better access to healthcare and wealthier individuals sustaining a higher-quality of living, for example greater access to healthy nutrition and preventative medicine (Semyonov *et al.*, 2013) which would in turn reduce CVD risk.

The Nagelkerke  $r^2$  indicates that the fully-adjusted model accounted for only 15% of variance in physiological health, implying that a considerable number of factors, influential to physiological health, were not included in the model. Moreover, although the Hosmer-Lemeshow (p=.279) indicates good model fit, because the interaction term was non-significant, and loneliness was not significantly associated with physiological health after controlling for wellbeing, the interaction between caregiving and loneliness had little or no impact on physiological health.

#### 5.2.4.2 Caregiving, loneliness and behavioural health

In model one, including only caregiving and loneliness, loneliness was negatively associated with frequent alcohol consumption (OR=0.63, 95CI=0.54-0.74) as was caregiving (OR=0.79, 95CI=0.67-

0.93) (Appendix I). The negative association between caregiving and alcohol consumption remained significant across all models after controlling for the caregiving-loneliness interaction and other covariates. However, the negative association between loneliness and alcohol consumption attenuated with the inclusion of wellbeing in the fully-adjusted model and was no longer statistically significant. This regression included 88% of the sample, with 12% (N=698) excluded due to missing data.

In the fully-adjusted model, caregivers were less likely to frequently consume alcohol compared to non-caregivers (OR=0.78, 95CI=0.65-0.94). Furthermore, the caregiving-loneliness interaction was positively associated with alcohol consumption; a combination of caregiving and loneliness was suggestive of a greater likelihood of frequent alcohol consumption. As shown in Figure 10, lonely caregivers, on average, consume alcohol more frequently than lonely non-caregivers. While, for lonely and non-lonely caregivers, mean alcohol consumption score was similar, for non-caregivers the difference between lonely and non-lonely individuals was much greater. Non-lonely, non-caregivers may have more time (no caregiving commitments) for social activities, and therefore may be more likely to spend time with friends (e.g. in a pub). This could explain why these individuals consumed alcohol more frequently, and were not lonely. Alternatively, the lonely non-caregivers may spend less time going out with friends, meaning that alcohol is consumed less frequently.



Figure 10: Interaction plot, alcohol consumption by loneliness and caregiving status Source: author's analysis of ELSA wave 8

Sex was negatively associated with alcohol consumption in the fully-adjusted model (Figure 11), whereas wealth and QoL were positively associated. Females had lower odds of frequent alcohol consumption, compared to males (OR=0.51, 95CI=0.45-0.58), which is consistent with published literature, whereby males consistently were found to drink more frequently and in higher quantities compared to females (Kirchner *et al.*, 2007; Hughes *et al.*, 2016). Alternatively, wealth was positively associated with frequent alcohol consumption: odds of frequent consumption for respondents in the highest quartile were three times higher than those in the poorest quartile (OR=3.00, 95CI=2.48-3.62). This result was largely unsurprising, as wealthier individuals may be able to afford to buy alcohol more frequently, whereas poorer individuals may be more conscious about unnecessary spending and opt to consume alcohol less frequently. Moreover, Moos *et al.* 



(2010) indicate a link between increased financial resources and higher likelihood of high-risk drinking behaviour among older adults.

Figure 11: Odds ratio plot, factors associated with alcohol consumption Source: author's analysis of ELSA wave 8

Overall, the caregiving-loneliness interaction suggests that the relationship between loneliness and alcohol consumption differs depending on caregiving status, but overall, the relationship is largely unexplained by these models. Although the Hosmer-Lemeshow statistic (p=.298) indicated a good model fit, the Nagelkerke  $r^2$  statistic highlighted that the fully-adjusted model accounted for only 12% of variance in frequency of alcohol consumption, thus there are many additional factors influencing alcohol consumption among the respondents.

#### 5.2.4.3 Caregiving, loneliness and health expectations

As outlined in section 4.3.3.1, the psychological health pathway involves future health expectations. The analysis was conducted in relation to the 'agree' response (expecting health to get worse with age), whereby comparison was drawn with the likelihood of a respondent 'disagreeing' (not expecting health to get worse with age) or responding with 'neither' (uncertainty about future health). In this regression included 11% (N=662) of the sample were excluded from the analysis due to missing data. Nagelkerke r<sup>2</sup> indicated that the fully-adjusted model explained only around 9% of variance in health expectations, which suggests that there are influential factors which have not been included. For example, current health is likely to have a substantial impact on health expectations. Controlling for current health status may have contributed to a greater explanation of variance, but also higher multicollinearity. Moreover, the focus of the research question was to determine the

impact of caregiving and loneliness on health expectations, not necessarily focus on the impact of health on health expectations. The following sections discuss the results from the multinomial logistic regression models (Appendix J).

#### Odds of disagreeing with health expectations, compared to agree

Health expectations were not significantly different for caregivers and non-caregivers. Furthermore, the caregiving-loneliness interaction was not statistically significant in any model. This suggests that caregiving, and the relationship between caregiving and loneliness, was not related to health expectations. Alternatively, lonely individuals were less likely to 'disagree', and more likely 'agree' than non-lonely individuals in models 1-3, although the significant association attenuated in the fully-adjusted model with the inclusion of wellbeing variables. This suggests that the association between wellbeing and health expectations may account for differences in loneliness. As loneliness is known, from previous literature, to be negatively associated with health outcomes (Mushtaq *et al.*, 2014; Friedler *et al.*, 2015; Ong *et al.*, 2016), it is likely that the lonely respondents may already be experiencing some negative health consequences, which may be contributing towards their health expectations.



Figure 12: Odds ratio plot, factors associated with disagreeing with health expectations Source: author's analysis of ELSA wave 8

In the fully-adjusted model (Figure 12) individuals with higher QoL scores demonstrated a more positive outlook on future health (OR=1.08, 95CI=1.07-1.10). Alternatively, the odds of respondents in the wealthiest quartile, reporting that they did not expect to be unhealthier when they were older, were lower than that of those in the poorest quartile (OR=0.63, 95CI=0.48-0.81). This suggests that wealthier individuals have fewer positive expectations for health, in comparison with poorer
individuals. In the physiological health analysis (section 5.2.4.1), wealthier individuals had lower odds of reporting CV conditions, and furthermore, in the LLTI analysis (section 5.2.4.5), wealthier individuals were less likely to report having LLTIs. This suggests that overall higher wealth quartiles had better health status, than lower quartiles. For this reason, wealthier individuals may be expected to have a more positive outlook on future health, but the results of this current analysis suggest otherwise. Findings among a sample of post-menopausal women also reported that lower SES was linked to more positive attitudes towards ageing (Kavirajan *et al.*, 2011), although this was not only limited to health expectations. Overall, literature is limited and inconclusive on the association between SES on health expectations. Amongst older adults with low SES, a link was found between ageing expectations and physical and mental health (Dogra *et al.*, 2015), whereas in other studies, no significant associations were reported between SES and expectations for ageing (Sarkisian *et al.*, 2002) or subjective quality adjusted life years (Rappange *et al.*, 2016).

# Odds of neither agreeing nor disagreeing with health expectations compared to agree

As with the previous section (disagree *versus* agree), caregivers' and non-caregivers' health expectations were not significantly different, nor was the caregiving-loneliness interaction significant in any model. This further clarifies that caregiving and the relationship between caregiving and loneliness were not related to health expectations within this sample. Similar to the 'disagree' response, loneliness was significantly negatively associated with uncertainty about health expectations across models 1-3, lonely respondents were more likely to 'agree' that they expect health to get worse as they age. However, this significance, again, attenuated with the inclusion of wellbeing in the fully-adjusted model (Figure 13).

Older (aged  $\geq$ 70) respondents were more likely to be uncertain about their future health, than younger (aged 50-59) respondents (OR=1.49, 95CI=1.17-1.89), but no significant difference was observed between ages 50-59 and 60-69. WHO (2018) suggests that although not exclusively a linear association, ageing is positively associated with increased risk of disease and decline in physical and mental capacity, therefore it could be, amongst the oldest group, that many already experience several health conditions. Therefore, older individuals may have adapted to the presence of health conditions, and if their health has remained relatively stable, they may not consider it likely to get worse. However, the age difference was not significant for the 'disagree' response, thus suggesting that the older individuals are more uncertain about health, than expecting it to decline.

In contrast to the 'disagree' results, significant differences were only observed between quartile two and the poorest wealth quartile (OR=1.49, 95CI=1.17-1.89). However, similar to the previous section (disagree *versus* agree), QoL was also positively associated with likelihood of reporting 'neither' over 'agree' (OR=1.07, 95CI=1.05-1.08). This supports the interpretation that individuals with lower QoL were more likely to have a more negative perception on future health.



Figure 13: Odds ratio plot, factors associated with neither agreeing nor disagreeing with health expectations Source: author's analysis of ELSA wave 8

## 5.2.4.4 Caregiving, loneliness, and depressive symptomatology

Despite the Nagelkerke  $r^2$  indicating that the fully-adjusted model accounted for the greatest amount of variance explained in depressive symptomatology ( $r^2$ =0.38), compared to other health conditions (e.g. physiological health  $r^2$ =0.15), the Hosmer-Lemeshow suggests that the fully-adjusted model was not a good fit (p=.002). Model three accounted for less variance (23%), but was a better fit (p=.358), so although adding wellbeing explained greater variance in depressive symptomatology, it did not improve model fit. There is evidence to suggest that increased numbers of predictors negatively influences the power of Hosmer-Lemeshow tests (Blanchard, 2016), but this was not an issue in previous analyses, containing the same number of variables, so the reduction in model fit is unlikely due to the additional predictors. Due to missing data, 12% (N=692) of the sample were excluded from the analysis.

Caregiving, and loneliness, were positively associated with depressive symptoms, and the caregiving-loneliness interaction was also significant indicating that the impact of loneliness on depressive symptoms differed for caregivers and non-caregivers. However, the conditional main effect of loneliness on depressive symptomatology was smaller in the fully-adjusted model after controlling for wellbeing (Appendix K). Odds of lonely individuals reporting depressive symptomatology were 1.8 times higher (OR=1.77, 95CI=1.41-2.23) than non-lonely individuals in the fully-adjusted model, compared to odds 5.4 times higher in model 3 (OR=5.43, 95CI=4.47-6.58).

This suggests that wellbeing may account for some of the relationship between loneliness and depressive symptomatology. In the fully-adjusted model, odds of caregivers reporting depressive symptoms were almost twice that of non-caregivers (OR=1.79, 95CI=1.34-2.31). The caregiving-loneliness interaction was negatively associated with depressive symptomatology, as depicted below (Figure 14), the combined impact of caregiving and loneliness was indicative of greater depressive symptoms.



Figure 14: Interaction plot, depressive symptomatology by loneliness and caregiving status Source: author's analysis of ELSA wave 8

All covariates in the fully-adjusted model, except cohabitation, were associated with depressive symptomatology (Figure 15). For instance, the oldest respondent group (aged  $\geq$ 70) were less likely to report depressive symptoms compared to the youngest group (aged 50-59) (OR=0.59, 95CI=0.45-0.77), however no significant difference was observed between ages 50-59 and 60-69. Evidence suggested that prevalence of depressive symptoms among older adults, although still substantial, are lower than both younger and middle-aged adults (Haigh *et al.*, 2018), therefore this finding aligns with previous literature.



Figure 15: Odds ratio plot, factors associated with depressive symptomatology Source: author's analysis of ELSA wave 8

Furthermore, both QoL and life satisfaction were negatively associated with depressive symptomatology (OR=0.89, 95CI=0.88=0.90 and OR=0.97, 95CI=0.96-0.99 respectively). Depression has previously been linked to negative self-evaluation (McCarthy and Morina, 2020). Furthermore, lower life satisfaction could be considered a negative evaluation, which may suggest why those with lower satisfaction and QoL were more likely to report depressive symptoms.

## 5.2.4.5 Caregiving, loneliness and limiting long-term illnesses

Initially (models 1-3), loneliness was positively associated with LLTIs: odds of lonely individuals reporting LLTIs were significantly higher than non-lonely. However, after controlling for wellbeing, this association was reversed (Appendix L). In the fully-adjusted model (Figure 16), which included 89% of the overall sample, odds of lonely individuals reporting LLTIs were lower than that of non-lonely, which is consistent with previous results (sections 5.2.2.2, and 5.2.3.2) whereby presence of LLTIs were linked to lower odds of loneliness. The fully-adjusted model accounted for around 28% of LLTI variance, which was an improvement from model three (17%).





Figure 16: Odds ratio plot, factors associated with LLTI Source: author's analysis of ELSA wave 8

Lonely individuals demonstrated lower odds of reporting LLTIs compared with non-lonely (OR=0.77, 95CI=0.63-0.94). However, the significant interaction indicated that the combined effect of caregiving and loneliness impacts the relationship between loneliness and LLTIs. Non-lonely caregivers reported LLTIs more commonly than non-lonely non-caregivers (Figure 17), whereas lonely caregivers reported fewer LLTIs compared to lonely non-caregivers.



*Figure 17: Interaction plot, LLTI by loneliness and caregiving status Source: author's analysis of ELSA wave 8* 

Cohabitation was linked to lower odds of reporting LLTIs (OR=0.72, 95CI=0.61-0.84). However, in all previous health measures, cohabitation was not significant. Individuals cohabiting with a spouse may be likely to receive assistance from their partner as part of their daily life and ongoing

relationship, which may reduce the perception of the long-term illness. As such, the respondent may find it less limiting because they receive help, and are therefore less likely to report LLTIs than those living alone.

As discussed previously (section 5.2.4.1), healthy worker effects may also explain the negative association between employment and LLTI. Odds of employed respondents reporting LLTIs were considerably lower than non-employed (OR=0.42, 95CI=0.35-0.50). However, those with LLTIs may be less able to work due to illness, rather than a causal relationship existing between employment and health. Furthermore, wealthier individuals may have greater access to healthcare, and typically report being healthier than poorer individuals. Thus, it was expected that odds of reporting LLTIs would also decrease between the wealth quartiles. As predicted, the odds of the wealthiest quartile reporting LLTIs were substantially lower than quartile one (OR=0.59, 95CI=0.48-0.72).

Finally, QoL, but not life satisfaction, was significantly negatively associated with LLTIs (OR=0.90, 95CI=0.89-0.92). When considering the differences between the wellbeing measures, SWLS asks respondents whether they would change their life or if they have the things they want (Diener *et al.*, 1985), whereas the CASP19 (QoL) scale contains questions about feeling full of energy and their health preventing them doing things (Sim *et al.*, 2011). Therefore, it is reasonable that CASP19 was more closely related to health measures, as there are health-related items within the questionnaire.

## 5.2.4.6 Caregiving, loneliness and quality of life

QoL was quantified in this analysis using CASP19: higher scores were indicative of greater QoL. As a continuous variable, a linear regression was required, instead of logistic regressions used for categorical health variables. Due to missing data, 10% (N=571) of the sample were excluded from the analysis. The fully-adjusted model (Table 16) was significant (F(17, 5348)=285.51, p<.001) with an R<sup>2</sup> of 0.48. Full regression table can be seen in Appendix M.

Variables	B (std. error)
Constant	41.26 (0.38)***
Caregiving	-0.82 (0.28)**
Loneliness	-7.51 (0.28)***
Caregiving-loneliness	-3.08 (0.63)***
Age	
60-69 years	1.47 (0.25)***
≥70 years	0.04 (0.15)
Sex: female	1.49 (0.19)***
Cohabitation	-0.13 (0.22)
In paid employment	0.43 (0.24)
Wealth	
Quartile 2	2.15 (0.27)***
Quartile 3	2.60 (0.27)***
Quartile 4	3.67 (0.28)***
LLTI	-3.85 (0.22)***
CES-D: depressive symptomatology	-6.22 (0.27)***
Physiological health: chronic CV condition(s)	-1.06 (0.20)***
Health expectations	
Neither agrees nor disagrees	2.38 (0.25)***
Disagrees	2.97 (0.28)***
Alcohol consumption >3 days per week	0.86 (0.20)***

Table 16: Fully-adjusted linear regression model examining how the interaction between caregiving and loneliness affects the relationships between caregiving, loneliness and quality of life

Notes: Linear regression, DV – CASP19; \*p<.05, \*\*p<.01, \*\*\*p<.001; source: author's

analysis of ELSA wave 8

Loneliness, caregiving, and the caregiving-loneliness interaction were negatively associated with QoL. Lonely individuals reported QoL scores, on average, 7.5 points lower than non-lonely, and QoL score was 0.8 lower for caregivers compared to non-caregivers. The combined effect of caregiving and loneliness, as identified through the significant caregiving-loneliness interaction (Figure 18), was suggestive of lower QoL. Although individually, both loneliness and caregiving were suggestive of poorer QoL, the difference in QoL scores was much greater between lonely caregivers and non-caregivers, than the non-lonely caregivers and non-caregivers.



*Figure 18: Interaction plot, quality of life by loneliness and caregiving status Source: author's analysis of ELSA wave 8* 

All included variables, except cohabitation and employment, were significantly associated with QoL in the fully-adjusted model. Moreover, some noteworthy results arose surrounding the health

covariates. As expected, LLTI, depressive symptomatology, and poor physiological health were negatively associated with QoL. Additionally, respondents who expected their health to worsen with age ('agree') were more likely to have lower QoL scores than those who 'disagreed' or were uncertain. Conversely, alcohol consumption was positively related to QoL, as respondents who frequently consumed alcohol reported higher QoL, compared to less frequent drinkers. As discussed previously (section 5.2.3.2), the context in which alcohol consumption takes place may be impacting the result. If respondents are consuming alcohol in pubs with friends, this is likely to enhance their QoL score, particularly as some of the CASP19 items refer to the company of others and choosing to do things they want to do (Sim *et al.*, 2011).

## 5.2.4.7 Caregiving, loneliness and life satisfaction

Similarly to CASP19, higher SWLS scores indicate greater life satisfaction. In model one, both caregiving and loneliness were significantly negatively associated with life satisfaction. However, in subsequent models, with the introduction of the caregiving-loneliness interaction and other control variables, the association between caregiving and life satisfaction attenuated (Appendix N). The volume of missing data was much less in this regression, only 8% of the sample were excluded from the analysis. The fully-adjusted model (Table 17) was significant overall (F(17, 5464)=175.39, p<.001), with an R<sup>2</sup> of 0.35.

Table 17: Fully-adjusted linear regression mode	el examining how the interaction of the interaction of the second s	ction between caregiving an
<u>Ionetiness affects the relationship.</u> Variables	<u>B (std_error)</u>	ness and life satisfaction
Constant	23.25 (0.30)***	
Caregiving	-0.12 (0.22)	
Loneliness	-4.91 (0.21)***	
Caregiving-loneliness	-1.38 (0.48)**	
Age	· · · · ·	
60-69 years	1.03 (0.19)***	
≥70 years	0.70 (0.11)***	
Sex: female	0.92 (0.15)***	
Cohabitation	2.16 (0.17)***	
In paid employment	-0.18 (0.19)	
Wealth		
Quartile 2	0.91 (0.21)***	
Quartile 3	1.25 (0.21)***	
Quartile 4	1.75 (0.22)***	
LLTI	-1.35 (0.17)***	
CES-D: depressive symptomatology	-3.54 (0.21)***	
Physiological health: chronic CV condition(s)	-0.23 (0.15)	
Health expectations		
Neither agrees nor disagrees	1.11 (0.19)***	
Disagrees	1.45 (0.22)***	
Alcohol consumption ≥3days per week	0.46 (0.16)**	
	* .001 .1.1	

Notes: Linear regression, DV – SWLS; \*p<.05, \*\*p<.01, \*\*\*p<.001; source: author's analysis of ELSA wave 8

Nevertheless, loneliness was negatively associated with life satisfaction: SWLS score was, on average, 4.9 points lower for lonely individuals, compared to non-lonely. Although the conditional main effect of caregiving was not significant once controlled for the interaction effect, sociodemographic characteristics, SES, and health, the caregiving-loneliness interaction was



statistically significant. Caregiving and loneliness combined was indicative of even lower life satisfaction than that of loneliness among non-caregivers (Figure 19).

Figure 19: Interaction plot, life satisfaction by loneliness and caregiving status Source: author's analysis of ELSA wave 8

In the fully-adjusted model, age, sex, cohabitation, wealth, health expectations, and alcohol consumption were positively associated with life satisfaction, whereas LLTI and depressive symptomatology were negatively associated with life satisfaction. Moreover, the consistent positive association between alcohol consumption and wellbeing (both QoL and life satisfaction), alongside the negative association with loneliness (lower odds of loneliness within the full sample, section 5.2.2.2 and among non-caregivers, section 5.2.3.2) supports the interpretation that the social aspect of alcohol consumption may be influencing these results. A more appropriate measure to determine the specific health and wellbeing effects of alcohol, may be volume of alcohol consumed, which would likely show more direct effects, in contrast to frequency, which appeared to be mediated by socialising occurring during the alcohol consumption. Furthermore, it is not evident whether respondents are consuming single or multiple drinks on each occurrence, which would also impact the relationship with health. Future research might consider this alternative, if using other data sources, but this information was largely unavailable for the sample within this thesis.

In contrast, physiological health was not significantly associated with life satisfaction. Although it could be reasoned that chronic CV conditions would reduce life satisfaction, it could also be argued that LLTIs would be more likely to influence life satisfaction than physiological health. For instance, respondents with a chronic CV condition may not be dissatisfied with their life, if the condition is managed sufficiently, as they may not finding it as restrictive as they do an illness they have specifically identified as limiting (LLTI).

# 5.2.4.8 Summary: how does the interaction between caregiving and loneliness affect the relationships between loneliness and health and wellbeing?

The same models were applied for each of the health and wellbeing variables, with the exception of including wellbeing controls in health regressions and vice versa, which enabled comparison to be drawn. As highlighted in each section, model fit and variance ranged across all of the measures. This highlights key differences and indicates that different aspects of health are not all influenced by the same factors. Tables 18 and 19 summarise the results across all of the health and wellbeing measures.

Overall, it was evident that loneliness and caregiving have differing relationships with different aspects of health and wellbeing. The results suggest lonely individuals were less likely to report LLTIs, but more likely to experience symptoms of depression and reduced wellbeing. In addition, caregivers were less likely to report poor physiological health and frequently consume alcohol, but more likely to exhibit depressive symptoms and lower QoL, compared to non-caregivers. Furthermore, the caregiving-loneliness interaction was significantly associated with alcohol consumption, LLTI, depressive symptomatology, and both wellbeing measures, suggesting the effect of loneliness on these outcomes differed between caregivers and non-caregivers.

animina and

<u>1 ubie 18. Summury</u>	oj significani resul	is joi KQIC. no	w abes the thierd	iciton between	caregiving and	l
loneli	ness affect the relat	tionships betwee	e <mark>n loneliness and</mark>	<mark>! health?</mark>		
XX 11	Physiological	Alcohol	Health expectations			
Variables	health	consumption	Neither agree nor disagree	Disagree	CES-D	LLTI
Caregiving	***	**	NS	NS	***	NS
Loneliness	NS	NS	NS	NS	***	*
Caregiving- loneliness	NS	**	NS	NS	***	***
Age: 60-69	***	NS	NS	NS	NS	***
$\geq 70$	***	NS	**	NS	***	***
Sex	***	***	NS	NS	***	NS
Cohabitation	NS	NS	NS	NS	NS	***
Paid employment	**	NS	NS	NS	***	***
Wealth: quartile 2	*	*	*	NS	NS	**
Quartile 3	**	***	NS	**	*	***
Quartile 4	***	***	NS	***	**	***
CASP19	***	**	***	***	***	***
SWLS	NS	NS	NS	NS	**	NS

Notes: NS non-significant, \*p<.05, \*\* p<.01, \*\*\* p<.001, source: author's analysis of ELSA wave 8

Table 19: Summary of significant results for RQ1c: how does the interaction betwee	<mark>n caregiving and</mark>
loneliness affect the relationships between loneliness and wellbeing?	

Variables	Quality of life	Life satisfaction
Caregiving	**	NS
Loneliness	***	***
Caregiving-loneliness	***	**
Age: 60-69	***	***
270	NS	***
Sex	***	***
Cohabitation	NS	***
Paid employment	NS	NS
Wealth: quartile 2	***	***
Quartile 3	***	***
Quartile 4	***	***
LLTI	***	***
CES-D: depressive symptomatology	***	***
Physiological health	***	NS
Health expectations: neither agree nor disagree	***	***
Health expectations: disagree	***	***
Alcohol consumption	***	**

Notes: NS non-significant, \*p<.05, \*\* p<.01, \*\*\* p<.001, source: author's analysis of ELSA wave 8

#### 5.2.5 Summary: how does loneliness differ for caregivers and non-caregivers?

Research question one addressed three key questions: (RQ1a) is caregiving associated with loneliness? (RQ1b) are the determinants of loneliness different for caregivers and non-caregivers? and (RQ1c) how does the interaction between caregiving and loneliness affect the relationships between loneliness, health, and wellbeing? Descriptive analyses indicated that there was little difference in the percentage of lonely caregivers, compared to lonely non-caregivers. Using the direct measure 11% of both caregivers and non-caregivers were lonely, whereas for the UCLA-LS, 20% of both groups were lonely. Similarly, 42% of non-caregivers and 41% caregivers 'agreed' that they expected to be lonelier as they age. Chi-squared analyses suggests that caregiving was independent from all of the loneliness measures, whereby no significant bivariate associations were found.

However, the results suggest that once controlled for theoretically important variables, caregiving was positively associated with loneliness, but only for the UCLA-LS, not the direct question or loneliness expectations, and this positive association attenuated with the inclusion of health and wellbeing controls. Additionally, the determinants of loneliness were found to significantly differ between caregivers and non-caregivers. For example, factors reported in literature, such as cohabitation or health-related predictors of loneliness were not significant in determining caregiver loneliness, which could indicate that factors associated with the caregiving role are more influential towards loneliness. Aspects of caregiving, including caregiver characteristics, the caregiving relationship and the caregiving role were analysed further in RQ2 (section 5.3).

Finally, the caregiving-loneliness interaction was significantly associated with three of the five health variables (depressive symptomatology, LLTI and alcohol consumption) and both wellbeing variables (QoL and life satisfaction). For the majority, the caregiving-loneliness interaction increased the likelihood of a negative outcome, i.e. more likely to frequently consume alcohol, report depressive symptoms, lower QoL, or lower life satisfaction. However, the caregiving-loneliness interaction

indicated that lonely caregivers were less likely to report LLTIs, compared to lonely non-caregivers. As discussed, this could be a healthy caregiver effect, but non-lonely caregivers were more likely to report LLTIs, than non-lonely non-caregivers. Therefore this is unlikely to be the full explanation. Further analysis into the time-order of caregiving, loneliness, and health (chapter six) sought to clarify this interpretation.

## 5.3 RQ2: how does caregiving affect loneliness, health, and wellbeing?

## 5.3.1 Descriptive statistics for the caregiving sample

This second research question analyses the caregiving sample (N=962). Subsequent to RQ1, in which differences were observed between caregivers and non-caregivers for the determinants of loneliness, and health and wellbeing outcomes, this section addresses differences within the caregiving group. Analysing caregiving-specific variables, such as factors relating to the caregiving relationship and role, may explain some of the differences observed between caregivers and non-caregivers, for example, identifying caregiving-specific determinants of loneliness. Descriptive statistics relating to characteristics, the caregiving relationship, role, and burden are presented below (Table 20). Further descriptive details, such as social networks, loneliness, health and wellbeing have been previously reported in Table 9 (section 5.2.1). The results from the bivariate analyses, including Chi-squared tests for categorical variables and Spearman's *r* for continuous variables are included in Appendix O.

Moreover, collinearity checks were all within accepted parameters: VIF typically ranged from 1-5, suggestive of some moderate correlations. However, VIF for spousal care was 6.9, suggesting stronger correlations, although this was still below the accepted parameters of 10. Correlations between all variables was checked using a correlation matrix. Correlations existed between the care recipient variables, which can be largely explained by care provision to multiple recipients (e.g. caregivers providing care to parents and others), whereby a stronger correlation would be expected. Additionally, as mentioned in section 5.2.1, correlations that existed between wellbeing variables were likely due to links between QoL and life satisfaction. However, evidence confirms that CASP19 and SWLS are distinct measures that measure separate aspects of wellbeing.

			Caregivers
			N (%)
		50-59	352 (36.6)
	Age (years)	60-69	311 (32.3)
		≥70	299 (31.1)
	Sov	Male	381 (39.6)
	Sex	Female	582 (60.4)
tics		Married/civil partnership	590 (61.4)
sris		Single, never married	54 (5.7)
icte	Marital status	Remarried	121 (12.6)
lara		Divorced/legally separated	146 (15.1)
r ch		Widowed	51 (5.3)
IVEI	Cohabitation	No	189 (19.6)
681	(spouse/partner)	Yes	774 (80.4)
Car	Daid amployment	No	630 (65.6)
<b>U</b>	Paid employment	Yes	330 (34.4)
		Quartile 1 (poorest)	257 (26.9)
Wealth		Quartile 2	237 (24.8)
		Quartile 3	219 (22.9)
		Quartile 4 (wealthiest)	242 (25.3)
		Spouse or partner	376 (39.1)
ng hip	Core reginigent	Adult-child	112 (11.7)
		Parent or parent-in-law	272 (28.2)
		Other relative, friend or neighbour	234 (24.4)
rela	Co residential core	No	469 (48.7)
	Co-residential care	Yes	493 (51.3)
50		Low (≤19hpw)	566 (59.7)
/ing	Caregiving intensity	High (≥20hpw)	200 (21.1)
givole		Round-the-clock (168hpw)	181 (19.1)
lare	Normhan of community in to	Single recipient	828 (86.1)
0	Number of care recipients	Multiple recipient	134 (13.9)
<u>в</u> –	Feels adequately appreciated	Agree	882 (92.2)
rivin aisa	for their caregiving role	Disagree	75 (7.8)
lareg appr	Feels satisfied with what	Agree	931 (97.1)
0	caregiving	Disagree	28 (2.9)

Table 20: Descriptive statistics for caregivers in ELSA wave eight

Notes: total percentage for care recipients surpasses 100% due to caregivers providing care to multiple recipients, source: author's analysis of ELSA wave 8

## 5.3.1.1 Caregiver characteristics

The mean age of caregivers was 65.2, with the largest percentage of caregivers aged 50-59 (37%). As implied by previous literature and other caregiving data (Dahlberg *et al.*, 2007; van Campen *et al.*, 2013; Lacey *et al.*, 2018), the caregiving sample was predominantly female (60%). Milligan and Morbey (2016) highlight that this is not the case among older caregivers, whereby male caregivers are more prevalent than female among caregivers aged  $\geq$ 65. However, as highlighted above, the age distribution of caregivers within ELSA data indicates a larger quantity of younger caregivers (aged 50-59), therefore a female majority was expected.

As previously identified (section 5.2.1.1), few caregivers identified as single, never married or widowed, this posed difficulties when running the analyses and contradictions arose with the cohabitation variable, whereby grouping into 'currently married' or 'not currently married' was not appropriate. Thus, the alternative measure relevant to relationship status was cohabitation (whether

residing with a spouse or partner), and among caregivers, 80% were cohabiting. The remaining caregiver characteristics measure SES through employment status and wealth. Employment alongside the caregiving role was less common (34%), and distribution across wealth quartiles was relatively balanced.

## 5.3.1.2 Caregiving relationship

Spousal caregivers were most common (39%), followed by care to parents/parents-in-law (28%). This is consistent with UK data, whereby 34% of caregivers aged  $\geq$ 50 provide spousal care, and 32% to parents (Colombo *et al.*, 2011). Spousal caregivers tend to be older (Broese van Groenou *et al.*, 2013), thus among this sample of caregivers (aged  $\geq$ 50), a high number of spousal caregivers would be expected. Care recipient variables were coded as dummy variables to account for caregivers of multiple recipients, therefore Chi-squared tests were conducted for each of the care recipient groups.

In this ELSA sample, over half of the caregivers lived with their care recipient (51%). Due to the high percentage of spousal caregivers, a large percentage of co-resident caregivers was expected, as it is typical for married couples, and spousal caregivers, to cohabit (Broese van Groenou *et al.*, 2013). Further crosstabulation analysis confirmed that most spousal caregivers (97%) lived with their care recipient, and similarly, most caregivers to adult-children were co-resident (72%). For all other recipients, it was more common that the caregiver lived separately: 17% of caregivers to parents/parents-in-law and 10% of caregivers to others reported living with the recipient.

## 5.3.1.3 Caregiving role

Although most commonly (60%), caregivers provided low-intensity ( $\leq$ 19hpw) care, there were a considerable number of caregivers providing high-intensity ( $\geq$ 20hpw) care (21%), and almost as many (19%) were providing round-the-clock care (168hpw).

The majority of caregivers (86%) provided care to only one recipient. Further analysis indicated that for caregivers to multiple recipients, the most common recipient combinations were multiple parents/parents-in-law (32%), and multiple others (27%). This suggests that the relationship with the care recipient influences whether, and to whom, additional care is provided. As such, providing care to an adult-child is likely due to a long-term disability. Therefore care needs may be greater and more complex than, for example, for parents, who may just require help with shopping. Complex care needs for a single recipient would limit caregivers' capacity to provide care to wider network members, as recipients with higher care needs are more dependent on their caregiver (Cottagiri and Sykes, 2019). This is reflected in the data, as only 16% of caregivers to multiple recipients provided care to an adult-child, compared to 51% providing care to parents/parents-in-law.

## 5.3.1.4 Caregiving burden

Most of the caregiving sample did not appear to experience heightened levels of burden when considering role satisfaction and perceived appreciation as proxy measures. Over 92% of caregivers felt adequately appreciated, and over 97% were satisfied with what they had gained from caregiving. Satisfaction and appreciation variables were included in preliminary variations of the regression models. However, upon evaluation, the small percentages of caregivers not feeling adequately appreciated (N=75, 8%) or lacking satisfaction (N=28, 3%) contributed towards large CIs and the author considered that the findings from these variables could be invalidated by these small numbers. Thus, these variables were excluded from the regression analyses.

#### 5.3.1.5 Social networks and living arrangements

Most caregivers lived in dual-person households (64%), with 27% living in households of three or more, and less than 10% living alone. Moreover, caregivers were inclined to have more close family relationships than friendships. The mean number of close family members was 4.9, compared to 3.7 for close friends.

#### 5.3.1.6 Loneliness, health, and wellbeing

Loneliness (UCLA-LS) was present among 20% of caregivers (N=191). However, caregivers' expectations for loneliness differed considerably: most (41%) 'agreed' that they expected to get lonelier with age, whereas 29% 'disagreed', and did not expect to get lonelier with age. A further 29% 'neither agreed nor disagreed', which was interpreted as uncertainty about future loneliness.

Health was determined through several measures, including SET health pathways (physiological, behavioural, and psychological) and depressive symptomatology (mental health). In contrast to RQ1, LLTI was not included in a health measure. Firstly, LLTI was not significantly associated with caregiver loneliness (section 5.2.3.1), and the caregiving-loneliness interaction results indicated that loneliness among caregivers was suggestive of lower odds of LLTI, compared to loneliness among non-caregivers (section 5.2.4.5). Arguably, LLTI is somewhat generic, in that it can refer to any illnesses that limit the individual, whereas SET pathways and depressive symptomatology refer to very specific areas of health, thus fostering a more unambiguous discussion on caregivers' health. Across most of these measures, a larger percentage of caregivers demonstrated good health. For the SET physiological and behavioural pathways, 40% reported having chronic CV conditions, and 29% reported consuming alcohol frequently. Alternatively, for health expectations, a substantial majority 'agreed' that they expected their health to get worse as they age (70%). Additionally, a larger portion of caregivers were uncertain about future health (17%) than in disagreement (13%). Finally, CES-D scores suggest that 24% of the caregiving sample showed depressive symptomatology, overall implying that caregivers were in relatively good health.

Higher CASP19 scores are indicative of greater QoL, and the highest possible score is 57. Average (mean) CASP19 score for caregivers was 39.9, indicating a relatively high QoL. The final wellbeing measure was SWLS. Scoring ranged from 5-35, whereby higher scores indicated greater satisfaction. Mean SWLS score among the caregiving sample was 24.9, suggesting high life satisfaction.

## 5.3.2 RQ2a: how does caregiving affect loneliness?

To examine the impact of caregiving on loneliness, caregiver characteristics, the caregiving relationship and the caregiving role were examined in relation to loneliness. A binary logistic regression, with the following models (Table 21) was conducted to determine whether caregiving-specific variables were associated with loneliness.

Model number	Variables included in the model
1	Characteristics: age, sex, cohabitation, employment, and wealth Relationship: care recipient, and co-residential care Role: caregiving intensity, and number of recipients
2	Model 1 + social networks
3	Model 2 + health and wellbeing

Source: author's own

Model one included the caregiving-specific variables outlined in section 4.3.1.1. This determined which aspects of caregiving were associated with loneliness. In model two, the addition of social networks highlighted the impact of close relationships with family members and friends on the relationship between caregiving and loneliness. Finally, model three included health and wellbeing controls, any changes observed to previously significant relationships after controlling for health and wellbeing would indicate that health and/or wellbeing were more strongly associated with loneliness, after controlling for caregiving-specific variables.

## 5.3.2.1 How does caregiving affect loneliness on the UCLA Loneliness Scale?

Of the 962 caregivers in the sample, 464 (48%) were excluded from this regression due to incomplete data. In the fully-adjusted model (Figure 20), being female, providing care to an adult-child, and frequent consumption of alcohol were positively associated with loneliness, whereas paid employment, larger numbers of close friendships and higher QoL scores were negatively associated.



Figure 20: Odds ratio plot, caregiving-specific factors associated with caregivers' loneliness Source: author's analysis of ELSA wave 8

Caregiver characteristics: in the first two models, age, employment and wealth were significantly negatively associated with loneliness (Appendix P). However, only employment remained significant after controlling for health and wellbeing. Employment was linked to 46% reduced odds of loneliness (OR=0.44, 95CI=0.20-0.96). As with the analysis of the full-sample, the opportunity to socialise in a work environment likely explains the reduction in loneliness. Moreover, in the fully-adjusted model, the odds of female caregivers experiencing loneliness were over three times higher than male caregivers (OR=3.13, 95CI=1.52-6.45). This is consistent with the results from RQ1b, where sex was a significant determinant of caregiver loneliness. Similarly, this result was only significant after controlling for health and wellbeing. It was considered that additional sex differences

that arose with the inclusion of health and wellbeing likely enhanced the relationship between sex and loneliness.

Caregiving relationship: co-residential care was positively associated with loneliness in model one, whereby living with the care recipient was indicative of a higher likelihood of loneliness. However, this relationship attenuated in model two with the inclusion of social networks. Furthermore, significant loneliness differences were observed between caregivers providing care to an adult-child, and spousal caregivers in models two and three, after controlling for social networks and health and wellbeing. In the fully-adjusted model, odds of loneliness for caregivers to an adult-child were over 2.5 times higher than for spousal caregivers (OR=2.55, 95CI=1.01-6.43). However, no significant differences were observed for caregivers to parents, or others, in comparison to spousal caregivers. One interpretation of these findings was that providing care to an adult-child is a long-term commitment, and frequently higher-intensity than other caregiving roles (Barnhart *et al.*, 2020). Care provision is most likely due to disability throughout their childhood and adulthood, which would limit the caregivers' availability to seek employment or socialise outside of the household. The caregivers to adult-children may have experienced lower numbers of friendships, which could also contribute towards loneliness.

Caregiving role: neither caregiving intensity nor the number of care recipients were significantly associated with loneliness.

Moreover, frequency of alcohol consumption was positively associated with loneliness (OR=2.37, 95CI=1.15-4.90). It is plausible to consider that frequent alcohol consumption may be a coping mechanism for loneliness and/or caregiving. As such, a US sample of persons aged  $\geq$ 45 found that lonely respondents tended to drink alcohol as a coping mechanism for loneliness. Consuming alcohol when lonely was more common among lonely respondents than non-lonely (Wilson and Moulton, 2010). Moreover, caregivers experiencing social or emotional burden are more likely to exhibit problematic drinking behaviours (Rospenda *et al.*, 2010). In contrast, in RQ1a, frequent alcohol consumption reduced the odds of loneliness within the full sample. However, as discussed, it was assumed that alcohol consumption may occur in social situations, particularly amongst non-caregivers. Therefore, in comparison to caregivers, who are more likely to consume alcohol as a coping mechanism, contrasting effects on loneliness were expected.

Improvement in model fit was observed across all three models, but overall, the fully-adjusted model was the best fit for the data, with a large Hosmer-Lemeshow p-value (p=.883), and accounting for 56% variance in loneliness. This suggests that health and wellbeing variables play a substantial part in explaining loneliness among caregivers. However, the results from RQ1b (section 5.2.3.1) would suggest that wellbeing may be more influential than health, as none of the health variables were previously significantly associated with caregiver loneliness.

## 5.3.2.2 Summary: how does caregiving affect loneliness?

In the fully-adjusted model (Table 22), for caregiver characteristics, sex (female) was positively associated with loneliness and employment negatively associated with loneliness. Additionally, for the caregiving relationship, only providing care to an adult-child showed a significant positive association with loneliness in comparison with spousal care. Moreover, contrary to expectation, none of the caregiving role variables were significantly associated with caregiver loneliness. It is important to consider the larger confidence intervals observed for variables, such as sex, and caring for an adult-child. A common explanation for large confidence intervals is often smaller sample sizes (Foster *et al.*, 2015). However, considerable efforts were made to minimise small cell counts, particularly among variables whereby category cell counts were <100. Therefore, this is likely not the case for sex, nor caring for an adult-child. An alternative explanation for the larger confidence intervals is variability. There are a large number of female caregivers, but the result demonstrates that loneliness varied, in that not all female caregivers were lonely.

Variables	Loneliness
Age: 60-69	NS
≥70	NS
Sex	**
Cohabitation	NS
Paid employment	*
Wealth: quartile 2	NS
Quartile 3	NS
Quartile 4	NS
Care recipient: adult-child	*
Parent or parent-in-law	NS
Other relative, friend or neighbour	NS
Lives with care recipient	NS
Caregiving intensity: high-intensity (≥20hpw)	NS
Round-the-clock care (168hpw)	NS
Multiple care recipients	NS
Close family relationships	NS
Close friendships	**
CES-D: depressive symptomatology	NS
Physiological health	NS
Health expectations: neither agree nor disagree	NS
Health expectations: disagree	NS
Alcohol consumption	*
CASP19	***
SWLS	NS

Table 22 · Summary	of significant	results for	RO2a · how does	careaiving affect	Ioneliness
<i>Lubie 22. Summuly</i>	O significant	results for 1	102u. now uses	curegiving uneci	ionenness:

Notes: NS non-significant, \*p<.05, \*\*p<.01, \*\*\*p<.001, source: author's analysis of ELSA wave 8

## 5.3.3 RQ2b: how does caregiving affect health?

To determine the impact of caregiving-specific variables on health, binary and multinomial logistic regressions were carried out with the SET health pathways (physiological, behavioural, and psychological) as DVs. The models for RQ2b were largely similar to those in RQ2a. However, model three included loneliness as a control in place of health measures (Table 23). These omissions were made because including health controls would influence the outcome when health variables were the DVs. It would be assumed that, even though CES-D measures different aspects of health than SET

health pathways, a relationship may be present because the variables were all health-related. As such, when examining the impact of caregiving on health, controlling for health was considered counterproductive.

Table 23: Regression models for RQ2b: how does caregiving affect health?			
Model number	Variables included in the model		
1	Characteristics: age, sex, cohabitation, employment, and wealth Relationship: care recipient, and co-residential care Role: caregiving intensity, and number of recipients		
2	Model 1 + social networks		
3	Model 2 + loneliness and wellbeing		

Source: author's own

In this research question, only the three SET health pathways are examined as outcomes. As health status was considered to be determinant of loneliness, the impact of LLTIs on loneliness was analysed, but LLTI was not significantly associated with caregiver loneliness (section 5.2.3.1). However, it was considered that this might be due to the more generic nature of the variable, in that LLTI can represent any condition the respondent perceives to be limiting. Thus, it was decided that the health measures in this caregiver-only analysis should look more specifically at distinct areas of health as opposed to a broad range of health outcomes. For example, physiological health in this study focuses on CV conditions only. Furthermore, the association between caregiving and mental health is robust and well-researched (Schulz *et al.*, 1997), and connections between caregiving and mental health have been consistently reported over several decades. Therefore, by restricting the analyses to the SET pathways, a more thorough discussion can occur around the impact of caregiving on physiological, behavioural and health expectations. This can potentially foster new understanding, clearer implications, and detailed areas for support for caregivers.

## 5.3.3.1 How does caregiving affect physiological health?

Some caregiver characteristics were significantly associated with physiological health, but factors relating to the relationship or role were not (Appendix Q). Caregiving characteristics negatively associated with poor physiological health were sex and employment (Figure 21). Moreover, being female compared to male (OR=0.43, 95CI=0.28-0.66), and employed compared to non-employed (OR=0.52, 95CI=0.32-0.87) were indicative of lower odds of poor physiological health. Whereas age was positively associated with poor physiological health; odds of caregivers aged 60-69 (OR=3.37, 95CI=1.92-5.93) or  $\geq$ 70 (OR=3.37, 95CI=1.79-6.38) experiencing physiological health conditions were over three times greater than age 50-59. The results are consistent with previous research, whereby female caregivers were more likely than males to experience serious medical illnesses and report poorer health (Navaie-Waliser *et al.*, 2002). This also aligns with previous findings (section 5.2.4.1) amongst the full sample; both females and employed individuals were less likely, whereas older ages were more likely, to report poor physiological health. This implies that the impact of sociodemographic characteristics and SES on physiological health does differ for caregivers and non-caregivers. Formerly, caregivers showed significantly lower odds of poor

physiological health compared to non-caregivers. From this, it may have been anticipated that significant associations between other caregiving-specific variables (e.g. relationship or role) may have explained this relationship. Yet none of the factors related to the caregiving relationship or role were significantly associated with the physiological health of caregivers. This therefore strengthens the argument for a healthy caregiver effect, in that none of the factors related to caregiving appeared to be protective of health, but caregivers' physiological health was generally better.

The fully-adjusted model explained around 32% of variance in physiological health, but the Hosmer-Lemeshow statistic was significant (p=.024), which would suggest poor model fit. In this analysis, 53% of caregivers were included, with 48% (n=457) removed due to missing data. This indicates that caregiving-specific variables, and therefore caregiving overall, when considered alongside social networks, loneliness and wellbeing, have limited influence in the explanation of caregivers' physiological health.



Figure 21: Odds ratio plot, caregiving-specific factors associated with caregivers' physiological health Source: author's analysis of ELSA wave 8

## 5.3.3.2 How does caregiving affect behavioural health?

Poor behavioural health was quantified by frequent alcohol consumption ( $\geq$ 3 days per week). In the fully-adjusted model (Figure 22), which, due to incomplete data, included 52% of the caregiving sample, the factors significantly negatively associated with alcohol consumption were sex, and provision of care to adult-children. Alternatively, wealth was positively associated with frequent alcohol consumption.



Figure 22: Odds ratio plot, caregiver-specific factors associated with caregivers' alcohol consumption Source: author's analysis of ELSA wave 8

Female caregivers demonstrated lower odds of frequent alcohol consumption, compared to male caregivers (OR=0.53, 95CI=0.33-0.84), which is consistent with non-caregiver research. It is well-reported that, globally, males drink more frequently, and in higher quantities, than females (Hughes *et al.*, 2016). Caregivers in all wealth quartiles were much more likely to drink alcohol frequently than those in the lowest quartile. As in previous discussions (section 5.2.4.2), increased wealth is likely reflective of higher discretionary income. Moreover, evidence on alcohol consumption among caregivers suggests that drinkers reported having higher household income (i.e. higher SES)

(Rospenda *et al.*, 2010). Compared to spousal caregivers, caregivers providing care to adult-children displayed lower odds of frequent alcohol consumption (OR=0.31, 95CI=0.12-0.77).

Alternatively, care provision to multiple recipients was positively associated with frequent alcohol consumption in models one and two (Appendix R) but was not significant in the fully-adjusted model. This suggests that caregivers to multiple recipients may consume alcohol as a coping mechanism, for perhaps loneliness experienced due to their caregiving commitments. Hence, once controlled for loneliness differences, this association was no longer significant.

Additionally, caregiver loneliness (OR=2.59, 95CI=1.27-5.30), and QoL (OR=1.09, 95CI=1.04-1.14), were positively associated with frequency of alcohol consumption. This suggests loneliness and wellbeing were associated with alcohol consumption, after accounting for caregiving-specific variables and social networks. However, the inclusion of loneliness and QoL variables in the fullyadjusted model attenuated the previously significant relationship between number of care recipients and alcohol consumption. Thus, associations between loneliness, QoL and alcohol consumption may have reduced the strength of relationships between caregiving-specific variables and alcohol consumption. This would explain the reduction in model fit observed between models two and three (Hosmer-Lemeshow: p=.839 and p=.022 respectively). Moreover, the greater explanation of variance in the fully-adjusted model (Nagelkerke  $r^2=0.24$ ) may suggest that the impact of loneliness and wellbeing on alcohol was greater than the impact of caregiving.

It could be considered contradictory that higher QoL would be indicative of more frequent alcohol consumption among caregivers. Previously it was interpreted that lonely individuals may seek to cope with loneliness by consuming alcohol, (e.g. by going out for a drink with friends), which increased socialisation and subsequently contributed towards higher QoL. However, this may not be possible for some caregivers, i.e. those providing care to an adult-child, as the complexity and duration of care is likely to be higher, leaving limited time for social activities. This could explain why the only significant difference in alcohol consumption was observed for caregivers to adult-children, as all other caregiving groups may have more free time to socialise. This might also elucidate why caregiving intensity was not significant, for example, the link between care recipient and intensity may account for any differences between intensity and alcohol consumption.

## 5.3.3.3 How does caregiving affect health expectations?

A multinomial logistic regression was used to compare caregivers' health expectations (Appendix S). The analytic sample represented 56% of the full caregiving sample, due to 44% (N=424) with incomplete data. The reference group ('agree') included individuals who expected their health to deteriorate with age. In comparison, 'disagree' indicated that caregivers did not expect their health to get worse and 'neither' was interpreted that the caregiver was unsure. As indicated in the descriptive statistics, the majority of caregivers expected health decline (70%) compared to 13% who did not, and 17% who were uncertain. Overall the fully-adjusted model did not explain a substantial

amount of variance in health expectations (Nagelkerke  $r^2=0.23$ ), but the likelihood ratio (Chi-squared=106.6, p<.001) suggests that the fully-adjusted model fit the data significantly better than an empty model.

## Odds of caregivers disagreeing with health expectations, compared to agree

In the fully-adjusted model (Figure 23), only provision of round-the-clock care was significant. The odds of respondents providing round-the-clock care 'disagreeing' were over twice that of respondents providing low-intensity care (OR=2.60, 95CI=1.07-6.31). However, caregivers with the capacity to provide round-the-clock care, are unlikely to be currently experiencing any major health concerns, which is likely to impact their perception of future health. Additionally, as their recipient is dependent upon them 24/7, they may not consider, or want to consider, the possibility of their own health deteriorating, as this may compromise their caregiving situation. Moreover, in contrast to findings from RQ1c, which indicated increased wealth was negatively associated with the 'disagree' response in the whole sample (section 5.2.4.3), there were no significant wealth differences in caregivers' health expectations. Therefore, it appears that the impact of wealth is less relevant to health expectations among caregivers, implying that when caregivers evaluate their future, they place less emphasis on SES.



## Figure 23: Odds ratio plot, caregiving-specific factors associated with caregivers disagreeing with health expectations

Source: author's analysis of ELSA wave 8

# Odds of caregivers neither agreeing nor disagreeing with health expectations compared to agree

Alternatively, employment, and QoL were positively associated, and relationship with the care recipient was negatively associated with the 'neither' response (Figure 24). Employed caregivers were less certain about future health, demonstrating higher odds of responding with 'neither', than non-employed caregivers (OR=2.02, 95CI=1.08-3.78). One interpretation could be that non-employed caregivers may be unable to work, due to the increased needs of their recipient. This may be negatively impacting caregivers' health, as high-intensity care has been shown to do (Lacey *et al.*, 2018). Therefore, non-employed caregivers may already experience health limitations, and thus be more likely to expect their health to deteriorate, compared to those who have the availability and capacity to work alongside their caregiving role. Moreover, research into health stereotypes in the general population (US-based) found no significant differences between employment groups (e.g. employed *versus* retired), suggesting that employment status, particularly retirement, had little influence on health stereotypes or expectations. This would advocate that factors outside of employment (i.e. caregiving-specific) might contribute towards caregivers' perceptions on future health.



Figure 24: Odds ratio plot, caregiving-specific factors associated with caregivers neither agreeing nor disagreeing with health expectations Source: author's analysis of ELSA wave 8

As discussed throughout the thesis, providing care to adult-children has, most likely, been a longterm commitment throughout the recipient's life, and may involve complex care needs due to disabilities. Caregivers to adult-children demonstrated significantly lower odds of responding with 'neither', compared to spousal caregivers (OR=0.10, 95CI=0.02-0.49). This suggests that this subgroup of caregivers was more likely to expect that their health would get worse with age. This could be for a number of reasons. For example, caregivers may already be experiencing health concerns. Lee *et al.* (2017b) found that caregivers of children with disabilities experienced greater odds of several chronic health conditions, compared to caregivers of children without disability. Alternatively, caregivers may recognise that over the course of providing care their health has deteriorated and they expect this to continue in line with their own ageing process.

Finally, higher QoL was positively associated with the 'neither' response, (OR=1.05, 95CI=1.00-1.10). Similarly to the analysis with the full sample (section 5.2.4.3), caregivers with greater wellbeing are likely to have a more positive outlook on health, or life in general. However, in contrast with the previous results, caregivers' QoL was only positively associated with the 'neither' response, not 'disagree', although this could still be considered a more positive outlook than 'agreeing'.

Therefore, the lack of **positive** association between QoL and 'disagreeing' may be attributed to the experience of caregiving, increasing uncertainty in caregivers with regard to their health.

## 5.3.3.4 Summary: how does caregiving affect health?

The factors associated with health were not consistent across the different health measures (Table 24). This indicates that differences between caregivers, or within the caregiving relationship and role, impact different aspects of health. Caregiver characteristics appear to have the greatest effect on health overall, as significant associations were observed across all three pathways. The strongest connection was observed for physiological health, whereby age was positively associated with, and sex and employment were negatively associated with poor physiological health. Moreover, providing care to an adult-child was negatively associated with both frequent alcohol consumption and neutral health expectations, whereas residing with the care recipient was not significantly associated with any of the health pathways. It was discussed that the reduced frequency of alcohol consumption and negative health expectations were likely linked to the high care needs of the recipient. For example, more intense and longer durations of care are linked to providing care to a dependent adult-child (Barnhart et al., 2020), which is likely to increase potential burden, alongside limiting availability to socialise, (e.g. going out with friends to consume alcohol). Alternatively, providing care to single, or multiple, recipients did not significantly impact SET health pathways among caregivers. Finally, caregiving intensity was only significantly positively associated with caregivers' health expectations, not behavioural or physiological health. Round-the-clock caregivers were more positive about future health expectations than low-intensity caregivers, which was interpreted as, perhaps, avoidance because they had 24/7 dependents, they may not want to consider the possibility of their own health deteriorating for the sake of the recipient.

The findings from RQ2b highlight that, of the SET health pathways, loneliness was only significantly positively associated with frequent alcohol consumption among caregivers. This corresponds with RQ1c, as the caregiving-loneliness interaction was only significant for behavioural health, not physiological or psychological. This new result (RQ2b) therefore emphasises that with the combined impact of caregiving, loneliness was associated with increased alcohol consumption, which, in turn, could contribute to further health concerns.

	<i>u (</i>	Alashal	Health expectations	
Variables	Physiological health	consumption	Neither agree nor disagree	Disagree
Age: 60-69	***	NS	NS	NS
_≥70	***	NS	NS	NS
Sex	***	**	NS	NS
Cohabitation	NS	NS	NS	NS
Paid employment	*	NS	*	NS
Wealth: quartile 2	NS	*	NS	NS
Quartile 3	NS	*	NS	NS
Quartile 4	NS	***	NS	NS
Care recipient: adult-child	NS	*	**	NS
Parent or parent-in-law	NS	NS	NS	NS
Other relative, friend or neighbour	NS	NS	NS	NS
Lives with care recipient	NS	NS	NS	NS
Caregiving intensity:				
High-intensity (≥20hpw)	NS	NS	NS	NS
Round-the-clock care (168hpw)	NS	NS	NS	*
Multiple care recipients	NS	NS	NS	NS
Close family relationships	NS	NS	NS	NS
Close friendships	NS	NS	NS	NS
Loneliness	NS	**	NS	NS
CASP19	**	***	*	NS
SWLS	NS	NS	NS	NS

Table 24: Summary of significant results for RO2b: how does caregiving affect health?

Notes: NS non-significant, \*p<.05, \*\*p<.01, \*\*\*p<.001, source: author's analysis of ELSA wave 8

## 5.3.4 RQ2c: how does caregiving affect wellbeing?

Linear regressions were carried out to investigate the impact of caregiving-specific variables on wellbeing (QoL and life satisfaction). The models below (Table 25) are comparable with the previous two research questions (RQ2a/b): models one and two remained consistent, but the controls in model three were loneliness and health, instead of wellbeing. This change was made because the wellbeing measures were used as DVs, and therefore were not used as control variables.

Table 25: Regression models for RQ2c: how does caregiving affect wellbeing?				
Model number	Variables included in the model			
1	Characteristics: age, sex, cohabitation, employment, and wealth Relationship: care recipient, and co-residential care Role: caregiving intensity, and number of recipients			
2	Model 1 + social networks			
3	Model 2 + loneliness and health			

Source: author's own

## 5.3.4.1 Caregivers' quality of life

Of the caregiving sample (N=962), 540 provided complete data and were included in this regression analysis; however, 442 (44%) were excluded due to missing data. The fully-adjusted model (Table 26) was significant (F(23, 516)=28.54, p<.001), with an R<sup>2</sup> of 0.56. This indicates that caregiving-specific variables, social networks, health, and loneliness explain 56% of variance in QoL among caregivers.

Variables	B (std. error)		
Constant	36.01 (1.76)***		
Age	· · · · · · · · · · · · · · · · · · ·		
60-69 years	1.50 (0.77)		
≥70 years	0.46 (0.46)		
Sex: female	1.43 (0.65)*		
Cohabitation	0.28 (0.88)		
In paid employment	0.88 (0.75)		
Wealth			
Quartile 2	3.60 (0.91)***		
Quartile 3	4.35 (0.95)***		
Quartile 4	4.53 (0.96)***		
Care recipients			
Adult-child	1.71 (1.07)		
Parent or parent-in-law	-0.01 (1.09)		
Other relative, friend or neighbour	-0.27 (1.20)		
Lives with the care recipient	-0.51 (1.05)		
Caregiving intensity			
High-intensity (≥20hpw)	-0.55 (0.85)		
Round-the-clock care (168hpw)	-0.92 (0.96)		
Multiple care recipients	2.94 (0.94)**		
Number of close relationships (family and child)	0.32 (0.11)**		
Number of close relationships (friends)	0.07 (0.13)		
UCLA-LS Loneliness	-10.01 (0.83)***		
CES-D: depressive symptomatology	-6.14 (0.84)***		
Physiological health: chronic condition(s)	-2.09 (0.68)**		
Health expectations			
Neither agrees nor disagrees	1.63 (0.81)*		
Disagrees	1.44 (0.98)		
Alcohol consumption ≥3 days per week	2.63 (0.72)***		
Note: Linear regression DV CASP10, $*\pi < 05$ $*\pi < 01$ $*\pi < 01$ $*\pi < 01$ course outhor's			

Table 26: Fully-adjusted linear regress	on model examining hov	v caregiving-specific	variables affect
caregivers' quality of life			

Notes: Linear regression, DV – CASP19; \*p<.05, \*\*p<.01, \*\*\*p<.001; source: author's

analysis of ELSA wave 8

For caregiver characteristics, female caregivers reported QoL scores 1.4 points higher than male caregivers. The largest difference was observed between the wealthiest and poorest quartile, whereby wealthier caregivers QoL was 4.5 points greater than poorer caregivers. Alternatively, there were no significant associations between the caregiving relationship variables (care recipient or co-residential care) and QoL in the fully-adjusted model. However, in model one (Appendix T), co-residential care was negatively associated with QoL (B=-3.1, p<.05). The introduction of social network variables in model two attenuated this relationship, which suggests that close relationships were more strongly related to QoL after accounting for the caregiving relationship. Finally, when considering the caregiving role, QoL was 2.9 points higher among caregivers to multiple recipients, compared to a single recipient. However, caregiving intensity was not associated with QoL in any model. Evidence suggests that caregiving can provide feelings of fulfilment or satisfaction (Quinn *et al.*, 2009). Thus, providing care to multiple individuals may provide a heightened sense of gratification than to a single recipient, as they are helping two people, not just one. This increased sense of satisfaction may translate into positive wellbeing.

Caregivers either 'want to', 'have to', or are 'able to' provide care, and these attitudes, beliefs and barriers determine care provision (Broese van Groenou and de Boer, 2016). 'Wanting' to provide

care is linked to feelings of altruism whereas 'having' to provide care may be more closely aligned with feelings of reciprocity and obligation. CASP19 considers aspects such as control and autonomy in relation to QoL. Therefore, it could be considered that individuals entering into caregiving roles because they 'want to', rather than because they 'have to', may experience greater feelings of control, contributing towards higher QoL. However, this may not directly link to the number of care recipients, nor was it accounted for within this analysis (i.e. it is unknown why caregivers in ELSA provide care). Verbakel *et al.* (2017) suggest that individuals with strong care attitudes are more likely to both provide care and provide care at higher intensities. Therefore, it could be deduced that caregivers with more altruistic motivations may be more likely to provide care to multiple recipients, which may be considered higher-intensity than a single recipient and, in turn, they may experience a greater return in fulfilment, increasing their QoL scores.

The negative association between loneliness and QoL was highly significant: QoL scores for lonely caregivers were 10 points lower than for non-lonely caregivers. Depressive symptomatology and poor physiological health were also negatively associated with QoL among caregivers. This aligns with previous literature, in that loneliness is known to reduce QoL among caregivers (Ekwall *et al.*, 2005; Vasileiou *et al.*, 2017). Furthermore, depression has been reported as the most common factor associated with reduced caregiver QoL (Farina *et al.*, 2017). In contrast, frequent alcohol consumption was associated with a 2.6 point increase in QoL score. As identified in section 5.3.2.1, the relationship between increased alcohol consumption and QoL among caregivers could be a coping mechanism. Thus, regular consumption of alcohol to relieve the stressors of caregiving may give the caregiver the illusion of higher QoL.

## 5.3.4.2 Caregivers' life satisfaction

For life satisfaction, the fully-adjusted model was also significant (F(23, 529)=15.88, p<.001), with an R<sup>2</sup> of 0.41, accounting for 41% variance in caregivers' life satisfaction (Table 27). This regression included 58% of the caregiving sample, 42% were excluded due to incomplete data. Significant positive associations were observed for cohabitation and wealth among the caregiver characteristics. However, neither of the caregiving relationship variables were significantly associated with life satisfaction in the fully-adjusted model (Appendix U). However, similarly to QoL, providing coresidential care was negatively associated with life satisfaction in model one (B=-1.88, p<.05), prior to the inclusion of social networks. This suggests that the conditional main effect of the care recipient relationship (e.g. spouse) does not directly impact life satisfaction after accounting for the caregivers' close family and friendship networks. This would imply that regardless of who care is provided to, caregivers gain satisfaction from their friends and family.

Variables	B (std. error)		
Constant	21.45 (1.38)***		
Age	\$		
60-69 years	0.47 (0.60)		
≥70 years	0.50 (0.36)		
Sex: female	0.84 (0.51)		
Cohabitation	3.07 (0.69)***		
In paid employment	0.16 (0.59)		
Wealth			
Quartile 2	1.74 (0.71)*		
Quartile 3	1.66 (0.74)*		
Quartile 4	2.27 (0.76)**		
Care recipients			
Adult-child	-0.33 (0.82)		
Parent or parent-in-law	0.82 (0.85)		
Other relative, friend or neighbour	-0.78 (0.94)		
Lives with the care recipient	-0.92 (0.83)		
Caregiving intensity			
High-intensity (≥20hpw)	0.47 (0.66)		
Round-the-clock care (168hpw)	0.61 (0.75)		
Multiple care recipients	1.76 (0.74)*		
Number of close relationships (family and child)	0.11 (0.08)		
Number of close relationships (friends)	-0.14 (0.10)		
UCLA-LS Loneliness	-5.51 (0.65)***		
CES-D: depressive symptomatology	-3.33 (0.66)***		
Physiological health: chronic condition(s)	-0.60 (0.53)		
Health expectations			
Neither agrees nor disagrees	1.38 (0.63)*		
Disagrees	1.55 (0.76)*		
Alcohol consumption ≥3days per week	0.78 (0.56)		
Notes Linear representation DV SWI Star (05 *** < 01 **** < 001 sevenes outhor's			

 Table 27: Fully-adjusted linear regression model examining how caregiving-specific variables affect

 caregivers' life satisfaction

Notes: Linear regression, DV – SWLS; \*p<.05, \*\*p<.01, \*\*\*p<.001; source: author's

analysis of ELSA wave 8

Furthermore, in the fully-adjusted model, care provision to multiple recipients was linked to 1.8 points higher SWLS scores than to a single recipient. However, again, caregiving intensity was not significantly associated with wellbeing. Research suggests that care to multiple recipients, in the form of sandwich caregiving, is often detrimental to wellbeing (Evans et al., 2019). However, sandwich care reflects provision to older parents and young children, or grandchildren (Falkingham et al., 2020). This form of caregiving is unlikely within this ELSA sample, because caregivers to grandchildren were excluded from the data (section 4.3.1.1). Moreover, due to the age range within the sample, care to a child likely represents that of dependent adult-children. Alternatively, compound caregivers (lifelong caregivers to disabled children who undertake additional caregiving responsibilities for another family member), are more likely to align with the ELSA sample. However, research with compound caregivers found no significant wellbeing differences for caregivers to individual and multiple recipients (Perkins, 2009). The results in this thesis suggest caregiving to multiple recipients is linked to higher life satisfaction, but, within this analytical sample, the most common caregiving combinations were multiple parents/parents-in-law and multiple others, which differs from the care provision of compound caregivers. Therefore, the literature on sandwich and compound caregiving may not be directly comparable to caregivers of

multiple recipients in ELSA. Similar to the QoL findings (section 5.3.4.1), caregiving intensity was also not significantly associated with life satisfaction.

Finally, caregiver loneliness was negatively associated with life satisfaction, SWLS scores were 5.5 points lower than non-lonely caregivers. This is consistent with the findings from RQ1c, where the combined effect of caregiving and loneliness demonstrated a negative impact on life satisfaction (section 5.2.4.7). Moreover, among the health controls, depressive symptomatology was negatively associated with SWLS score (B=-3.3, p<.001), and caregivers who 'disagreed' that their health would get worse with age reported on average scores 1.6 higher on the SWLS than caregivers who 'agreed'. This suggests that negative expectations, and depressive symptoms are linked to lower satisfaction. Within the literature, research highlights a connection between positive views and life satisfaction (Cummins and Nistico, 2002; Wu, 2009). Further evidence suggests that individuals reorganise their priorities within life according to their health status, which has sequential effects on improving satisfaction (Wu *et al.*, 2009). Relating this evidence to caregivers, it could be considered that while providing care, typically due to recipients' poor health, the caregiver may be influenced to evaluate their own health and lifestyle. Thus, a caregiver may become more appreciative of their health status and consequently adopt a more positive outlook (positive health expectations) subsequently improving their life satisfaction.

## 5.3.4.3 Summary: how does caregiving affect wellbeing?

While there was some overlap in the findings for QoL and life satisfaction, there were also several differences (Table 28). Consistencies between the two wellbeing outcomes included: positive associations for wealth, providing care to multiple recipients and health expectations, and negative associations for depressive symptomatology and loneliness.

Other than wealth, the relationships between caregiver characteristics and wellbeing differed between the measures. For example, female caregivers showed higher QoL, but not life satisfaction, and cohabitation was positively associated with caregivers' life satisfaction but not QoL. Furthermore, neither of the caregiving relationship variables (recipient or co-residential care) were significantly associated with wellbeing in the fully-adjusted models. Similarly, caregiving intensity was not associated with either wellbeing measure. However, providing care to multiple recipients was positively associated with both. These discrepancies highlight that QoL, and life satisfaction are separate constructs, representing different aspects of wellbeing.

Table 28: Summary of significant results for RQ2c: how does caregiving affect wellbeing?				
Variables	Quality of life	Life satisfaction		
Age: 60-69	NS	NS		
≥70	NS	NS		
Sex	*	NS		
Cohabitation	NS	***		
Paid employment	NS	NS		
Wealth: quartile 2	***	*		
Quartile 3	***	*		
Quartile 4	***	**		
Care recipient: adult-child	NS	NS		
Parent or parent-in-law	NS	NS		
Other relative, friend or neighbour	NS	NS		
Lives with care recipient	NS	NS		
Caregiving intensity: high-intensity (≥20hpw)	NS	NS		
Round-the-clock care (168hpw)	NS	NS		
Multiple care recipients	**	*		
Close family relationships	**	NS		
Close friendships	NS	NS		
CES-D: depressive symptomatology	***	***		
Physiological health	**	NS		
Health expectations: neither agree nor disagree	*	*		
Health expectations: disagree	NS	*		
Alcohol consumption	***	NS		
Loneliness	***	***		

Notes: NS non-significant, \*p<.05, \*\*p<.01, \*\*\*p<.001, source: author's analysis of ELSA wave 8

#### 5.3.5 Summary: how does caregiving affect loneliness, health, and wellbeing?

Research question two investigated how caregiving-specific variables impacted loneliness, health, and wellbeing among the sample of caregivers. These analyses were important to understand firstly, what factors may be contributing towards increased caregiver loneliness, and identify caregiving-specific determinants of loneliness. The results could also aid the explanation of differences observed between caregivers and non-caregivers in the health and wellbeing outcomes. Caregiver characteristics (age, sex, cohabitation, employment status and wealth), the caregiving relationship (care recipient and co-residential care) and the caregiving role (caregiving intensity and number of recipients) were analysed in relation to loneliness (RQ2a), SET health pathways (RQ2b), and wellbeing (RQ2c). The descriptive analyses (section 5.3.1) identified that caregivers were generally female, younger (aged 50-59), non-employed, and cohabiting. Additionally, the sample predominantly compromised spousal caregivers, and those providing co-residential care, or care at lower intensities ( $\leq$ 19hpw). Finally, there were substantially more caregivers to a single recipient, compared to multiple recipients.

It was hypothesised that due to the discrepancies in the determinants of loneliness between caregivers and non-caregivers observed in RQ1b (section 5.2.3), caregiving-specific variables would play a large role in explaining caregiver loneliness. However, there was little evidence to support this hypothesis, in that the only caregiving-specific variable significantly positively associated with loneliness was care provision to adult-children in comparison to spousal care. Otherwise, significant factors included sex, employment, friendship networks, alcohol consumption and QoL which largely replicates the findings from RQ1b. One discrepancy was the significance of alcohol consumption. Previously not a significant determinant of caregiver loneliness in RQ1b, after controlling for caregiving-specific variables, frequent alcohol consumption was positively associated with

caregivers' loneliness. As discussed throughout the thesis, alcohol consumption is likely to be a coping mechanism among caregivers.

Moreover, the results indicated that caregiver loneliness was positively associated with frequent alcohol consumption, and negatively associated with QoL and life satisfaction, but not associated with physiological health or health expectations. However, simultaneity was present among the majority of the aforementioned relationships (alcohol consumption and QoL). To illustrate, loneliness was positively associated with frequency of alcohol consumption, but also frequent alcohol consumption was positively associated with loneliness, indicating relationships in multiple directions. In contrast, the relationship between life satisfaction and loneliness was not reciprocal. As such, loneliness was indicative of reduced SWLS scores, but life satisfaction was not significantly associated with alcohol consumption and QoL, a level of simultaneity was expected between life satisfaction and loneliness. However, the observed results suggest that reverse causality may be present, in that rather than reduced life satisfaction leading to loneliness, loneliness was influential on SWLS scores. Conversely, further analysis (section 6.2.2) refutes this inference, by highlighting that both wellbeing measures at wave two were negatively associated with loneliness 12-years later (wave eight).

Finally, the results from RQ2 suggest the relationships between loneliness, health, and wellbeing among caregivers are multidimensional and complex. As such, being female was positively associated with loneliness, but negatively associated with frequent alcohol consumption. These results are slightly contradictory, considering that loneliness and frequency of alcohol consumption were positively related to one another (i.e. more frequent alcohol consumption among caregivers was linked to higher odds of loneliness and vice versa). Therefore, potential interactions may be present, for example, between sex and loneliness in relation to alcohol consumption. The complexity of the relationships will be investigated further in the following chapter: the follow-up analyses can clarify aspects such as the direction of some relationships.

## 5.4 Chapter summary

Chapter five presented the results of the cross-sectional analyses. A combination of binary logistic, multinomial logistic, and linear regressions were conducted to examine loneliness, health and wellbeing differences between caregivers and non-caregivers (RQ1) and among the caregiving sample at wave eight (RQ2). The findings from RQ1 suggest that caregiving is positively associated only with loneliness measured by UCLA-LS, not a direct question, but health and wellbeing controls accounted for the loneliness differences between caregivers and non-caregivers in the fully-adjusted model. There is evidence to suggest that the determinants of loneliness vary for caregivers and non-caregivers and that the impact of loneliness on health and wellbeing is different for caregivers across a number of health measures. Further analysis within the caregiving sample highlighted differences that arose within the caregiving group in relation to loneliness, health, and wellbeing. Predominantly,

these differences were impacted by caregiver characteristics, but there was evidence to suggest that aspects of the caregiving relationship and role also played a part. The following chapter (six) presents the follow-up analyses conducted between wave two and eight, and across waves two to eight, comparing longer-term effects on loneliness for both caregivers and non-caregivers, and identifying how changes to the caregiving relationship and role influence loneliness.
# Chapter 6 Twelve year follow-up

### 6.1 Chapter outline

Chapter six reports the follow-up results, addressing the final research question: what are the longerterm effects of caregiving on loneliness? Section 6.2.1 discusses the descriptive analyses of the baseline variables (wave two), followed by the descriptive analyses of the caregiving sample (across all waves). Bivariate analyses for the baseline variables in relation to wave eight loneliness, and for caregiving-specific variables in relation to loneliness occurring after the start of care provision are included in Appendices W-X.

First, wave two caregivers and non-caregivers are compared (section 6.2.2), then caregivers at any wave with non-caregivers (section 6.2.3) to indicate whether caregiving has a longer-term (12-year) effect on loneliness, or whether the association is shorter-term. Finally, short-term, intermittent, and long-term caregivers are contrasted (section 6.2.4) to identify how changes throughout the caregiving experience can impact loneliness over a 12-year period.

# 6.2 RQ3: What are the longer-term effects of caregiving on loneliness?

#### 6.2.1 Descriptive statistics for the longitudinally-weighted sample

#### 6.2.1.1 Characteristics of caregivers and non-caregivers

This section discusses the descriptive statistics (Table 29), for the longitudinally-weighted sample of individuals that provided a response to the caregiving question (N=2954) at baseline (wave two). Moreover, the bivariate correlations with wave eight loneliness and loneliness at any wave are presented. Similarly to the cross-sectional analyses, checks were conducted between the predictors and DVs. For all variables, VIF and tolerance statistics were within accepted parameters (Field, 2013). Some moderate correlations were apparent between the variables, the strongest of which being the relationship between cohabitation and household size (r=-0.62).

Table 29: Descriptive	statistics of the longitudinal sample a	<u>t baseline (ELSA wave tv</u>	vo)
		Non-caregivers	Caregivers
		N (%)	N (%)
Care provision (wave	e two)	2566 (86.6)	389 (13.2)
	50-59	1064 (41.5)	180 (46.3)
Age (years)	60-69	984 (38.3)	139 (35.7)
	≥70	518 (20.2)	70 (18.0)
Sov	Male	1222 (47.6)	115 (29.6)
Sex	Female	1343 (52.4)	274 (70.4)
	Married/civil partnership	1575 (61.4)	267 (68.6)
	Single, never married	115 (4.5)	8 (2.1)
Marital status	Remarried	316 (12.3)	39 (10.0)
	Divorced/legally separated	276 (10.8)	42 (10.8)
	Widowed	283 (11.0)	33 (8.5)
Cohabitation	No	573 (22.7)	73 (19.0)
(spouse/partner)	Yes	1949 (77.3)	312 (81.0)
Det de sur le sur sur t	No	1466 (57.2)	224 (57.7)
Paid employment	Yes	1099 (42.8)	164 (42.3)
	Quartile 1 (poorest)	624 (24.6)	109 (28.3)
XX7 1/1	Quartile 2	644 (25.4)	91 (23.6)
wealth	Quartile 3	636 (25.1)	91 (23.6)
	Quartile 4 (wealthiest)	631 (24.9)	94 (24.4)
	Single person household	488 (19.0)	47 (12.1)
Household size	Dual-person household	1533 (59.7)	250 (64.3)
	Large household (≥3)	545 (21.2)	92 (23.7)
	Not lonely	2123 (83.7)	315 (82.2)
UCLA-LS	Lonely	414 (16.3)	68 (17.8)
T 1'	Agree	951 (37.7)	146 (38.2)
Loneimess	Neither agree nor disagree	665 (26.3)	102 (26.7)
expectations	Disagree	909 (36.0)	134 (35.1)
	No chronic CV conditions	1701 (66.3)	271 (69.8)
Physiological	$\geq 1$ chronic CV condition	864 (33.7)	117 (30.2)
Health expectations	Agree	1721 (68.6)	272 (70.5)
	Neither agree nor disagree	418 (16.7)	67 (17.4)
	Disagree	371 (14.8)	47 (12.2)
Alcohol	<3 days per week	1596 (63.0)	249 (65.0)
consumption	≥3 days per week	937 (37.0)	134 (35.0)
	No depressive symptomatology	2129 (83.6)	315 (81.6)
CES-D	Depressive symptomatology	417 (16.4)	71 (18.4)
W - 111	CASP19, mean (SD)	43.9 (8.1)	42.9 (8.2)
Wellbeing	SWLS, mean (SD)	26.6 (5.9)	25.7 (6.2)

Source: author's analysis of the longitudinal sample at ELSA wave 2

At baseline, 11% of the total sample were caregivers (N=389). Consistent with the literature (Dahlberg *et al.*, 2007; van Campen *et al.*, 2013), caregivers were predominantly female. Male presence is typically higher among older (aged >65) caregivers (Milligan and Morbey, 2016); however, within this thesis, there are fewer male caregivers, this is likely due to the age distribution. As such, the largest portion of caregivers were aged 50-59 (46%). Furthermore, most caregivers were cohabiting (81%), and commonly not in paid employment (58%). Among the non-caregivers (87%) at baseline, the sociodemographic characteristics were largely similar to caregivers. There were more females (52%) than males, although less of a female majority than amongst caregivers. The majority of non-caregivers were also cohabiting (77%), were aged 50-59 (42%) and not in paid employment (57%). A lesser proportion of non-caregivers were in the lowest wealth quintile (N=624, 25%) compared to caregivers (28%), suggesting that caregivers were slightly more socioeconomically disadvantaged. Among both caregivers and non-caregivers at baseline, it was more common to reside

in a dual-person household (64% and 60% respectively). Due to the high prevalence of married and cohabiting individuals, it is likely that the majority of the dual-person households represented couples.

As shown in Table 29, 18% of caregivers and 16% of non-caregivers were lonely at baseline. Furthermore, crosstabulation showed that loneliness altered over the 12-year period (Appendix V). Of the individuals lonely at wave two, 50% remained lonely at wave eight, whereas only 12% of respondents not lonely at wave two, were lonely at wave eight. This suggested that baseline loneliness may be linked to a higher likelihood of loneliness at follow-up. Additionally, almost as many respondents did not expect to become lonelier with age, as those that did. Descriptive analysis showed that, at baseline, 38% of both caregivers and non-caregivers 'agreed' that they expected to become lonelier with age, moreover, comparable percentages of both caregivers (35%) and non-caregivers (36%) 'disagreed'.

Among both caregivers and non-caregivers, the majority of the sample appeared healthy at baseline. Most reported no chronic CV conditions (70% and 66% respectively) and did not consume alcohol frequently ( $\geq$ 3 days per week) (65% *versus* 63%). Additionally, the majority of both groups were without depressive symptomatology: 82% of caregivers and 84% of non-caregivers. However, a large percentage of both caregivers (71%) and non-caregivers (69%) demonstrated a negative perception on future health, whereby most agreed that they expected to experience poorer health as they grew older. Wellbeing at wave two was not substantially different between caregivers and non-caregivers. On average compared to non-caregivers, caregivers reported CASP19 scores 1 point lower, and SWLS scores 1 point lower. Overall, the baseline data suggested that caregiver health and wellbeing was not largely different, arguably only poorer than that of non-caregivers by negligible amounts; nevertheless, caregivers presented with a slightly higher prevalence depressive symptomatology, and marginally lower QoL and life satisfaction scores. Results from the bivariate analyses can be seen in Appendix W.

#### 6.2.1.2 Characteristics of the caregiving sample

Table 30 shows the sample of caregivers who reported providing care at any point between waves two and eight; this represented 42% (N=1445) of the total longitudinally-weighted sample. Among the caregivers in the sample, most provided care in only one wave (N=633, 44%), or in two or more consecutive waves (N=594, 41%), although a minority reported providing care in two or more non-consecutive waves (N=218, 15%).

		N (%)
Care provision	Caregiver (any wave)	1445 (100.0)
Duration of care	Short-term	633 (43.8)
	Intermittent	218 (15.1)
	Long-term	594 (41.1)
	Ever cared for spouse	551 (38.1)
Cara reginigant	Ever cared for adult-child	126 (8.7)
Care recipient	Ever cared for parent/parent-in-law	382 (26.4)
	Ever cared for others	501 (34.7)
Co-residential	Never lived with the care recipient	789 (54.8)
care	Ever lived with the care recipient	651 (45.2)
Caragiving	Ever provided low-intensity care	1075 (74.4)
intonsity	Ever provided high-intensity care	485 (33.6)
	Ever provided round-the-clock care	365 (25.3)
Number of care	Never cared for multiple recipients	924 (63.9)
recipients	Ever cared for multiple recipients	521 (36.1)
	Stable (always agrees)	196 (15.5)
	Stable (always neither)	71 (5.6)
Loneliness expectations	Stable (always disagree)	186 (14.6)
	Negative change	316 (24.9)
	Positive change	237 (18.7)
	Fluctuating	262 (20.7)
Changes to mental health	Stable (never reports depressive symptoms)	909 (67.1)
	Reduction in depressive symptoms	139 (10.3)
	Increase in, or persistent depressive symptoms	187 (13.8)
	Fluctuating	119 (8.8)

Table 30: Frequencies of caregiving-specific variables, and changes to mental health and loneliness expectations for caregivers in the longitudinal sample

Notes: valid % of caregivers, source: author's analysis of ELSA waves 2-8

The most common care recipients were spouses (N=551, 38%), followed by others (N=501, 35%), parents/in-laws (N=382, 26%) and adult-children (N=126, 9%). The total percentages exceed 100% due to respondents providing to multiple recipients, or different recipients over their caregiving role. Less than half of caregivers reported ever living with the care recipient (N=651, 45%); this likely due to the high proportion of caregivers to others, which represents other relatives, neighbours, and friends.

Similarly to the care recipient dummy variables, the total percentage of intensity variables exceeds 100% due to multiple caregiving episodes and changes to intensity. Low-intensity care was the most commonly reported intensity, 1075 (74%) caregivers identified that they had provided care at  $\leq$ 19hpw. However, around one third of caregivers (N=485, 34%) provided high-intensity care ( $\geq$ 20hpw) and one quarter (N=365, 25%) of caregivers provided round-the-clock care (168hpw). Across the waves, caregiving for a single recipient was more prevalent than to multiple recipients; just over one third of caregivers (N=521, 36%) reported ever providing care to multiple recipients.

Changes to loneliness expectations and depressive symptomatology were computed between three time-points (section 4.3.8). More caregivers changed their loneliness expectations (N=815, 64%) compared to those with stable expectations at each time-point (N=453, 36%). A quarter of caregivers reported a negative change in loneliness expectations (N=316, 25%), which indicated that as caregivers aged, they were more likely to expect future loneliness. Furthermore, changes to health were determined using the CES-D variable. Among the caregivers, the highest proportion never reported depressive symptoms (N=909, 67%). Results from the bivariate analyses can be seen in Appendix X.

#### 6.2.2 RQ3a: is caregiving in wave two associated with loneliness at wave eight?

To determine if caregiving was associated with a longer-term effect on loneliness, a 12-year followup was conducted. In binary logistic regressions, caregiving at wave two was the main explanatory variable, with loneliness (UCLA-LS) at wave eight as the DV; all control variables were obtained at baseline (wave two). Two regressions were conducted to compare the results when (i) individuals who were lonely at baseline were excluded; and (ii) controlling for individuals who were lonely at baseline. Table 31 details the models applied in the analyses.

Model number         Variables included in the model           1         Caregiving status (wave two)           2         Model 1 + loneliness*, loneliness expectations, sociodemographic characteristics, socioeconomic status and social networks           2         Model 2 + health and wallhairs	<u> Table 31: Regress</u>	ion models for RQ3a: is caregiving in wave two associated with lonelir	iess at wave e	<mark>eight?</mark>
<ol> <li>Caregiving status (wave two)</li> <li>Model 1 + loneliness*, loneliness expectations, sociodemographic characteristics, socioeconomic status and social networks</li> <li>Model 2 + health and wallhairg</li> </ol>	Model number	Variables included in the model		
<ul> <li>2 Model 1 + loneliness*, loneliness expectations, sociodemographic characteristics, socioeconomic status and social networks</li> <li>2 Model 2 + health and wallhairs</li> </ul>	1	Caregiving status (wave two)		
2. Model 2 health and wellbeing	2	Model 1 + loneliness*, loneliness expectations, sociodemographic characteristics, socioeconomic status and social networks		
5 Model 2 + health and wendering	3	Model 2 + health and wellbeing		

Notes: \*in regression ii only, source: author's own

Model one included caregiving at wave two, determining whether providing care at baseline was associated with loneliness at the 12-year follow-up (wave eight). In model two, control variables were added for loneliness expectations, sociodemographic characteristics, SES, and social networks. Separate regressions were conducted excluding those lonely at baseline (i) and controlling for those lonely at baseline (ii); whereby comparison was drawn between the long-term effect of caregiving on 'new' loneliness, not present at the time of caregiving, and the long-term effect of caregiving on later loneliness, after accounting for those who had already reported loneliness at baseline. Finally, model three included baseline health and wellbeing variables; after controlling for health and wellbeing, any changes to the relationship between caregiving and loneliness, after controlling for caregiving and other factors. All control variables in this regression were measured at baseline and thus treated as time-invariant.

# 6.2.2.1 The association between caregiving and loneliness over 12-years (i) excluding those lonely at baseline

A binary logistic regression was carried out to determine if caregiving was linked to loneliness at a 12-year follow-up. Respondents who identified as lonely at baseline were excluded from the analyses, the remaining sample was based on a CCA, excluding 17% (N=496) of the longitudinally-weighted sample (N=2954). Caregiving in wave two was not significantly associated with loneliness at wave eight in any model (Appendix Y); this would suggest that caregiving was not linked to long-term effects on loneliness over a 12-year period. In the fully-adjusted model (Figure 25), which controlled for loneliness expectations, sociodemographic characteristics, SES, social network, health, and wellbeing, only age, sex, and wellbeing (both QoL and life satisfaction) were linked to long-term loneliness outcomes.



#### Figure 25: Odds ratio plot, wave two factors associated with wave eight loneliness, excluding baseline loneliness Source: author's analysis of ELSA waves 2 and 8

Respondents aged 60-69 (OR=1.74, 95CI=1.21-2.50) and  $\geq$ 70 (OR=2.25, 95CI=1.40-3.60) at baseline had significantly higher odds of reporting loneliness at wave eight than those aged 50-59. The respondents in these age categories at wave two would be aged 72-81 or  $\geq$ 82 when reporting loneliness (wave eight). This contradicts the evidence from the cross-sectional analyses. As such, age was not significantly associated with loneliness among the full sample (section 5.2.2.2), nor was it a significant determinant of loneliness among caregivers or non-caregivers when analysed separately (section 5.2.3). Similarly, after controlling for caregiving variables, social networks, health, and wellbeing (section 5.3.2), age was not linked to loneliness among the cross-sectional caregiving sample. Particularly among caregivers, cross-sectionally age was not significant after controlling for the respondents' current health and wellbeing status. This suggested that previously significant loneliness differences between the age groups were explained by health and wellbeing differences. In the follow-up analyses, age was positively associated with later loneliness, after controlling for baseline health and wellbeing. However, health variables reported at baseline would

not account for changes to health experienced over the 12-year period between baseline and followup (see section 7.5.3 for further discussion of this limitation). Therefore, as older adults are more likely to experience poorer health, it could be health deterioration (e.g. health changes between waves two and eight) experienced by the older respondents that is contributing towards the significant positive association between age and loneliness. This would imply that poor health and loneliness may occur simultaneously, or that the effect occurs in a shorter time-period, rather than poor health leading to long-term loneliness outcomes.

Females had 46% higher odds of loneliness at follow-up than males (OR=1.46, 95CI=1.08-1.97). Previous evidence on sex differences and loneliness is largely inconclusive. Some literature suggests prevalence of loneliness is higher among females (Pinquart and Sörensen, 2001). However, seminal work evaluating the validity of the UCLA-LS suggested no sex differences when applying this loneliness scale, which was supported in an early review of sex differences between the UCLA-LS and direct measures, whereby few studies reported a significant difference, and of those that did, males appeared to be lonelier (Borys and Perlman, 1985). More contemporary evidence which utilised data from the BBC Loneliness Experiment also reported that males were lonelier than females (Barreto et al., 2020). In the cross-sectional analyses, being female was a significant determinant of loneliness among caregivers, but not non-caregivers. Moreover, analyses on the full sample found no significant difference cross-sectionally for UCLA-LS loneliness between males and females (section 5.2.2.2). The positive association between sex and loneliness at follow-up, when excluding those lonely at baseline, suggests that over the 12-year period, females were more likely to become lonely than males, however this contradicts the cross-sectional result, which suggests that, at wave eight, among the full sample, being female was not associated with a higher likelihood of loneliness.

Additionally, as none of the wave two health controls (physiological health, health expectations, alcohol consumption or depressive symptomatology) were significantly associated with wave eight loneliness, this supports the aforementioned discussion that health does not appear to generate a long-term effect on loneliness, after accounting for sociodemographic and socioeconomic characteristics. Previous results suggest cross-sectional associations between loneliness and health, particularly for depressive symptomatology, which was not apparent in this follow-up analysis. In contrast, both wellbeing measures were significantly associated with long-term loneliness outcomes, those with higher QoL and life satisfaction at baseline were significantly less likely to be lonely at follow-up. This could suggest that wellbeing has a stronger and more enduring negative association with loneliness, than that of health.

# 6.2.2.2 The association between caregiving and loneliness over 12-years (ii) controlling for loneliness at baseline

In contrast to the previous regression, whereby baseline lonely individuals were excluded, a further binary logistic regression was conducted controlling for baseline loneliness. This analytic sample represented 70% (N=2074) of the overall sample. This would determine whether caregiving, after controlling for baseline loneliness was associated with later loneliness. Caregiving in wave two was not significantly associated with loneliness at wave eight in any model (Appendix Z), however baseline loneliness was strongly positively associated with follow-up loneliness. In the fully-adjusted mode (Figure 26), individuals who were lonely at baseline, on average, were four times more likely to be lonely at follow-up (OR=4.13, 95CI=3.08-5.54). This suggests that loneliness may be persistent, or highly likely to reoccur, regardless of caregiving responsibilities, and that caregiving shows no link to long-term loneliness outcomes, even amongst those previously lonely.



# Figure 26: Odds ratio plot, wave two factors associated with wave eight loneliness, controlling for baseline loneliness

Source: author's analysis of ELSA waves 2 and 8

Respondents not expecting to get lonelier with age, when asked at baseline, were less likely to be lonely at follow-up than those who expected loneliness to occur (OR=0.68, 95CI=0.51-0.93). However, this negative association was not significant in the previous regression when baseline

lonely individuals were excluded (section 6.2.2.1). Loneliness expectations were more pertinent to lonely individuals. To illustrate, those already lonely may be more likely to have a negative perception on future loneliness than non-lonely individuals. This would explain why there was no significant difference in loneliness outcomes between the loneliness expectations categories when baseline lonely individuals were removed, i.e. a smaller percentage of individuals in the sample expected to become lonelier with age, because they were within the baseline lonely category, and therefore excluded. Crosstabulations support this (Appendix AA): among the respondents lonely at baseline, a substantially higher proportion of individuals expected to get lonelier with age than the non-lonely, 67% compared to 32%.

After controlling for baseline loneliness, being female was no longer positively associated with loneliness at follow-up. This suggests that the previously significant differences between males and females in RQ3a(i), may be accounted for by loneliness differences at baseline. As such, when lonely individuals at baseline were included, there were no significant differences in the loneliness outcomes between males and females. This might suggest that by removing the baseline lonely respondents, the balance of males and females within the sample was skewed, i.e. more females were lonely at baseline than males. Nevertheless, the lack of a significant relationship between sex and loneliness over the 12-year follow-up suggests that after accounting for caregiving, baseline loneliness, and all of the other factors, the occurrence of loneliness does not differ due to sex. However, after controlling for loneliness at baseline, age remained positively associated with loneliness at follow-up, respondents aged  $\geq$ 70 were almost two times as likely to be lonely at follow-up, compared to respondents aged 50-59 (OR=1.98, 95CI=1.35-2.89). Baseline health remained non-significant, so it is reasonable to consider that, as discussed in the previous regression, health changes over the 12-year period which are likely to affect the older age groups more, could be contributing towards the significance seen between age and loneliness.

Wellbeing remained negatively associated with loneliness, after controlling for baseline loneliness. As such, individuals with higher QoL and life satisfaction were less likely to be lonely at follow-up, after accounting for the respondents who were experiencing loneliness at baseline. As well as asserting that a long-term relationship is present between wellbeing and loneliness, the results might also imply that loneliness and poor wellbeing are comorbidities. To illustrate, mean CASP19 and SWLS scores were considerably lower for lonely respondents at wave two, compared to non-lonely. Therefore, poor wellbeing, in conjunction with loneliness, could contribute towards a greater likelihood of persistent or reoccurring loneliness.

# 6.2.2.3 Summary: is caregiving in wave two associated with loneliness at wave eight?

When comparing the results from (i) excluding baseline loneliness with the results from (ii) controlling for baseline loneliness, the variance explained by the final models differs substantially.

The Nagelkerke  $R^2$  statistics demonstrate that the fully-adjusted models explained around 9% (i) and 23% (ii) of variance in wave eight loneliness, respectively. Given that the difference between the analyses was the inclusion of previously lonely individuals, the result suggests that baseline loneliness explains a large amount of variance in follow-up loneliness. However, both of the final models were deemed a good fit, from the Hosmer-Lemeshow test (p=.109 and p=.062 respectively). Table 32 contrasts the significant variables from each of the analyses.

 Table 32: Summary of significant results for RQ3a: is caregiving in wave two associated with loneliness at

 wave eight?

Variables	RQ3a(i) Excluding baseline loneliness	RQ3a(ii) Controlling for baseline loneliness
Caregiving	NS	NS
Loneliness	-	***
Loneliness expectations: neither agree nor disagree	NS	NS
Loneliness expectations: disagree	NS	**
Age 60-69	**	**
270	**	*
Sex	*	NS
Cohabitation	NS	NS
Employment	NS	NS
Wealth Q2	NS	NS
Wealth Q3	NS	NS
Wealth Q4	NS	NS
Household: dual-person	NS	NS
Household: large (≥3)	NS	NS
Physiological health	NS	NS
Health expectations: neither	NS	**
Health expectations: disagree	NS	NS
Alcohol consumption	NS	NS
Depressive symptomatology	NS	NS
CASP19	**	*
SWLS	**	**

Notes: NS non-significant, \*p<.05, \*\* p<.01, \*\*\* p<.001, source: author's analysis of ELSA waves 2 and 8

The positive association between baseline age and follow-up loneliness, and negative association between baseline wellbeing and loneliness at the 12-year follow-up remained consistent both when excluding, and accounting for, baseline loneliness, whereas the main differences were sex, loneliness, and health expectations. Being female was only positively associated with later loneliness when baseline loneliness was excluded, this implies sex differences in the experience of loneliness at baseline, which are then accounted for in the second regression, hence the attenuation of significance. Alternatively, expectations for future loneliness and health were only significantly negatively associated with follow-up loneliness after controlling for baseline loneliness. As discussed previously, respondents' perceptions on future loneliness appear to be linked to their current experience of loneliness, whereby the proportion of baseline lonely respondents expecting to become lonelier with age was over twice the proportion of non-lonely respondents with similar expectations (67% compared to 32%), therefore by deleting the baseline lonely respondents from the sample in RQ3a(i), this also removed a large number of respondents expecting to be lonelier.

In contrast, only respondents who 'neither agreed nor disagreed' that they expected their health to deteriorate with age significantly differed from those who 'agreed'. Neutral health expectations were linked to lower odds of loneliness when compared with negative expectations (agree), but no

significant difference was observed between agree and disagree responses. It is challenging to speculate or interpret the meaning of this result, because it is not evident what is meant when the respondent cannot agree or disagree with the health statement. As such, research suggests that the interpretation of this middle category differs substantially from neutral, to no opinion, to equal or both, to neither, and to unsure (Nadler et al., 2015), and further, it should be interpreted differently to non-response or unknown (Baka et al., 2012). When researched in the context of Greek local elections, the middle category was reportedly selected due to a lack of knowledge or indifference on the subject, indecisiveness, or ambivalence, or to challenge the assumptions of the statement (Baka et al., 2012). In the context of the health statement, the latter seems unlikely because 'disagree' would inherently challenge the assumption that old age was linked to poor health. However, it is plausible that the respondents either felt they did not know enough to speculate on their future health, for example, they might consider advancements in medical healthcare, and consider whether recent or upcoming developments might impact themselves. Alternatively respondents might be undecided, they may have experienced periods of both good and poor health and therefore be uncertain about what the future may hold in regard to their health. Therefore, without accurately knowing why a respondent selected 'neither', it is difficult to draw conclusion about the link to loneliness.

Overall, there was no evidence to support a long-term association between caregiving at wave two and loneliness at wave eight. The strongest predictor of loneliness at follow-up appeared to be baseline loneliness, suggesting that loneliness was largely persistent or reoccurring. The subsequent section explores the relationship between caregiving and loneliness across all waves to determine if care provision is linked to experiencing later loneliness.

#### 6.2.3 RQ3b: is caregiving at any wave associated with later loneliness?

The cross-sectional results suggested that prior to controlling for health and wellbeing (section 5.2.2.2), caregiving was positively associated with loneliness; this was under the circumstances where caregiving and loneliness were present at the same time-point. Moreover, subsequent to RQ3a, which examined whether caregiving at baseline (wave two) was associated with loneliness at follow-up (wave eight), this research question augments the previous analyses by questioning whether caregiving at any point over the 12-year period was associated with experiencing later loneliness. Duration of care was used as the primary explanatory variable when examining loneliness occurring after care provision, comparing short-term caregivers (only one wave) with long-term (two or more consecutive waves) and intermittent caregivers (two or more non-consecutive waves). All further controls were measured at baseline (wave two). Table 33 shows the models applied in the binary logistic regression for RQ3b.

Table 33: Regr	ession models for RQ3b: is caregiving at any wave associated with	ever being lonely?
Model number	Variables included in the model	
1	Duration of care	
2	Model 1 + loneliness expectations, sociodemographic characteristics, socioeconomic status and social networks	
3	Model 2 + health and wellbeing	
Source: author's o	own	
Model one inc	cluded only the duration of care variable, which characterised	caregivers as short-term,
long-term or i	intermittent. The purpose of this was to identify whether di	fferent durations of care
provision were	e associated with experiencing later loneliness. Loneliness that	t occurred prior to or was
first reported a	at the same time as caregiving was excluded to ensure all lone	liness occurred after care
<mark>provision.</mark> M	odel two controlled for sociodemographic characteristic	s, SES, and loneliness

expectations. Individuals were asked at baseline whether they expected to get lonelier as they aged. Controlling for expectations tests the self-fulfilling prophecy theory in SET. In the final model, health and wellbeing variables at baseline were included to account for previous health and wellbeing, which may also be linked to the occurrence of loneliness over the waves. All control variables in this regression were measured at baseline and thus treated as time-invariant.

### 6.2.3.1 The association between caregiving duration and later loneliness

Long-term care provision was significantly positively associated with loneliness, compared to shortterm care, when the data were examined over all waves (2-8) (Appendix BB). Of the 1445 respondents who indicated providing care in at least one wave, 507 (35%) were excluded due loneliness occurring before, or starting in the same wave as care provision, leaving an analytic sample of 938 caregivers. Of those 938, a further 10% (N=92) were excluded from the regression analyses due to incomplete data. In the fully-adjusted model after controlling for loneliness expectations, sociodemographic characteristics, SES, household size, and health and wellbeing, odds of long-term caregivers experiencing loneliness were 1.8 times higher than short-term caregivers (OR=1.82, 95CI=1.17-2.85). In contrast to RQ3a, whereby no long-term association was found between caregiving and loneliness (section 6.2.2), this result suggested that the impact of caregiving on loneliness may be shorter-term, i.e. caregivers may experience loneliness during their role, or for a period succeeding their role, but this did not extend to the 12-year follow-up. The Nagelkerke R<sup>2</sup> statistic illustrated that the fully-adjusted model explains 15% of variance in loneliness across the waves, and the Hosmer-Lemeshow test confirmed that the model is a good fit for the data (p=.130).

In addition to duration of care, loneliness expectations were also significantly negatively associated with loneliness outcomes (Figure 27). Compared to respondents who agreed that they expected to become lonelier with age at baseline, caregivers who disagreed were 40% less likely (OR=0.60, 95CI=0.37-0.98) to be lonely after care provision. This supports the relevance of age-related stereotypes (Kornadt, 2016), and advocates that internalised stereotypes can become self-fulfilling prophecies with negative consequences (Levy *et al.*, 2009; Meisner and Levy, 2016).



Figure 27: Odds ratio plot, associations between loneliness and duration of care Source: author's analysis of ELSA waves 2-8

None of the included baseline controls, except QoL, were significantly associated with loneliness occurring after care provision. In some circumstances, variables may have changed between baseline (wave two) and the loneliness occurring (any wave) which could account for the lack of significant association. For example, in comparison to the 12-year follow up (section 6.2.2), age was no longer significantly positively associated with loneliness; compared to ages 50-59, there was no significant difference in prevalence of loneliness among the 60-69 or  $\geq$ 70 age categories. This would suggest, as in the cross-sectional analyses, that there was no association between age and loneliness, or that this was better explained by other factors that may have changed over the 12-year period such as health. This could be considered a limitation of the approach to treat control variables as time invariant (section 7.5.3).

As indicated above, QoL remained negatively associated with loneliness, higher QoL scores at baseline were associated with a lower likelihood of loneliness occurring after the start of care provision (OR=0.93, 95CI=0.90-0.96). This is partially consistent with the literature and the previous regressions. To illustrate, in the previous research question, both baseline wellbeing measures were negatively associated with follow-up loneliness in the full longitudinally-weighted sample, whereas in this analysis, only baseline QoL was significant. Therefore, this would suggest that lower QoL is

linked to both shorter and longer-term effects on loneliness among caregivers and non-caregivers, whereas the evidence suggests that the link between life satisfaction and loneliness is apparent only among non-caregivers. As such, cross-sectionally, life satisfaction was not significantly associated with UCLA-LS among the caregiving sample (section 5.2.3.1 and 5.3.2.1) but was significantly negatively associated with UCLA-LS among the full sample (caregivers and non-caregivers, section 5.2.2.2) and non-caregivers (section 5.2.3.2); this appears to remain consistent in the follow-up analyses, whereby life satisfaction was previously significant (section 6.2.2) among the full sample, but not among the current caregiving sample.

#### 6.2.3.2 Summary: is caregiving at any wave associated with later loneliness?

Duration of care provision was positively associated with UCLA-LS loneliness. Caregivers providing care for two or more consecutive waves (long-term) demonstrated higher odds of reporting loneliness than caregivers for a single wave (short-term). Alternatively, intermittent care was not associated with increased odds of loneliness compared to short-term care. Thus, loneliness was more likely to occur for long-term caregivers than for caregivers providing care over shorter periods of time, regardless of whether it was the caregivers' first episode of care provision or a subsequent episode of care provision at a later date. This emphasises that the positive association between caregiving and loneliness may be linked to continuing care provision over a prolonged period, rather than multiple caregiving episodes. Moreover, wave two caregiving was not significantly associated with wave eight loneliness (section 6.2.2). This suggests that loneliness may occur during the caregiving role particularly for long-term caregivers, or after the caregiving role and may persist for a short-term period, but there is little evidence to support long-lasting effects of caregiving on loneliness over periods such as 12-years.

Alongside duration of care, factors negatively associated with experiencing loneliness after care provision were negative expectations for future loneliness, and QoL, both measured at baseline. This result aligns with the theory of self-fulfilling prophecies and internalised age-stereotypes in SET, in that expecting to become lonelier was linked to an increased likelihood of a loneliness outcome. However, no significant associations were found for any of the health measures, indicating that, for example, presence of chronic CV conditions at baseline had limited influence on later loneliness. As identified previously, this could be a limitation of adopting time invariant controls, for full discussion see section 7.5.3. The following research question analyses the caregiving sample further, investigating how the caregiving relationship and role contribute to later loneliness and how changes to loneliness expectations and mental health impact loneliness.

# 6.2.4 RQ3c: how are the caregiving relationship, and role, associated with loneliness for short-term *versus* long-term, or intermittent caregivers?

Question 3c seeks to answer whether aspects of the caregiving relationship and role impact loneliness experienced by caregivers. As such, comparison is drawn between the duration of care provision, the caregiving relationships (relationship to the recipient and whether the caregiver lives with them), and the caregiving role (caregiving intensity and the number of care recipients). Table 34 shows the models applied in the binary logistic regression for RQ3c.

 Table 34: Regression models for RQ3c: how do changes to the caregiving relationship and role impact

 loneliness among caregivers?

Model number	Variables included in the model
1	Duration of care
2	Model 1 + care recipient, co-residential care, caregiving intensity, and number of care recipients
3	Model 3 + changes to loneliness expectations and mental health

Source: author's own

In this regression, the main explanatory variable was duration of care. Caregivers who had provided long-term care ( $\geq 2$  consecutive waves) and cared intermittently ( $\geq 2$  non-consecutive waves) were compared with short term caregivers (caregivers providing care in only one wave). This will identify whether there is a difference in loneliness between caregivers providing care at one time point, those who care over consecutive waves for longer time-periods, and caregivers who provide care on multiple separate occasions. Model two introduces the caregiving relationship and role variables. As some of the caregivers had only provided care at one time-point, it was not possible to look at changes to these variables over the caregiving role; short-term caregivers would always be in the 'stable' categories, and this would likely affect the results. Therefore, the caregiving variables were measured as, for example, 'ever provided high-intensity care' or 'ever provided care to multiple recipients'. Significant results would indicate whether that particular aspect of caregiving was associated with loneliness after controlling for other caregiving-specific variables. In the final model, control variables were added for loneliness expectations, and changes to mental health (details in section 4.3.8). By controlling for changes to mental health and loneliness expectations over the 12-year follow-up, this identified whether, after accounting for caregiving variables, mental health and loneliness expectations were more strongly associated with loneliness.

# 6.2.4.1 The association between the caregiving relationship, and role, with loneliness.

The odds of loneliness were not significantly different for intermittent or long-term caregivers, in comparison with short-term caregivers in the fully-adjusted model (Figure 28). This could have been due to the smaller sample size (N=523), in this final regression analysis, 44% of the potential sample (N=938 caregivers, either non-lonely or loneliness occurring after care provision only) were excluded due to missing data. However in the initial model (Appendix CC), before controlling for other

caregiving-specific factors, significant difference was observed between long-term and short-term care provision. Long-term caregivers were 1.9 times more likely to be lonely than short-term caregivers (OR=1.87, 95CI=1.16-3.00). This aligns with the result from RQ3b, in that longer periods of care were linked to increased odds of loneliness. However the attenuation of the significant result after controlling for variables related to the caregiving relationship and role suggest that other caregiving-specific factors were more strongly associated with caregiver loneliness after accounting for care duration. Overall, the result suggested that loneliness was not wholly dictated by the duration of the caregiving role. Intensity of care was also an influential factor. In the fully-adjusted model, odds of loneliness were significantly higher for caregivers providing high-intensity (OR=1.88, 95CI=1.11-3.19), or round-the-clock care (OR=1.94, 95CI=1.05-3.59) compared to low-intensity care.

Previous results from the cross-sectional analyses suggested that caregiving to an adult-child was linked to increased odds of loneliness (section 5.3.2.1), this positive association was not present in the follow-up analyses. Therefore, while care to an adult-child appeared to be strongly linked to loneliness that occurred simultaneously with the provision of care, the significant positive association between caregiving intensity, and previously caregiving duration, may better explain the longer-term relationship between care and loneliness. Moreover, Barnhart *et al.* (2020) indicates that caregivers to an individual with an intellectual or developmental disability (adult-child) are more likely to provide care over a longer duration (25-years compared to national US average of four-years) and at higher-intensity (57hpw compared to national US average of 24hpw). Thus the provision of high-intensity care, over longer durations is reflective of the care provided to adult-children, consequently the intensity and duration variables may account for the care relationship and have contributed towards the lack of significance.

The provision of co-residential care was not associated with loneliness among caregivers. As such, caregivers living with the care recipient did not experience a greater or lesser likelihood of loneliness in comparison to caregivers living separately from the care recipient. It is unknown from the data whether those providing co-residential care had additional support with the caregiving role; for example, caregivers living with and providing care to their spouse, whether they had adult-children outside, or within the home to assist with care needs. This would likely reduce the likelihood of loneliness if there were additional caregivers to share the burden. Caregivers who were ever required to give round-the-clock care indicated that their caregiving responsibility was constant, this suggests limited availability for social activities or breaks from the caregiving role. This would also require being available to meet care needs during the night and suggests co-residential care was highly likely. The significance between round-the-clock care and loneliness may account for this and explain why no significant difference was seen between co-residential care and loneliness. Finally, caregivers to multiple recipients were no more or less lonely than the caregivers to single recipients.



Figure 28: Odds ratio plot, associations between loneliness and duration of care, caregiving-specific factors and changes to loneliness expectations and mental health Source: author's analysis of ELSA waves 2-8

Changes to loneliness expectations were negatively associated with loneliness, and changes to mental health, as measured by depressive symptomatology, were positively associated with loneliness among caregivers. Compared to caregivers who always expected to become lonelier with age (always agree), those who demonstrated a positive change in expectations, or fluctuating change in expectations were less likely to experience loneliness. Interestingly, the odds of loneliness for caregivers always disagreeing, i.e. never expecting to grow lonelier with age, were not significantly different from those always agreeing, i.e. always expecting to grow lonelier with age.

When considering changes to depressive symptomatology over the 12-year period, compared to the reference group (stable: no depressive symptoms), persistent or an increase in depressive symptoms (OR=4.42, 95CI=2.39-8.17), and fluctuating (OR=2.73, 95CI=1.25-5.98) depressive symptoms were indicative of higher odds of loneliness. However, there was no significant difference in loneliness odds between caregivers who reported a reduction in depressive symptoms and the reference group. From these results, it is conceivable that depression and loneliness may be comorbid, and most frequently occur within similar time-periods, otherwise the 'reduction' group would also demonstrate heightened odds of loneliness due to having previously experienced depressive symptoms.

# 6.2.4.2 Summary: how do changes to the caregiving relationship and role impact loneliness among caregivers?

Overall, the results in RQ3c suggested that the main aspect of the caregiving relationship and role that influences loneliness was the caregiving intensity. Previously in the cross-sectional analyses the care recipient, predominantly caregiving for an adult-child appeared the most influential. However,

as discussed previously, recipient and intensity are likely be linked, as it is known that care to adultchildren has a higher average of duration and intensity, compared to other care recipient groups. The Nagelkerke R<sup>2</sup> indicated that 16% of variance in caregiver loneliness was explained by the fullyadjusted model, and the Hosmer-Lemeshow statistic (p=0.729) suggested the model was a good fit for the data. Some of the unexplained variance may be accounted for by caregiver characteristics such as sex, age, or employment. In the cross-sectional caregiver-only analysis (section 5.3.2.1), sex was positively associated, and employment negatively associated with caregivers' loneliness. Additionally, in the previous follow-up regression analyses (sections 6.2.2 and 6.2.3), age has been shown to be significantly positively associated with loneliness over the 12-year period. Baseline caregiver characteristics were not controlled for in these analyses, due to the focus on changes to mental health and loneliness expectations; the small numbers of caregivers reporting change between the waves meant that splitting by sex or age category would prove unreliable, leaving much smaller subgroups for comparison or leading to overfitting of the regression model. Future research, to build on the contribution of this thesis, might consider a similar approach with a larger caregiving sample, which would permit additional controls.

#### 6.2.5 Summary: what are the longer-term effects of caregiving on loneliness?

There were inconsistencies between the follow-up and cross-sectional findings among caregivers. Overall, the results within this thesis suggest that caregiving to an adult-child, was positively associated with loneliness when examined cross-sectionally. However, the follow-up analyses highlighted significant positive associations between both duration, and intensity, of care, and loneliness. Literature suggests a substantial overlap between these caregiving-specific factors, in that caregivers to adult-children provided care at higher-intensities and longer durations that other caregiving groups. Alternatively, this may be that the impact of caregiving intensity has longer-term effects on loneliness; for example, caregivers providing high-intensity or round-the-clock care may not experience loneliness at the same time as they are providing care (as would be indicated in cross-sectional analyses), but that this may develop over longer durations of care, or subsequent to the caregiving role due to diminished social circles while caregiving.

The initial follow-up analysis (RQ3a) suggested that the association between caregiving and loneliness does not extend over the 12-year period, indicating that the relationship may be shorterterm. There was no evidence of increased loneliness odds among the wave two caregivers after the 12-year follow-up. However, provision of long-term care at any point between waves two and eight was positively associated with loneliness occurring after care provision. A strong positive relationship was observed between loneliness at baseline and loneliness at follow-up, which would advocate loneliness as a trait characteristic, enduring rather than temporary like a state characteristic. Arguably, the positive association between caregiving and loneliness appeared to be short-term. Therefore, this suggested loneliness in itself may be more likely to be consistent or reoccurring, irrespective of caregiving, but that caregiving does not appear to have long-lasting effects on loneliness.

Expectations for future loneliness appeared influential to long-term loneliness, particularly in the analysis controlling for baseline loneliness (RQ3a(ii)), and when examining loneliness occurring after the start of care provision (RQ3b). The results from this thesis support previous research into loneliness as a self-fulfilling prophecy, see Pikhartova *et al.* (2016).

The impact of sex was unclear, while significant in RQ3a(i), significance between sex and loneliness attenuated when baseline loneliness was controlled for in RQ3a(ii). Additionally when examined among caregivers over all waves (2-8), sex at baseline was not significantly associated with experiencing later loneliness (occurring after the start of care provision). This indicates that, overall, females may be more likely to be lonely, but that other factors such as loneliness or care provision may be stronger predictors of future loneliness than sex. As such, males who were previously lonely may be more likely to continue to be, or to be lonely again due to previous loneliness; sex would not be considered a protective factor for reoccurring loneliness among males. Moreover, although care provision was not associated with loneliness at 12-year follow up, there was evidence for shorter-term effects. Therefore, the significant positive association observed between females and loneliness in RQ3a(i) may be better explained by other factors occurring between the two time-points. For example, females have been shown to be more likely to be caregivers, therefore the females at baseline may become caregivers between waves three and eight, which could contribute towards the later reported loneliness.

Among caregivers, the duration of care was not statistically relevant for loneliness after controlling for other caregiving-specific factors such as intensity and the caregiving relationship (RQ3c). However, when controlling only for caregivers' sociodemographic characteristics, SES and health and wellbeing at baseline (RQ3b), characteristics long-term care provision was indicative of increased odds of later loneliness. The main factors influencing longitudinal loneliness among caregivers appeared to be caregiving intensity; the provision of high-intensity or round-the-clock care resulted in odds almost twice as high as low-intensity caregivers. The results from RO3c also established that demonstrating a positive change in loneliness expectations, or even fluctuating expectations over the waves, had a beneficial impact on loneliness, in that caregivers with this aforementioned shift in expectations were less likely to become lonely. An increase in, or persistent depressive symptoms influenced loneliness as expected, with higher odds compared to never reporting depressive symptoms. However, the result for fluctuating depressive symptoms was less predictable; unlike for loneliness expectations, fluctuating symptoms resulted in higher odds of loneliness. This suggests that depression and loneliness may be comorbid, rather than one leading to the other, as it is plausible that loneliness occurred when depressive symptoms were present for those who were categorised as fluctuating.

## 6.3 Chapter summary

Chapter six has summarised the findings from the follow-up analyses. Four binary logistic regressions were carried out comparing loneliness among wave two caregivers and non-caregivers, and among short-term, long-term, and intermittent caregivers across waves 2-8. The results suggested that the positive association between caregiving and loneliness may be acute rather than chronic, highlighting a critical period for loneliness intervention (during the caregiving role). However, there is also evidence to suggest that loneliness, irrespective of caregiving, may be enduring. Therefore, both caregivers and non-caregivers who have been previously lonely, are at risk of chronic or reoccurring loneliness. The discussion in the subsequent chapter reflects on both the cross-sectional and follow-up results in relation to theory and literature, highlighting the original contribution of this thesis. Furthermore the final chapter reflects upon limitations of this thesis, considers the policy implications of the findings, and provides suggestions for future research.

# Chapter 7 Discussion and conclusion

### 7.1 Chapter outline

This discussion chapter addresses the main aims of the thesis. It is structured as follows: the association between caregiving and loneliness (section 7.2), the determinants of loneliness (section 7.3), and the health and wellbeing consequences (section 7.4). Furthermore, the limitations of the thesis will be discussed (section 7.5), before concluding and providing policy and research recommendations (section 7.6.).

## 7.2 Association between caregiving and loneliness

When comparing caregivers with non-caregivers in the ELSA sample, the results of the crosssectional analysis suggested that the positive association between caregiving and loneliness was conditional on the loneliness measure. As such, caregiving was not significantly associated with the direct loneliness question but was positively associated with UCLA-LS loneliness. Furthermore, the positive association between caregiving and UCLA-LS loneliness attenuated with the inclusion of health and wellbeing controls, which suggests that health and/or wellbeing are more strongly associated with loneliness, after accounting for caregiving. Nevertheless, in the unadjusted model, the odds of caregivers reporting loneliness were 1.35 times higher than non-caregivers. This finding is comparable with the ONS statistic, based on CLS data from 10,256 adults across England, which indicated that caregivers were around 37% more likely to report loneliness than non-caregivers (Pyle and Evans, 2018). Alternatively, caregivers' expectations for future loneliness did not significantly differ from non-caregivers; therefore despite the positive association between caregiving and loneliness, caregiving was not indicative of a negative outlook on future loneliness. As such, caregivers were no more, or less, likely than non-caregivers to expect loneliness as they age. Previous evidence using ELSA data demonstrated that loneliness expectations were linked to future loneliness (Pikhartova et al., 2016), although this was not compared for caregivers and non-caregivers. However, there was no current evidence found whereby loneliness expectations were the outcome variable. Therefore, the knowledge around what contributes towards future loneliness expectations is limited. But the findings from this thesis suggest that caregiving is not a contributing factor. This would imply that the loneliness experienced by caregivers is not the result of an internalised stereotype and may be related to factors specifically associated with caregiving.

However, when examining caregiver characteristics, the caregiving relationship, and the caregiving role in relation to loneliness, the cross-sectional results suggested that role factors (e.g. intensity or number of recipients) may play a lesser part than caregivers' sociodemographic and socioeconomic characteristics or relationship with the care recipient. Once controlled for social networks, health, and wellbeing, female caregivers compared to male, non-employed compared to employed, and

caregivers to a dependent adult-child compared to spousal caregivers, had higher odds of loneliness. Whereas caregiving intensity and number of care recipients were not significantly associated with caregiver loneliness in any model. These findings align with certain aspects of caregiver literature, but also give additional insight into how these caregiving factors tie together. As such, caregiving literature suggests female caregivers are more likely to be lonely (Beeson, 2003; Vitaliano *et al.*, 2003; Soylu *et al.*, 2016). However, in previous literature this was not examined simultaneously with other caregiver characteristics, or factors relating to the caregiving relationship and role.

Within the literature on caregiver loneliness, the discussion is centralised around sex, with a large body of evidence (e.g. Beeson, 2003; Soylu et al., 2016; Victor et al., 2020) indicating that female caregivers are lonelier than males (section 3.4.2.1). Alternatively, only evidence from Bramboeck et al. (2020) was found to dispute this, suggesting male caregivers were lonelier. The results from this thesis also suggested female caregivers were lonelier than males, when examined among other traditional loneliness determinants (section 5.2.3.1), and caregiving-specific factors (section 5.3.2.1) supporting the majority of previous evidence. The literature review also identified co-residential care, and caregiver-recipient relationships as important factors related to loneliness. Bramboeck et al. (2020) found co-residential caregivers to be lonelier than non-residential, and Victor et al. (2020) reported that spousal caregivers were lonelier than caregivers to other family members or friends. Within this thesis co-residential care was only significantly positively associated with caregiver loneliness before controlling for social networks, suggesting that an external network outside of the caregiving role, particularly close relationships with friends, may be vital for those providing care to someone within their household. Moreover, the results suggest that after controlling for other caregiving-specific factors, and caregivers' health and wellbeing, those providing care to adultchildren experienced greater loneliness than spousal caregivers. Literature suggests that caregivers to adult-children provide care at considerably higher intensities (Barnhart et al., 2020), and over longer durations (Anderson et al., 2018), which may account for the heightened loneliness.

However, in comparison to research into caregiver characteristics (sex), the body of evidence examining the links between factors related to the caregiving relationship or role and loneliness is substantially smaller. Evidence from this thesis conflicts with previous evidence on the caregiving relationship and role highlighting a need for further robust research into these areas and less emphasis on sex differences which are well-established. Additionally, among caregiving literature, intensity is a substantial factor linked to caregivers' health and wellbeing, but much less is reported about the effects of caregiving intensity on loneliness. This thesis found no significant association between caregiving intensity and loneliness cross-sectionally but found evidence of significant longer-term associations. However, the significance of adult-child caregiving may account for some of the intensity differences between the caregivers, i.e. if the majority of caregivers to adult-children were providing high-intensity or round-the-clock care, this significant difference may explain why no significant difference was seen for the caregiving intensity variable. Nevertheless, caregiving duration, arguably another method of determining caregiving intensity, was significantly positively

associated with loneliness. Thus specific studies examining the impact of caregiving intensities on loneliness, both short-term and long-term are vital to further clarify the relationships between these variables.

When considering the long-term associations of caregiving and loneliness, the findings suggest that the positive association between caregiving and loneliness may be more acute than chronic. As such, caregiving was not associated with loneliness at 12-year follow-up, but long-term caregiving (for two or more waves) over the 12-year period was positively associated with loneliness occurring after care provision. No published literature was found reporting the lasting effects of caregiving, therefore little comparison can be drawn between these findings and other data sources. Moreover, there is a substantial lack of clarity about how long during, or after the caregiving role caregivers might experience loneliness. This thesis provides an insight suggesting that the impact of caregiving on loneliness is less than 12-years, but further research is essential to identify whether the effects are lasting, and how long for. Nevertheless, the findings highlight the need for immediate intervention, or prevention strategies to ensure loneliness does not endure, nor reoccur, as previous loneliness among caregivers was shown to increase the odds of loneliness at 12-year follow-up by over four times.

## 7.3 Determinants of loneliness

Investigation into the determinants of loneliness identified that largely, the factors related to loneliness among caregivers are different to non-caregivers. However, loneliness expectations, an indicator of SET psychological health, was a significant determinant of loneliness for both caregivers and non-caregivers. The cross-sectional results indicated that those who did not expect to be lonelier with age, were also less likely to be lonely at the same time-point. Moreover, there was evidence of longer-term effects; positive loneliness expectations at baseline were negatively associated with loneliness scores 12-years later when controlled for baseline loneliness. Pikhartova et al. (2016) have previously demonstrated using ELSA data that expecting to be lonelier with age was positively associated with loneliness 2-6 years later. Therefore, this thesis augments this conclusion by demonstrating that loneliness expectations are also associated with more immediate impact on loneliness and showing the impact of loneliness expectations persists over longer time-periods. SET suggests that age-stereotypes relate to a range of health-related areas (Levy, 2009; Meisner and Levy, 2016), predominantly research considers stereotypes of physical and mental health decline. Aside from the cited work from Pikhartova et al. (2016), limited research has been conducted on internalised loneliness stereotypes and the implications of loneliness expectations. Thus, this thesis provides additional context of the impact of loneliness expectations over both shorter and longer time-periods.

The determinants of loneliness exclusive to non-caregivers were predominantly health and wellbeing related. In contrast, factors associated with loneliness among caregivers included sex, employment,

and closer relationships with friends. Largely, the determinants match previous literature; however there is limited literature in caregiving samples, so little comparison can be drawn. The discrepancies between caregivers and non-caregivers could aid the identification of those at higher risk of loneliness. As such, female and non-employed caregivers are more likely to be lonely, and similarly non-caregivers living alone with poor mental health. These individuals are most likely to experience loneliness and thus would benefit from loneliness interventions. The findings also emphasise a clear need for further research, as anticipated from the literature review, the majority of determinants of loneliness for caregivers were different from the non-caregivers, less prominence appears to be placed on health and wellbeing as determinants of loneliness for caregivers. Among dementia caregivers, previously reported determinants of loneliness were social networks, cohabitation, depression, life satisfaction and QoL (Victor et al., 2020). These findings were almost mirrored in the present thesis, with the exception of depression and life satisfaction, which were only positively and negatively (respectively) associated with loneliness among non-caregivers, and the significance of employment for caregivers. Although arguably, the impact of employment on caregiver loneliness may be related to the social network which accompanies being employed. Results demonstrated that a larger network of close friends was negatively associated with loneliness, whereas a larger network of close family members was not significantly associated with loneliness. Research on relationships (section 3.2.2) and social networks (section 3.2.3) highlights the importance of relationship quality, and it was suggested by Stephens et al. (2011) that the individuals making up the network was also relevant for loneliness. Family-based networks have been linked to an increased risk of loneliness, as they provide little other social support (*ibid*); moreover, family networks may be indicative of a restricted network, whereby the individual may have reduced connections with the wider network due to health. Therefore, the finding that close friends have a significant positive impact on loneliness, but close family do not, aligns with the conclusion of Stephens et al. (2011) that family networks may provide minimal social support.

A key theme identified within the literature review was the importance of SES as a determinant of loneliness. Overall, previous evidence suggested that lower SES was positively associated with loneliness, whereby loneliness was highest among the poorest (Pinquart and Sörensen, 2001; Niedzweidz *et al.*, 2016), non-employed (Algren *et al.*, 2020), and least educated (Pinquart and Sörensen, 2001; McRae *et al.*, 2009). The results of this thesis generally upheld these conclusions in that poorer and non-employed respondents demonstrated higher odds of loneliness. Employment was negatively associated with loneliness among the full sample and amongst caregivers when examined separately. However, the relationship between wealth and loneliness appears to be moderated by health and wellbeing. Among caregivers (sections 5.2.3.1 and 5.3.2.1) the negative association between wealth and loneliness attenuated after the inclusion of health and wellbeing controls. Furthermore among the full sample, health and wellbeing controls affected the direction of the relationship between wealth and loneliness. Initially, those in lower wealth quartiles were lonelier than wealthier respondents (section 5.2.2.2), which aligns with the published evidence. However,

after controlling for health and wellbeing effects, this ceased to apply. The results demonstrated that, without the additional health and wellbeing consequences that are typically associated with poverty (for example, due to healthcare inaccessibility, or other unattainable health-promoting lifestyle factors, such as healthier diets), poorer respondents were less lonely in comparison to wealthier individuals. Thus implying that the connection between SES and loneliness is not necessarily direct, but that poverty-related health and wellbeing inequalities may have a substantial influence on loneliness among those in lower socio-economic brackets. This implication may have been less apparent using only employment as an indicator of SES, the conclusion may have been that health and wellbeing were stronger determinants of loneliness rather than considering health as a moderator in the relationship between SES and loneliness. Thus the emphasis placed on the importance of appropriate context-specific SES variables in the literature review (section 3.2.4) is confirmed, but also the results of this thesis would advocate for the use of multiple SES indicators and the inclusion of health-related factors when examining SES and loneliness.

The long-term factors associated with loneliness over the 12-year period also differed from the crosssectional findings. Baseline age was positively associated with, and QoL and life satisfaction at baseline were negatively associated with, loneliness at follow-up, both when controlling for and excluding those lonely at baseline. Age was not significantly associated with loneliness in the fullyadjusted models of any of the cross-sectional analyses, typically any significant association attenuated once controlled for health and wellbeing. This is likely because poor health is often more common with older age groups, therefore the inclusion of health in the cross-sectional analysis likely accounts for the positive association between age and loneliness. However, in the follow-up analysis, baseline health and wellbeing are controlled for, rather than health status concurrent with the experience of loneliness. Thus, because baseline health was not associated with later loneliness, this implies that **poor** health may be more likely to co-occur with loneliness, not precede over such a long period, therefore age may be reflective of later health decline which is then linked to loneliness. To clarify, the long-term positive association with age may only be significant because health data were not included at follow-up, which is likely to have attenuated the significant result, as was evident in the cross-sectional analysis. Furthermore, the results suggested that over a 12-year period, wellbeing (QoL and life satisfaction) were stronger long-term predictors of loneliness than health. This suggests that wellbeing could have long-term effects on loneliness, whereas the evidence does not suggest the same for health. Cross-sectional results found wellbeing was negatively associated with loneliness, and loneliness was negatively associated with wellbeing, indicating that poor wellbeing and loneliness may also be comorbid.

When considering the results over all waves (section 6.2.3), QoL continued to be negatively associated with loneliness, but no other health or wellbeing measure was significant. As baseline depressive symptoms were not significant in predicting loneliness 12-years later, nor over a shorter-period (loneliness occurring in any wave, but after care provision) this suggests a simultaneous experience of depression and loneliness, which highlights a need for immediate intervention for both

caregivers and non-caregivers with depressive symptoms as they demonstrate higher likelihood of experiencing loneliness at the same time. Nevertheless, this further supports the conclusion that wellbeing is a stronger long-term predictor of loneliness than health. Additionally, age was not significantly associated with loneliness in RQ3b, indicating that age had significant long-term impact on loneliness, but may not be associated with loneliness over a shorter time-period. As previously discussed, this significant relationship may be attributed to age-related health changes which occurred between baseline and follow-up.

## 7.4 Health and wellbeing consequences of loneliness

One of the key findings from applying SET as a health theory was the significant negative association between loneliness and frequent alcohol consumption. The interpretation that increased frequency of alcohol consumption actually represented socialisation, more than the impact of alcohol itself, was further supported by the positive association between QoL and alcohol consumption. Evidence from Dunbar et al. (2017) corroborates these findings, they found those who drank frequently, particularly those drinking in the same establishment ('their local'), experienced substantial social benefits and enhanced wellbeing compared to non-drinkers. Alternatively, the impact of loneliness on alcohol consumption was different for caregivers and non-caregivers, in that caregivers were overall less likely to drink frequently, but loneliness among caregivers was positively associated with frequent alcohol consumption. This implies that, for caregivers, alcohol may be a coping mechanism. There is little evidence on the connection between caregiving and alcohol, but no literature was found on alcohol consumption among lonely caregivers. Evidence from Gottschalk et al. (2020) demonstrated caregivers had more responsible drinking habits, but did not examine the additional impact of loneliness, as within this thesis. Moreover, Rospenda et al. (2010) suggested that alcohol was consumed by caregivers due to stress from caregiving; their results indicated social burden (relationship strain outside of the caregiving role) and emotional burden (negative emotions towards the care recipient) were the main factors related to alcohol consumption among caregivers. Particularly, social burden could be linked to experiences of loneliness, which supports the findings from this thesis that lonely caregivers were more likely to report frequent alcohol consumption.

Among the full sample, loneliness was associated with all three SET health pathways prior to controlling for wellbeing; positively associated with poor physiological health, and negatively associated with frequent alcohol consumption, and neutral or positive health expectations, suggesting that wellbeing was a stronger predictor of health after accounting for loneliness. However, given that the CASP19 measure for QoL contains several health-related questions, a strong relationship could have been expected. Furthermore, after controlling for wellbeing (QoL and life satisfaction), loneliness was negatively associated with reporting LLTIs. This suggests that wellbeing may account for the previously observed negative association between LLTIs and loneliness, which also implies that the connection between wellbeing and health may be stronger than that of loneliness and health.

The findings identify strong relationships between wellbeing and loneliness, and loneliness and wellbeing, and highlight the existence of simultaneous relationships between health and wellbeing. However, the relationships between loneliness and health, and health and loneliness, appear to be largely influenced by wellbeing. Further analysis is required to clarify the relationships between loneliness and health in a longitudinal context and to examine the mediating effects of wellbeing.

Finally, the caregiving-loneliness interaction suggested that the impact of loneliness on health and wellbeing often differed for caregivers and non-caregivers. Where caregivers appeared healthier than non-caregivers, a healthy caregiver effect was proposed, in that caregivers may be able to provide care because their health permits them, those with poorer health may be less likely to demonstrate a capacity to care. Moreover, with the example of LLTIs, it could be considered that non-lonely noncaregivers may have less of an awareness of their health in comparison to caregivers, therefore illnesses may have been under-reported in this group. Whereas in contrast, because loneliness is known to be linked to poor health, non-caregivers who were lonely may become increasingly aware of their health and therefore more likely to report LLTIs. In contrast, the combined effect of caregiving and loneliness was indicative of greater negative consequences for mental health and wellbeing. These findings were consistent with the caregiver-only analysis. The implications of these results are that the impact of caregiving and loneliness appear to have greater impact on mental health and wellbeing in comparison to the physical health measures. This aligns with previous discussion in this thesis, that the relationship between loneliness and wellbeing is stronger than that of loneliness and health. Moreover, these findings expand on previous literature which indicates that, separately, caregiving (section 3.4.2.2) and loneliness (section 3.4.1) are positively associated with negative health and wellbeing outcomes, by highlighting a greater impact on health and wellbeing when caregiving and loneliness are combined.

### 7.5 Limitations

One of the main considerations of this section is justifying the decision to conduct the analysis using ELSA data. As identified in section 4.2.2.1, ELSA was deemed more appropriate than comparable UK surveys (e.g. UKHLS), due to the longitudinal availability of loneliness data. The advantages of ELSA were the cross-sectional and longitudinal capacity to examine caregiving, loneliness, health, and wellbeing, with little variance in the measurement tools between the waves. However, there are recognised limitations to the ELSA dataset, which will be addressed in relation to the variables (section 7.5.1), the sample (section 7.5.2), and methods (section 7.5.3).

#### 7.5.1 Variable limitations

Within this thesis, caregiving was measured based on provision during the past week; in early waves (2-5) there was a prerequisite question about caregiving in the past month (section 4.3.1), which was not consistent with later waves. One limitation with caregiving over the past week, is that there may

be a likelihood of undetected caregivers providing care less often (e.g. less than once per week), or those who had not provided care recently (i.e. more than a week ago). But debatably, this demographic of caregivers may firstly, be less likely to experience the negative outcomes related to caregiving due to a reduced intensity, and secondly, for those who had provided care, but not recently, it would be difficult to relate the caregiving to other variables. As such, in cross-sectional analysis, the collected data are occurring within the same time-point, if the respondent was lonely at the point of interview, but had provided care over a month ago, these are, arguably, not occurring simultaneously. In contrast, in the CLS, care responsibilities were not measured exclusively, and were categorised under informal volunteering (Gov.UK, 2020), which is less frequent than ELSA, and no supplementary data was available on caregiving intensity and recipients. Moreover, HSE only included caregiving responsibilities in the most recent wave (2019) (NHS Digital, 2019), which would restrict any longitudinal analysis from being conducted.

Another limitation that arose from the caregiving variable, was the inability to distinguish between short- and long-term caregivers in the cross-sectional analyses. While this was computed in the longitudinal analyses by checking the caregiving variable across all waves, there were no data collected on the duration of care provision for use cross-sectionally. Additionally, the method used to compute duration of care accounts only for those providing care within the week of data collection at each ELSA wave, therefore it is unknown whether care provision continued consistently, or intermittently during the two-year period between each wave. Furthermore, as described in section 4.3.8.2, the loneliness occurring after care provision variable was computed for the regression analyses in RQ3b and RQ3c. The implications that arose from the creation of this variable, such as the exclusion of lonely caregivers in wave two and any caregivers in wave eight, resulted in further reductions to the sample size. Additionally, due to the time gaps between data collection waves (i.e. two-years), it was not possible to control for changes that occur between reporting caregiving in one wave, then loneliness in the next. For example, other lifestyle changes may have occurred, such as bereavement which could contribute towards loneliness more so than the caregiving, so while caregiving occurred chronologically before loneliness, it cannot be guaranteed that the caregiving was the sole contributor.

Limitations could also be identified for some of the health variables, namely LLTI and alcohol consumption. As such, for LLTI, as a self-reported long-standing illness, there is potential for significant discrepancies between respondent interpretations. As no further information was collected on what this illness was: what respondent A considers to be a limiting illness may be considerably different to respondent B. Additionally, no guidance was given regarding 'long-standing', what one recipient might report after a month, another might not report until a year. The results from RQ1 also suggested that it is not explicitly clear what the LLTI variable represents. As such, LLTI was negatively associated with QoL and life satisfaction, but was also negatively associated with both directly reporting loneliness and UCLA-LS. Furthermore, LLTI was positively associated with non-caregiver loneliness prior to controlling for wellbeing, then negatively associated once controlling

for wellbeing. Simultaneously, loneliness was linked to higher odds of LLTI until controlled for wellbeing, then the odds were significantly lower. This could indicate an interaction effect between wellbeing and LLTI, which would require further investigation to fully understand the variable.

Alcohol consumption was determined by how often the respondent had an alcoholic drink in the past 12 months. When asking respondents to recall behaviour over such a long period, there will be an element of recall bias, or particularly for sensitive topics such as alcohol consumption, respondents may provide a more socially desirable answer. It could be considered that frequency of alcohol consumption may not reflect poor behavioural health in the same way that quantity of alcohol might, as such, one respondent might more drink frequently, but in small quantities, and another might drink on fewer occasions, but in considerably larger amounts. Data were collected from around 53% of the total sample in wave eight on quantities of alcohol consumed over the past week, but in previous waves (e.g. wave two), the data collected on alcohol consumption differed substantially, asking about specific beverages (sherry, spirits, beer, and wine) but this was based on their 'heaviest' day and would not be comparable with the wave eight data. Moreover, generalising alcohol consumption based on one 'heavy' day in the past week would arguably not accurately reflect typical alcohol consumption in the way that average frequency over a year would. Similarly, limitations arose by using only alcohol consumption to reflect behavioural health, rather than a range of behavioural health factors for example smoking and exercise. As identified in section 4.3.3.1 problems arose in the analyses when including an aggregated behavioural health variable. However, future research might consider focussing solely on behavioural health aspects, excluding variables such as LLTI and physiological health to allow for a more in-depth analyses of the relationships between loneliness, caregiving and behavioural health. Additionally, in this analysis, the alcohol variable may have been a proxy for social interaction, as the correlations with loneliness might suggest. Therefore, the frequency of alcohol consumption, although largely not representative of typical health-related alcohol measures, was still a valid variable to analyse alongside loneliness.

Additional variable limitations have been highlighted throughout the thesis, particularly in relevant methodology sections. For example, within the physiological health variable, which was computed to represent CV conditions relevant to SET, dementia was incorporated due to evidence that dementia can be CV-related (section 4.3.3.1). However, it could be argued that dementia should not have been included as it is not exclusively a CV condition. To align with other SET research, CV responses such as specific heart rate and blood pressure measurements could have been analysed alongside CV conditions. However, further constraints with the ELSA data restrict the ability to include this information. While nurse data were collected in wave eight, including a range of blood pressure readings, this was only available for a subset of the sample (N=3525), around 42% of those who completed the main questionnaire (N=8445) (ELSA, no date [c]), which would have been largely detrimental to the sample size within this thesis.

Limitations also arose for the social network control variables, whereby only the size of the network could be accounted for rather than other important aspects such as relationship quality. Relationship quality was highlighted as an important factor influencing loneliness, but other than identifying individuals that the respondent was 'close' to, or shared a household with, no further inference could be drawn about the quality of each relationship and no further data were available. Other studies that examined relationship quality (e.g. Hsieh and Hawkley, 2018) considered elements of relationship quality such as support and strain. Typically these would refer to spousal relationships, but collecting data on strain within caregiving relationships would provide insight towards levels of burden experienced by caregivers.

Finally, within this thesis it was not possible to capture information on caregiving burden. As proposed in the conceptual framework, experiences of burden are likely to mediate caregivers feelings of loneliness. It could be hypothesised that those reporting higher levels of burden would feel more isolated and unsupported in their caregiving role and thus more likely to be lonely. However, no formal measures of burden are included in the ELSA data. It was intended that feelings of achievement within the caregiving role and appreciation from others would be applied as proxy measures of burden, but the small proportions of caregivers feeling under-appreciated or dissatisfied with achievement, restricted the inclusion of this variable in any of the analyses (section 5.3.1.4). Thus constituting a limitation within this thesis, as aspects of burden could not be examined or controlled for in the regression models.

#### 7.5.2 Sample limitations

As identified in section 4.3.5, a lack of ethnic diversity is a significant limitation of the ELSA dataset, the majority (>95%) of the total sample are White, and all other ethnic categories are combined as non-White. This limits the generalisability of the results to White England residents aged  $\geq$ 50. Although evidence from 2011 Census data suggests that in England, 93% of over 50s were of White ethnicity (author's own calculation from Nomis, 2011), therefore the ELSA sample may be reflective of the general population, but the small number of ethnic minorities grouped together leave limited scope for analysis.

Furthermore, as discussed throughout the literature review (chapter three) and caregiving theories (chapter two), caregiving appraisal, including aspects such as satisfaction and burden are key elements when researching the impact of caregiving. Depicted in Table 20 (section 5.3.1), only 3% of caregivers were dissatisfied with what they had gained from caregiving, and 8% did not feel adequately appreciated. Although not typical caregiver appraisal measures (e.g. Zarit burden interview), these variables were to be considered a proxy measure for burden. However, the small sample sizes meant that conducting robust statistical analyses was not possible, therefore the current results could not be controlled for burden and the implications of burden within the ELSA caregiving sample could not be researched. To examine the impact of caregiver burden on loneliness, health and

wellbeing, an alternative data source would need to be used, or additional questionnaire items included in later waves of ELSA. Finally, when computing the change variables (section 4.3.8) for RQ3c, the majority of caregivers did not change their co-residential status, intensity, or recipient number between the measured waves. However, this does not necessarily mean unequivocally that no changes occurred. For example, ELSA data are collected every two-years, therefore several changes may have taken place between data collection points. Moreover, due to caregivers providing care to multiple recipients, and the complexity of the variable over the multiple waves, it was not possible to accurately compute changes in care recipient, therefore the results from RQ3c regarding changes should be interpreted with caution.

#### 7.5.3 Methodological limitations

Due to the CCA approach, in some cases, large percentages of the samples were excluded from analyses due to data being missing or incomplete. As such, as high as 64% of the sample were removed from RQ3c (section 6.2.4). However, on average, the percentage excluded were much lower, as few as 8% were excluded from a regression analysis in RQ1c (section 5.2.4.7). Nevertheless, in all circumstances, the sample sizes were deemed sufficient enough to conduct the proposed analyses. Alternative methods were considered, for example imputation, however, the lack of certainty around the mechanisms behind the missing data, i.e. at random or not at random, raised concerns of bias when applying imputation methods. Furthermore, arguably, due to the subjective nature of some of the variables (e.g. expectations for loneliness and health), imputation could be considered less appropriate, in that other variables within the model may not be able to accurately predict an individual's opinion.

Another methodological limitation was the treatment of all control variables in the follow-up analyses as time-invariant. This method is common in similar research using the same data source and is typically employed to avoid overly complex regression models. In RQ3a and RQ3b, all controls were measured at baseline, this meant that changes to variables such as cohabitation or employment were not accounted for. For example, those who were cohabiting, or in employment at baseline may have divorced or retired during the period between baseline and follow-up, which likely would contribute to loneliness. Future research examining the longer-term effects of caregiving on loneliness should aim to account for changes to sociodemographic and socioeconomic variables in addition to changes to mental health and loneliness expectations (as in RQ3c). This identified limitation may explain the lack of significance between control variables and the outcome variable, in that responses provided to, for example, cohabitation or health status, 12-years prior to the follow-up were no longer accurate or relevant.

#### 7.5.4 Limitations summary

Overall, although there are some limitations to the ELSA dataset, in comparison with other UKbased datasets (e.g. UKHLS, HSE, CLS) it was the most suitable for conducting the proposed analysis. Furthermore, the identified limitations with selected variables are likely to be evident in most SDA projects due to the authors' inability to design data collection tools to fit their research questions exactly. This, however, is outweighed by the advantages of SDA (section 4.2.1.1), whereby a large representative sample is readily available for analysis. Quality of data could be improved by alterations to ELSA variables in later waves, but broadly, when using self-completion questionnaires the interpretation of respondents will likely always differ and therefore the clarity of more generic questions, such as LLTI could always be questioned. In contrast, the methodological limitations could be addressed in future research, applying imputation methods, and accounting for different change variables, or time-varying controls to determine the effects on the overall results.

### 7.6 Conclusion

This thesis addressed three key research questions. Firstly, how does loneliness differ for caregivers and non-caregivers? Overall, the findings indicated that caregiving was positively associated with loneliness, but the loneliness measure was important; caregivers were more likely to be lonely on the UCLA-LS but not to directly report loneliness. Moreover, the determinants of loneliness were different for caregivers and non-caregivers, for example the relationships between health and loneliness were significant among non-caregivers but not caregivers. Due to the large discrepancy in determinants identified in RQ1, it was hypothesised that caregiving-specific variables played a larger role in explaining caregivers' loneliness. However, the caregiver results (RQ2) confirmed the importance of sociodemographic and socioeconomic characteristics, sex and employment as determinants, and only the inter-personal relationship between caregiver and care recipient was found to significantly impact loneliness, not co-residential care, caregiving intensity or number of recipients.

Secondly, how did caregiving affect loneliness, health, and wellbeing? Caregivers demonstrated reduced odds of poor physiological and behavioural health, which have been discussed throughout as a healthy caregiver effect. However, the odds of depressive symptoms were considerably higher for caregivers, and they experienced significantly lower QoL and life satisfaction. This suggests that the impact of caregiving may be more substantial for mental health and wellbeing than for physical health.

Finally, what were the longer-term effects of caregiving on loneliness? Over a 12-year period caregiving at baseline was not significantly associated with loneliness at follow-up, which suggests that the impact of caregiving is acute rather than chronic, highlighting a need for early intervention to prevent the possibility of detrimental effects on health and wellbeing that were indicated in the

cross-sectional analysis. However, further research is essential to determine the longer-term effects of loneliness on health and wellbeing among caregivers. There was a strong positive association between previous loneliness and loneliness at follow-up, indicating that loneliness is likely to endure or reoccur over an individual's lifetime. Again, this provides a precedent for early intervention, or prevention strategies for loneliness, as initial development of loneliness becomes a risk factor for loneliness in the future.

Largely, the findings in this thesis aligned with the conceptual framework proposed initially (section 2.2), whereby relationships were evident between caregiving and loneliness directly and through other factors, and both loneliness and caregiving impacted health and wellbeing, including through the interpretation of the SET pathways. However, as discussed throughout the thesis, caregiving was often negatively associated with poor health outcomes, which were interpreted as a healthy caregiver effect. In essence, caregivers may have better health initially, permitting them to undertake a caregiving role, rather than caregiving having a positive impact on health. An additional arrow has therefore been included in the conceptual framework (Figure 29) to demonstrate this effect, but further analysis (detailed ahead) would be required to clarify this theory.



Figure 29: Amended conceptual framework Source: author's own

## 7.6.1 Original contribution

As outlined initially in section 3.5.1.1, this thesis makes a significant contribution to current caregiving and loneliness knowledge in several ways. Firstly, the results demonstrate that determinants of loneliness are substantially different for caregivers than non-caregivers, as identified in the research gaps (section 3.5.1), there was limited investigation into the determinants of loneliness for caregivers. Victor *et al.* (2020) also highlighted this limitation and contributed by examining prevalence and predictors of loneliness in dementia caregivers; however, this thesis expands on those findings by conducting a direct comparison of loneliness determinants in caregiving and non-caregiving samples and including a greater range of caregivers, i.e. not limited to dementia

caregivers, and providing a more detailed division of care recipients. The comparison with noncaregivers emphasises the substantial difference between the two subgroups and highlights a need for further research into the determinants of loneliness among caregivers to clarify the current findings.

Additionally, evidence suggested that the reliability and validity of loneliness measures differs by sociodemographic characteristics, such as age and sex (Pinquart and Sörensen, 2011; Nicolaisen and Thorsen, 2014a), but there is no evidence attesting to the use of direct or indirect loneliness measures among caregivers. Research on loneliness among caregivers uses a range of direct and indirect measures, including common scales such as UCLA-LS and DJGLS with differing results. The results from this thesis confirm that among the same sample of caregivers, loneliness prevalence significantly differs depending on the measure applied. This identifies that it is not sociodemographic characteristics alone, that influence the likelihood of directly reporting loneliness, but that further consideration should be made into which loneliness measure is used within caregiving samples.

Finally, as discussed within the literature review (section 3.4.2), literature tends to examine the impact of caregiving on loneliness, health, and wellbeing separately, despite a large body of evidence in non-caregiving populations (section 3.4.1) identifying connections between loneliness and health, and loneliness and wellbeing. Little was known about the impact of loneliness on health and wellbeing among caregivers, and more research is required to identify the impact of health and wellbeing on loneliness among caregivers (section 7.6.3). This thesis highlighted through use of a caregiving-loneliness interaction that the relationships between loneliness and alcohol consumption, depressive symptomatology, LLTI, QoL and life satisfaction differed for caregivers and non-caregivers. Moreover, because the relationship between loneliness and health, and loneliness and wellbeing differ for caregivers, loneliness, health, and wellbeing should be researched simultaneously rather than generic health and wellbeing implications of loneliness from non-caregiving populations being applied to caregivers.

#### 7.6.2 Contribution to theory

CIT demonstrates how a caregiver's identity changes throughout the caregiving role; as the role becomes more complex, the caregiver identifies less with the original relationship they once held with the recipient, and more with their role as caregiver (section 2.1.1.1). While identity could not be directly examined in this thesis, due to limitations within ELSA data, it was considered that caregiving intensity may serve as an appropriate alternative, representing caregiving context (Montgomery and Kosloski, 2009). As such, those providing higher-intensity, or more specifically round-the-clock care are more likely to be in later phases of CIT whereby they would be expected to associate more as a caregiver than a spouse, for example. In relation to this thesis, it was postulated that those in later CIT phases may experience heightened loneliness due to the feeling of losing the significant attachment (emotional loneliness) that was present in the pre-caregiving relationship.

Largely, the results from this thesis demonstrate that CIT is applicable to caregiver loneliness. In the cross-sectional, caregiver-only analysis (section 5.3.2.1), although caregiving intensity was not significantly associated with loneliness, care provision to an adult-child was positively associated with loneliness. As discussed previously (section 7.2), care to adult-children is known to be of higherintensities, therefore it is possible that the relationship between caregiver-recipient relationships and loneliness accounts for some of the relationship between intensity and loneliness. Additionally, caregivers to adult-children may find that they associate less with the parental relationship, and more with being a caregiver, as their child reaches adulthood; this would be when typical child-care tasks usually cease, but for these individuals care requirements continue, if not escalate. Thus caregivers to adult-children may be representative of caregivers in later phases of CIT. With regard to the longitudinal results, the provision of long-term care was significantly positively associated with loneliness, whereas intermittent care was non-significant (section 6.2.3.1). This could indicate caregivers moving through the phases of CIT. Those providing short-term care, or intermittent care may be less likely to experience contextual changes within the caregiving role, due to the shorter time-periods; thus their identity as a caregiver, and the caregiver-recipient relationship, remain relatively stable. Finally, in section 6.2.4.1 'ever providing' high-intensity, or round-the-clock care was positively associated with loneliness. While the caregivers may not currently be providing this level of care, the phases of CIT are not exhaustive, in that caregivers can experience the phases in different orders. Caregivers who had previously, or were currently providing, higher-intensity care represent the later phases of CIT and demonstrated higher odds of loneliness. The literature on CIT highlights implications for supporting caregivers (Montgomery, 2009), but the focus is on the congruence of identity, not instrumental support with their role. However, interpretations from this thesis imply that the CIT phases may be useful, if implemented in caregiving support services. For example, as a self-rated monitoring tool to observe changes to caregiver identity, and highlight those at higher risk of loneliness, and potentially other detrimental health and wellbeing consequences (later CIT phases) to relevant service providers.

Moreover, SET provides a health framework (section 2.1.2), indicating that internalised stereotypes can have consequences, in the form of self-fulfilling prophecies, for physiological, psychological, and behavioural health pathways (Levy, 2009; Meisner and Levy, 2016). Throughout this thesis health measures have been included for all three pathways. However, caregiving specifically does not appear to be strongly linked to increased likelihood of internalised negative stereotypes. Crosssectionally, caregiving was not significantly associated with loneliness expectations (section 5.2.2.2) nor health expectations (section 5.2.4.3). Alternatively, caregiving was linked to more positive outcomes related to physiological and behavioural health pathways and the caregiving loneliness interaction indicated that the relationship between loneliness and behavioural health was significantly different for caregivers and non-caregivers. As discussed throughout the thesis these results have been attributed to a healthy caregiver effect, or for example, the use of alcohol as a coping mechanism for caregivers, *versus* alcohol as a social activity for non-caregivers.

This thesis examined health and loneliness related age-stereotypes on loneliness. There was evidence of longer-term impacts of stereotypes and expectations particularly for loneliness. Longitudinally, among the full sample negative loneliness expectations (disagree) at baseline were positively associated with loneliness 12-years later. However, the results for health expectations were less clear when linked to loneliness outcomes, further research is required to determine the long-term effects of health expectations on health-related outcomes. Additionally, there was evidence of long-term self-fulfilling loneliness prophecies among the caregiving sample. Negative loneliness expectations at baseline were positively associated with loneliness occurring after the start of care provision, and changes to loneliness expectations, both positive and fluctuating, were negatively associated with later loneliness. This would suggest that any form of positive expectation may have the ability to counteract previous negative beliefs. While SET provided a clear framework for examining health outcomes among caregivers and non-caregivers, the results from this thesis do not provide enough evidence to indicate a strong affiliation between caregiving and all of the SET pathways. To clarify, from the caregiving-loneliness interaction analyses, only the impact of loneliness on behavioural health was found to differ between caregivers and non-caregivers. This is applicable only to alcohol consumption and further research should be conducted to examine other aspects of behavioural health before drawing further conclusions about the connection between caregiving, loneliness, and SET. Finally, the occurrence of negative age-stereotypes and self-fulfilling loneliness prophecies appeared common among both caregivers and non-caregivers. Additional research examining loneliness expectations as an outcome, with caregiving-specific factors as explanatory variables would aid to explain if caregiving influenced the development of specific age-stereotypes.

#### 7.6.3 Recommendations for policy and further research

Several policy recommendations can be deduced from the results of this thesis. Namely, the use of SET as a health theory identified specific areas of health associated with caregiving and loneliness among caregivers. This allowed for more tailored policy recommendations regarding the health of caregivers. The impact of caregiving on alcohol consumption differed depending on whether they were lonely or not. It is considered that this increased frequency of alcohol consumption among lonely caregivers is likely to be a coping mechanism as evidence has previously linked caregiver burden with unhealthy drinking behaviours (Rospenda *et al.*, 2010). This, therefore, is a specific area of behavioural health shown to be particularly relevant in caregiving populations; caregivers should have access to support and guidance promoting alternative, healthy coping strategies. For example, signposting towards relevant support groups, and access to professional services. These may help address specific issues such as alcohol consumption, or more generally, for concerns they have about their caregiving role.

A recent briefing paper on informal care (Powell *et al.*, 2020) suggested that caregivers who were more likely to experience physical ill health were those providing round-the-clock care, palliative care, or caregivers with socioeconomic difficulties. However, there was limited detail as to what
#### Chapter 7

physical ill health entailed. Whereas, the results of this thesis would suggest that specific areas such as physiological health are particularly relevant to caregiving samples, but broader measures such as LLTI were not significantly associated with caregiving. A 2018/19 NHS mandate detailed that caregivers were to be identified and given access to information and advice about support (Powell et al., 2020). However, this support appears limited, as in the case of caregivers to disabled children, their local authorities needed only consider services the caregiver is assessed for (*ibid*). Thus, if the advice and support to caregivers is provided around more general health, and caregivers are assessed using broad health measures, rather than more specific areas such as physiological or behavioural health (e.g. alcohol consumption) as identified, these support services are likely to be ineffective. It is recommended that caregivers have a more thorough health screening to identify which services should be recommended. Alternatively, the results from this thesis suggested that caregivers of adultdependent children were more likely to be lonely than spousal caregivers, additionally, caregiving as a whole was only significantly positively associated with UCLA-LS loneliness, not the direct question. Relating these findings to the aforementioned policy, if the caregivers' loneliness is assessed by a direct measure, and the individual is therefore not identified as needing loneliness support, the advice and support services would not be considered. It is recommended that when caregivers are identified, they are given access to information about all of the available support services and how to access these, regardless of an assessment. This would give the caregivers all of the available knowledge and reduce the need for caregivers to be reassessed each time that support may be needed.

Moreover, in the carers action plan 2018-2020, research was outlined for areas including the cost and benefit of informal care, the experiences of caregivers and factors that might influence propensity to care (Department of Health and Social Care (DHSC), 2018). Moreover, it was identified that caregivers should be considered in the cross-government loneliness strategy: they were identified as a vulnerable group for loneliness and £500,000 funding was pledged to build caregiver-friendly communities, services, and interventions via the Carer Innovations Fund (DCMS, 2018). This funding was extended to £5 million, to support projects with positive effects on health, wellbeing, employment, loneliness, and respite (DHSC, 2019), but was later suspended in 2020 with the intention to "support carers via other means in due course" (Carers UK, 2020, paragraph 1). Additionally, in the Loneliness Annual Report, published January 2020, little update was provided on the support for caregivers other than work from Sport England and Carers UK, targeting barriers around sport and physical activity for lonely caregivers (HM Government, 2020). While physical activity is arguably important for loneliness, as such, Shankar et al. (2011) identified that lonely individuals were less likely to be physically active. This is not limited to caregivers, as evidence from Gottschalk et al. (2020) found caregivers to be more physically active. In contrast, both physiological health and alcohol consumption were found, in this thesis, to be related to caregiving, alongside depressive symptoms and QoL. Therefore, it is recommended to reinstate funded projects to support caregiver loneliness, health, and wellbeing through methods aside from physical activity.

#### Chapter 7

Furthermore, to expand upon the contribution of this thesis, recommendations can also be made for further research. Such research could determine the long-term effects of caregiving on health and wellbeing outcomes to supplement the information gathered on long-term effects of caregiving on loneliness. Moreover, within this thesis, wave two health and wellbeing variables were used as baseline controls for wave eight loneliness. To further disentangle the relationships between caregiving, loneliness, health, and wellbeing, additional longitudinal research could be conducted using previous loneliness and wellbeing to examine future health, and previous loneliness and health to determine future wellbeing in caregiving samples. This would clarify the directions of the relationships and aid researchers to map out which aspect occurs first, i.e. loneliness, poor health, or reduced wellbeing.

Furthermore, there was limited evidence from these analyses to suggest that loneliness persisted 12years after care provision, additional research using alternative methods, for example, survival analysis, could determine the length of caregiving effects. To clarify, this could estimate how long after initiating caregiving roles, the onset of loneliness or poor health and wellbeing might occur. Alternatively, similar research could be conducted to identify how long these consequences persist. Finally, as discussed previously, a healthy caregiver effect may have been observed in several of the analyses, additional longitudinal analysis examining the health and wellbeing of caregivers prior to undertaking a caregiving role, for example, examining health and wellbeing in the wave prior to the first wave of care, would aid to confirm this theory.

#### **Appendix A** Confirmation of ethical approval

Thu 02/11/2017 13:31 ERGO <ergo@soton.ac.uk> Your Ethics Submission (Ethics ID:30736) has been reviewed and approved

To Davison L.

Submission Number: 30736 Submission Name: Carers and loneliness in the English Longitudinal Study of Ageing (SDA - Secondary Data Analysis) This is email is to let you know your submission was approved by the Ethics Committee.

Comments None <u>Click here to view your submission</u> Coordinator: Lisa Davison

ERGO : Ethics and Research Governance Online http://www.ergo.soton.ac.uk

DO NOT REPLY TO THIS EMAIL

# Appendix B Bivariate associations between sociodemographic characteristics, socioeconomic factors, social networks, loneliness, health, wellbeing and the UCLA-LS and direct loneliness question in the full sample

	$\frac{\text{UCLA-LS}}{\sqrt{2}}$ df S			Di	rect loneline	ess
	$\chi^2$	df	Sig.	$\chi^2$	df	Sig.
Age	15.80	3	.000	27.68	1	.000
Sex	4.68	1	.030	17.78	1	.000
Cohabitation	451.16	1	.000	606.98	1	.000
In paid employment	19.86	1	.000	62.72	1	.000
Wealth	181.94	3	.000	147.32	1	.000
Household size	230.01	2	.000	412.46	2	.000
UCLA-LS				1339.95	1	.000
Loneliness expectations	417.12	2	.000	327.28	2	.000
LLTI	135.27	1	.000	142.63	1	.000
Physiological health	21.62	1	.000	39.17	1	.000
Health expectations	69.34	2	.000	47.24	2	.000
Alcohol consumption	39.02	1	.000	26.62	1	.000
CES-D	632.37	1	.000	1064.69	1	.000
	r		Sig.	r		Sig.
Close relationships with family	-0.11	.000		-0.06	.000	
Close relationships with friends	-0.11	.000		-0.09	.000	
CASP19	-0.44	.000		-0.31	.000	
SWLS	-0.42	.000		-0.30	.000	

Source: author's analysis of ELSA wave 8

#### Appendix C RQ1a is caregiving associated with directly reporting loneliness?

		Model 1			Model 2			Model 3			Model 4	
Variables	$E_{vp}(\mathbf{P})$	95%	6 CI	$E_{vp}(\mathbf{P})$	95%	6 CI	$E_{VD}(\mathbf{P})$	95%	6 CI	$E_{VD}(\mathbf{P})$	95%	6 CI
	Ехр(Б)	Lower	Upper	Ехр(Б)	Lower	Upper	Exp(B)	Lower	Upper	Exp(B)	Lower	Upper
Constant	0.102***			0.439***			0.682			4.216**		
Caregiving	0.921	0.662	1.279	1.041	0.732	1.482	1.130	0.790	1.615	0.829	0.552	1.246
Age												
60-69 years				0.839	0.588	1.199	0.831	0.577	1.198	1.095	0.725	1.653
≥70 years				0.714	0.484	1.054	0.700	0.465	1.054	0.964	0.601	1.549
Sex, female				0.844	0.648	1.098	0.870	0.667	1.135	0.763	0.562	1.036
Cohabitation				0.136***	0.103	0.180	0.162***	0.109	0.241	0.254***	0.161	0.399
In paid employment				0.552**	0.391	0.779	0.543**	0.383	0.769	0.898	0.603	1.336
Wealth												
Quartile 2				1.052	0.755	1.465	1.125	0.804	1.574	1.895**	1.275	2.816
Quartile 3				0.816	0.570	1.169	0.866	0.602	1.245	1.926**	1.250	2.967
Quartile 4				0.684	0.459	1.019	0.748	0.499	1.122	1.761*	1.094	2.834
Household size												
Dual-person household							0.710	0.473	1.066	0.507**	0.318	0.809
Large household ( $\geq$ 3)							0.811	0.496	1.325	0.467*	0.261	0.835
Number of close relationships (family and child)							0.942*	0.895	0.991	0.965	0.914	1.019
Number of close relationships (friends)							0.918**	0.863	0.977	0.952	0.890	1.017
LLTI										0.711*	0.506	0.999
Physiological health: chronic CV condition(s)										1 1 3 9	0.837	1 550
Health expectations										11107	0.007	11000
Neither agrees nor disagrees										0.772	0.483	1 234
Disagrees										1.020	0.405	1.234
Alcohol consumption >3 days per week										0.866	0.615	1 218
CES D: depressive symptomatology										7.0/1***	5.058	0.801
CASP10										0.056***	0.022	0.000
										0.930****	0.933	0.980
SWLS										0.952**	0.923	0.982

Notes: binary logistic regression, N=3236, DV – direct loneliness question; \*p<.05, \*\*p<.01, \*\*\*p<.001; source: author's analysis of ELSA wave 8

### Appendix D RQ1a is caregiving associated loneliness on the UCLA loneliness scale?

		Model 1			Model 2			Model 3		N	Iodel 4	
Variables	$E_{vn}(\mathbf{R})$	95%	6 CI	$\mathbf{Evn}(\mathbf{R})$	95%	6 CI	$Evp(\mathbf{R})$	95%	6 CI	$E_{vn}(\mathbf{R})$	95%	5 CI
	Ехр(Б)	Lower	Upper	Ехр(Б)	Lower	Upper	Exp(B)	Lower	Upper	Exp(B)	Lower	Upper
Constant	0.189***			0.554***			1.273			682.407***		
Caregiving	1.349*	1.066	1.707	1.454**	1.131	1.869	1.463**	1.132	1.892	1.014	0.743	1.385
Age												
60-69 years				0.800	0.615	1.039	0.844	0.643	1.107	1.151	0.831	1.593
≥70 years				0.635**	0.469	0.859	0.734	0.534	1.010	0.824	0.556	1.222
Sex, female				0.964	0.788	1.179	1.007	0.821	1.236	1.177	0.920	1.505
Cohabitation				0.247***	0.199	0.306	0.167***	0.122	0.229	0.259***	0.178	0.379
In paid employment				0.722*	0.560	0.931	0.725*	0.560	0.940	1.124	0.820	1.540
Wealth												
Quartile 2				0.666**	0.511	0.868	0.692**	0.528	0.908	1.196	0.861	1.662
Quartile 3				0.567***	0.429	0.749	0.593***	0.446	0.787	1.443*	1.020	2.040
Quartile 4				0.602***	0.453	0.800	0.667**	0.498	0.894	1.894**	1.317	2.723
Household size												
Dual-person household							1.640**	1.157	2.324	1.219	0.805	1.847
Large household (≥3)							1.785**	1.196	2.663	1.001	0.620	1.615
Number of close relationships (family and child)							0.934**	0.897	0.972	0.966	0.924	1.010
Number of close relationships (friends)							0.901***	0.858	0.945	0.976	0.927	1.029
LLTI										0.632**	0.478	0.837
Physiological health: chronic CV condition(s)										0.939	0.731	1.207
Health expectations												
Neither agrees nor disagrees										1.169	0.834	1.638
Disagrees										0.994	0.662	1.492
Alcohol consumption ≥3days per week										0.760*	0.578	0.998
CES-D: depressive symptomatology										1.681**	1.253	2.254
CASP19										0.869***	0.851	0.887
SWLS										0.939***	0.916	0.962

Notes: binary logistic regression, N=3213, DV – UCLA-LS; \*p<.05, \*\*p<.01, \*\*\*p<.001; source: author's analysis of ELSA wave 8

Appendix E	<b>RQ1a does careg</b>	iving impact	t loneliness expecta	ations?
------------	------------------------	--------------	----------------------	---------

		el 1			Model 2							
Variables	Neither agree nor disagree			Γ	Disagree		Neither ag	ree nor c	lisagree	Disagree		
	$\mathbf{E}_{\mathbf{v}\mathbf{n}}(\mathbf{D})$	959	% CI	$E_{\rm res}({\bf D})$	95%	6 CI	$E_{\rm res}({\bf D})$	95%	6 CI	$\mathbf{E}_{\mathbf{r},\mathbf{r}}(\mathbf{D})$	95%	6 CI
	Ехр(Б)	LCI	UCI	Ехр(Б)	LCI	UCI	схр(б)	LCI	UCI	Ехр(Б)	LCI	UCI
Caregiving	0.995	0.841	1.177	1.034	0.874	1.223	0.976	0.823	1.158	0.992	0.836	1.177
Age												
60-69 years							1.146	0.968	1.356	1.218*	1.025	1.447
≥70 years							1.026	0.846	1.245	1.294*	1.063	1.575
Sex, female							0.911	0.802	1.033	1.005	0.885	1.142
Cohabitation							1.374***	1.187	1.591	1.643***	1.414	1.909
In paid employment							0.939	0.798	1.104	1.023	0.867	1.207
Wealth												
Quartile 2							1.166	0.973	1.396	1.071	0.895	1.282
Quartile 3							1.147	0.957	1.373	0.941	0.784	1.129
Quartile 4							0.998	0.831	1.198	0.842	0.701	1.012

Notes: multinomial logistic regression, N=3259, DV – loneliness expectations; LCI – Lower confidence interval, UCI – Upper confidence interval;\*p<.05, \*\*p<.01, \*\*\*p<.001, source: author's analysis of ELSA wave 8

#### Abbreviations

#### RQ1a does caregiving impact loneliness expectations? (continued)

			Мо	del 3					Moc	lel 4		
	Neith	ner agree	nor	D	isagree		Neithe	er agree	nor	D	isagree	
Variables	(	disagree			8		di	sagree			8	
	Fvn(B)	959	% CI	Evn(B)	95%	6 CI	Evp(B)	95%	6 CI	Evn(B)	95%	6 CI
	Lxp(D)	LCI	UCI	Lxp(D)	LCI	UCI	Lxp(D)	LCI	UCI	Lxp(D)	LCI	UCI
Caregiving	0.934	0.747	1.166	0.980	0.784	1.224	1.000	0.790	1.266	1.154	0.900	1.481
Age												
60-69 years	1.113	0.890	1.392	1.311*	1.043	1.648	1.022	0.808	1.292	1.213	0.941	1.564
≥70 years	0.957	0.734	1.247	1.315*	1.002	1.726	0.866	0.651	1.153	1.361	0.997	1.856
Sex, female	0.862	0.731	1.018	0.906	0.766	1.071	0.832*	0.696	0.995	0.798*	0.660	0.965
Cohabitation	1.214	0.887	1.661	1.824**	1.294	2.572	1.184	0.840	1.670	1.530*	1.022	2.290
In paid employment	0.881	0.714	1.089	1.061	0.856	1.315	0.742*	0.593	0.930	0.872	0.684	1.110
Wealth												
Quartile 2	1.075	0.847	1.363	0.971	0.765	1.232	0.844	0.650	1.095	0.628**	0.476	0.828
Quartile 3	1.062	0.834	1.351	0.955	0.750	1.215	0.829	0.636	1.081	0.578***	0.437	0.765
Quartile 4	0.850	0.666	1.085	0.712**	0.557	0.911	0.640**	0.487	0.842	0.390***	0.292	0.523
Household size												
Dual-person household	1.335	0.933	1.909	0.891	0.607	1.307	1.206	0.823	1.769	0.933	0.601	1.449
Large household (≥3)	1.276	0.863	1.886	1.022	0.677	1.541	1.236	0.811	1.883	1.200	0.749	1.923
Number of close relationships (family and child)	1.024	0.993	1.056	1.076***	1.045	1.108	1.022	0.989	1.056	1.060**	1.025	1.096
Number of close relationships (friends)	1.046**	1.012	1.081	1.095***	1.061	1.130	1.029	0.995	1.064	1.057**	1.023	1.093
LLTI							1.159	0.934	1.438	1.728***	1.373	2.174
Physiological health: chronic CV condition(s)							1.213*	1.009	1.459	1.126	0.924	1.372
Health expectations												
Neither agrees nor disagrees							1.548***	1.223	1.959	1.711***	1.340	2.183
Disagrees							1.528**	1.127	2.072	3.254***	2.447	4.326
Alcohol consumption ≥3days per week							0.987	0.818	1.191	0.917	0.750	1.120
CES-D: depressive symptomatology							0.517***	0.394	0.680	0.600**	0.438	0.822
CASP19							1.036***	1.020	1.053	1.111***	1.091	1.131
SWLS							0.985	0.966	1.005	0.991	0.969	1.014

Notes: multinomial logistic regression, N=3259, DV – loneliness expectations; LCI – Lower confidence interval, UCI – Upper confidence interval; \*p<.05, \*\*p<.01, \*\*\*p<.001; source: author's analysis of ELSA wave 8

#### **Appendix F RQ1b** what are the determinants of caregivers' loneliness?

		Model 1			Model 2			Model 3		]	Model 4	
Variables	$Evp(\mathbf{R})$	95%	5 CI	$Evp(\mathbf{R})$	95%	6 CI	$Evp(\mathbf{B})$	95%	6 CI	$Evn(\mathbf{R})$	95%	b CI
	Exh(P)	Lower	Upper	Exp(D)	Lower	Upper	Exp(D)	Lower	Upper	Exp(B)	Lower	Upper
Constant	0.189***			0.217***			0.669			269.295***		
Age												
60-69 years	0.601	0.341	1.060	0.461*	0.241	0.882	0.476*	0.240	0.943	0.715	0.319	1.598
≥70 years	0.627	0.355	1.106	0.359**	0.178	0.727	0.426*	0.199	0.914	0.626	0.249	1.576
Sex, female	1.029	0.632	1.676	1.373	0.808	2.331	1.685	0.964	2.945	3.061**	1.487	6.303
Cohabitation	0.565*	0.326	0.979	0.718	0.389	1.326	0.404*	0.176	0.926	0.770	0.270	2.190
LLTI	1.685*	1.012	2.805	1.299	0.754	2.236	1.311	0.747	2.302	0.653	0.322	1.326
Loneliness expectations												
Neither agrees nor disagrees	0.776	0.470	1.279	0.732	0.432	1.241	0.801	0.464	1.382	1.316	0.681	2.543
Disagrees	0.079***	0.032	0.196	0.088***	0.035	0.219	0.100***	0.039	0.255	0.182**	0.061	0.538
Public transport use												
1-3 times per month	0.789	0.448	1.391	0.763	0.414	1.408	0.758	0.402	1.427	0.994	0.474	2.084
1-3 times per week	1.593	0.805	3.155	1.192	0.572	2.484	1.063	0.494	2.284	1.093	0.441	2.711
Every day or nearly everyday	2.022	0.884	4.628	1.092	0.431	2.766	0.727	0.254	2.078	0.583	0.137	2.487
In paid employment				0.238***	0.124	0.458	0.227***	0.113	0.455	0.398*	0.180	0.878
Wealth												
Quartile 2				0.524	0.257	1.069	0.644	0.305	1.364	0.857	0.346	2.126
Quartile 3				0.378**	0.176	0.810	0.448*	0.203	0.988	0.953	0.363	2.498
Quartile 4				0.403**	0.191	0.850	0.522	0.239	1.137	1.405	0.539	3.660
Household size												
Dual-person household							1.597	0.505	5.056	0.729	0.181	2.937
Large household (≥3)							2.273	0.669	7.720	1.188	0.270	5.226
Number of close relationships (family and child)							0.887*	0.790	0.995	0.961	0.852	1.084
Number of close relationships (friends)							0.803**	0.691	0.934	0.843*	0.718	0.990
Physiological health: chronic CV condition(s)										1.627	0.816	3.244
Health expectations												
Neither agrees nor disagrees										0.593	0.228	1.540
Disagrees										1.122	0.360	3.497
Alcohol consumption $\geq$ 3 days per week										2.051	0.990	4.249
CES-D: depressive symptomatology										0.839	0.394	1.788
CASP19										0.854***	0.808	0.902
SWLS										0.943	0.886	1.003

Notes: binary logistic regression, N=503, DV – UCLA-LS; \*p<.05, \*\*p<.01, \*\*\*p<.00; source: author's analysis of ELSA wave 8

#### Appendix G RQ1b what are the determinants of non-caregivers' loneliness?

		Model 1			Model 2			Model 3		I	Model 4	
Variables	$Evp(\mathbf{R})$	95%	6 CI	$Evp(\mathbf{B})$	95%	6 CI	$Evp(\mathbf{R})$	95%	6 CI	$E_{vp}(\mathbf{R})$	95%	o CI
	Exp(B)	Lower	Upper	Exp(B)	Lower	Upper	Exp(D)	Lower	Upper	Exb(p)	Lower	Upper
Constant	0.145***			0.132***			0.192***			199.159***		
Age												
60-69 years	0.958	0.717	1.280	1.071	0.778	1.473	1.139	0.818	1.587	1.416	0.964	2.080
≥70 years	0.707*	0.521	0.959	0.846	0.578	1.238	0.980	0.657	1.462	1.049	0.655	1.678
Sex, female	0.965	0.760	1.227	0.972	0.764	1.237	0.982	0.770	1.252	1.040	0.783	1.380
Cohabitation	0.208***	0.162	0.267	0.227***	0.176	0.294	0.147***	0.100	0.217	0.214***	0.137	0.334
LLTI	1.938***	1.507	2.492	1.864***	1.434	2.424	1.863***	1.431	2.425	0.709*	0.507	0.990
Loneliness expectations												
Neither agrees nor disagrees	0.273***	0.206	0.361	0.269***	0.203	0.356	0.270***	0.204	0.359	0.297***	0.215	0.410
Disagrees	0.120***	0.084	0.173	0.116***	0.080	0.167	0.120***	0.083	0.174	0.210***	0.139	0.318
Public transport use												
1-3 times per month	0.853	0.644	1.129	0.856	0.644	1.136	0.876	0.658	1.166	1.045	0.751	1.455
1-3 times per week	0.907	0.639	1.286	0.929	0.654	1.321	0.922	0.647	1.313	1.033	0.683	1.561
Every day or nearly everyday	1.175	0.772	1.788	1.148	0.752	1.752	1.165	0.760	1.786	1.448	0.885	2.370
In paid employment				1.206	0.871	1.669	1.180	0.849	1.640	1.435	0.974	2.113
Wealth												
Quartile 2				0.751	0.543	1.038	0.753	0.541	1.049	1.004	0.684	1.472
Quartile 3				0.702*	0.501	0.985	0.718	0.510	1.011	1.392	0.933	2.077
Quartile 4				0.650*	0.458	0.923	0.684	0.478	0.979	1.468	0.961	2.244
Household size												
Dual-person household							1.702*	1.123	2.580	1.397	0.872	2.239
Large household (≥3)							2.002**	1.233	3.251	1.215	0.698	2.116
Number of close relationships (family and child)							0.965	0.923	1.010	0.977	0.929	1.027
Number of close relationships (friends)							0.947	0.897	1.000	1.009	0.951	1.072
Physiological health: chronic CV condition(s)										0.861	0.644	1.153
Health expectations												
Neither agrees nor disagrees										1.740**	1.174	2.578
Disagrees										1.415	0.891	2.246
Alcohol consumption ≥3days per week										0.607**	0.443	0.833
CES-D: depressive symptomatology										1.910***	1.351	2.700
CASP19										0.884***	0.862	0.906
SWLS										0.925***	0.899	0.952

Notes: binary logistic regression, N=2701, DV – UCLA-LS; \*p<.05, \*\*p<.01, \*\*\*p<.00; source: author's analysis of ELSA wave 8

Appendix H	<b>RQ1c</b> how does the interaction between caregiving and loneliness affect the
<b>relationshi</b> j	ps between caregiving, loneliness and physiological health?

		Model 1			Model 2			Model 3			Model 4	
Variables	$E_{\rm res}({\bf D})$	95%	6 CI	$E_{\rm res}({\bf D})$	95%	6 CI	$E_{\rm wm}({\bf D})$	95%	6 CI	$E_{\rm res}({\bf D})$	95%	6 CI
	Ехр(Б)	Lower	Upper	схр(б)	Lower	Upper	Ехр(Б)	Lower	Upper	Ехр(Б)	Lower	Upper
Constant	0.760***			0.769***			1.085			4.515***		
Caregiving	0.753***	0.648	0.875	0.699***	0.590	0.829	0.750**	0.626	0.898	0.712***	0.594	0.854
Loneliness	1.442***	1.256	1.656	1.360***	1.169	1.582	1.232*	1.044	1.453	0.881	0.732	1.060
Caregiving-loneliness				1.423	0.983	2.062	1.466	0.996	2.159	1.340	0.905	1.982
Age												
60-69 years							1.631***	1.393	1.909	1.729***	1.473	2.028
≥70 years							3.365***	2.809	4.033	3.383***	2.815	4.065
Sex, female							0.674***	0.600	0.758	0.694***	0.616	0.781
Cohabitation							0.935	0.813	1.075	0.917	0.794	1.059
In paid employment							0.738***	0.636	0.856	0.800**	0.688	0.929
Wealth												
Quartile 2							0.719***	0.608	0.851	0.816*	0.687	0.969
Quartile 3							0.646***	0.546	0.764	0.748**	0.629	0.888
Quartile 4							0.484***	0.407	0.575	0.595***	0.498	0.711
CASP19										0.960***	0.951	0.970
SWLS										1.012	0.998	1.025

Notes: binary logistic regression, N=5275, DV – physiological health chronic CV conditions; \*p<.05, \*\*p<.01, \*\*\*p<.001; source: author's analysis of ELSA wave 8

### Appendix I RQ1c how does the interaction between caregiving and loneliness affect the relationships between caregiving, loneliness and alcohol consumption?

		Model 1			Model 2			Model 3			Model 4	
Variables	$E_{\rm res}({\bf D})$	95%	6 CI	$\mathbf{E}_{\mathbf{rr}}(\mathbf{D})$	95%	6 CI	$E_{\rm res}({\bf D})$	95%	6 CI	$E_{\rm res}({\bf D})$	95%	6 CI
	Exp(B)	Lower	Upper	Exp(B)	Lower	Upper	Exp(B)	Lower	Upper	Exp(B)	Lower	Upper
Constant	0.564***			0.570***			0.639			0.298***		
Caregiving	0.791**	0.674	0.930	0.733**	0.613	0.876	0.765**	0.635	0.921	0.780**	0.648	0.940
Loneliness	0.629***	0.538	0.735	0.586***	0.493	0.696	0.706***	0.587	0.850	0.844	0.689	1.034
Caregiving-loneliness				1.541*	1.021	2.326	1.689*	1.104	2.585	1.790**	1.167	2.744
Age												
60-69 years							1.024	0.873	1.202	0.998	0.850	1.172
≥70 years							0.926	0.768	1.118	0.925	0.765	1.118
Sex, female							0.519***	0.459	0.586	0.509***	0.450	0.576
Cohabitation							1.076	0.925	1.252	1.058	0.906	1.236
In paid employment							1.129	0.968	1.316	1.098	0.940	1.282
Wealth												
Quartile 2							1.309**	1.081	1.585	1.230*	1.013	1.493
Quartile 3							2.030***	1.688	2.441	1.893***	1.569	2.283
Quartile 4							3.306***	2.753	3.971	2.995***	2.480	3.618
CASP19										1.015**	1.004	1.025
SWLS										1.006	0.992	1.020

Notes: binary logistic regression, N=5239, DV – behavioural health frequency of alcohol consumption; \*p<.05, \*\*p<.01, \*\*\*p<.001; source: author's analysis of ELSA wave 8

### Appendix J RQ1c how does the interaction between caregiving and loneliness affect the relationships between caregiving, loneliness and health expectations?

			Mo	odel 1					Ν	Iodel 2		
Variables	Neither ag	gree nor d	isagree		Disagree		Neither	agree no	r disagree		Disagree	
V arrables	Evn(R)	95%	6 CI	$Evp(\mathbf{R})$	95	% CI	$Evp(\mathbf{B})$	9	5% CI	$Evp(\mathbf{R})$	959	% CI
	Exp(B)	LCI	UCI	Exp(B)	LCI	UCI	Exp(B)	LCI	UCI	Exp(B)	LCI	UCI
Caregiving	0.979	0.810	1.184	1.053	0.850	1.304	1.051	0.856	1.291	1.121	0.894	1.406
Loneliness	0.606***	0.502	0.732	0.425***	0.333	0.542	0.648***	0.529	0.794	0.459***	0.353	0.596
Caregiving-loneliness							0.639	0.367	1.111	0.606	0.294	1.247
			Mo	odel 3					Ν	Iodel 4		
Variables	Neither ag	gree nor d	isagree		Disagree		Neither	agree no	r disagree		Disagree	
v unuoreis	Exp(B)	95%	6 CI	Exp(B)	95	% CI	Exp(B)		95% CI	Exp(B)	959	6 CI
	Enp(E)	LCI	UCI	Enp(D)	LCI	UCI	Enp(D)	LCI	UCI	Enp(E)	LCI	UCI
Caregiving	1.098	0.892	1.351	1.115	0.887	1.403	1.208	0.974	1.500	1.168	0.912	1.496
Loneliness	0.663***	0.536	0.820	0.461***	0.351	0.606	1.112	0.863	1.433	0.967	0.704	1.328
Caregiving-loneliness	0.685	0.393	1.196	0.612	0.296	1.266	0.515	0.259	1.022	0.819	0.386	1.734
Age												
60-69 years	1.155	0.952	1.401	0.926	0.739	1.160	1.063	0.868	1.302	0.904	0.712	1.148
≥70 years	1.483**	1.186	1.854	1.158	0.898	1.493	1.490**	1.174	1.891	1.261	0.958	1.659
Sex, female	0.939	0.814	1.083	1.034	0.878	1.218	0.875	0.752	1.017	0.978	0.822	1.164
Cohabitation	0.884	0.745	1.049	1.057	0.865	1.292	0.882	0.732	1.062	1.012	0.814	1.258
In paid employment	1.342**	1.115	1.615	0.971	0.782	1.206	1.188	0.978	1.444	0.872	0.693	1.098
Wealth												
Quartile 2	1.370**	1.115	1.684	1.197	0.952	1.505	1.256*	1.004	1.572	0.977	0.763	1.251
Quartile 3	1.182	0.957	1.459	0.996	0.787	1.260	0.985	0.784	1.237	0.697**	0.540	0.899
Quartile 4	1.361**	1.104	1.677	0.936	0.736	1.190	1.050	0.836	1.320	0.625***	0.482	0.812
CASP19							1.067***	1.052	1.081	1.082***	1.065	1.100
SWLS							0.998	0.980	1.016	1.005	0.984	1.027

Notes: multinomial logistic regression, N=5275, DV – health expectations; LCI – Lower confidence interval, UCI – Upper confidence interval; \*p<.05, \*\*p<.01, \*\*\*p<.001; source: author's analysis of ELSA wave 8

Appendix K	<b>RQ1c</b> how does the interaction between caregiving and loneliness affect th	e
<b>relationshi</b>	os between caregiving, loneliness and mental health (CES-D)?	

Model 1		Model 2				Model 3		Model 4				
Variables	$E_{\rm res}({\bf D})$	95%	6 CI	$\mathbf{E}_{\mathbf{v}\mathbf{r}}(\mathbf{D})$	95%	5 CI	$\mathbf{E}_{\mathbf{v}\mathbf{r}}(\mathbf{D})$	95% CI		$E_{\rm vm}({\rm D})$	95%	5 CI
	Ехр(Б)	Lower	Upper	Ехр(Б)	Lower	Upper	Exp(B)	Lower	er Upper	Ехр(Б)	Lower	Upper
Constant	0.109***			0.105***			0.136***			24.193***		
Caregiving	1.717***	1.418	2.079	2.029***	1.617	2.547	1.975***	1.560	2.501	1.791***	1.389	2.310
Loneliness	5.899***	5.023	6.928	6.527***	5.455	7.810	5.425***	4.471	6.583	1.774***	1.414	2.226
Caregiving-loneliness				0.598*	0.400	0.895	0.473**	0.309	0.723	0.340***	0.209	0.551
Age												
60-69 years							0.613***	0.490	0.765	0.788	0.617	1.006
≥70 years							0.519***	0.408	0.661	0.589***	0.450	0.770
Sex, female							1.602***	1.359	1.890	2.057***	1.714	2.468
Cohabitation							0.816*	0.681	0.978	0.861	0.703	1.055
In paid employment							0.331***	0.265	0.413	0.430***	0.339	0.547
Wealth												
Quartile 2							0.577***	0.466	0.715	0.824	0.650	1.044
Quartile 3							0.470***	0.376	0.586	0.759*	0.595	0.968
Quartile 4							0.341***	0.267	0.435	0.683**	0.523	0.892
CASP19										0.890***	0.878	0.903
SWLS										0.974**	0.957	0.992

Notes: binary logistic regression, N=5245, DV – CES-D; \*p<.05, \*\*p<.01, \*\*\*p<.001; source: author's analysis of ELSA wave 8

Appendix L	<b>RQ1c</b> how does the interaction between caregiving and loneliness affect the
<b>relationshi</b> p	os between caregiving, loneliness and limiting long-term illness?

		Model 1		Model 2				Model 3		Model 4			
Variables	$E_{\rm res}({\bf D})$	95% CI		$\mathbf{E}_{\mathbf{v}\mathbf{n}}(\mathbf{D})$	95%	6 CI	$E_{\rm res}({\bf D})$	95% CI		$\mathbf{E}_{\mathbf{v}\mathbf{r}}(\mathbf{D})$	95%	6 CI	
	Ехр(Б)	Lower	Upper	схр(б)	Lower	Upper	Ехр(Б)	Lower	Upper	Ехр(Б)	Lower	Upper	
Constant	0.373***			0.368***			0.655***			33.617***			
Caregiving	1.085	0.926	1.272	1.171	0.979	1.400	1.253*	1.036	1.515	1.124	0.920	1.373	
Loneliness	2.190***	1.901	2.524	2.318***	1.984	2.707	1.883***	1.587	2.235	0.767*	0.627	0.937	
Caregiving-loneliness				0.717	0.491	1.047	0.603*	0.405	0.900	0.444***	0.287	0.687	
Age													
60-69 years							1.221*	1.019	1.463	1.515***	1.250	1.837	
≥70 years							1.541***	1.267	1.875	1.667***	1.351	2.058	
Sex, female							0.953	0.839	1.082	1.056	0.923	1.209	
Cohabitation							0.746***	0.645	0.863	0.717***	0.612	0.840	
In paid employment							0.352***	0.296	0.417	0.415***	0.347	0.497	
Wealth													
Quartile 2							0.546***	0.457	0.651	0.717**	0.594	0.867	
Quartile 3							0.468***	0.392	0.559	0.664***	0.549	0.803	
Quartile 4							0.349***	0.290	0.421	0.588***	0.481	0.718	
CASP19										0.904***	0.894	0.915	
SWLS										1.011	0.996	1.026	

Notes: binary logistic regression, N=5275, DV – LLTI;\*p<.05, \*\*p<.01, \*\*\*p<.001; source: author's analysis of ELSA wave 8

### **Appendix M** RQ1c how does the interaction between caregiving and loneliness affect the relationships between caregiving, loneliness and QoL?

			Model 1		Model 2					
Variables	Unstand	ardised	Standardised		Unstand	lardised	Standardised			
Vallables	В	Std. Error	Beta	Sig.	В	Std. Error	Std. Beta			
Constant	43.964	0.129		0.000***	43.862	0.131		0.000***		
Caregiving	-1.791	0.293	-0.072	0.000***	-1.162	0.327	-0.047	0.000***		
Loneliness	-11.699	0.276	-0.500	0.000***	-11.161	0.302	-0.477	0.000***		
Caregiving-loneliness					-3.212	0.737	-0.062	0.000***		
			Model 3				Model 4	-		
Variables	Unstand	ardised	Standardised		Unstanc	lardised	Standardised			
v allables	В	Std. Error	Beta	Sig.	В	Std. Error	Beta	Sig.		
Constant	38.167	0.389		0.000***	41.263	0.384		0.000***		
Caregiving	-1.269	0.312	-0.051	0.000***	-0.819	0.278	-0.033	0.003**		
Loneliness	-10.105	0.299	-0.431	0.000***	-7.506	0.276	-0.320	0.000***		
Caregiving-loneliness	-2.616	0.703	-0.050	0.000***	-3.077	0.625	-0.059	0.000***		
Age										
60-69 years	1.594	0.277	0.082	0.000***	1.466	0.247	0.076	0.000***		
≥70 years	0.040	0.161	0.004	0.805	0.042	0.146	0.004	0.775		
Sex, female	1.127	0.209	0.061	0.000***	1.485	0.188	0.081	0.000***		
Cohabitation	0.368	0.252	0.018	0.144	-0.125	0.224	-0.006	0.577		
In paid employment	2.065	0.267	0.110	0.000***	0.425	0.242	0.023	0.079		
Wealth										
Quartile 2	3.334	0.298	0.155	0.000***	2.152	0.266	0.100	0.000***		
Quartile 3	4.063	0.299	0.192	0.000***	2.600	0.269	0.123	0.000***		
Quartile 4	5.645	0.301	0.269	0.000***	3.674	0.276	0.175	0.000***		
					-3.850	0.218	-0.193	0.000***		
CES-D: depressive symptomatology					-6.218	0.272	-0.253	0.000***		
Physiological health: chronic CV condition(s)					-1.058	0.195	-0.057	0.000***		
Health expectations							0.000			
Neither agrees nor disagrees					2.378	0.247	0.098	0.000***		
Disagrees					2.972	0.284	0.107	0.000***		
Alcohol consumption ≥3days per week					0.864	0.201	0.044	0.000***		

Notes: linear regression, N=5366, DV – CASP19; \*p<.05, \*\*p<.01, \*\*\*p<.001; source: author's analysis of ELSA wave 8

<b>A</b>			Model 1		Model 2					
	Unstand	ardised	Standardised		Unstandardised		Standardised			
variables		Std. Error	Beta	Sig.	В	Std. Error	Beta	Sig.		
Constant	26.960	0.092		0.000***	26.917	0.094		0.000***		
Caregiving	-0.463	0.210	-0.026	0.027*	-0.196	0.235	-0.011	0.405		
Loneliness	-7.671	0.196	-0.468	0.000***	-7.448	0.215	-0.454	0.000***		
Caregiving-loneliness					-1.298	0.519	-0.036	0.012*		
			Model 3	-			Model 4	-		
Variables	Unstand	ardised	Standardised		Unstan	dardised	Standardised			
v arabies	В	Std. Error	Beta	Sig.	В	Std. Error	Beta	Sig.		
Constant	21.944	0.280		0.000***	23.254	0.298		0.000***		
Caregiving	-0.399	0.227	-0.023	0.079	-0.117	0.217	-0.007	0.591		
Loneliness	-6.231	0.215	-0.380	0.000***	-4.913	0.213	-0.300	0.000***		
Caregiving-loneliness	-1.103	0.499	-0.031	0.027*	-1.379	0.476	-0.038	0.004**		
Age										
60-69 years	1.181	0.201	0.086	0.000***	1.025	0.193	0.075	0.000***		
≥70 years	0.795	0.116	0.117	0.000***	0.699	0.113	0.103	0.000***		
Sex, female	0.697	0.150	0.054	0.000***	0.917	0.146	0.071	0.000***		
Cohabitation	2.383	0.181	0.165	0.000***	2.157	0.173	0.150	0.000***		
In paid employment	0.589	0.194	0.044	0.002**	-0.184	0.188	-0.014	0.328		
Wealth										
Quartile 2	1.451	0.215	0.096	0.000***	0.906	0.207	0.060	0.000***		
Quartile 3	1.932	0.216	0.129	0.000***	1.252	0.209	0.084	0.000***		
Quartile 4	2.660	0.218	0.179	0.000***	1.753	0.215	0.118	0.000***		
					-1.347	0.169	-0.096	0.000***		
CES-D: depressive symptomatology					-3.535	0.211	-0.205	0.000***		
Physiological health: chronic condition(s)					-0.230	0.151	-0.018	0.129		
Health expectations					1.107	0.100	0.044	0.000		
Neither agrees nor disagrees					1.106	0.192	0.064	0.000***		
Disagrees					1.454	0.221	0.074	0.000***		
Alcohol consumption $\geq 3$ days per week					0.459	0.156	0.033	0.003**		

## Appendix N RQ1c how does the interaction between caregiving and loneliness affect the relationships between caregiving, loneliness, and life satisfaction?

Notes: linear regression, N=5482, DV – SWLS; \*p<.05, \*\*p<.01, \*\*\*p<.001; source: author's analysis of ELSA wave 8

Appendix O Bivariate associations between caregivers' characteristics, the caregiving relationship, caregiving role, social networks, loneliness, health, wellbeing, and loneliness (wave eight) in the caregiving sample

		-LS	
	$\chi^2$	df	Sig.
Age	7.18	2	.028
Sex	0.00	1	.967
Cohabitation	55.48	1	.000
In paid employment	13.51	1	.000
Wealth	43.82	3	.000
Cares for a spouse	0.26	1	.608
Cares for an adult-child	8.14	1	.004
Cares for parents or in-laws	5.98	1	.014
Cares for others	0.35	1	.557
Co-residential care	9.07	1	.003
Caregiving intensity	5.35	2	.069
Number of care recipients	0.06	1	.801
Caregivers' perceived appreciation	25.74	1	.000
Caregivers' role satisfaction	20.93	1	.000
Household size	7.05	2	.030
Loneliness expectations	57.92	2	.000
Physiological health	8.77	1	.003
Health expectations	24.52	2	.000
Alcohol consumption	0.001	1	.981
CES-D	62.58	1	.000
	r		Sig.
Close relationships with family	-0.19	.000	)
Close relationships with friends	-0.24	.000	)
CASP19	-0.49	.000	)
SWLS	-0.42	.000	)

Source: author's analysis of ELSA wave 8

Appendix P RQ2a how does c	aregivi	ing aff	ect lon	eliness	on the	e UCL	A Lone	liness	<b>Scale</b> ?
	Model 1				Model 2		Model 3		
Variables	$\mathbf{E}_{mn}(\mathbf{D})$	95% CI		$\mathbf{E}_{mn}(\mathbf{D})$	95% CI		$\mathbf{E}_{\mathrm{res}}(\mathbf{D})$	95%	6 CI
	Exp(B)	Lower	Upper	Exp(B)	Lower	Upper	Exp(B)	Lower	Upper
Constant	0.230**			1.185			273.879***		
Age									
60-69 years	0.400**	0.213	0.750	0.363**	0.189	0.700	0.640	0.289	1.419
≥70 years	0.264***	0.134	0.520	0.257***	0.126	0.521	0.527	0.212	1.312
Sex, female	1.505	0.892	2.540	1.630	0.948	2.801	3.126**	1.515	6.451
Cohabitation	0.522	0.268	1.019	0.426*	0.215	0.844	0.575	0.239	1.383
In paid employment	0.229***	0.121	0.434	0.207***	0.107	0.403	0.438*	0.201	0.955
Wealth									
Quartile 2	0.613	0.311	1.209	0.727	0.356	1.482	0.939	0.382	2.310
Quartile 3	0.325**	0.157	0.674	0.383*	0.180	0.815	1.152	0.442	3.007
Quartile 4	0.499	0.248	1.007	0.598	0.286	1.251	1.535	0.584	4.031
Care recipients									
Adult-child	2.102	0.988	4.472	2.536*	1.165	5.516	2.550*	1.012	6.425
Parent or parent-in-law	0.786	0.334	1.852	0.636	0.257	1.572	0.665	0.225	1.969
Other relative, friend or neighbour	1.826	0.712	4.681	1.325	0.495	3.545	1.085	0.307	3.830
Lives with the care recipient	2.744*	1.243	6.057	1.934	0.833	4.491	1.358	0.490	3.762
Caregiving intensity									
High-intensity (≥20hpw)	0.890	0.457	1.730	0.772	0.387	1.541	0.928	0.400	2.154
Round-the-clock care (168hpw)	1.031	0.509	2.089	1.166	0.557	2.442	1.290	0.549	3.033
Multiple care recipients	0.721	0.317	1.643	0.911	0.398	2.084	1.413	0.482	4.143
Number of close relationships (family and child)				0.931	0.837	1.035	0.985	0.874	1.110
Number of close relationships (friends)				0.750***	0.650	0.865	0.797**	0.680	0.933
CES-D: depressive symptomatology							0.587	0.275	1.254
Physiological health: chronic CV condition(s)							1.552	0.793	3.036
Health expectations									
Neither agrees nor disagrees							0.682	0.261	1.782
Disagrees							0.517	0.166	1.610
Alcohol consumption ≥3days per week							2.369*	1.145	4.904
CASP19							0.852***	0.808	0.898
SWLS							0.954	0.896	1.016

Notes: binary logistic regression, N=498, DV – UCLA-LS; \*p<.05, \*\*p<.01, \*\*\*p<.001; source: author's analysis of ELSA wave 8

		Model 1			Model 2		Model 3			
Variables	$\mathbf{F}_{\mathbf{n}}(\mathbf{D})$	95	% CI	$\mathbf{F}_{\mathbf{n}}(\mathbf{D})$	95%	6 CI		95%	% CI	
	Exp(B)	Lower	Upper	Exp(B)	Lower	Upper	Exp(B)	Lower	Upper	
Constant	2.369			2.646			7.718*			
Age										
60-69 years	2.670***	1.563	4.560	2.680***	1.569	4.579	3.374***	1.919	5.932	
≥70 years	2.690**	1.484	4.878	2.711**	1.492	4.925	3.374***	1.785	6.379	
Sex, female	0.416	0.273	0.633	0.417	0.273	0.634	0.425***	0.276	0.656	
Cohabitation	0.610	0.346	1.077	0.610	0.346	1.078	0.630	0.345	1.151	
In paid employment	0.434**	0.266	0.706	0.434**	0.266	0.706	0.524*	0.316	0.870	
Wealth										
Quartile 2	0.905	0.497	1.647	0.911	0.500	1.659	1.109	0.597	2.059	
Quartile 3	0.636	0.350	1.156	0.642	0.353	1.167	0.929	0.492	1.756	
Quartile 4	0.533*	0.290	0.980	0.533*	0.289	0.984	0.775	0.404	1.486	
Care recipients										
Adult-child	0.678	0.337	1.362	0.683	0.339	1.373	0.695	0.339	1.423	
Parent or parent-in-law	0.517	0.241	1.113	0.513	0.238	1.107	0.526	0.242	1.142	
Other relative, friend or neighbour	1.116	0.498	2.505	1.089	0.481	2.464	1.065	0.463	2.451	
Lives with the care recipient	1.414	0.697	2.872	1.376	0.671	2.822	1.253	0.606	2.591	
Caregiving intensity										
High-intensity (≥20hpw)	1.613	0.931	2.793	1.603	0.925	2.778	1.610	0.914	2.833	
Round-the-clock care (168hpw)	0.951	0.515	1.758	0.955	0.516	1.766	0.875	0.465	1.646	
Multiple care recipients	0.958	0.506	1.812	0.958	0.506	1.816	1.014	0.528	1.947	
Number of close relationships (family and child)				0.985	0.918	1.058	1.010	0.939	1.086	
Number of close relationships (friends)				0.995	0.917	1.079	1.034	0.950	1.125	
UCLA-LS: Loneliness							1.551	0.844	2.851	
CASP19							0.939**	0.905	0.974	
SWLS							1.042	0.993	1.093	

Notes: binary logistic regression, N=505, DV – physiological health chronic CV conditions; significance \*p<.05, \*\*p<.01, \*\*\*p<.001; source: author's analysis of ELSA wave 8
Appendix R RQ2b how does caregiving affect alcohol consumption	?
--	---

		Model 1			Model 2		Model 3			
Variables	$\mathbf{E}_{\mathbf{v}}\mathbf{n}(\mathbf{D})$	95	% CI	$\mathbf{E}_{\mathbf{v}}\mathbf{n}(\mathbf{D})$	95%	6 CI	$E_{\rm res}({\bf D})$	95%	% CI	
	Exp(B)	Lower	Upper	Exp(B)	Lower	Upper	Exp(B)	Lower	Upper	
Constant	0.892			0.588			0.029***			
Age										
60-69 years	1.414	0.833	2.403	1.391	0.818	2.367	1.332	0.770	2.305	
≥70 years	0.966	0.500	1.865	0.908	0.467	1.767	1.006	0.503	2.013	
Sex, female	0.586*	0.374	0.919	0.577*	0.367	0.907	0.527**	0.329	0.843	
Cohabitation	0.844	0.446	1.598	0.840	0.441	1.597	0.872	0.442	1.720	
In paid employment	1.024	0.612	1.712	0.994	0.592	1.669	1.090	0.633	1.879	
Wealth										
Quartile 2	3.515**	1.604	7.704	3.561**	1.623	7.815	2.747*	1.245	6.060	
Quartile 3	3.593**	1.626	7.943	3.571**	1.610	7.920	2.289*	1.015	5.161	
Quartile 4	7.624***	3.535	16.442	7.582***	3.497	16.439	5.049***	2.294	11.115	
Care recipients										
Adult-child	0.342*	0.141	0.828	0.320*	0.131	0.781	0.310*	0.124	0.774	
Parent or parent-in-law	0.797	0.371	1.709	0.836	0.390	1.791	0.813	0.367	1.803	
Other relative, friend or neighbour	0.466	0.208	1.047	0.475	0.210	1.077	0.446	0.193	1.034	
Lives with the care recipient	0.513	0.240	1.094	0.559	0.261	1.196	0.596	0.269	1.322	
Caregiving intensity										
High-intensity (≥20hpw)	0.712	0.374	1.355	0.714	0.374	1.364	0.737	0.377	1.441	
Round-the-clock care (168hpw)	1.663	0.847	3.264	1.593	0.808	3.143	1.637	0.800	3.349	
Multiple care recipients	1.916*	1.036	3.547	1.912*	1.031	3.546	1.592	0.839	3.020	
Number of close relationships (family and child)				1.026	0.951	1.107	1.007	0.931	1.089	
Number of close relationships (friends)				1.067	0.981	1.161	1.055	0.967	1.150	
UCLA-LS: Loneliness							2.594**	1.270	5.296	
CASP19							1.087***	1.041	1.135	
SWLS							0.982	0.931	1.036	

Notes: binary logistic regression, N=503, DV – behavioural health frequency of alcohol consumption; \*p<.05, \*\*p<.01, \*\*\*p<.001; source: author's analysis of ELSA wave 8

Disagree

0.781 0.758

0.534

0.595

0.254

0.528

0.176

0.726

0.128

0.632

0.953

1.072

0.602

0.819

0.881

0.132

0.986

0.989

95% CI LCI U

UCI 3.921

4.804

2.992

2.620

1.469

2.703

1.161

4.835

1.352 6.334

3.962

4.713

6.309

3.583

1.041

1.116

1.165

1.098

1.149

		uuus	, cai	USI I	me d			alui		<mark>JUUU</mark>						
			Mod	lel 1					Mod	iel 2			Model 3			
Variables	Nei	ither agre r disagree	e e	Γ	Disagree		Ne no	ither agre or disagre	ee e	]	Disagree		Nei noi	ither agre r disagree	e >	
	Exp	95%	6 CI	Exp	95%	5 CI	Exp	95%	6 CI	Exp	95%	6 CI	Exp	95%	6 CI	Exp
	(B)	LCI	UCI	(B)	LCI	UCI	(B)	LCI	UCI	(B)	LCI	UCI	(B)	LCI	UCI	(B)
Age																
60-69 years	1.150	0.737	1.794	1.317	0.784	2.212	0.973	0.553	1.710	1.758	0.825	3.744	0.976	0.521	1.828	1.750
$\geq$ 70 years	1.038	0.595	1.811	1.382	0.760	2.513	0.987	0.497	1.959	2.246	0.976	5.168	1.210	0.556	2.631	1.908
Sex, female	1.152	0.796	1.667	1.090	0.720	1.649	1.113	0.687	1.803	0.940	0.530	1.668	1.108	0.654	1.877	0.994
Cohabitation	1.289	0.745	2.230	2.115*	1.140	3.924	1.097	0.541	2.223	1.542	0.683	3.481	1.361	0.569	3.254	1.236
In paid employment	1.956**	1.257	3.045	0.901	0.535	1.519	1.907*	1.089	3.338	1.470	0.737	2.931	2.023*	1.082	3.783	1.249
Wealth																
Quartile 2	1.031	0.600	1.774	0.768	0.426	1.385	1.330	0.668	2.651	1.118	0.501	2.498	1.588	0.712	3.543	0.610
Quartile 3	1.528	0.887	2.632	1.086	0.610	1.934	2.025*	1.003	4.089	2.025	0.937	4.375	2.138	0.929	4.921	1.194
Quartile 4	1.144	0.663	1.975	0.697	0.383	1.266	1.452	0.715	2.949	0.813	0.334	1.978	1.595	0.706	3.606	0.452
Care recipients																
Adult-child	0.417*	0.203	0.856	1.304	0.677	2.512	0.393*	0.161	0.958	1.274	0.530	3.065	0.102**	0.021	0.492	1.873
Parent or parent-in-law	0.436	0.223	0.854	0.841	0.397	1.781	0.608	0.256	1.447	0.480	0.156	1.478	0.513	0.199	1.320	0.416
Other relative, friend or neighbour	0.742	0.374	1.471	1.960	0.926	4.152	0.686	0.265	1.775	1.829	0.621	5.385	0.560	0.200	1.568	2.000
Lives with the care recipient	1.037	0.572	1.878	1.228	0.639	2.361	1.319	0.570	3.054	1.155	0.428	3.117	1.218	0.472	3.145	1.368
Caregiving intensity																

### **Appendix S RQ2b** how does caregiving affect health expectations?

High-intensity (≥20hpw)

Round-the-clock care (168hpw)

Number of close relationships (family and child)

Number of close relationships (friends)

Multiple care recipients

UCLA-LS: Loneliness

CASP19

SWLS

0.810

1.448

1.676

0.474

0.862

0.958

1.385

2.433

2.933

1.803\*

2.251\*\*

1.396

1.044

1.259

0.741

3.113

4.026

2.630

Notes: multinomial logistic regression, N=538, DV - health expectations; LCI - Lower confidence interval, UCI - Upper confidence interval; \*p<.05, \*\*p<.01, \*\*\*p<.001; source: author's analysis ELSA wave 8

0.960

1.284

0.771

1.002

1.040

0.499

0.655

0.350

0.927

0.949

1.845

2.517

1.701

1.084

1.139

2.016

2.258\*

1.672

0.945

1.026

0.946

1.001

0.739

0.847

0.920

4.299

5.093

3.783

1.055

1.143

1.295

1.144

0.673

0.972

1.032

0.586

1.048\*

0.994

0.650

0.527

0.287

0.892

0.938

0.242

1.002

0.936

2.578

2.485

1.579

1.060

1.136

1.420

1.097

1.054

2.119

2.600\*

1.469

0.924

0.991

0.391

1.041

1.066

## Appendix T RQ2c how does caregiving affect QoL?

Model 1						Ν	Model 2		Model 3				
	Unstand	ardised	Standardised		Unstan	dardised	Standardised		Unstar	ndardised	Standardised		
Variables	coeffi	cients	coefficients	Sig	coeff	ricients	coefficients	Sig	coef	ficients	coefficients	Sig	
	В	Std. Error	Beta	5.5.	В	Std. Error	Beta	5.5.	В	Std. Error	Beta	5.6.	
Constant	30.738	1.903		0.000***	26.377	2.013		0.000***	36.009	1.761		0.000***	
Age													
60-69 years	2.962	0.952	0.138	0.002**	2.768	0.927	0.129	0.003**	1.495	0.765	0.070	0.051	
≥70 years	1.623	0.565	0.149	0.004**	1.470	0.551	0.135	0.008**	0.463	0.458	0.043	0.313	
Sex, female	0.885	0.809	0.043	0.274	0.622	0.789	0.030	0.431	1.428	0.649	0.069	0.028*	
Cohabitation	2.237	1.115	0.087	0.045*	2.259	1.084	0.088	0.038*	0.280	0.879	0.011	0.750	
In paid employment	3.989	0.931	0.194	0.000***	3.929	0.906	0.191	0.000***	0.879	0.753	0.043	0.244	
Wealth													
Quartile 2	5.954	1.148	0.255	0.000***	5.710	1.118	0.245	0.000***	3.602	0.909	0.155	0.000***	
Quartile 3	8.320	1.178	0.358	0.000***	7.936	1.148	0.341	0.000***	4.353	0.950	0.187	0.000***	
Quartile 4	8.143	1.178	0.357	0.000***	7.859	1.151	0.345	0.000***	4.534	0.964	0.199	0.000***	
Care recipients													
Adult-child	-0.129	1.338	-0.004	0.923	-0.456	1.305	-0.015	0.727	1.706	1.073	0.055	0.112	
Parent or parent-in-law	-0.076	1.378	-0.003	0.956	0.552	1.346	0.025	0.682	-0.006	1.086	0.000	0.996	
Other relative, friend or neighbour	-1.474	1.493	-0.062	0.324	-0.567	1.470	-0.024	0.700	-0.270	1.196	-0.011	0.821	
Lives with the care recipient	-3.064	1.316	-0.153	0.020*	-1.803	1.300	-0.090	0.166	-0.509	1.049	-0.025	0.628	
Caregiving intensity													
High-intensity (≥20hpw)	-0.852	1.074	-0.035	0.428	-0.659	1.046	-0.027	0.529	-0.547	0.845	-0.023	0.518	
Round-the-clock care (168hpw)	-0.976	1.210	-0.036	0.420	-1.219	1.182	-0.045	0.303	-0.922	0.955	-0.034	0.334	
Multiple care recipients	2.879	1.183	0.104	0.015*	2.868	1.153	0.104	0.013*	2.940	0.937	0.106	0.002**	
Number of close relationships (family and child)					0.472	0.132	0.143	0.000***	0.316	0.107	0.096	0.003**	
Number of close relationships (friends)					0.443	0.157	0.113	0.005**	0.067	0.128	0.017	0.603	
UCLA-LS Loneliness									-10.014	0.829	-0.402	0.000***	
CES-D: depressive symptomatology									-6.141	0.840	-0.235	0.000***	
Physiological health: chronic condition(s)									-2.086	0.680	-0.102	0.002**	
Health expectations													
Neither agrees nor disagrees									1.631	0.809	0.062	0.044*	
Disagrees									1.440	0.981	0.046	0.143	
Alcohol consumption $\geq$ 3 days per week									2.629	0.716	0.116	0.000***	

Notes: linear regression, N=540, DV - CASP19; \*p<.05, \*\*p<.01, \*\*\*p<.001; source: author's analysis of ELSA wave 8

## Appendix U RQ2c how does caregiving affect life satisfaction?

	Model 1					Ν	Iodel 2		Model 3				
	Unstand	lardised	Standardised		Unstan	dardised	Standardised		Unstar	ndardised	Standardised		
Variables	coeffi	cients	coefficients	Sig.	coeff	ficients	coefficients	Sig	coef	ficients	coefficients	Sig.	
	В	Std. Error	Beta	515.	В	Std. Error	Beta	515.	В	Std. Error	Beta	515.	
Constant	17.659	1.301		0.000***	16.275	1.411		0.000***	21.451	1.383		0.000***	
Age													
60-69 years	1.257	0.655	0.086	0.055	1.190	0.653	0.081	0.069	0.469	0.600	0.032	0.435	
≥70 years	1.248	0.386	0.169	0.001**	1.216	0.386	0.165	0.002**	0.504	0.359	0.068	0.161	
Sex, female	0.457	0.556	0.032	0.412	0.388	0.554	0.027	0.484	0.837	0.508	0.059	0.100	
Cohabitation	4.174	0.762	0.243	0.000***	4.181	0.759	0.243	0.000***	3.073	0.685	0.179	0.000***	
In paid employment	1.794	0.639	0.128	0.005**	1.797	0.637	0.128	0.005**	0.158	0.587	0.011	0.789	
Wealth													
Quartile 2	2.983	0.786	0.189	0.000***	2.932	0.783	0.186	0.000***	1.738	0.709	0.110	0.015*	
Quartile 3	3.725	0.809	0.235	0.000***	3.648	0.807	0.230	0.000***	1.661	0.743	0.105	0.026*	
Quartile 4	3.949	0.817	0.252	0.000***	3.935	0.818	0.252	0.000***	2.273	0.763	0.145	0.003**	
Care recipients													
Adult-child	-1.395	0.899	-0.068	0.121	-1.488	0.897	-0.072	0.098	-0.326	0.819	-0.016	0.691	
Parent or parent-in-law	0.893	0.949	0.060	0.347	1.073	0.949	0.072	0.259	0.817	0.853	0.055	0.338	
Other relative, friend or neighbour	-1.183	1.028	-0.074	0.250	-0.816	1.036	-0.051	0.431	-0.782	0.938	-0.049	0.404	
Lives with the care recipient	-1.880	0.908	-0.138	0.039*	-1.502	0.919	-0.110	0.103	-0.920	0.826	-0.068	0.266	
Caregiving intensity													
High-intensity (≥20hpw)	0.462	0.736	0.028	0.531	0.550	0.734	0.033	0.454	0.470	0.661	0.029	0.477	
Round-the-clock care (168hpw)	0.516	0.833	0.028	0.536	0.470	0.832	0.026	0.572	0.613	0.748	0.034	0.413	
Multiple care recipients	1.556	0.823	0.081	0.059	1.568	0.821	0.082	0.057	1.764	0.744	0.092	0.018*	
Number of close relationships (family and													
child)					0.199	0.092	0.088	0.031*	0.112	0.083	0.050	0.175	
Number of close relationships (friends)					0.052	0.110	0.019	0.637	-0.141	0.100	-0.053	0.158	
UCLA-LS Loneliness									-5.507	0.650	-0.328	0.000***	
CES-D: depressive symptomatology									-3.332	0.658	-0.189	0.000***	
Physiological health: chronic CV													
condition(s)									-0.602	0.531	-0.043	0.258	
Health expectations													
Neither agrees nor disagrees									1.383	0.616	0.079	0.025*	
Disagrees									1.552	0.762	0.073	0.042*	
Alcohol consumption $\geq$ 3 days per week									0.777	0.561	0.050	0.167	

Notes: linear regression, N=553, DV – SWLS; \*p<.05, \*\*p<.01, \*\*\*p<.001; source: author's analysis of ELSA wave 8

### Appendix V Crosstabulation of UCLA loneliness responses at waves two and eight

	UCL	A-LS Wave 8
UCLA-LS Wave 2	Not Lonely	Lonely
	N (%)	N (%)
Not Lonely	2988 (88.4)	275 (11.6)
Lonely	234 (49.8)	236 (50.2)

Notes: % of UCLA at wave two, source: author's analysis of ELSA waves 2 and 8

# Appendix W Bivariate associations between baseline (wave two) variables and wave eight loneliness in the longitudinally-weighted sample

Pagalina (waya two) yariahlag		UCLA-LS (wave	e 8)
Baseline (wave two) variables	$\chi^2$	df	Sig.
Caregiving	0.39	1	.533
Age	12.24	2	.003
Sex	13.62	1	.000
Cohabitation	58.22	1	.000
In paid employment	8.77	1	.007
Wealth	36.52	3	.000
Household size	43.85	2	.000
UCLA-LS	394.58	1	.000
Loneliness expectations	75.41	2	.000
	r		Sig.
CASP19	-0.27	.000	
SWLS	-0.25	.000	

Source: author's analysis of ELSA waves 2 and 8

### Appendix X Bivariate associations between the variables computed for the follow-up analyses and loneliness occurring after the start of care provision, in the longitudinally-weighted caregiving sample

	UCLA-LS (lone start o	eliness occu f care provis	rring after the sion)
	$\chi^2$	df	Sig.
Duration of care	7.97	2	.019
Ever provided care to an adult-child	11.94	1	.001
Ever provided care to parents or in-laws	0.12	1	.726
Ever provided care to others	0.94	1	.332
Ever provided co-residential care	2.22	1	.137
Ever provided high-intensity care	6.95	1	.008
Ever provided round-the-clock care	10.04	1	.002
Ever provided care to multiple recipients	0.48	1	.490
Changes to loneliness expectations	14.01	5	.016
Changes to depressive symptomatology	125.43	3	.000

Source: author's analysis of ELSA waves 2-8

# Appendix Y RQ3a(i) is caregiving associated with loneliness 12-years later, excluding those lonely at baseline

		Model 1			Model 2		Model 3			
Variables	$\mathbf{E}_{max}(\mathbf{D})$	95	% CI	$\mathbf{E}_{mn}(\mathbf{D})$	95%	% CI	$\mathbf{E}_{\mathrm{res}}(\mathbf{D})$	95%	% CI	
	Exp(B)	Lower	Upper	Exp(B)	Lower	Upper	Exp(B)	Lower	Upper	
Constant	0.122***			0.112***			1.472			
Caregiving (wave two)	1.231	0.840	1.803	1.179	0.796	1.746	1.137	0.763	1.696	
Loneliness expectations										
Neither agree nor disagree				0.854	0.604	1.207	0.982	0.687	1.405	
Disagree				0.608**	0.437	0.846	0.809	0.571	1.148	
Age										
60-69 years				1.435*	1.004	2.050	1.735**	1.206	2.495	
≥70 years				1.654*	1.045	2.617	2.248**	1.404	3.600	
Sex, female				1.354*	1.013	1.809	1.461*	1.081	1.973	
Cohabitation				0.975	0.471	2.018	1.049	0.498	2.210	
In paid employment				1.047	0.745	1.472	1.209	0.855	1.708	
Wealth										
Quartile 2				1.044	0.704	1.548	1.295	0.859	1.952	
Quartile 3				0.988	0.666	1.466	1.226	0.813	1.849	
Quartile 4				0.505**	0.323	0.790	0.686	0.427	1.103	
Household size										
Dual-person household				0.876	0.403	1.903	0.941	0.428	2.070	
Large household (≥3)				0.868	0.377	1.998	0.917	0.392	2.143	
Physiological health: chronic CV condition(s)							0.953	0.698	1.294	
Health expectations										
Neither agrees nor disagrees							0.649	0.419	1.004	
Disagrees							0.807	0.514	1.267	
Alcohol consumption ≥3days per week							1.182	0.875	1.596	
CES-D: depressive symptomatology							1.196	0.789	1.813	
CASP19							0.961**	0.939	0.985	
SWLS							0.956**	0.927	0.987	

Notes: binary logistic regression, N=2458, DV – UCLA-LS (wave 8);\*p<.05, \*\*p<.01, \*\*\*p<.001; source: author's analysis of ELSA waves 2 and 8

# **Appendix Z** RQ3a(ii) is caregiving associated with loneliness 12-years later, controlling for loneliness at baseline

		Model 1			Model 2		Model 3			
Variables	$E_{\rm wm}({\rm D})$	95	% CI	$E_{\rm reg}({\bf D})$	95%	6 CI	$E_{\rm we}({\bf D})$	95%	% CI	
	Exp(B)	Lower	Upper	Exp(B)	Lower	Upper	Exp(B)	Lower	Upper	
Constant	0.208			0.126***			0.882			
Caregiving (wave two)	1.105	0.820	1.488	1.101	0.794	1.526	1.064	0.762	1.484	
UCLA-LS Loneliness				6.136***	4.684	8.039	4.130***	3.080	5.538	
Loneliness expectations										
Neither agree nor disagree				0.783	0.590	1.041	0.864	0.646	1.156	
Disagree				0.577***	0.431	0.772	0.685*	0.505	0.928	
Age										
60-69 years				1.242	0.931	1.657	1.445*	1.075	1.941	
≥70 years				1.594**	1.103	2.302	1.976***	1.352	2.887	
Sex, female				1.180	0.933	1.493	1.206	0.944	1.540	
Cohabitation				0.981	0.589	1.634	1.029	0.608	1.740	
In paid employment				1.002	0.759	1.323	1.121	0.845	1.488	
Wealth										
Quartile 2				1.139	0.831	1.560	1.312	0.948	1.814	
Quartile 3				1.119	0.808	1.550	1.315	0.939	1.841	
Quartile 4				0.679**	0.474	0.973	0.893	0.611	1.304	
Household size										
Dual-person household				0.905	0.533	1.537	0.888	0.517	1.527	
Large household (≥3)				1.025	0.576	1.823	0.970	0.538	1.749	
Physiological health: chronic condition(s)							0.978	0.764	1.252	
Health expectations										
Neither agrees nor disagrees							0.690*	0.482	0.989	
Disagrees							0.948	0.653	1.376	
Alcohol consumption $\geq$ 3 days per week							0.935	0.725	1.205	
CES-D: depressive symptomatology							1.220	0.898	1.656	
CASP19							0.970**	0.951	0.989	
SWLS							0.973*	0.949	0.997	

Notes: binary logistic regression, N=2074, DV – UCLA-LS (wave 8); \*p<.05, \*\*p<.01, \*\*\*p<.001; source: author's analysis of ELSA waves 2 and 8

# Appendix AA Crosstabulation of responses to loneliness expectations and UCLA-LS loneliness at wave two

		Loneliness expectations Wave 2										
UCLA-LS	Agree	Neither agree, nor disagree	Disagree									
Wave 2	N (%)	N (%)	N (%)									
Not lonely	768 (31.9)	654 (27.1)	988 (41.0)									
Lonely	319 (67.4)	104 (22.0)	50 (10.6)									

Notes: % of UCLA-LS, source: author's analysis of the longitudinal sample at baselines (ELSA wave 2)

### Appendix BB RQ3b is caregiving at any wave associated with later loneliness?

		Model 1			Model 2		Model 3			
	Evp(B)	95%	6 CI	$Evp(\mathbf{R})$	95%	6 CI	$E_{vp}(\mathbf{R})$	95%	6 CI	
	Exp(D)	Lower	Upper	Exp(B)	Lower	Upper	Exb(p)	Lower	Upper	
Constant	0.186***			0.259**			9.428**			
Caregiving duration (reference: short-term)										
Long-term	2.088**	1.376	3.167	1.954**	1.272	3.003	1.823**	1.167	2.847	
Intermittent	1.221	0.678	2.199	1.093	0.595	2.009	1.007	0.536	1.893	
Loneliness expectations										
Neither agree nor disagree				0.658	0.401	1.082	0.722	0.430	1.213	
Disagree				0.489**	0.309	0.772	0.604*	0.373	0.979	
Age										
60-69 years				1.163	0.720	1.878	1.389	0.844	2.287	
≥70 years				1.076	0.564	2.054	1.230	0.624	2.422	
Sex, female				1.425	0.939	2.162	1.505	0.973	2.328	
Cohabitation				0.660	0.228	1.904	0.557	0.180	1.721	
In paid employment				0.754	0.474	1.199	0.826	0.513	1.331	
Wealth										
Quartile 2				0.667	0.386	1.152	0.830	0.465	1.480	
Quartile 3				0.703	0.415	1.192	0.896	0.513	1.565	
Quartile 4				0.444**	0.252	0.784	0.634	0.345	1.163	
Household size										
Dual-person household				0.925	0.286	2.992	1.060	0.306	3.668	
Large household $(\geq 3)$				1.256	0.364	4.331	1.439	0.389	5.324	
Physiological health: chronic CV condition(s)							0.639	0.402	1.015	
Health expectations										
Neither agrees nor disagrees							0.864	0.492	1.517	
Disagrees							0.925	0.507	1.686	
Alcohol consumption $\geq$ 3 days per week							0.923	0.594	1.434	
CES-D: depressive symptomatology							1.433	0.770	2.669	
CASP19							0.931***	0.901	0.962	
SWLS							0.988	0.944	1.035	

Notes: binary logistic regression, N=846, DV – UCLA-LS (loneliness occurring after the start of care provision); \*p<.05, \*\*p<.01, \*\*\*p<.001; source: author's analysis of ELSA waves 2-8

# Appendix CC RQ3c how are the caregiving relationship, and role, associated with loneliness for short-term *versus* long-term, or intermittent caregivers?

	Model 1			Model 2			Model 3		
		95% CI		_	95% CI			95% CI	
	Exp(B)	Lower	Upper	Exp(B)	Lower	Upper	Exp(B)	Lower	Upper
Constant	0.253***			0.165***			0.251***		
Caregiving duration (reference: short-term)									
Long-term	1.867*	1.160	3.004	1.317	0.746	2.325	1.194	0.654	2.181
Intermittent	1.162	0.576	2.345	1.017	0.488	2.118	0.811	0.370	1.775
Ever provided care to (reference: spouse)									
Adult-child				0.980	0.383	2.507	0.919	0.352	2.394
Parents /in-law				1.190	0.688	2.058	1.406	0.788	2.508
Others				1.144	0.673	1.945	1.203	0.693	2.087
Ever provided co-residential care				1.069	0.608	1.881	1.219	0.672	2.210
Ever provided (reference: low-intensity care)									
High-intensity care				1.707*	1.033	2.821	1.880*	1.108	3.192
Round-the-clock care				2.109*	1.177	3.778	1.943*	1.050	3.593
Ever provided care to multiple recipients				0.786	0.462	1.336	0.745	0.425	1.306
Changes to loneliness expectations (reference: stable – always agree)									
Stable – always neither							0.193	0.028	1.333
Stable – always disagree							0.233	0.038	1.443
Negative – disagree to agree/neither OR neither to agree							0.250	0.051	1.230
Positive – agree to neither/disagree OR neither to disagree							0.144*	0.028	0.756
Fluctuating							0.198*	0.041	0.960
Changes to CES-D (reference: stable - never reports depressive symptoms)									
Reduction in depressive symptoms							1.622	0.673	3.908
Increase in or persistent depressive symptoms							4.422***	2.393	8.174
Fluctuating							2.734*	1.251	5.975

Notes: binary logistic regression, N=523, DV – UCLA-LS (loneliness occurring after the start of care provision); \*p<.05, \*\*p<.01, \*\*\*p<.001; source: author's analysis of ELSA waves 2-8

## **Glossary of Terms**

Capacity to care	Caregivers' physical ability to care (e.g. due to adequate health, or availability).
Caregiver	An individual who provides unpaid assistance to someone (typically a family member, friend or neighbour) with disabilities, or chronic physical or mental health conditions.
Caregiver characteristics	The sociodemographic and socioeconomic characteristics of caregivers.
Caregiving relationship	The inter-personal relationship between caregiver and care recipient, and whether the care provided is co-residential or non-residential.
Caregiving role	The intensity of the care provided and the number of care recipients a caregiver provides care to.
Compound caregiver	Caregivers who undertake an additional caregiving role (e.g. to a spouse, or sibling) alongside providing care to a dependent child with developmental or intellectual disabilities.
Co-residential care	Care provided to a care recipient living within the same household.
Filial care	Care provided by an adult-child to an older parent.
Healthy caregiver effect	Caregivers demonstrating better health than non-caregivers because, principally, their health permits them to provide care, not exhibiting good health due to their caregiving role.
Healthy worker effect	Employed individuals demonstrating better health than non- employed because, principally, their health permits them to work, not exhibiting good health due to their employment.
High-intensity care	Care provided for 20 or more hours per week.
Invisible care	Observations or unseen tasks related to caregiving (e.g. monitoring symptoms).
Low-intensity care	Care provided for less than 20-hours per week.
Non-residential care	Care provided to a care recipient living in a different household.
Parental care	Care provided by parents to their children, typically adult- children with developmental or intellectual disability.
Round-the-clock care	Care provided 168-hours per week, i.e. 24-hours per day, 7 days per week.
Sandwich caregiver	Caregivers to both a surviving parents and a young child or grandchild.
Spousal care	Care provided to a spouse.
Visible care	Caregiving behaviours or actions which are more easily perceived by others (e.g. provision of personal care tasks).

Abajo, M., Rodríguez-Sanz, M., Malmusi, D., Salvador, M. and Borrell, C. (2017). Gender and socio-economic inequalities in health and living conditions among co-resident informal caregivers: a nationwide survey in Spain. *Journal of Advanced Nursing*. 73(3), 700-715.

Abete, P., Della-Morte, D., Gargiulo, G., Basile, C., Langellotto, A., Galizia, G., Testa, G., Canonico, V., Bonaduce, D. and Cacciatore, F. (2014). Cognitive impairment and cardiovascular diseases in the elderly. A heart-brain continuum hypothesis. *Ageing Research Reviews*. 18(1), 41-52.

Age UK (2017). Half of UK workers aged 40-64 won't have enough money to retire when they reach their State Pension Age. [*Online*]. https://www.ageuk.org.uk/latest-press/articles/2017/october/not-enough-money-to-retire/ (19/05/2021).

Age UK. (2018). All the Lonely People: Loneliness in Later Life. [*Online*]. https://www.ageuk.org.uk/globalassets/age-uk/documents/reports-and-publications/reports-and-briefings/loneliness/loneliness-report\_final\_2409.pdf (16/12/2020).

Akpinar, B., Küçükgüçlü, Ö. and Yener, G. (2011). Effects of gender on burden among caregivers of Alzheimer's patients. *Journal of Nursing Scholarship*. 43(3), 248-254.

Algren, M.H., Ekholm, O., Nielsen, L., Ersbøll, A.K., Bak, C.K. and Andersen, P.T. (2020). Social isolation, loneliness, socioeconomic status, and health-risk behaviour in deprived neighbourhoods in Denmark: a cross-sectional study. *SSM – Population Health.* 10(1), https://doi.org/10.1016/j.ssmph.2020.100546.

Alin, A. (2010). Multicollinearity. *Wiley Interdisciplinary Reviews: Computational Statistics*. 2(3), 370-374.

Alley, D.E., Putney, N.M., Rice, M. and Bengtson, V.L. (2010). The increasing use of theory in social gerontology: 1990-2004. *The Journals of Gerontology Series B: Psychological Sciences and Social Sciences*. 65(5), 583-590.

Anders, J. and Dorsett, R. (2017). What young English people do once they reach schoolleaving age: a cross-cohort comparison for the last 30 years. *Longitudinal and Life Course Studies*. 8(1), 75-103.

Anderson, L., Hewitt, A., Pettingell, S., Lulinski, A., Taylor, M. and Reagan, J. (2018). Family and individual needs for disability supports (v.2) community report 2017. Minnesota: Research and Training Center on Community Living, Institute on Community Integration, University of Minnesota.

Anderson, L.A., Edwards, V.J., Pearson, W.S., Talley, R.C., McGuire, C. and Andresen, E.M. (2013). Adult caregivers in the United States: characteristics and differences in well-being, by caregiver age and caregiving status. *Preventing Chronic Disease*. 10(1), 1-5.

Angrisani, M., Kapteyn, A. and Schuh, S. (2015). Measuring household spending and payment habits: the role of "typical" and "specific" timeframes in survey questions. In Carroll, C.D., Crossley, T.F. and Sabelhaus, J. (eds.) *Improving the Measurement of Consumer Expenditures*. United States of Amrica: University of Chicago Press. 414-440.

Anthony, K.P., Geldhof, G.J. and Mendez-Luck, C.A. (2017). Characterizing intensity among Mexican-origin women caregivers. *The Gerontologist*. 57(6), 1084-1092.

Araujo, T., Wonneberger, A., Neijens, P. and de Vreese, C. (2017). How much time do you spend online? Understanding and improving the accuracy of self-reported measures of internet use. *Communication Methods and Measures*. 11(3), 173-190.

Arslantas, H., Adana, F., Ergin, F.A., Kayar, D. and Acar, G. (2015). Loneliness in elderly people, associated factors and its correlation with quality of life: a field study from Western Turkey. *Iranian Journal of Public Health*. 44(1), 43-50.

Asi, Y. and Williams, C. (2020). A woman's (unpaid) work: global perspectives on gender, healthcare, and caregiving. In Fiedler, B.A. (ed.) *Three facets of public health and paths to improvements*. Jacksonville, Florida: Academic Press. 261-294.

Aurora, C.L. and Elena, M.C. (2019). Binary logistic regression analysis: the indicators underlying the granting of a high value personal loan. *Academic Journal of Economic Studies*. 5(2), 193-200.

Bailis, D.S., Segall, A. and Chipperfield, J.G. (2003). Two views of self-rated general health status. *Social Science and Medicine*. 56(2), 203-217.

Baka, A., Figgou, L. and Triga, V. (2012). 'Neither agree, nor disagree': a critical analysis of the middle answer category in voting advice applications. *International Journal of Electronic Governance*. 5(3-4), 244-263.

Banks, J.W., Muriel, A. and Smith, J.P. (2011). Attrition and health in ageing studies: Evidence from ELSA and HRS. *Longitudinal and Lifecourse Studies*. 2(2), http://dx.doi.org/10.14301/llcs.v2i2.115.

Banks, J., Batty, G.D., Nazroo, J., Oskala, A. and Steptoe, A. (2018). The dynamics of ageing: evidence from the English Longitudinal Study of Ageing 2002-2016. London: Institute for Fiscal studies.

Barnhart, W.R., Ellsworth, D.W., Robinson, A.C., Myers, J.V., Andridge, R.R. and Havercamp, S.M. (2020). Caregiving in the shadows: National analysis of health outcomes and intensity and duration of care among those who care for people with mental illness and for people with developmental disabilities. *Disability and Health Journal*. 13(1), https://doi.org/10.1016/j.dhjo.2019.100837.

Barreto, M., Victor, C., Hammond, C., Eccles, A., Richins, M.T. and Qualter, P. (2020). Loneliness around the world: age, gender, and cultural differences in loneliness. *Personality and Individual Differences*. 169. https://doi.org/10.1016/j.paid.2020.110066.

Bartlett, J.W., Carpenter, J.R., Tilling, K. and Vansteelandt, S. (2014). Improving upon the efficiency of complete case analysis when covariates are MNAR. *Biostatistics*. 15(4), 719-730.

Bauer, J.M. and Sousa-Poza, A. (2015). Impacts of informal caregiving on caregiver employment, health, and family. *Population Ageing*. 8(3), 113-145.

Beaumont, J. (2011). Measuring national well-being: a discussion paper on domains and measures. Newport: Office for National Statistics.

Bédard, M., Kuzik, R., Chambers, L., Molloy, D., Dubois, S. and Lever, J.A. (2005). Understanding burden differences between men and women caregivers: the contribution of care-recipient problem behaviours. *International Psychogeriatrics*. 17(1), 99-118.

Beeson, R.A. (2003). Loneliness and depression in spousal caregivers of those with Alzheimer's disease versus non-caregiving spouses. *Archives of Psychiatric Nursing*. 17(3), 135-143.

Benjamin, E.J., Muntner, P., Alonso, A., Bittencourt, M.S., Callaway, C.W., Carson, A.P., Chamberlain, A.M., Chang, A.R., Cheng, S., Das, S.R., Delling, F.N., Djousse, L., Elkind, M.S.V., Ferguson, J.F., Fornage, M., Jordan, L.C., Khan, S.S., Kissela, B.M., Knutson, K.L., Kwan, T.W., Lackland, D.T., Lewis, T.T., Lichtman, J.H., Longenecker, C.T., Loop, M.S., Lutsey, P.L., Martin, S.S., Matsushita, K., Moran, A.E., Mussolino, M.E., O'Flaherty, M., Pandey, A., Perak, A.M., Rosamond, W.D., Roth, G.A., Sampson, U.K.A., Satou, G.M., Schroeder, E.B., Shah, S.H., Spartano, N.L., Stokes, A., Tirschwell, D.L., Tsao, C.W., Turakhia, M.P., VanWagner, L.B., Wilkins, J.T., Wong, S.S. and Virani, S.S. (2019). Heart disease and stroke statistics – 2019 update. *Circulation*. 139(1), 56-528.

Ben-Zur, H. (2012). Loneliness, optimism, and well-being among married, divorced, and widowed individuals. *Journal of Psychology*. 146(1-2), 23-36.

Bergman, E.J. and Pulling, B.W. (2020). Caregiving of the other adult. In Avers, D. and Wong, R. (eds.) *Guccione's Geriatric Physical Therapy*. 4<sup>th</sup> edn. Missouri: Elsevier. 265-282.

Bevans, M.F. and Sternberg, E.M. (2012). Caregiving burden, stress, and health effects among family caregivers of adult cancer patients. *Journal of the American Medical Association*. 307(4), 398-403.

Bibb, S.C.G. (2007). Issues associated with secondary analysis of population health data. *Applied Nursing Research*. 20(2), 94-99.

Biggs, S., Lowenstein, A. and Hendricks, J. (2003). *The Need for Theory: Critical Approaches to Social Gerontology*. New York: Baywood Publishing Company.

Blanchard, C.T. (2016). Assessing whether the number of predictors affect the power of the Hosmer-Lemeshow test in large data sets. (Doctoral dissertation, The University of Texas School of Public Health).

Blazer, D. (2020). Social isolation and loneliness in older adults – a mental health/ public health challenge. *JAMA Psychiatry*. 77(10), 990-991.

Böger, A. and Huxhold, O. (2018). Age-related changes in emotional qualities of the social network from middle adulthood into old age: how do they relate to the experience of loneliness? *Psychology and Aging.* 33(3), 482-496.

Boling, P.A. (2010). Effects on policy, reimbursement, and regulation on home health care. In Olsen, S. (ed.) *The Role of human factors in home health care: workshop summary.* Washington DC: National Academies Press. 275-302.

Borys, S. and Perlman, D. (1985). Gender differences in loneliness. *Personality and Social Psychology Bulletin.* 11(1), 63-74.

Bouchard, K., Greenman, P. S., Pipe, A., Johnson, S. M. and Tulloch, H. (2019). Reducing caregiver distress and cardiovascular risk: a focus on caregiver-patient relationship quality. *Canadian Journal of Cardiology*. 35(10), 1409-1411.

Bramboeck, V., Moeller, K., Marksteiner, J. and Kaufmann, L. (2020). Loneliness and burden perceived by family caregivers of patients with Alzheimer's disease. *American Journal of Alzheimer's Disease and Other Dementias*. 35(1), 1-8.

Braveman, P.A., Cubbin, C., Egerter, S., Chideya, S., Marchi, K.S., Metzler, M. and Posner, S. (2005). Socioeconomic status in health research: one size does not fit all. *JAMA*. 294(22), 2879-2888.

Bremer, P., Challis, D., Hallberg, I.R., Leino-Kilpi, H., Saks, K., Vellas, B., Zwakhalen, S.M.G. and Sauerland, D. (2017). Informal and formal care: substitutes or complements in care

for people with dementia? Empirical evidence for 8 European countries. *Health Policy*. 121(6), 613-622.

British Heart Foundation. (2019). Vascular dementia. [*Online*]. https://www.bhf.org.uk/informationsupport/conditions/vascular-dementia (17/12//2020).

Brodaty, H. and Green, A. (2002). Defining the role of the caregiver in Alzheimer's disease treatment. *Drugs and Aging*. 19(12), 891-898.

Broese van Groenou, M.I. and de Boer, A. (2016). Providing informal care in a changing society. *European Journal of Ageing*. 13(3), 271-279.

Bryman, A. (2016). Social Research Methods. 5th edn. Oxford: Oxford University Press.

Bu, F., Steptoe, A. and Fancourt, D. (2020). Who is lonely in lockdown? Cross-cohort analyses of predictors of loneliness before and during the COVID-19 pandemic. *Public Health.* 186(1), 31-34.

Buecker, S., Denissem J.J.A. and Luhmann, M. (2020). A propensity-score matched study of changes in loneliness surrounding major life events. *Journal of Personality and Social Psychology*. https://doi.org/10.1037/pspp0000373

Burger, J.M. (1995). Individual differences in preference for solitude. *Journal of Research in Personality*. 29(1), 85-108.

Burholt, V. and Scharf, T. (2014). Poor health and loneliness in later life: the role of depressive symptoms, social resources, and rural environments. *The Journals of Gerontology Series B: Psychological Sciences and Social Sciences.* 69(2), 311-324.

Byram, E. (2018). Late-life challenges in caregiving for an adult child with a developmental disability. *Journal of the American Society on Aging*. 42(3), 9-14.

Cacioppo, J.T., Hawkley, L.C., Crawford, L.E., Ernst, J.M., Burleson, M.H., Kowalewski, RB., Malarkey, W.B., van Cauter, E. and Berntson, G.G. (2002). Loneliness and health: potential mechanisms. *Psychosomatic Medicine*. 64(3), 407-417.

Calasanti, T. and King, N. (2007). Taking 'women's work' 'like a man': Husbands' experiences of care work. *The Gerontologist*. 47(4), 516-527.

Campaign to End Loneliness. (no date). Measuring your impact on loneliness in later life. [*Online*]. https://www.campaigntoendloneliness.org/wp-content/uploads/Loneliness-Measurement-Guidance1.pdf (17/12/2020).

Campbell, L.D. (2010). Sons who care: examining the experience and meaning of filial caregiving for married and never married sons. *Canadian Journal on Aging*. 29(1), 73-84.

Canham, S.L., Mauro, P.M., Kaufmann, C.N. and Sixsmith, A. (2016). Association of alcohol use and loneliness frequency among middle-aged and older adult drinkers. *Journal of Aging and Health.* 28(2), 267-284.

Caputo, J., Pavalko, E.K. and Hardy, M.A. (2016). The long-term effects of caregiving on women's health and mortality. *Journal of Marriage and Family*. 78(5), 1382-1398.

Card, A.J. (2017). Moving beyond the WHO definition of health: a new perspective for an aging world and the emerging era of value-based care. *World Medical Health Policy*. 9(1), 127-137.

Carers UK. (2018). State of Caring 2018. [*Online*.] https://www.carersuk.org/images/Downloads/SoC2018/State-of-Caring-report-2018.pdf (17/12/2020).

Carers UK. (2020). Carers innovation fund suspended. [*Online*]. http://www.carersuk.org/news-and-campaigns/news/carers-innovation-fund-suspended (28/09/2020).

Carers Week. (2020). Carers Week 2020 research report: the rise in the number of unpaid carers during the coronavirus (COVID-19) outbreak. [*Online*]. https://www.carersweek.org/images/CW%202020%20Research%20Report%20WEB.pdf (16/12/2020).

Carmichael, F., Charles, S. and Hulme, C. (2010). Who will care? Employment participation and willingness to supply informal care. *Journal of Health Economics*. 29(1), 182-190.

Carr, E., Murray, E.T., Zaninotto, P., Cadar, D., Head, J., Stansfeld, S. and Stafford, M. (2018). The association between informal caregiving and exit from employment among older workers: prospective findings from the UK Household Longitudinal Study. *The Journals of Gerontology Series B: Psychological Sciences and Social Sciences*. 73(7), 1253-1262.

Casado, B. and Sacco, P. (2012). Correlates of caregiver burden among family caregivers of older Korean Americans. *The Journals of Gerontology Series B: Psychological Sciences and Social Sciences*. 67(3), 331-336.

Cavallero, P., Morino-Abbele, F. and Bertocci, B. (2007). The social relations of the elderly. *Archives of Gerontology and Geriatrics*. 44(1), 97-100.

Chanfreau, J., Lloyd, C., Byron, C., Roberts, C., Craig, R., De Feo, D. and McManus, S. (2013). Predicting wellbeing. London: NatCen Social Research.

Chang, H.Y., Chiou, C.J. and Chen, N.S. (2010). Impact of mental health and caregiver burden on family caregivers' physical health. *Archives of Gerontology and Geriatrics*. 50(3), 267-271.

Chang, L. and Krosnick, J.A. (2003). Measuring the frequency of regular behaviours: comparing the 'typical week' to the 'past week'. *Sociological Methodology*. 33(1), 55-80.

Chari, A.V., Engberg, J., Ray, K.N. and Mehrotta, A. (2015). The opportunity costs of informal eldercare in the United States: new estimates from the American Time Use Survey. *Health Services Research*. 50(3), 871-882.

Chen, L., Zhao, N., Fan H. and Coyte, P.C. (2017). Informal care and labor marker outcomes: evidence from Chinese married women. *Research on Aging*. 39(2), 345-371.

Chiao, C.Y., Wu, H.S. and Hsiao, C.Y. (2015). Caregiver burden for informal caregivers of patients with dementia: A systematic review. *International Nursing Review*. 62(3), 340-350.

Chou, K.R. (2000). Caregiver burden: a concept analysis. *Journal of Paediatric Nursing*. 15(6), 398-407.

Ciccarelli, N. and van Soest, A. (2018). Informal caregiving, employment status and work hours of the 50+ population in Europe. *The Economist.* 166(3), 363-396.

Clark, A.M., Reid, M.E., Morrison, C.E., Capewell, S., Murdoch, D.L. and McMurray, J.J. (2008). The complex nature of informal care in home-based heart failure management. *Journal of Advanced Nursing*. 61(4), 373-383.

Clemens, S., Phelps, A., Oldfield, Z., Blake, M., Oskala, A., Marmot, M., Rogers, N., Banks, J., Steptoe, A. and Nazroo, J. (2019). *English Longitudinal Study of Ageing: Waves 0-8, 1998-2017.* [Data collection]. 30<sup>th</sup> edn. UK Data Service. SN:5050.

Cohen, S.A., Sabik, N.J., Cook, S.K., Azzoli, A.B. and Mendez-Luck, C.A. (2019). Differences within differences: gender inequalities and caregiving intensity vary by race and ethnicity in informal caregivers. *Journal of Cross-Cultural Gerontology*. 34(3), 245-263.

Cohen-Mansfield, J., Hazan, H., Lerman, Y. and Shalom, V. (2016). Correlates and predictors of loneliness in older-adults: a review of quantitative results informed by qualitative insights. *International Psychogeriatrics*. 28(4), 557-576.

Colombo, F., Llena-Nozal, A., Mercier, J. and Tjadens, F. (2011). Help wanted? Providing and paying for long-term care. OECD Health Policy Studies: OECD Publishing. http://dx.doi.org/10.1787/9789264097759-en

Cook, S.K. and Cohen, S.A. (2018). Sociodemographic disparities in adult child informal caregiving intensity in the United States: results from the new National Study of Caregiving. *Journal of Gerontological Nursing*. 44(9), 15-20.

Corey, K.L. and McCurry, M.K. (2018). When caregiving ends: the experiences of former family caregivers of people with dementia. *The Gerontologist.* 58(2), 87-96.

Costa, N., Ferlicoq, L., Derumeaux-Burel, H., Rapp, T., Garnault, V., Gillette-Guyonnet, S., Andrieu, S., Vellas, B., Lamure, M., Grand, A. and Moliner, L. (2013). Comparison of informal care time and costs in different age-related dementias: a review. *BioMed Research International*. 2013, https://doi.org/10.1155/2013/852368.

Cottagiri, SA and Sykes, P. (2019). Systematic review in public health. Key health impacts and support systems for informal carers in the UK: A thematic review. *Journal of Health and Social Sciences*. doi10.19204/2019/kyhl11.

Courtin, E. and Knapp, M. (2017). Social isolation, loneliness and health in old age: a scoping review. *Health and Social Care in the Community*. 25(3), 799-812.

Crimmins, E.M., Kim, J.K. and Solé-Auró, A. (2010). Gender differences in health: results from SHARE, ELSA and HRS. *European Journal of Public Health.* 21(1), 81-91.

Crosnoe, R. and Elder Jr., G.H. (2015). Life course: sociological aspects. *International Encyclopedia of the Social and Behavioural Sciences*. 14(1), 80-84.

Cummins, R.A. and Nistico, H. (2002). Maintaining life satisfaction: the role of positive cognitive bias. *Journal of Happiness Studies*. 3(1), 37-69.

Dafoe, W.A. and Colella, T.J.F. (2016). Loneliness, marriage and cardiovascular health. *European Journal of Preventive Cardiology*. 23(12), 1242-1244.

Dahlberg, L. and McKee, K.J. (2014). Correlates of social and emotional loneliness in older people: evidence from an English community study. *Aging and Mental Health.* 18(4), 504-514.

Dahlberg, L., Demack, S. and Bambra, C. (2007). Age and gender of informal carers: a population-based study in the UK. *Health and Social Care in the Community*. 15(5), 439-445.

Dahlberg, L., Agahi, N. and Lennartsson, C. (2018a). Lonelier than ever? Loneliness of older people over two decades. *Archives of Gerontology and Geriatrics*. 75(1), 96-103.

Dahlberg, L., Andersson, L. and Lennartsson, C. (2018b). Long-term predictors of loneliness in old age: results of a 20-year national study. *Aging and Mental Health*. 22(2), 190-196.

Dahlrup, B., Ekström, H., Nordell, E. and Elmståhl, S. (2015). Coping as a caregiver: a question of strain and its consequences on life satisfaction and health-related quality of life. *Archives of Gerontology and Geriatrics*. 61(2), 261-270.

Danilovich, M., Xiang, X.L. and Pinto, D. (2017). Factors that influence self-reported health changes with caregiving. *Journal of Ageing and Health*. 29(8), 1444-1458.

Daoud, J.I. (2017). Multicollinearity and regression analysis. *Journal of Physics: Conference Series*. 949, 012009.

Darin-Mattson, A., Fors, S. and Kåreholt, I. (2017). Different indicators of socioeconomic status and their relative importance as determinants of health in old age. *International Journal for Equity in Health.* 16(173), 1-11.

Davies, N., Walker, N., Hopwood, J., Iliffe, S., Rait, G. and Walters, K. (2019). A "separation of worlds": the support and social networks of family carers of people with dementia at the end of life, and possible role of the internet. *Health and Social Care in the Community*. 27(4), 223-232.

[DCMS] Department of Culture, Media and Sport. (2017). Community life online and paper survey technical report 2016-2017. London: HM Government.

[DCMS] Department of Culture, Media and Sport. (2018). A connected society: a strategy for tackling loneliness. London: HM Government.

De Jong Gierveld, J. (1998). A review of loneliness: concept and definitions, determinants and consequences. *Reviews in Clinical Gerontology*. 8(1), 73-80.

De Jong Gierveld, J. and van Tilburg, T. (2006). A 6-item scale for overall, emotional, and social loneliness. *Research on Aging*. 28(5), 582-598.

De Jong Gierveld, J. and van Tilburg, T. (2010). The de Jong Gierveld short scales for emotional and social loneliness: tested on data from 7 countries in the UN generations and gender surveys. *European Journal of Aging*. 7(2), 121-130.

De Jong Gierveld, J. and van Tilburg, T. (2016). Social isolation and loneliness. *Encyclopedia* of Mental Health. 4(2), 175-178.

De Jong Gierveld, J., Broese van Groenou, M., Hoogendoorn, A.W. and Smit, J.H. (2009). Quality of marriages in later life and emotional and social loneliness. *The Journals of Gerontology Series B: Psychological Sciences and Social Sciences*. 64(4), 497-506.

De Jong Gierveld, J., Keating, N. and Fast, J.E. (2015). Determinants of loneliness among older adults in Canada. *Canadian Journal of Aging*. 34(2), 125-136.

De Koning, J.L., Stathi, A. and Richards, S. (2017). Predictors of loneliness and different types of social isolation of rural-living older adults in the United Kingdom. *Ageing and Society*. 37(10), 2012-2043.

De Labra, C., Millán-Calenti, J.C., Buján, A., Núñez-Naveira, L., Jensen, A.M., Peersen, M.C., Mojs, E., Samborski, W. and Maseda, A. (2015). Predictors of caregiving satisfaction in informal caregivers of people with dementia. *Archives of Gerontology and Geriatrics*. 60(3), 380-388.

Demakakos, P., Biddulph, J.P., Bobak, M. and Marmot, M.G. (2016). Wealth and mortality at older ages: a prospective cohort study. *Journal of Epidemiology and Community Health*. 70(4), 346-353.

De Oliveira, G.R., Neto, J.F., de Camargo, S.M., Lucchetti, A.L.G., Espinha, D.C.M. and Lucchetti, G. (2015). Caregiving across the lifespan: comparing caregiver burden, mental health, and quality of life. *Psychogeriatrics*. 15(2), 123-132.

Department of Health. (2014). The relationship between wellbeing and health. [*Online*]. https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment\_data/fi le/295474/The\_relationship\_between\_wellbeing\_and\_health.pdf (17/12/2020).

De Roos, A., van der Grond, J., Mitchell, G. and Westenberg, J. (2017). Magnetic resonance imaging of cardiovascular function and the brain: Is dementia a cardiovascular-driven disease? *Circlulation*. 135(22), 2178-2195.

[DHSC] Department of Health and Social Care. (2018). Carers action plan: 2018-2020. London: HM Government.

[DHSC] Department of Health and Social Care. (2019). £5 million fund to develop support for unpaid carers. [*Online*]. https://www.gov.uk/government/news/5-million-fund-to-develop-support-for-unpaid-carers (17/12/2020).

Diabetes UK. (no date). Diabetes and heart disease. [*Online*]. https://www.diabetes.org.uk/guide-to-diabetes/complications/cardiovascular\_disease (17/12/2020).

Diener, E.D., Emmons, R.A., Larsen, R.J. and Griffin, S. (1985). The satisfaction with life scale. *Journal of Personality Assessment*. 49(1), 71-75.

Diener, E.D., Suh, E.M., Lucas, R.E. and Smith, H.L. (1999). Subjective well-being: three decades of progress. *Psychological Bulletin.* 125(2), 276-302.

Di Gessa, G. and Grundy, E. (2017). The dynamics of paid and unpaid activities among people aged 50-69 in Denmark, France, Italy, and England. *Research on Aging*. 39(9), 1013-1038.

Dodeen, H. (2015). The effects of positively and negatively worded items on the factor structure of the UCLA loneliness scale. *Journal of Psychoeducational Assessment*. 33(3), 259-267.

Dodge, R., Daly, A.P., Huyton, J. and Sanders, L.D. (2012). The challenge of defining wellbeing. *International Journal of Wellbeing*. 2(3), 222-235.

Dogra, S., Al-Sahab, B., Manson, J. and Tamim, H. (2015). Aging expectations are associated with physical activity and health among older adults of low socioeconomic status. *Journal of Aging and Physical Activity*. 23(2), 180-186.

Dolan, P., Layard, R. and Metcalfe, R. (2011). Measuring subjective well-being for public policy. Newport: Office for National Statistics.

Domènech-Abella, J., Lara, E., Rubio-Valera, M., Olaya, B., Moneta, M.V., Rico-Uribe, L.A., Ayuso-Mateos, J.L., Mundó, J. and Haro, J.M. (2017a). Loneliness and depression in the elderly: the role of social network. *Social Psychiatry and Psychiatric Epidemiology*. 52(4), 381-390.

Domènech-Abella, J., Mundó, J., Lara, E., Moneta, M. V., Haro, J. M. and Olaya, B. (2017b). The role of socio-economic status and neighbourhood social capital on loneliness among older adults: evidence from the Sant Boi Aging Study. *Social Psychiatry and Psychiatric Epidemiology*. 52(10), 1237-1246.

Domènech-Abella, J., Mundó, J., Haro, J.M. and Rubio-Valera, M. (2019). Anxiety, depression, loneliness and social network in the elderly: Longitudinal associations from The Irish Longitudinal Study on Ageing (TILDA). *Journal of Affective Disorders*. 246(1), 82-88.

Donkor, E.S. (2018). Stroke in the 21<sup>st</sup> Century: a snapshot of the burden, epidemiology, and quality of life. *Stroke Research and Treatment*. 2018, https://doi.org/10.1155/2018/3238165.

Dos Santos-Orlandi, A.A., Brigola, A.G., Ottaviani, A.C., Luchesi, B.M., Souza, E.N., de Moura, F.G., Zacarin, J.D.F., Terassi, M., de Oliveira, N.A. and Pavarini, S.C.L. (2019). Elderly caregivers of the elderly: frailty, loneliness and depressive symptoms. *The Brazilian Journal of Nursing*. 72(2), 86-96.

Drink Aware. (2020). How much alcohol is too much? [*Online*]. https://www.drinkaware.co.uk/alcohol-facts/alcoholic-drinks-units/how-much-is-too-much/ (17/12/2020).

Dunbar, R.I.M., Launay, J., Wlodarski, R., Robertson, C., Pearce, E., Carney, J. and MacCarron, P. (2017). Functional benefits of (modest) alcohol consumption. *Adaptive Human Behaviour and Physiology*. 3(2), 118-133.

Ekwall, A.K. and Hallberg, I.R. (2007). The association between caregiving satisfaction, difficulties and coping among older family caregivers. *Journal of Clinical Nursing*. 16(5), 832-844.

Ekwall, A.K., Sivberg, B. and Hallberg, I.R. (2005). Loneliness as a predictor of quality of life among older caregivers. *Journal of Advanced Nursing*. 49(1), 23-32.

Elder Jr., G.H. (1994). Time, human agency, and social change: perspectives on the life course. *Social Psychology Quarterly*. 57(1), 4-15.

ELSA. (2019a). Confidentiality. [*Online*]. https://www.elsa-project.ac.uk/confidentiality (17/12/2020).

ELSA. (2019b). Study documentation. [*Online*]. https://www.elsa-project.ac.uk/study-documentation (17/12/2020).

ELSA. (no date [a]). Financial derived variables user guide. [*Online*]. https://www.ucl.ac.uk/epidemiology-health-care/sites/epidemiology-healthcare/files/financial\_derived\_variables\_user\_guide\_waves\_1to9.pdf (19/05/2021).

ELSA. (no date [b]). Health and lifestyles of people aged 50 and over: self-completion questionnaire. [*Online*]. https://www.ucl.ac.uk/drupal/site\_iehc/sites/iehc/files/elsa\_self-comp\_questionnaire\_w8.pdf (16/12/2020).

ELSA. (no date [c]). UK data archive data dictionaries. [Online]. https://www.elsaproject.ac.uk/study-documentation (25/06/2021).

Emerson, E., Fortune, N., Llewellyn, G. and Stancliffe, R. (2020). Loneliness, social support, social isolation and wellbeing among working age adults with and without disability: cross sectional study. *Disability and Health Journal*. 14(1) https://doi.org/10.1016/j.dhjo.2020.100965.

ESRC. (2019). ESRC's Secondary Data Analysis Initiative (SDAI) Factsheet. [*Online*]. https://esrc.ukri.org/files/funding/funding-opportunities/sdai-factsheet/ (17/12/2020).

Evans, K.L., Millsteed, J., Richmond, J.E., Falkmer, M., Falkmer, T. and Girdler S.J. (2019). The impact of within and between role experiences on role balance outcomes for working sandwich generation women. *Scandinavian Journal of Occupational Therapy*. 26(3), 184-193.

Fagerland, M.W. and Hosmer, D.W. (2017). How to test for goodness of fit in ordinal logistic regression models. *The Stata Journal*. 17(3), 668-686.

Falkingham, J., Evandrou, M., Qin, M. and Vlachantoni, A. (2020). Informal care provision across multiple generations in China. *Ageing and Society*. 40(9), 1978-2005.

Fancourt, D. and Steptoe, A. (2018). Cultural engagement predicts changes in cognitive function in older adults over a 10 year period: findings from the English Longitudinal Study of Ageing. *Scientific Reports.* 8(1), 1-8.

Farina, N., Page, T.E., Daley, S., Brown, A., Bowling, A., Basset, T., Livingston, G., Knapp, M., Murray, J. and Banerjee, S. (2017). Factors associated with the quality of life of family carers of people with dementia: a systematic review. *Alzheimer's and Dementia*. 13(5), 572-581.

Feenstra, M., van Munster, B.C., MacNeil Vroomen, J.L., de Rooij, S.E. and Smidt, N. (2020). Trajectories of self-rated health in an older general population and their determinants: the lifelines cohort study. *BMJ Open.* 10(2), 1-11.

Fernández-Ballesteros, R., Bustillos, A., Santacreu, M., Schettini, R., Díaz-Veiga, P. and Huici, C. (2016). Is older adult care mediated by caregivers' cultural stereotypes? The role of competence and warmth attribution. *Clinical Interventions in Aging*. 11(1), 545-552.

Field, A. (2013). *Discovering statistics using IBM SPSS statistics*. 4<sup>th</sup> edn. London: Sage Publications Ltd.

Finch, J. and Mason, J. (1990). Gender, employment and responsibilities to kin. *Work, Employment and Society*. 4(3), 349-367.

Finch, J. and Mason, J. (1991). Obligations of kinship in contemporary Britain: is there normative agreement?. *The British Journal of Sociology*. 42(3), 345-367.

Forgeard, M.J.C., Jayawickreme, E., Kern, M.L. and Seligman, M.E.P. (2011). Doing the right thing: measuring wellbeing for public policy. *International Journal of Wellbeing*. 1(1), 79-106.

Foster, L., Diamond, I. and Jefferies, J. (2015). *Beginning statistics: an introduction for social scientists.* 2<sup>nd</sup> edn. London: Sage Publications Ltd.

Friedemann, M.L. and Buckwalter, K.C. (2014). Family caregiver role and burden related to gender and family relationships. *Journal of Family Nursing*. 20(3), 313-336.

Friedler, B., Crapser, J. and McCullough, L. (2015). One is the deadliest number: the detrimental effects of social isolation on cerebrovascular diseases and cognition. *Acta Neuropathologica*. 129(4), 493-509.

Fuino, M. and Wagner, J. (2020). Duration of long-term care: socio-economic factors, type of care interactions and evolution. *Insurance: Mathematics and Economics*. 90(1), 151-168.

Galanaki, E. (2004). Are children able to distinguish among the concepts of aloneness, loneliness, and solitude?. *International Journal of Behavioural Development*. 28(5), 435-443.

Garbarski, D. (2016). Research synthesis: research in and prospects for the measurement of health using self-rated health. *Public Opinion Quarterly*. 80(4), 977-997.

García-Mochón, L., Peña-Longobardo, L.M., del Río-Lozano, M., Oliva-Moreno, J., Larrañaga-Padilla, I. and del Mar García-Calvente, M. (2019). Determinants of burden and satisfaction in informal caregivers: two sides of the same coin? The CUIDAR-SE Study. *International Journal of Environmental Research and Public Health*. 16(22), 4378-4390.

Gérain, P. and Zech, E. (2019). Informal caregiver burnout? Development of a theoretical framework to understand the impact of caregiving. *Frontiers in Psychology*. 10(1), 1748-1760.
Gerino, E., Rolle, L., Sechi, C. and Brustia, P. (2017). Loneliness, resilience, mental health, and quality of life in old age: a structural equation model. *Frontiers in Psychology*. 8(2003), 1-12.

Gilligan, M., Karraker, A. and Jasper, A. (2018). Linked lives and cumulative inequality: a multigenerational family life course framework. *Journal of Family Theory and Review*. 10(1), 111-125.

Golden, J., Conroy, R.M., Bruce, I., Denihan, A., Greene, E., Kirby, M. and Lawlor, B.A. (2009). Loneliness, social support networks, mood and wellbeing in community-dwelling elderly. *International Journal of Geriatric Psychiatry*. 24(7), 694-700.

Gottschalk, S., König, H.H. and Brettschneider, C. (2020). The association between informal caregiving and behavioural risk factors: a cross-sectional study. *International Journal of Public Health*. 65(1), 911-921.

Gov.UK. (2018). Maximum weekly working hours. [*Online*]. https://www.gov.uk/maximum-weekly-working-hours (16/12/2020).

Gov.UK. (2020). Community life survey 2019/2020. [*Online*]. https://www.gov.uk/government/statistics/community-life-survey-201920 (02/10/2020).

Gov.UK. (no date). Get a legal separation. [*Online*]. https://www.gov.uk/legal-separation (17/12/2020).

Green, M., Iparraguirre, J., Davidson, S., Rossall, P. and Zaidi, A. (2017). A summary of Age UK's index of wellbeing in later life. London: Age UK.

Greenstone, H.S.R., Wootton, R.E., Abdellaoui, A., Denys, D., Verweij, K.J.H., Munafò, M.R. and Treur, J.L. (2019). Bi-directional effects between loneliness and substance use: evidence from a Mendelian randomisation study. *medRxiv*. https://doi.org/10.1101/19006767.

Greenwood, N. and Smith, R. (2016). The oldest carers: a narrative review and synthesis of the experiences of carers aged over 75 years. *Maturitas*. 94(1), 161-172.

Griswold, M.E., Talluri, R., Zhu, X., Su, D., Tingle, J., Gottesman, R.F., Deal, J., Rawlings, A.M., Mosley, T.H., Windham, B.G. and Bandeen-Roche, K. (2021). Reflection on modern methods: shared-parameter models for longitudinal studies with missing data. *International Journal of Epidemiology*. doi.org/10.1093/ije/dyab086

Groarke, J.M., Berry, E., Graham-Wisener, L., McKenna-Plumley, P., McGlinchey, E. and Armour, C. (2020). Loneliness in the UK during the COVID-19 pandemic: cross-sectional results from the COVID-19 psychological wellbeing study. https://doi.org/10.31234/osf.io/j2pce.

Grundy, E. and Holt, G. (2001). The socioeconomic status of older adults: how should we measure it in studies of health inequalities?. *Journal of Epidemiology and Community Health*. 55(12), 895-904.

Hajek, A. and König, H.H. (2017). The association of falls with loneliness and social exclusion: evidence from the DEAS German Ageing Survey. *BMC Geriatrics*. 17(204), 1-11.

Hajek, A. and König, H.H. (2020). Which factors contribute to loneliness among older Europeans? Findings from the Survey of Health, Ageing and Retirement in Europe: Determinants of loneliness. *Archives of Gerontology and Geriatrics*. 89(1), 1-4.

Hamer, M., Bell, J.A., Sabia, S., Batty, G.D. and Kivimäki, M. (2015). Stability of metabolically healthy obesity over 8 years: the English Longitudinal Study of Ageing. *European Journal of Endocrinology*. 173(5), 703-708.

Hasworth, S.B. and Cannon, M.L. (2015). Social theories of ageing: a review. *Disease-A-Month.* 61(11), 475-479.

Hawkley, L.C., Hughes, M.E., Waite, L.J., Masi, C.M., Thisted, R.A. and Cacioppo, J.T. (2008). From social structural factors to perceptions of relationship quality and loneliness: The Chicago Health, Aging, and Social Relations Study. *The Journal of Gerontology Series B: Psychological Sciences and Social Sciences*. 63(6), 375-384.

Hawkley, L.C. and Kocherginsky, M. (2018). Transitions in loneliness among older adults: a 5-year follow-up in the national social life, health, and aging project. *Research on Aging.* 40(4), 365-387.

Hays, R.D. and DiMatteo, M.R. (1987). A short-form measure of loneliness. *Journal of Personality Assessment*. 51(1), 69-81.

Heinrich, L.M. and Gullone, E. (2006). The clinical significance of loneliness: a literature review. *Clinical Psychology Review*. 26(6), 695-718.

Hendricks, J. (2012). Considering life course concepts. *The Journals of Gerontology Series B: Psychological Sciences and Social Sciences*. 67(2), 226-231.

Hendricks-Lalla, A. and Pretorius, C. (2020). The male familial caregiver experience for caring for persons with Alzheimer's disease from low socioeconomic status: A South African Perspective. *Dementia.* 19(3), 618-639.

Hiel, L., Beenackers, M.A., Renders, C.M., Robroek, S.J.W., Burdof, A. and Croezen, S. (2015). Providing personal informal care to older European adults: should we care about the caregivers' health? *Preventive Medicine*. 70(1), 64-68.

Hill, R.A. and Dunbar, R.I.M. (2003). Social network size in humans. *Human Nature*. 14(1), 53-72.

HM Government. (2020). Loneliness annual report: the first year. London: HM Government.

Holmen, K., Ericsson, K. and Winblad, B. (2000). Social and emotional loneliness among nondemented and demented elderly people. *Archives of Gerontology and Geriatrics*. 31(3), 177-192.

Holt-Lunstad, J., Smith, T.B. and Layton, J.B. (2010). Social relationships and mortality risk: a meta-analytic review. *PLoS Medicine*. 7(7), 1-20.

Holwerda, T.J., Deeg, D.J.H., Beekman, A.T.F., van Tilburg, T.G., Stek, M.L., Jonker, C. and Schoevers, R.A. (2014). Feelings of loneliness, but not social isolation, predict dementia onset: results from the Amsterdam Study of the Elderly (AMSTEL). *Journal of Neurology, Neurosurgery, and Psychiatry.* 85(2), 135-142.

Honigh-de Vlaming, R., Haveman-Nies, A., Bos-Oude Groeniger, I., de Groot, L. and van't Veer, P. (2014). Determinants of trends in loneliness among Dutch older people over the period 2005-2010. *Journal of Aging and Health*. 26(3), 422-440.

Howel, D. (2012). Interpreting and evaluating the CASP-19 quality of life measure in older people. *Age and Ageing*. 41(5), 612-617.

Hsieh, N. and Hawkley, L.C. (2016). Loneliness and well-being in older adult marriages: the dyadic influences of marital quality. *Population Association of America 2016 Annual Meeting*. 1-5.

Hsieh, N. and Hawkley, L.C. (2018). Loneliness in the older adult marriage: associations with dyadic aversion, indifference, and ambivalence. *Journal of Social and Personal Relationships*. 35(10), 1319-1338.

Hughes, M.E., Waite, L.J., Hawkley, L.C. and Cacioppo, J.T. (2004). A short scale for measuring loneliness in large surveys: results from two population-based studies. *Research on Aging.* 26(6), 655-672.

Hughes, N., Locock, L. and Ziebland, S. (2013). Personal identity and the role of 'carer' among relatives and friends of people with multiple sclerosis. *Social Science and Medicine*. 96(1), 78-85.

Hughes, R.A., Heron, J., Sterne, J.A.C. and Tilling, K. (2019). Accounting for missing data in statistical analyses: multiple imputation is not always the answer. *International Journal of Epidemiology*. 48(4), 1294-1304.

Hughes, T.L., Wilsnack, S.C. and Kantor, L.W. (2016). Influence of gender and sexual orientation on alcohol use and alcohol related problems. *Alcohol Research: Current Reviews*. 38(1), 121-132.

Humble, Á.M., Seidel, A.J., Yorgason, J.B. and Redden, M. (2020). A 10-year portrait of theorizing in family gerontology: making the mosaic visible. *Journal of Family Theory and Review*. 12(2), 147-163.

Hwang, B., Luttik, M.L., Dracup, K. and Jaarsma, T. (2010). Family caregiving for patients with heart failure: types of care provided and gender differences. *Journal of Cardiac Failure*. 16(5), 398-403.

Iecovich, E. (2016). Do caregiving burden and satisfaction predict loneliness in older care recipients? *Ageing and Mental Health.* 20(4), 441-449.

Institute of Fiscal Studies. (no date). English Longitudinal Study of Ageing (ELSA) User Guide for the Wave 1 Core Dataset Version 3. [*Online*]. https://www.ifs.org.uk/elsa/user\_guides/wave\_1\_user\_guide.pdf (17/12/2020).

Jackson, S.E., Hackett, R.A. and Steptoe, A. (2019). Associations between age discrimination and health and wellbeing: cross-sectional and prospective analysis of the English Longitudinal Study of Ageing. *Lancet Public Health.* 4(4), 200-208.

Jacob, L., Smith, L., Jackson, S. E., Shin, J. I., Haro, J. M., Vancampfort, D., Stubbs, B. and Koyanagi, A. (2020). Informal caregiving and physical activity among 204,315 adults in 38 low- and middle-income countries: a cross-sectional study. *Preventive Medicine*. 132, https://doi.org/10.1016/j.ypmed.2020.106007.

Jennings, J. (1987). Elderly parents as caregivers for their adult dependent children. *Social Work*. 32(5), 430-433.

Jessen, M.A.B., Pallesen, A.V.J., Kriegbaum, M. and Kristiansen, M. (2017). The association between loneliness and health – a survey-based study among middle-aged and older adults in Denmark. *Aging and Mental Health*. 22(10), 1338-1343.

Jopling, K. (2015). Promising approaches to reducing loneliness and isolation in later life. [*Online*.] https://www.campaigntoendloneliness.org/wp-content/uploads/Promising-approaches-to-reducing-loneliness-and-isolation-in-later-life.pdf (16/12/2020).

Juntunen, K., Salminen, A.L., Tormakangas, T., Tillman, P., Leinonen, K. and Nikander, R. (2018). Perceived burden among spouse, adult child, and parent caregivers. *Journal of Advanced Nursing*. 74(10), 2340-2350.

Karim, J., Weisz, R., Bibi, Z. and Rehman, S.U. (2015). Validation of the eight-item center for epidemiologic studies depression scale (CES-D) among older adults. *Current Psychology*, 34(4), 681-692.

Kapteyn, A. (2008). Comparison between SHARE, ELSA, and HRS. In Börsch-Supan, A., Brugiavini, A., Jürges, H., Kapteyn, A., Mackenbach, J., Siegrist, J. and Weber, G. *First results from the Survey of Health, Ageing and Retirement in Europe (2004-2007): Starting the Longitudinal Dimension*. Mannheim: MEA. 23-64.

Kavirajan, H., Vahia, I.V., Thompson, W.K., Depp, C., Allison, M. and Jeste, D.V. (2011). Attitude toward own aging and mental health in post-menopausal women. *Asian Journal of Psychiatry*. 4(1), 26-30.

Keating, N. and Eales, J. (2017). Social consequences of family care of adults: a scoping review. *International Journal of Care and Caring*. 1(2), 153-173.

Kelfve, S., Fors, S. and Lennartsson, C. (2017). Getting better all the time? Selective attrition and compositional changes in longitudinal and life-course studies. *Longitudinal and Life Course Studies*. 8(1), 104-119.

Kemperman, A., van den Berg, P., Weijs-Perée, M. and Uijtdewillegen, K. (2019). Loneliness of older adults: social network and the living environment. *International Journal of Environmental Research and Public Health*. 16(3), 1-16.

Kietzman, K.G., Benjamin, A.E. and Matthias, R.E. (2008). Of family, friends, and strangers: caregiving satisfaction across three types of paid caregivers. *Home Health Care Services Quarterly*. 27(2), 100-120.

Kirchner, J.E., Zubritsky, C., Cody, M., Coakley, E., Chen, H., Ware, J.H., Oslin, D.W., Sanchez, H.A., Durai, U.N.B., Miles, K.M., Llorente, M.D., Costantino, G. and Levkoff, S. (2007). Alcohol consumption among older adults in primary care. *Journal of General Internal Medicine*. 22(1), 92-97.

Kobayashi, L.C. and Steptoe, A. (2018). Social Isolation, Loneliness, and Health Behaviours at Older Ages: Longitudinal Cohort Study. *Annals of Behavioural Medicine*. 52(7), 582-593.

Kornadt, A.E. (2016). Do age stereotypes as social role expectations for older adults influence personality development?. *Journal of Research in Personality*. 60(1), 51-55.

Kristensen, K., König, H.H. and Hajek, A. (2019). The association of multimorbidity, loneliness, social exclusion and network size: findings from the population-based German Ageing Survey. *BMC Public Health* 19(1), 1383-1392.

Kropf, N.P. and Kelly, T.B. (1995). Stigmatized and perpetual parents: older parents caring for adult children with life-long disabilities. *Journal of Gerontological Social Work*. 24(1-2), 3-16.

Kumagai, N. (2017). Distinct impacts of high intensity caregiving on caregivers' mental health and continuation of caregiving. *Health Economics Review*. 7(1), 15-27.

Lacey, R.E., McMunn, A. and Webb, E.A. (2018). Informal caregiving and metabolic markers in the UK household longitudinal study. *Maturitas*. 109(1), 97-103.

Lacey, R.E., McMunn, A. and Webb, E.A. (2019). Informal caregiving patterns and trajectories of psychological distress in the UK Household Longitudinal Study. *Psychological Medicine*. 49(10), 1652-1660.

La Grow, S., Neville, S., Alpass, F. and Rodgers, V. (2012). Loneliness and self-reported health among older persons in New Zealand. *Australian Journal on Ageing*. 31(2), 121-123.

Lang, S.F. and Fowers, B.J. (2019). An expanded theory of Alzheimer's caregiving. *American Psychologist*. 74(2), 194-206.

Larkin, M. (2009). Life after caring: the post-caring experiences of former carers. *British Journal of Social Work*. 39(6), 1026-1042.

Lawton, M.P., Kleban, M.H., Moss, M., Rovine, M. and Glicksman, A. (1989). Measuring caregiving appraisal. *The Journals of Gerontology Series B: Psychological Sciences and Social Sciences*. 44(3), 61-71.

Lee, K., Martin, P. and Poon, L.W. (2017a). Predictors of caregiving burden: impact of subjective health, negative affect, and loneliness of octogenarians and centenarians. *Ageing and Mental Health*. 21(11), 1214-1221.

Lee, M. H., Park, C., Matthews, A.K. and Hsieh, K. (2017b). Differences in physical health, and health behaviours between family caregivers of children with and without disabilities. *Disability and Health Journal*. 10(4), 565-570.

Leszko, M., Iwanski, R. and Bugajska, B. (2020). Prisoners of care: the experience of loneliness among caregivers of individuals with Alzheimer's disease. *Paedagogia Christiana*. 45(1), 171-184.

Leung, G.T.Y., de Jong Gierveld, J. and Lam, L.C.W. (2008). Validation of the Chinese translation of the 6-item de Jong Gierveld loneliness scale in elderly Chinese. *International Psychogeriatrics*. 20(6), 1262-1272.

Levin, K.A. (2006). Study design III: cross-sectional studies. *Evidence-Based Dentistry*. 7(1), 24-25.

Levy, B.R. (1996). Improving memory in old age through implicit self-stereotyping. *Journal of Personality and Social Psychology*. 71(6), 1092-1107.

Levy, B.R. (2009). Stereotype embodiment: a psychosocial approach to aging. *Current Directions in Psychological Science*. 18(6), 332-336.

Levy, B.R. and Myers, L.M. (2004). Preventive health behaviours influenced by self-perceptions of aging. *Preventive Medicine*. 39(3), 625-629.

Levy, B.R., Hausdorff, J.M., Hencke, R. and Wei, J.Y. (2000). Reducing cardiovascular stress with positive self-stereotypes of aging. *The Journals of Gerontology Series B: Psychological Sciences and Social Sciences*. 55(4), 205-213.

Levy, B.R., Slade, M.D., May, J. and Caracciolo, E.A. (2006). Physical recovery after acute myocardial infarction: positive age self-stereotypes as a resource. *International Journal of Aging and Human Development*. 62(4), 285-301.

Levy, B.R., Zonderman, A.B., Slade, M.D. and Ferrucci, L. (2009). Age stereotypes held earlier in life predict cardiovascular events in later life. *Psychological Science*. 20(3), 296-298.

Li, L. and Lee, Y. (2020). Caregiving choice and caregiver-receiver relation: effects on psychological well-being of family caregivers in Canada. *Canadian Journal on Aging*. 39(4), 634-646.

Linley, P.A., Maltby, J., Wood, A.M., Osborne, G. and Hurling, R. (2009). Measuring happiness: the higher order factor structure of subjective and psychological well-being measures. *Personality and Individual Differences*. 47(8), 878-884.

Lodder, P. (2013). To impute or not impute: That's the question. *Advising on Research Methods: Selected Topics*. Huizen, Wedekuil: Johannes van Kessel Publishing, 1-7.

Long, C.R., Seburn, M., Averill, J.R. and More, T.A. (2003). Solitude experiences: Varieties, settings, and individual differences. *Personality and Social Psychology Bulletin*. 29(5), 578-583.

Losada, A., Márquez-González, M., García-Ortiz, L., Gómez-Marcos, M.A., Fernández-Fernández, V. and Rodríguez-Sánchez, E. (2012). Loneliness and mental health in a representative sample of community-dwelling Spanish older adults. *The Journal of Psychology*. 146(3), 277-292.

Lynn, P. (2011). A fishy story: the roles of rods and nets in maintaining representative longitudinal survey samples. *Longitudinal and Life Course Studies*. 2(2), 98-100.

Lyons, J.G., Cauley, J.A. and Fredman, L. (2015). The effect of transitions in caregiving status and intensity on perceived stress among 992 female caregivers and non-caregivers. *The Journals of Gerontology Series A: Biological Sciences and Medical Sciences*. 70(8), 1018-1023.

Mann, E., Hinrichsen, G.A. and Baharlou, S. (2020). Depression in older adults: principles of diagnosis and management. In Chunn, A. (ed.) *Geriatric Practice: A Competency Based Approach to Caring for Older Adults*. Switzerland: Springer Publishing Company. 213-220.

Marcén, M. and Molina, J.A. (2012). Informal caring-time and caregiver satisfaction. *The European Journal of Health Economics*. 13(6), 683-705.

Mason, W.M. and Wolfinger, N.H. (2001). Cohort analysis. In Smelser, N.J. and Baltes, P.B. (eds.) *International Encyclopedia of Social and Behavioural Sciences*. Amsterdam: Elsevier Science. 2189-2194.

McCartney, G., Popham, F., McMaster, R. and Cumbers, A. (2019). Defining health and health inequalities. *Public Health*. 172(1), 22-30.

McCarthy, P.A. and Morina, N. (2020). Exploring the association of social comparison with depression and anxiety: a systematic review and meta-analysis. *Clinical Psychology and Psychotherapy*. 27(5), 640-671.

McCusker, K. and Gunaydin, S. (2015). Research using qualitative, quantitative or mixed methods and choice based on the research. *Perfusion*. 30(7), 537-542.

McDonald, J., Swami, N., Pope, A., Hales, S., Nissim, R., Rodin, G., Hannon, B. and Zimmermann, C. (2018). Caregiver quality of life in advanced cancer: qualitative results from a trial of palliative care. *Palliative Medicine*. 32(1), 69-78.

McDonnell, E.I. and Ryan, A. (2013). Male caregiving in dementia: a review and commentary. *Dementia*. 12(2), 238-250.

McHugh, M.L. (2013). The chi-square test of independence. *Biochemia Medica*. 23(2), 143-149.

McKay, M.T., Konowalczyk, S., Andretta, J.R. and Cole, J.C. (2017). The direct and indirect effect of loneliness on the development of adolescent alcohol use in the United Kingdom. *Addictive Behaviors Reports.* 6(1), 65-70.

McRae, C., Fazio, E., Hartsock, G., Kelley, L., Urbanski, S. and Russell, D. (2009). Predictors of loneliness in caregivers of persons with Parkinson's disease. *Parkinsonism and Related Disorders*. 15(8), 554-557.

Meisner, B.A. and Levy, B.R. (2016). Age stereotypes' influence on health: stereotype embodiment theory. In Bengtson, V.L. and Settersten, R.A. (eds.) *Handbook of Theories of Aging*. 3<sup>rd</sup> edn. New York: Springer Publishing Company. 259-275.

Mentzakis, E., McNamee, P. and Ryan, M. (2009). Who cares and how much: exploring the determinants of co-residential care. *Review of Economics of the Household*. 7(3), 283-303.

Milligan, C. and Morbey, H. (2016). Care, coping and identity: older men's experiences of spousal caregiving. *Journal of Aging Studies*. 38(1), 105-114.

Mitchell, J. (2004). The difference between gender and sexual difference. In Matthis, I. (ed.) *Dialogues on Sexuality, Gender and Psychoanalysis.* London: Routledge. 67-78.

Mochari-Greenberger, H. and Mosca, L. (2012). Caregiver burden and nonachievement of healthy lifestyle behaviours among family caregivers of cardiovascular disease patients. *American Journal of Health Promotion.* 27(2), 84-89.

Montgomery, R.J.V. (2007). Caregiver Identity Theory Overview. [*Online*.] https://www.researchgate.net/publication/265260328\_Caregiver\_Identity\_Theory\_Overview (17/12/2020).

Montgomery, R.J.V. and Kosloski, K. (2009). Caregiving as a process of changing identity: Implications for caregiver support. *Generations*. 33(1), 47-52.

Montgomery, R.J.V., Kwak, J. and Kosloski, K.D. (2016). Theories Guiding Support Services for Family Caregivers. In Bengtson, V.L. and R.A. Settersten, R.A. (eds.) *Handbook of Theories of Aging*. 3<sup>rd</sup> edn. New York: Springer Publishing Company. 443-462.

Mortensen, J., Dich, N., Lange, T., Ramlau-Hansen, C.H., Head, J., Kivimaki, M., Leineweber, C. and Rod, N.H. (2017). Weekly hours of informal caregiving and paid work, and the risk of cardiovascular disease. *The European Journal of Public Health*. 28(4), 743-747.

Moos, R.H., Brennan, P.L., Schutte, K.K. and Moos, B.S. (2010). Social and financial resources and high-risk alcohol consumption among older adults. *Alcoholism: Clinical and Experimental Research*. 34(4), 646-654.

Moyle, W., Kellett, U., Ballantyne, A. and Gracia, N. (2011). Dementia and loneliness: an Australian perspective. *Journal of Clinical Nursing*. 20(9-10), 1445-1453.

Mushtaq, R., Shoib, S., Shah, T. and Mushtaq, S. (2014). Relationship between loneliness, psychiatric disorders and physical health? A review on the psychological aspects of loneliness. *Journal of Clinical and Diagnostic Research*. 8(9), 1-4.

Nadler, J.T., Weston, R. and Voyles, E.C. (2015). Stuck in the middle: the use and interpretation of mid-points in items on questionnaires. *The Journal of General Psychology*. 142(2), 71-89.

NatCen. (2014). English Longitudinal Study of Ageing: wave six interview questionnaire – 2012 to 2013. [Online].

http://doc.ukdataservice.ac.uk/doc/5050/mrdoc/pdf/5050\_elsa\_wave\_6\_documentation.pdf (16/12/2020).

NatCen. (2018a). Interviewer questionnaire: wave 8 – 2016/17, version 1 (May 2018). [*Online*]. ucl.ac.uk/drupal/site\_iehc/sites/iehc/files/elsa\_questionnaire\_w8.pdf (16/12/2020).

NatCen. (2018b). User guide to the main interview datasets: waves 1-8. [*Online*]. http://doc.ukdataservice.ac.uk/doc/8375/mrdoc/pdf/8375\_waves\_1-8\_interviewer\_data\_user\_guide\_v01.pdf (16/12/2020).

National Alliance for Caregiving and AARP Public Policy Institute. (2015). Caregiving in the US. [*Online*]. https://www.aarp.org/content/dam/aarp/ppi/2015/caregiving-in-the-united-states-2015-report-revised.pdf (17/12/2020).

Navaie-Waliser, M., Spriggs, A. and Feldman, P.H. (2002). Informal caregiving: differential experiences by gender. *Medical Care*. 40(12), 1249-1259.

Neto, F. (2014). Psychometric analysis of the short-form UCLA Loneliness Scale (ULS-6) in older adults. *European Journal of Ageing*. 11(4), 313-319.

Ngamaba, K.H., Panagioti, M. and Armitage, C.J. (2017). How strongly related are health status and subjective well-being? Systematic review and meta-analysis. *European Journal of Public Health*. 27(5), 879-885.

Nguyen, T.T., Ryan, R.M. and Deci, E.L. (2018). Solitude as an approach to affective self-regulation. *Personality and Social Psychology Bulletin*. 44(1), 92-106.

NHS. (2017). Causes: vascular dementia. [*Online*]. https://www.nhs.uk/conditions/vascular-dementia/causes/ (17/12/2020).

NHS. (2018). The risks of drinking too much. Retrieved from: https://www.nhs.uk/live-well/alcohol-support/the-risks-of-drinking-too-much/ (17/12//2020).

NHS. (2019). Physical activity guidelines for older adults. Retrieved from: https://www.nhs.uk/live-well/exercise/physical-activity-guidelines-olderadults/?tabname=exercise-tips#guidelines-for-older-adults-aged-65-and-over (09/06/2020).

NHS Digital. (2019). Health survey for England content. [*Online*]. https://digital.nhs.uk/data-and-information/areas-of-interest/public-health/health-survey-for-england---health-social-care-and-lifestyles/survey-content (17/12/2020).

[NICE] National Institute for Health and Care Excellence. (2016). Mental wellbeing and independence for older people. [*Online*.] https://www.nice.org.uk/guidance/qs137 (17/12/2020).

Nicolaisen, M. and Thorsen, K. (2014a). Loneliness among men and women – a five year follow-up study. *Aging and Mental Health*. 18(2), 194-206.

Nicolaisen, M. and Thorsen, K. (2014b). Who are lonely? Loneliness in different age groups (18-81 years old), using two measures of loneliness. *International Journal of Aging and Human Development*. 78(3), 229-257.

Niedzwiedz, C.L., Richardson, E.A., Tunstall, H., Shortt, N.K., Mitchell, R.J. and Pearce, J.R. (2016). The relationship between wealth and loneliness among older people across Europe: is social participation protective?. *Preventive Medicine*. 91(1), 24-31.

Nomis. (2011). LC2109EWIs – Ethnic group by age. [*Online*]. http://www.nomisweb.co.uk/census/2011/LC2109EWLS/view/2092957699?rows=c\_ethpuk11 &cols=c\_age (02/10/2020).

Nomis. (2018a). Provision of unpaid care. [*Online*]. https://www.nomisweb.co.uk/census/2011/qs301uk (17/12/2020).

Nomis. (2018b). Provision of unpaid care by general health by sex by age. [*Online*]. https://www.nomisweb.co.uk/census/2011/LC3301EW/view/2092957699?rows=c\_age&cols=c\_carer (17/12/2020).

O'Donnell, P., O'Donovan, D. and Elmusharaf, K. (2018). Measuring social exclusion in healthcare settings: a scoping review. *International Journal for Equity in Health*. 17(15), 1-16.

Oldenkamp, M., Bültmann, U., Wittek, R.P.M., Stolk, R.P., Hagendoorn, M. and Smidt, N. (2018a). Combining informal care and paid work: the use of work arrangements by working

adult-child caregivers in the Netherlands. *Health and Social Care in the Community*. 26(1), 122-131.

Oldenkamp, M., Hagedoorn, M., Wittek, R.P.M., Stolk, R.P. and Smidt, N. (2018b). Chapter Six: Positive and negative caregiving experiences: the role of the quality of the informal caregiver-care recipient relationship and intrinsic caregiving motivations. In Oldenkamp, M. (ed.) *Caregiving experiences of informal caregivers: The importance of characteristics of the informal caregiver, care recipient, and care situation.* Groningen: University of Groningen. 117-146.

Oldfield, Z. (2018). ELSA financial derived variables. [*Online*]. https://www.ucl.ac.uk/epidemiology-health-care/sites/epidemiology-healthcare/files/zoe\_oldfield\_presentation\_4.pdf (19/05/2021).

Ong, A.D., Uchino, B.N. and Wethington, E. (2016). Loneliness and health in older adults: a mini-review and synthesis. *Gerontology*. 62(4), 443-449.

[ONS] Office for National Statistics. (2013). 2011 Census analysis: unpaid care in England and Wales, 2011 and comparison with 2001. [Online].

https://www.ons.gov.uk/peoplepopulationandcommunity/healthandsocialcare/healthcaresystem /articles/2011censusanalysisunpaidcareinenglandandwales2011andcomparisonwith2001/2013-02-15 (17/12/2020).

[ONS] Office for National Statistics. (2018a). Mapping the loneliness measurement landscape. [*Online*].

https://www.ons.gov.uk/peoplepopulationandcommunity/wellbeing/compendium/nationalmeas urementofloneliness/2018/mappingthelonelinessmeasurementlandscape (17/12/2020).

[ONS] Office for National Statistics. (2018b). Recommended national indicators of loneliness. [*Online*].

https://www.ons.gov.uk/peoplepopulationandcommunity/wellbeing/compendium/nationalmeas urementofloneliness/2018/recommendednationalindicatorsofloneliness (17/12/2020).

[ONS] Office for National Statistics. (2018c). Testing of loneliness questions in surveys. [*Online*].

https://www.ons.gov.uk/peoplepopulationandcommunity/wellbeing/compendium/nationalmeas urementofloneliness/2018/testingoflonelinessquestionsinsurveys (17/12/2020).

O'Reilly, D., Connolly, S., Rosato, M. and Patterson, C. (2008). Is caring associated with an increased risk of mortality? A longitudinal study. *Social Science and Medicine*. 67(8), 1282-1290.

Oxford University Press. (2020). Oxford Learner's Dictionaries: Simultaneity. [*Online*]. https://www.oxfordlearnersdictionaries.com/definition/english/simultaneity (17/12/2020).

Parliamentary Office of Science and Technology. (2018). Unpaid care. [*Online*]. https://post.parliament.uk/research-briefings/post-pn-0582/ (17/12/2020).

Penning, M.J., Liu, G. and Chou, P.H.B. (2014). Measuring loneliness among middle-aged and older adults: the UCLA and de Jong Gierveld loneliness scales. *Social Indicators Research*. 118(3), 1147-1166.

Perelli-Harris, B. and Styrc, M. (2017). Mental well-being differences in cohabitation and marriage: the role of childhood selection. *Journal of Marriage and Family*. 80(1), 239-255.

Perkins, E.A. (2009). *Caregivers of adults with intellectual disabilities: the relationship of compound caregiving and reciprocity to quality of life*. (Doctoral dissertation, University of South Florida).

Perlman, D. and Peplau, L.A. (1981). Toward a social psychology of loneliness. In Duck, S. and Gilmour, R. (eds.) *Personal Relationships: personal relationships in disorder*. London: Academic Press Inc. 31-56.

Petrie, K. and Kirkup, J. (2018). Caring for carers: the lives of family carers in the UK. [*Online*]. http://www.smf.co.uk/wp-content/uploads/2018/07/Caring-for-Carers.pdf (17/12/2020).

Phillips, J., Ajrouch, K.J. and Hillcoat-Nallétamby, S. (2010). *Key concepts in social gerontology*. London: Sage Publications.

Pikhartova, J., Bowling, A. and Victor, C. (2016). Is loneliness in later life a self-fulfilling prophecy? *Aging and Mental Health.* 20(5), 543-549.

Pillemer, K. and Suitor, J.J. (2014). Who provides care? A prospective study of caregiving among adult siblings. *The Gerontologist*. 54(4), 589-598.

Pinquart, M. and Sörensen, S. (2001). Influences on loneliness in older adults: a meta-analysis. *Basic and Applied Social Psychology*. 23(4), 245-266.

Pinquart, M. and Sörensen, S. (2011). Gender differences in self-concept and psychological well-being in old age: a meta-analysis. *The Journals of Gerontology Series B: Psychological Sciences and Social Sciences.* 56(4), 195-213.

Powell, A., Francis-Devine, B., Foster, D., Thurley, D., Roberts, N., Loft, P., Harker, R., McInnes, R., Danechi, S., Kennedy, S. and Powell, T. (2020). Informal carers. Briefing paper number 07756. London: House of Commons.

Pyle, E. and Evans, D. (2018). Loneliness – what characteristics and circumstances are associated with feeling lonely? [*Online*]. https://www.basw.co.uk/system/files/resources/basw\_51624-8.pdf (16/12/2020).

Quinn, C. and Toms, G. (2019). Influence of positive aspects of dementia caregiving on caregivers' well-being: a systematic review. *The Gerontologist*. 59(5), 584-596.

Quinn, C., Clare, L. and Woods, B. (2009). The impact of the quality of relationship on the experiences and wellbeing of caregivers of people with dementia. *Ageing and Mental Health*. 13(2), 143-154.

Quinn, C., Nelis, S.M., Martyr, A., Victor, C., Morris, R.G. and Clare, L. (2019). Influence of positive and negative dimensions of dementia caregiving on caregiver well-being and satisfaction with life: findings from the IDEAL study. *American Journal of Geriatric Psychiatry*. 27(8), 838-848.

Quirke, E., König, H.H. and Hajek, A. (2019). Association between caring for grandchildren and feelings of loneliness, social isolation and social network size: a cross-sectional study of community dwelling adults in Germany. *BMJ Open.* 9(12), 1-8.

Rappange, D.R., Brouwer, W.B.F. and van Exel, J. (2016). A long life in good health: subjective expectations regarding length and future health-related quality of life. *European Journal of Health Economics.* 17(5), 577-589.

Richard, A., Rohrmann, S., Vandeleur, C.L., Schmid, M., Barth, J. and Eichholzer, M. (2017). Loneliness is adversely associated with physical and mental health and lifestyle factors: results from a Swiss national survey. *PLoS One.* 12(7), https://doi.org/10.1371/journal.pone.0181442.

Rius-Ottenheim, N., Kromhout, D., van der Mast, R.C., Zitman, F.G., Geleijnse, J.M. and Giltay, E.J. (2012). Dispositional optimism and loneliness in older men. *International Journal of Geriatric Psychiatry*. 27(2), 151-159.

Robards, J., Vlachantoni, A., Evandrou, M. and Falkingham, J. (2015). Informal caring in England and Wales– Stability and transition between 2001 and 2011. *Advances in Life Course Research*. 24(1), 21-33.

Roberto, K.A., Blieszner, R. and Allen, K.R. (2006). Theorizing in family gerontology: new opportunities for research and practice. *Family Relations*. 55(5), 513-525.

Rospenda, K.M., Minich, L.M., Milner, L.A. and Richman, J.A. (2010). Caregiver burden and alcohol use in a community sample. *Journal of Addictive Diseases*. 29(3), 314-324.

Russell, D., Peplau, L.A. and Cutrona, C.E. (1980). The revised UCLA Loneliness Scale: concurrent and discriminant validity evidence. *Journal of Personality and Social Psychology*. 39(3), 472-480.

Russell, D., Peplau, L.A. and Ferguson, M.L. (1978). Developing a measure of loneliness. *Journal of Personality Assessment.* 42(3), 290-294.

Rutherford, A. and Bowes, A. (2014). Networks of informal caring: a mixed-methods approach. *Canadian Journal on Aging*. 33(4), 473-487.

Rutherford, A. and Bu, F. (2018). Issues with the measurement of informal care in social surveys: evidence from the English Longitudinal Study of Ageing. *Ageing and Society*. 38(12), 2541-2559.

Ryff, C.D. (1989). Happiness is everything, or is it? Explorations on the meaning of psychological wellbeing. *Journal of Personality and Social Psychology*. 57(6), 1069-1081.

Ryff, C.D. (2014). Psychological well-being revisited: advances in the science and practice of eudaimonia. *Psychotherapy and Psychosomatics*. 83(1), 10-28.

Sanders, S. and Power, J. (2009). Roles, responsibilities, and relationships among older husbands caring for wives with progressive dementia and other chronic conditions. *Health and Social Work*. 34(1), 41-51.

Sarkisian, C.A., Hays, R.D. and Mangione, C.M. (2002). Do older adults expect to age successfully? The association between expectations regarding aging and beliefs regarding healthcare seeking among older adults. *Journal of the American Geriatrics Society*. 50(11), 1837-1843.

Sassler, S. and Lichter, D.T. (2020). Cohabitation and marriage: complexity and diversity in union-formation patterns. *Journal of Marriage and Family*. 82(1), 35-61.

Schmidt, A.E., Ilinca, S., Schulmann, K., Rodrigues, R., Principi, A., Barbabella, F., Sowa, A., Golinowska, S., Deeg, D. and Galenkamp, H. (2016). Fit for caring: factors associated with informal care provision by older caregivers with and without multimorbidity. *European Journal of Ageing*. 13(2), 103-113.

Schrank, B., Ebert-Vogel, A., Amering, M., Masel, E.K., Neubauer, M., Watzke, H., Zehetmayer, S. and Schur, S. (2016). Gender differences in caregiver burden and its determinants in family members of terminally ill cancer patients. *Psycho-Oncology*. 25(7), 808-814.

Schulz, R., Beach, S.R., Czaja, S.J., Martire, L.M. and Monin, J.K. (2020). Family caregiving for older adults. *Annual Review of Psychology*. 71(1), 635-659.

Schulz, R., Newsom, J., Mittelmark, M., Burton, L., Hirsch, C. and Jackson, S. (1997). Health effects of caregiving: The caregiver health effects study: An ancillary study of the cardiovascular health study. *Annals of Behavioural Medicine*. 19(2), 110-116.

Sedgwick, P. (2014). Cross sectional studies: advantages and disadvantages. *BMJ*. 348, https://doi.org/10.1136/bmj.g2276.

Segrin, C., McNelis, M. and Pavlich, C.A. (2018). Indirect effects of loneliness on substance use through stress. *Health Communication*. 33(5), 513-518.

Semyonov, M., Lewin-Epstein, N. and Maskileyson, D. (2013). Where wealth matters more for health: the wealth-health gradient in 16 countries. *Social Science and Medicine*. 81(1), 10-17.

Sequeira, C. (2013). Difficulties, coping strategies, satisfaction and burden in informal Portuguese caregivers. *Journal of Clinical Nursing*. 22(3-4), 491-500.

Shankar, A., McMunn, A., Banks, J. and Steptoe, A. (2011). Loneliness, social isolation, and behavioural and biological health indicators in older adults. *Health Psychology*. 30(4), 377-385.

Shankar, A., Hamer, M., McMunn, A. and Steptoe, A. (2013). Social isolation and loneliness: relationships with cognitive function during 4 years of follow-up in the English Longitudinal Study of Ageing. *Psychosomatic Medicine*. 75(2), 161-170.

Shim, B., Barroso, J. and Davis, L.L. (2012). A comparative qualitative analysis of stories of spousal caregivers of people with dementia: negative, ambivalent, and positive experiences. *International Journal of Nursing Studies*. 49(2), 220-229.

Shiovitz-Ezra, S. and Ayalon, L. (2012). Use of direct versus indirect approaches to measure loneliness in later life. *Research on Aging*. 34(5), 572-591.

Sim, J., Bartlam, B. and Bernard, M. (2011). The CASP-19 as a measure of quality of life in old age: evaluation of its use in a retirement community. *Quality of Life Research*. 20(7), 997-1004.

Sin, J., Murrells, T., Spain, D., Norman, I. and Henderson, C. (2016). Wellbeing, mental health knowledge and caregiving experiences of siblings of people with psychosis, compared to their peers and parents: an exploratory study. *Social Psychiatry and Psychiatric Epidemiology*. 51(9), 1247-1255.

Smith, K.J. and Victor, C. (2019). Typologies of loneliness, living alone and social isolation, and their associations with physical and mental health. *Ageing and Society*. 39(8), 1709-1730.

Smith, K.V. and Goldman, N. (2007). Socioeconomic differences in health among older adults in Mexico. *Social Science and Medicine*. 65(7), 1372-1385.

Smith, T., Saunders, A. and Heard, J. (2020). Trajectory of psychosocial measures amongst informal caregivers: case-controlled study of 1375 informal caregivers from the English Longitudinal Study of Ageing. *Geriatrics*. 5(26), 1-17.

Soylu, C., Ozaslan, E., Karaca, H. and Ozkan, M. (2016). Psychological distress and loneliness in caregiver of advanced oncological inpatients. *Journal of Health Psychology*. 21(9), 1896-1906.

Steffick, D.E. (2000). Documentation of affective functioning measures in the Health and Retirement Study. [*Online*.] http://hrsonline.isr.umich.edu/sitedocs/userg/dr-005.pdf (17/12/2020).

Stephens, C., Alpass, F., Towers, A. and Stevenson, B. (2011). The effects of types of social networks, perceived social support, and loneliness on the health of older people: accounting for the social context. *Journal of Aging and Health.* 23(6), 887-911.

Steptoe, A. Breeze, E., Banks, J. and Nazroo, J. (2013a). Cohort profile: the English Longitudinal Study of Ageing. *International Journal of Epidemiology*. 42(6), 1640-1648.

Steptoe, A., Shankar, A., Demakakos, P. and Wardle, J. (2013b). Social isolation, loneliness, and all-cause mortality in older men and women. *Proceedings of the National Academy of Sciences of the United States of America.* 110(15), 5797-5801.

Štípková, M. (2019). Marital status, close social network and loneliness of older adults in the Czech Republic. *Ageing and Society*. 1-15. doi:10.1017/S0144686X19001442.

Stockemer, D. (2019). Quantitative methods for the social sciences: a practical introduction with examples in SPSS and Stata. Switzerland: Springer International Publishing.

Swinkels, J., van Tilburg, T., Verbakel, E. and Broese van Groenou, M. (2017). Explaining the gender gap in the caregiving burden of partner caregivers. *The Journals of Gerontology: Series B: Psychological Sciences and Social Sciences*. 74(2), 309-317.

Tatangelo, G., McCabe, M., Macleod, A. and You, E. (2018). "I just don't focus on my needs." The unmet health needs of partner and offspring caregivers of people with dementia: A qualitative study. *International Journal of Nursing Studies*. 77(1), 8-14.

Taylor, R., Conway, L., Calderwood, L., Lessof, C., Cheshire, H., Cox, K. and Scholes, S. (2007). Health, wealth and lifestyles of the older population in England: the 2002 English Longitudinal Study of Ageing technical report. [*Online*]. https://www.ifs.org.uk/elsa/report03/w1\_tech.pdf (17/12/2020).

Tengland, P.A. (2007). A two-dimensional theory of health. *Theoretical Medicine and Bioethics*. 28(4), 257-284.

Thomas, C. (1993). De-constructing concepts of care. Sociology. 27(4), 649-669.

Tiilikainen, E. and Seppanen, M. (2017). Lost and unfulfilled relationships behind emotional loneliness in old age. *Ageing and Society*. 37(5), 1068-1088.

Tilvis, R.S., Routasalo, P., Karppinen, H., Strandberg, T.E., Kautiainen, H. and Pitkala, K.H. (2012). Social isolation, social activity and loneliness as survival indicators in old age; a nationwide survey with a 7-year follow-up. *European Geriatric Medicine*. 3(1), 18-22.

Tokunaga, M. and Hashimoto, H. (2017). The socioeconomic within-gender gap in informal caregiving among middle-aged women: evidence from a Japanese nationwide study. *Social Science and Medicine*. 173(1), 48-53.

Tomás, J.M., Pinazo-Hernandis, S., Donio-Bellegarde, M. and Hontangas, P.M. (2017). Validity of the de Jong Gierveld loneliness scale in Spanish older population: competitive structural models and item response theory. *European Journal of Ageing*. 14(4), 429-437.

Torres, J.L., Lima-Costa, M.F., Marmot, M. and de Oliveira, C. (2016). Wealth and disability in later life: the English Longitudinal Study of Ageing (ELSA). *PLoS One*. 11(11), https://doi.org/10.1371/journal.pone.0166825.

Tosi, M. and Grundy, E. (2019). Work—family lifecourses and later-life health in the United Kingdom. *Ageing and Society*. 1-27. doi:10.1017/S0144686X19001752.

Tough, H., Brinkhof, M.W., Siegrist, J. and Fekete, C. (2017). Subjective caregiver burden and caregiver satisfaction: the role of partner relationship quality and reciprocity. *Archives of Physical Medicine and Rehabilitation*. 98(10), 2042-2051.

Tough, H., Brinkhof, M.W., Siegrist, J. and Fekete, C. (2018). The impact of loneliness and relationship quality on life satisfaction: a longitudinal dyadic analysis in persons with physical disabilities and their partners. *Journal of Psychosomatic Research*. 110(1), 61-67.

Tranmer, M., Murphy, J., Elliot, M. and Pampaka, M. (2020). Multiple linear regression (2<sup>nd</sup> edn.). *Cathie Marsh Institute Working Paper 2020-01*. https://hummedia.manchester.ac.uk/institutes/cmist/archive-publications/working-papers/2020/2020-1-multiple-linear-regression.pdf.

Understanding Society. (2019). Main survey: questionnaires. [*Online*]. https://www.understandingsociety.ac.uk/documentation/mainstage/questionnaires (17/12/2020).

Valtorta, N.K., Kanaan, M., Gilbody, S. and Hanratty, B. (2018). Loneliness, social isolation and risk of cardiovascular disease in the English Longitudinal Study of Ageing. *European Journal of Preventative Cardiology*. 25(13), 1387-1396.

Van Campen, C., de Boer, A.H. and Iedema, J. (2013). Are informal caregivers less happy than non-caregivers? Happiness and the intensity of caregiving in combination with paid and voluntary work. *Scandinavian Journal of Caring Sciences*. 27(1), 44-50.

Van den Berg, P., Kemperman, A., de Kleijn, B. and Borgers, A. (2016). Ageing and loneliness: the role of mobility and the built environment. *Travel Behaviour and Society*. 5(1), 48-55.

Van den Broek, T. and Grundy, E. (2018). Parental health limitations, caregiving and loneliness among women with widowed parents: longitudinal evidence from France. *European Journal of Ageing*. 15(4), 369-377.

Van den Broek, T. and Grundy, E. (2020). Does long-term care coverage shape the impact of informal care-giving on quality of life? A difference-in-difference approach. *Ageing and Society*. 40(6), 1291-1308

Vasileiou, K., Barnett, J., Barretim, M., Vines, J., Atkinson, M., Lawson, S. and Wilson, M. (2017). Experiences of loneliness associated with being an informal caregiver: a qualitative investigation. *Frontiers in Psychology*. 8(585), 1-11.

Verbakel, E., Tamlagsronning, S., Winstone, L., Fjaer, E.L. and Eikemo, T.A. (2017). Informal care in Europe: Findings from the European Social Survey (2014) special module on the social determinants of health. *European Journal of Public Health*. 27(supplement 1), 90-95.

Victor, C.R. (2005). The social context of ageing: a textbook of gerontology. London: Routledge.

Victor, C.R. (2010). Ageing, health and care. Bristol: Policy Press.

Victor, C.R. and Bowling, A. (2012). A longitudinal analysis of loneliness among older people in Great Britain. *The Journal of Psychology*. 146(3), 313-331.

Victor, C.R., Bowling, A., Bond, J. and Scambler, S. (2003). Loneliness, social isolation and living alone in later life. *ESRC: Growing Older Programme, Research Findings*. 17(1), 1-4.

Victor, C.R., Rippon. I., Quinn, C., Nelis, S.M., Martyr, A., Hart, N., Lamont, R. and Clareon, L. (2020). The prevalence and predictors of loneliness in caregivers of people with dementia: Findings from the IDEAL programme. *Ageing and Mental Health*. 1-7. https://doi.org/10.1080/13607863.2020.1753014.

Victor, C.R., Scambler, S.J., Marston, L., Bond, J. and Bowling, A. (2005). Older people's experiences of loneliness in the UK: does gender matter? *Social Policy and Society*. 5(1), 27-38.

Victor, C.R. and Yang, K. (2012). The prevalence of loneliness among adults: a case study of the United Kingdom. *The Journal of Psychology*. 146(1-2), 85-104.

Vincent, J.A., Phillipson, C.R. and Downs, M. (2006). The futures of old age. London: Sage Publications Ltd.

Vitaliano, P.P., Zhang, J. and Scanlan, J.M. (2003). Is caregiving hazardous to one's physical health? A meta-analysis. *Psychological Bulletin*. 129(6), 946-972.

Vlachantoni, A., Robards, J., Falkingham, J. and Evandrou, M. (2016). Trajectories of informal care and health. *SSM – Population Health*. 2(1), 495-501.

Vonneilich, N., Jöckel, K.H., Erbel, R., Klein, J., Dragano, N., Siegrist, J. and von dem Knesebeck, O. (2012). The mediating effect of social relationships on the association between socioeconomic status and subjective health – results from the Heinz Nixdorf Recall cohort study. *BMC Public Health*. 12(1), 285-295.

Von Soest, T., Luhmann, M., Hansen, T. and Gerstorf, D. (2020). Development of loneliness in midlife and old age: its nature and correlates. *Journal of Personality and Social Psychology*. 118(2), 388-406.

Wade, K.F., Marshall, A., Vanhoutte, B., Wu, F.C., O'Neill, T.W. and Lee, D.M. (2017). Does pain predict frailty in older men and women? Findings from the English Longitudinal Study of Ageing (ELSA). *The Journals of Gerontology Series A: Biological Sciences and Medical Sciences*. 72(3), 403-409.

Walker, A.J., Shin, H.Y. and Bird, D.N. (1990). Perceptions of relationship change and caregiver satisfaction. *Family Relations: An Interdisciplinary Journal of Applied Family Studies*. 39(2), 147-152.

Walzenbach, S. (2019). Hiding sensitive topics by design? An experiment on the reduction of social desirability bias in factorial surveys. *Survey Research Methods*. 13(1), 103-121.

Wang, S., Cheung, D.S.K., Leung, A.Y.M. and Davidson, P.M. (2020a). Factors associated with caregiving appraisal of informal caregivers: a systematic review. *Journal of Clinical Nursing*. 29(17-18), 3201-3221.

Wang, Y., Li, J., Ding, L., Feng, Y., Tang, X., Sun, L. and Zhou, C. (2020b). The effect of socioeconomic status on informal caregiving intensity of adult females for parents: evidence from a national study in China. *BMC Geriatrics*. https://doi.org/10.21203/rs.3.rs-17557/v3.

Warner, D.F. and Kelley-Moore, J. (2012). The social context of disablement among older adults: does marital quality matter for loneliness?. *Journal of Health and Social Behaviour*. 53(1), 50-66.

Weeks, L.E., Bryanton, O., Kozma, A. and Nilsson, T. (2008). Well-being of mid- and laterlife mothers of children with developmental disabilities. *Journal of Women and Aging*. 20(1-2), 115-130.

Weiss, R.S. (1973). Loneliness: The experience of emotional and social isolation. Cambridge: MIT Press.

Wenger, G.C., Scott, A. and Patterson, N. (2000). How important is parenthood? Childlessness and support in old age in England. *Ageing and Society*. 20(2), 161-182.

Wharton, T. and Zivin, K. (2017). Relationships among caregiving, income, gender, and health: A cross-sectional examination of a representative sample of older Americans. *Journal of Social Service Research*. 43(1), 141-148.

White, J., Zaninotto, P., Walters, K., Kivimäki, M., Demakakos, P., Biddulph, J., Kumari, M., de Oliveira, C., Gallacher, J. and Batty, G.D. (2018). Duration of depressive symptoms and mortality risk: The English Longitudinal Study of Ageing (ELSA). *British Journal of Psychiatry*. 208(4), 337-342.

[WHO] World Health Organisation. (1998). Health Promotion Glossary. [*Online*]. https://www.who.int/healthpromotion/about/HPR%20Glossary%201998.pdf (17/12/2020).

[WHO] World Health Organisation. (2006). Chapter IX: diseases of the circulatory system. [*Online*]. https://icd.who.int/browse10/2019/en#/IX (17/12/2020).

[WHO] World Health Organisation. (2018). Ageing and Health. [*Online*]. https://www.who.int/news-room/fact-sheets/detail/ageing-and-health (17/12/2020).

Wiggins, R.D., Netuveli, G., Hyde, M., Higgs, P. and Blane, D. (2008). The evaluation of a self-enumerated scale of quality of life (CASP-19) in the context of research on ageing: A combination of exploratory and confirmatory approaches. *Social Indicators Research*, 89(1), 61-77.

Wilcox, R.R. (2016). *Introduction to robust estimation and hypothesis testing*. 4<sup>th</sup> edn. Amsterdam: Academic Press.

Williams, A.M. Wang, L. and Kitchen, P. (2014). Differential impacts of care-giving across three caregiver groups in Canada: end-of-life care, long-term care and short-term care. *Health and Social Care in the Community*. 22(2), 187-196.

Wilson, C. and Moulton, B. (2010). Loneliness among older adults: A national survey of adults 45+. *Washington DC: AARP*.

Wimo, A., Von Strauss, E., Nordberg, G., Sassi, F. and Johansson, L. (2002). Time spent on informal and formal care giving for persons with dementia in Sweden. *Health Policy*. 61(3), 255-268.

Windle, P.E. (2010). Secondary data analysis: is it useful and valid? *Journal of Peri Anesthesia Nursing*. 25(5), 322-324.

Winefield, H.R., Gill, T.K., Taylor, A.W. and Pilkington, R.M. (2012). Psychological wellbeing and psychological distress: is it necessary to measure both? *Psychology of Well-Being: Theory, Research and Practice.* 2(1), 3-16.

Wong, J.D. and Shobo, Y. (2017). Types of family caregiving and daily experiences in midlife and late adulthood: the moderating influences of marital status and age. *Research on Aging*. 39(6), 719-740.

Wu, C.H. and Yao, G. (2008). Psychometric analysis of the short-form UCLA Loneliness Scale (ULS-8) in Taiwanese undergraduate students. *Personality and Individual Differences*. 44(8), 1792-1771.

Wu, C.H. (2009). Enhancing quality of life by shifting importance perception among life domains. *Journal of Happiness Studies*. 10(1), 37-47.

Wu, C.H., Tsai, Y.M. and Chen, L.H. (2009). How do positive views maintain life satisfaction? *Social Indicators Research*. 91(2), 269-281.

Young, H. and Grundy, E. (2008). Longitudinal perspectives on caregiving, employment history and marital status in midlife in England and Wales. *Health and Social Care in the Community*. 16(4), 388-399.

Yu, Y., Li, T.X., Li, Y.L., Qiu, D., Xi, S.J., Xiao, S.Y. and Tebes, J.K. (2020). A crosssectional study on spouse and parent differences in caregiving experiences of people living with schizophrenia in rural China. *BMC Psychiatry*. 20(226), 1-13.

Zhang, Y. and Bennett, M.R. (2019). *The likelihood of being a carer in adult life*. In Carers UK. Will I care? The likelihood of being a carer in adult life. [*Online*]. http://www.carersuk.org/images/News\_campaigns/CarersRightsDay\_Nov19\_FINAL.pdf (16/12/2020).

Zwar, L., König, H.H. and Hajek, A. (2018). The impact of different types of informal caregiving on cognitive functioning of older caregivers: evidence from a longitudinal, population-based study in Germany. *Social Science and Medicine*. 214(1), 12-19.